

COMMISSION WORK SESSION 100 E STREET, COLUMBIA CITY, OR 97018 JULY 24, 2024 6:00 P.M.

The Port of Columbia County Commission Meeting will be in person. In accordance with state law, the meeting will be accessible via telephone or Zoom. Members of the public may attend the meeting electronically by: https://us02web.zoom.us/j/81715459481

Call-In: (253) 215-8782 Meeting ID: 817 1545 9481 Passcode: 206881

- I. CALL MEETING TO ORDER (President, Brian Fawcett)
 - A. Flag Salute
 - B. Roll Call

II. ADDITIONS TO AGENDA

- III. CONSENT AGENDA (Items marked with an asterisk (*) are adopted by a single motion unless a Commissioner requests otherwise.)
 - A. * Approval of Minutes: June 12, 2024

IV. COMMENTS FROM THE PUBLIC

(Limited to 2 min. per person unless prior authorization is obtained)

V. NEW BUSINESS

- A. Columbia Pacific Bio-Refinery Quarterly Update
- B. Railroad Ave Discussion with DEQ
- C. <u>Resolution 2024-15</u> APPROVING BLUE HERON SEPTIC LEASE
- D. <u>Resolution 2024-16</u> Lacey Tolles APPROVING AIRPORT PAVEMENT MAINTENANCE BID

VI. EXECUTIVE DIRECTOR'S REPORT

VII. COMMISSIONER REPORTS

VIII. EXECUTIVE SESSION

The Board will hold an Executive Session to consider exempt public records, including written legal advice from Port General Counsel, which is privileged under ORS 40.225 and exempt from disclosure under ORS 192.355(9)(a) and ORS 192.660 (2)(f).

Kevin Jones

Elliot Levin

Amy Bynum



IX. ADJOURNMENT

Upcoming Meetings

| July 29 | 5:00 p.m. | Scappoose Airport Advisory Committee: |
|---------|-----------|---|
| | | Scappoose Airport - New Hangar Building |
| | | 34090 Sky Way Drive |
| | | Scappoose, OR 97056 |
| Aug 14 | 8:30 a.m. | Port Commission Meeting |

Community Events with Port Booth

| Aug 9 | Movies by Moonlight - Heritage Park in Scappoose |
|--------|--|
| Aug 10 | Columbia City Celebration |

Agenda times and order of items listed are estimated and are subject to change without notice. This facility is ADA-accessible. If you need special accommodation, please contact the Port office at (503) 397-2888 or TTY (800) 735-1232, at least 48 hours before the meeting.

Pursuant to ORS 192.640 (1) the Port of Columbia County Commission reserves the right to consider and discuss, in either Open Session or Executive Session, additional subjects which may arise after the agenda is published.



COMMISSION MEETING MINUTES JUNE 12, 2024 100 E. STREET COLUMBIA CITY, OREGON 97018

The Port of Columbia County held a Board meeting at 8:30 a.m. on Wednesday, June 12, 2024 at the Port office and via Zoom (*) video conferencing with the following present:

| <u>Commissioners</u> | | <u>Staff</u> | |
|----------------------|----------------------------|--------------------|-----------------------------------|
| Robert Keyser | President | Sean P. Clark | Executive Director |
| Chip Bubl | Secretary | Amy Bynum | Deputy Executive Director |
| Nancy Ward | 2nd Vice President | Robert Salisbury | Port General Counsel |
| Nick Sorber | Treasurer | Guy Glenn, Jr. | Executive Finance Manager |
| Brian Fawcett | Vice President | Miriam House | Operations Manager |
| | | Elizabeth Millager | Operations Coordinator |
| <u>Guests</u> | | Elliot Levin | North County Ops. & Terminal Mgr. |
| Chris Iverson | Hagan Hamilton Insurance | Gina Sisco | Comm. & External Affairs Mgr. |
| Jan Schollenberger | Columbia City | Christa Burns | Administrative Assistant II |
| Kim Karber | Interim Columbia City Mgr. | Lacey Tolles | Data Resource & Project Spec. |
| Ralph Culpepper | Scappoose | Amy Bynum | Deputy Executive Director |
| Natasha Parvey | NXTClean Fuels | Brittany Scott * | Finance Assistant |
| Kevin Jones * | CPBR / Global | Sydell Cotton * | Assistant Finance Manager |
| Alta Lynch * | Scappoose | Susan Tolleshaug* | Administrative Assistant |
| Scott Keillor * | WSP | Mary Laitala | Administrative Assistant |
| Dan Serres * | Columbia Riverkeeper | | |
| | | | |

President Robert Keyser called the Port of Columbia County Budget Hearing to order at 8:30 a.m. All Commissioners were present.

Budget Hearing

Commission President Robert Keyser opened the Budget Hearing to take comments from the public. There were no public comments on the proposed budget and the Budget Hearing closed at 8:32 a.m.

The Port of Columbia County regular Board meeting was then called to order at 8:33 a.m. by Mr. Keyser.

Additions To Agenda

Robert Keyser asked the Commission if there were any additions to the agenda. The Commission added the Executive Director Performance Evaluation to Executive Session.

Consent Agenda

Mr. Keyser asked for a motion to approve the consent agenda. Nancy Ward moved; Brian Fawcett seconded a motion to adopt consent agenda items A, B and C: April 10, 2024 and May 22, 2024 Commission Meeting Minutes, May 22, 2024 Budget Committee Minutes, April 2024 Finance Report, and Approval of Check Register (A) and electronic payments in the total amount of \$459,061.57. Motion carried unanimously.

Comments From the Public

Brady Preheim, St. Helens resident, commented on Resolution 2024-09. Mr. Preheim stated that he would support the Port tax because the Port needs the money, but also needs the rental revenue from NEXT. He suggested giving NEXT a 30-day notice to begin making full rent payments.



Dan Serres informed the Commission that the online participants could not hear what happened before the Consent Agenda. Mr. Keyser responded that the Executive Director Performance Evaluation was added to Executive Session under "Additions to Agenda."

New Business

Resolution 2024-09

SETTING THE PORT TAX

Port Executive Director Sean Clark presented Resolution 2024-09. The Port held a Budget Committee Meeting on May 8, 2024. At that time, the Budget Committee voted in favor of setting the Port tax at \$0.0400 per \$1,000 of assessed value. The next step is for the Commission to vote and, if voting to set the tax at zero, Resolution 2024-09 will need to be amended. Robert Keyser stated that at the time of the Budget Committee meeting, he voted yes on the \$0.0400 tax rate in order to move it forward to discussion. However, he does not support levying the Port tax this year. Mr. Keyser said that he had a conversation with Commissioner Ward about putting together a committee to look at water access, dredging, and marina issues, and to make sure that stakeholders in the community are involved. They would like to determine whether it makes sense to add the tax back into the Port's budget next year. Nancy Ward reinforced Mr. Keyer's comments and added that if there is any type of change to the tax structure, the Port needs outreach to the community. Ms, Ward also thanked Chip Bubl for his letter to the newspaper. Commissioner Bubl stated that he supports the tax and agreed with Mr. Preheim regarding the NEXT lease. Mr. Bubl pointed out that it has only been four years since the tax was removed from the Port's budget, so at some point the public voted to levy the tax. Additionally, the public has never voted to remove the tax. Mr. Bubl said that it is important for the Port to start taxing again for fiscal stability, and that he would support levying the tax at \$0.08 per \$1,000. Nick Sorber stated that he does not support the tax at this time and said the Port has already done public outreach to educate the community on how the Port operates, specifically at Scappoose Bay. Mr. Sorber echoed levying the full rate if the Port tax is brought back. Brian Fawcett stated that he is also opposed to reinstating the Port tax this year in an environment where most taxes were voted down. Mr. Fawcett thinks it is important for the Port to continue educating the public about the marina, water access, and local recreation. He pointed out that the tax is a small amount that would make a big impact on bringing more value to the community. Mr. Fawcett looks forward to having those conversations over the next year and reevaluating the tax in 2025.

Robert Keyser called for a motion. Brian Fawcett moved to amend Resolution 2024-09 to set the Port tax rate at \$0.0 per \$1,000 of assessed value; Nick Sorber seconded the motion. Commissioners Keyser, Ward, Sorber and Fawcett voted yes. Commissioner Bubl voted no. Motion carried 4-1.

Resolution 2024-10

ADOPTING THE FY 2024-25 BUDGET, MAKING APPROPRIATIONS

Sean Clark presented Resolution 2024-10. Mr. Clark explained that the Port has followed the legal budget process. The Budget Committee met on May 8, 2024, and a Budget Hearing was held at the start of today's meeting to take public comment. There were no public comments on the proposed budget. Following the adoption of the FY 2024-25 Budget, the adopted budget will be submitted to the County by the June 30, 2024 deadline.

President Keyser called for a motion to approve Resolution 2024-10. Brian Fawcett moved; Chip Bubl seconded a motion to adopt Resolution 2024-10, adopting the fiscal year 2024-2025 Budget in the total amount of \$17,564,969. Motion carried unanimously, 5-0.



Resolution 2024-11

ROTATING THE NOMINATION PROCESS FOR THE ELECTION OF PRESIDENT

Port General Counsel Robert Salisbury gave a PowerPoint presentation on Officer Elections: Rotating the Presidency. Mr. Salisbury explained the officer elections process and went over the order of priority for nominations. The Commission President is selected based on majority vote at the first meeting in July each year, subject to accepting a nomination. The Vice President is automatically selected based upon the Commissioner second in order of priority. Mr. Salisbury noted that a board member must have served on the Commission for at least one year in order to be eligible for President or Vice President. The top priority in the new Policy for choosing the Port President would be the person with the most seniority as Commissioner who has never served as President. Second priority would be the person with the most seniority and the longest period of time since serving as President, and third priority would be the person with the longest time since serving as President. Mr. Salisbury informed the Commission that he consulted with two statistical and analytics experts and they agreed that the Policy was very well-written and comprehensive. The only question that came up was how to resolve ties. The Commission considered this question and decided to leave that open to a majority vote or other process to be decided later. Mr. Keyser stated that this assumes all of the candidates accept the nomination, and Mr. Salisbury confirmed this would just be for nominations and, if a candidate declined the nomination it would go to the next in line based on seniority. Mr. Fawcett asked who would be next in line for Vice President if the first in priority says no, and Mr. Salisbury replied that the first in line for President would then be given the chance to become Vice President.

President Keyser called for a motion to approve Resolution 2024-11. Nancy Ward moved; Brian Fawcett seconded a motion to adopt Resolution 2024-11. Motion carried unanimously, 5-0. Port Policy 008 and Mr. Salisbury's PowerPoint are on file at the Port Office.

Resolution 2024-12

APPROVING AN AGRICULTURAL LEASE WITH CRAIG COLEMAN AT PORT WESTWARD INDUSTRIAL PARK

Elliot Levin, North County Terminal & Operations Manager, presented Resolution 2024-12, which would authorize the Port to enter into an agricultural Lease with Craig F. Coleman Inc. for 118 acres at Port Westward. Mr. Levin displayed a map showing the location of the parcel and said that the Port bought the land in 2009 from Lower Columbia Tree Farm. Under that purchase agreement, Lower Columbia Tree Farm retained timber rights to the land; thus, it had not previously been leased to a third party although Columbia River Ranch has licensed the land periodically to graze cattle. The trees were recently harvested, and the land is now available for lease. Mr. Levin informed the Commission that over the past few months, the Port and Mr. Coleman began negotiating a lease. Rent for the first two years will be \$30.00 per acre per year. This rate compares with that of other Port agricultural leases and recognizes that the land will need time to become productive. During the third year of the lease, the rent increases to \$150.00 per acre per year. If the lease extends beyond the initial 4-year term, rent increases again to \$175.00 per acre per year. This lease structure recognizes the work needed to improve the land while bringing the rent up to the market rate of the land. Next, Mr. Levin introduced Craig Coleman to discuss his plans for the property and answer any questions from the Commission. Mr. Coleman stated that he has been in the farming business since 1982 and has a farm in Hermiston, OR which his son now oversees. They primarily grow blueberries among about fifteen other crops depending on the season. Mr. Coleman said he is relocating to Columbia County and had purchased about 300 acres in the Beaver Drainage District last summer (2023). He plans to grow blueberries at Port Westward and hopes to have the crop started in the Spring of 2025. Mr. Keyser asked if he had considered the 200 acres by the boatyard. Mr. Coleman responded that he did and that he is interested in acquiring more land once the trees are harvested. Ms. Ward asked what brought him to this area. Mr. Coleman responded that the land appears to be undeveloped and that he wants to bring high value to the community.



President Keyser called for a motion to approve Resolution 2024-12. Brian Fawcett moved; Nick Sorber seconded a motion to adopt Resolution 2024-12. The motion carried unanimously.

Old Business

Marina Update

Miriam House, Port Operations Manager, provided an update on the Scappoose Bay Marine Park. Ms. House stated that the Upland construction project is moving along on schedule with design plans at 100 percent. When the Port receives the building permit, the Invitation to Bid (ITB) will go out. Ms. House complimented KPFF, the consulting engineers, for doing such great work on the drawings. Port maintenance staff completed the building of two new pay stations except for the metal roofs which will be installed on site during construction to accommodate the electrical wiring. The only unplanned addition was the replacement of an old fire hydrant. The project came in under budget and the Port received a grant in the amount of \$408,000. The electric vehicle chargers (EV) have been installed at the marina in partnership with the Columbia River Public Utilities District (CRPUD). Ms. House showed a photo of vehicles already being successfully charged. The Port hired a contractor to perform maintenance on the nature trails to increase pedestrian safety. This recent maintenance included filling potholes, removing tree roots, and re-paving sections of the trail loop, at a cost of \$26,000. Ms. House reported areas of the main truck and trailer parking lot that also need the same treatment, and the plan is to time those repairs with the Upland construction for less interruption to park visitors. Ms. House also reported that Metro-Watch security service is on duty at the marina and at Trestle Beach. Nancy Ward inquired about the 10-minute parking spots being removed. Ms. House confirmed removal of those spaces because they were not being used appropriately. Marine park guests often exceed the 10-minute time limit, and some remained parked there all day. Ms. Ward pointed out that we want people to use the facility, especially locals, yet the monitoring of parking spaces is a challenge for staff. Brian Fawcett agreed and suggested taking it to the Marina Advisory Board. The Commission agreed and gave consensus to have a discussion on parking and monitoring at the next Marina Advisory Board meeting. Lastly, Mr. Clark gave kudos to Miriam House and the maintenance team and said he appreciates their efficiency.

Airport Update

Interim Airport Manager Lacey Tolles gave a PowerPoint presentation on current projects at the Scappoose Airport. First, Ms. Tolles informed the Commission that there are no pending thru-the-fence permit applications. Ms. Tolles continued by stating that Northeast Electric is wrapping up Phase 1 of the Emergency Generator Project, and both generator pads have been placed at Transwestern Aviation and the Airport Lighting Vault (ALV). The tanks and generators are on order. Ms. Tolles also said the trench has been dug for infrastructure improvements on the east side of Airport Road. The Port would like to thank Ed Freeman and his Moore Road Expansion Project for providing the trench at no cost. As part of that process, the Port will be installing five concrete bases and laying conduit in preparation for future lighting and a natural gas line. The Port has ordered two lighted runway closure lights that will be housed on site at Scappoose Airport for quicker access. Ms. Tolles also announced that the Port is moving forward with an Invitation to Bid (ITB) on the Westside Pavement Maintenance Project. Commissioner Keyser asked about the status of the East side hangar demolition project which was included in the budget for fiscal year 2024-25. Ms. Tolles responded that demolition of the East side hangers is a priority and is being discussed with the Scappoose Advisory Committee.

Lignetics/Neighbors Update

Port Executive Director Sean P. Clark updated the Commission on Lignetics' progress with DEQ and stated that he had engaged Nate Stice, area representative for the Governor's Regional Solutions Team. The DEQ process requires approval of the air permit modification modeling before Lignetics can submit the air contaminant discharge permit application. Mr. Clark stated that he received a message



from Mike Sale, VP of Lignetics, confirming that Lignetics had submitted the air modeling for the new cyclone and is still waiting on approval from DEQ. As a response to the unannounced inspection by DEQ for the storm water permit, DEQ provided a list of items for Lignetics to address by July 16. Sawdust left outdoors is required to be covered, and Lignetics has made a focused effort on reducing inventory. Mr. Clark stated that Lignetics will most likely not be rerouting the truck traffic due to the required changes to their operations and the expense of moving the scale. Nick Sorber stated that the road in the back needs repairs. Nancy Ward mentioned that the truck location had been moved in the past and asked if moving the scale is the main issue. Mr. Clark replied that he would find out. Mr. Sorber said that trucks pulling out on the highway with no traffic light is also a concern. Mr. Clark added that Port tenant Clearwater Paper also uses that road to and from Longview, WA. Ms. Ward asked if it is a private road. Mr. Clark responded that the Port owns the road. The Port will continue to monitor any spillage from the trucks. Ms. Ward then invited Columbia City resident Jan Schollenberger to come forward to speak on the record. Ms. Schollenberger said the neighborhood appreciates the Port being proactive and helping to make progress. She said she is disappointed to hear about truck traffic not being rerouted, as the problem is not just the spillage from the trucks but also the congestion in residential streets. She also pointed out several items that were planned to be completed by September 2024 are now being pushed into 2025. The residents do not want to endure two more summers of the dust. The neighbors would like a timeline update from Lignetics. Ms. Schollenberger thanked the Port for its continued support and said that another community meeting will be held sometime this summer.

Donations Policy Discussion

Communications and External Affairs Manager Gina Sisco came forward to provide information on the Port's Donation Policy, as the Commission had previously requested a policy review. Under current policy, the Port does not contribute money or items of cash value to support charitable organizations. Ms. Sisco brought this up at the Commission's request for discussion and guidance on possibly revising the policy to create a fund for nonprofit donations requests. Brian Fawcett stated that it would be beneficial to the public and that including it in next year's budget process seems like a reasonable approach. Nick Sorber said the Commission would need to decide on an amount and how many awards would be given out annually. Nancy Ward added that it would need to be on a rotating basis for recipients. Ms. Sisco inquired about guidelines and an application and screening process. Mr. Bubl mentioned that earlier materials stated that donations must be related to Port functions and benefiting Port properties and economic development. This would be separate from community outreach funds for FY 2024-25. The Commission gave consensus to form a committee that includes two Commissioners to consider revising the resolution. Commissioners Fawcett and Bubl volunteered to serve on the Committee.

Executive Director's Report

Executive Director Sean Clark introduced Mary Laitala with Cardinal Services and announced that Port Administrative Assistant Susan Tolleshaug will be retiring. Mr. Clark stated that Ms. Laitala would be filling that role and welcomed her to the Port. Mr. Clark said that he attended the Pacific Northwest Waterways (PNWA) Conference in Clarkston, WA June 4 – 6 and one of the key takeaways is the rising costs affecting projects and budgets across the board. Mr. Clark and Amy Bynum toured SportCopter, a commercial tenant at Scappoose Airport, on Monday, June 10. Mr. Clark informed the Commission that Elliot Levin gave a presentation to the Oregon Freight Advisory Council as part of the review process for a Connect Oregon grant. Mr. Clark said he attended the Columbia County Business Alliance Board Meeting and had a round table with Columbia Economic Team (CET) to discuss Port specific issues. This Friday, June 14, Mr. Clark and Commissioner Fawcett will go on a Milton Creek walk. Mr. Clark and Commissioner Keyser will meet with Nate Stice from the Governor's Regional Solutions Team on June 20 at the Port office to talk about storm water issues and then take a tour of Port Westward. Mr. Clark asked the Commissioner Fawcett said he will



join the Port Westward tour and Mr. Sorber will go on the Milton Creek walk. The Port will have a booth at Citizen's Day in the Park on June 22 and staff will attend a Tenant Luncheon at the Scappoose Airport on June 25. The Executive Director's Report is on file at the Port Office.

Commissioner Reports

Brian Fawcett reported that he has been having conversations around the budget and tax lately, as well as discussing a few marina items. Mr. Fawcett said he appreciates the Port staff's work and willingness to hear feedback.

Nick Sorber commented that he also appreciates staff efforts. Mr. Sorber said he is excited to hear about the agricultural lease and hopes that it works out.

Chip Bubl stated that he wrote a letter to the newspaper and had conversations with several people including a few Commissioners.

Nancy Ward reported being invited to join a group to get acquainted with the four people selected as finalists for Scappoose City Administrator. Ms. Ward suggested the Port look into the availability of federal funding for infrastructure improvements.

Robert Keyser reported that he was recently out of state. Mr. Keyser mentioned 200 acres that are available in Clatskanie for potential future development of blueberries and other crops.

Executive Session

The Board held an Executive Session to consider exempt public records, including written legal advice from Port General Counsel, which is privileged under ORS 40.225 and exempt from disclosure under ORS 192.355(9)(a) and ORS 192.660(2)(f).

THERE BEING NO FURTHER BUSINESS BEFORE THE COMMISSION, THE MEETING ADJOURNED AT 10:10 A.M.

President

<u>July 24, 2024</u>

Date Adopted by Commission

Secretary



2024-15 STAFF REPORT

Blue Heron Septic & Drain Services New Lease

| DATE: | July 24, 2024 |
|-------|---|
| TO: | Port Commission |
| FR: | Amy Bynum, Real Estate & Business Development Manager |
| RE: | Blue Heron Septic & Drain Services New Lease |

Discussion

Blue Heron Septic & Drain Services ("Blue Heron") has been serving St. Helens and the surrounding area for years and has built a strong reputation throughout our community for their reliable service. Blue Heron has been operating their business under a home occupation license within the City of St. Helens and has plans to expand. Approval of the attached Lease is an important step for the company.

Blue Heron is well-suited for Railroad Avenue as they do not require a public-facing facility and they understand the environmental concerns at Railroad Avenue and the fact that the Department of Environmental Quality ("DEQ") has oversight of this property. DEQ has been notified and approves of the intended use.

Summary of Lease Terms:

- 1. <u>Premises</u>: 3,500 square feet of office and shop space, 21,500 sq ft of parking lot and covered outdoor parking.
- 2. <u>Purpose</u>: Non-loaded heavy machinery and truck storage, fleet maintenance, office and administrative activities.
- 3. <u>Term</u>: Five-year lease with two successive three-year renewals.
- 4. <u>Basic Rent</u>: \$2,592 per month and includes annual CPI adjustments & a security deposit. This is a Triple Net Lease. Some charges such as maintenance of the property, property taxes and insurance will be billed separately.

Recommendation

Adopt Resolution No. 2024-15, authorizing the Executive Director to execute a new Lease with Blue Heron Septic & Drain Services.

RESOLUTION 2024-15

A RESOLUTION APPROVING A LEASE WITH BLUE HERON SEPTIC & DRAIN SERVICES AT RAILROAD AVENUE INDUSTRIAL PARK

WHEREAS, Blue Heron Septic & Drain Services, ("Blue Heron") a local sanitation company, has been working with Port Staff to negotiate a lease to support their business operations; and

WHEREAS, Blue Heron has been serving St. Helens and the surrounding area for years and has built a strong reputation throughout our community for their reliable service; and

WHEREAS, Blue Heron is interested in expanding their business and have successfully negotiated a lease for the property, shop building, and covered parking area, on terms agreeable to both parties; and

WHEREAS, Port staff recommends the adoption of the lease, which is attached with its appropriate exhibits; now, therefore

BE IT RESOLVED by the Board of Commissioners of the Port of Columbia County as follows:

The Board authorizes the Executive Director to execute the attached Lease with Blue Heron Septic & Drain Services.

PASSED AND ADOPTED this 24th day of July 2024, by the following vote:

AYES: _____ NAYS: _____

Port of Columbia County

ABSTAIN:

By:

President

Attested By:

Secretary

LEASE

BETWEEN THE

PORT OF COLUMBIA COUNTY

AND

BLUE HERON SEPTIC & DRAIN SERVICES

Blue Heron Septic and Drain Services Lease

TABLE OF CONTENTS

| 1. AGF | REEMEN | T TO LEASE; DESCRIPTION OF PREMISES | 8 |
|---------------------------------|----------------------|---|-------|
| 1.1 | Use of the Premises | | |
| 8 | 1.1.1 | Access | |
| 1.2 | Complian | nce with the Law | 8 |
| 2. TER | M | | 9 |
| 2.1 | Initial Le | ase Term | 9 |
| 2.2 | Lease Ye | ar | 9 |
| 2.3 | Extensior | 1 Options | 9 |
| | 2.3.1 | Lessee's Notice of Intent to Extend | 9 |
| | 2.3.2 | Conditions of Extension | 9 |
| 2.4 | Holdover | · | 10 |
| 3. REN | T | | 10 |
| 3.1 | Basic Rei | nt and Additional Rent | 10 |
| | 3.1.1 | Attorney Fees and Staff Costs to Be Paid by Lessee During Term of Lease | 11 |
| 3.2 | Rent Adj | ustment | ,11 |
| | 3.2.1 | Fair Market Value | 11 |
| | 3.2.2 | Effective Date of Adjustment; Payment of Adjustments | .12 |
| 3.3 | Rent Payments 1 | | 13 |
| | 3.3.1 | Due Dates | .13 |
| 3.4 | Delinquency Charge13 | | 13 |
| 3.5 | Returned Checks13 | | 13 |
| 3.6 | Acceptance of Rent13 | | .13 |
| 3.7 | Security Deposit14 | | .14 |
| 3.8 | Taxes14 | | .14 |
| 3.9 | Non-Ren | nonstrance | .15 |
| 4. LESSEE'S OTHER OBLIGATIONS15 | | | |
| 4.1 | Construc | tion of Improvements | .15 |
| Blue H | Heron Sep | tic and Drain Services Lease P | age 2 |

| | 4.1.1 | Port Appro | oval and Ownership15 | |
|--------|-------------------------------|--------------|-----------------------------------|--|
| | 4.1.2 | Wetlands I | Prevention15 | |
| 4.2 | Ownershi | p of Improv | /ements | |
| 4.3 | Signs | Signs | | |
| 4.4 | No Liens | | | |
| 4.5 | Utilities a | nd Services | 317 | |
| 4.6 | Safety Re | quirements | | |
| 4.7 | Fire Safet | ty | | |
| 4.8 | Security. | | | |
| 4.9 | Waste, R | emoval and | Demolition | |
| 4.10 | Lessee's | Maintenan | ce Obligations | |
| 4.11 | Port's Ma | aintenance (| Obligations18 | |
| 4.12 | Conduct | of Business | | |
| 5. POR | RT AUTH | ORITY AN | D OBLIGATIONS | |
| 5.1 | Quiet Enj | joyment | | |
| 5.2 | Delivery | of Premises | | |
| 5.3 | Condition | n of Premise | es19 | |
| 5.4 | Port Cons | struction an | d Repair Obligation on Premises19 | |
| 5.5 | Port and Third-Party Access19 | | | |
| 5.6 | Remediat | tion Access | | |
| 6. ENV | IRONMI | ENTAL MA | ANAGEMENT AND COMPLIANCE19 | |
| 6.1 | Definition | ns | | |
| | 6.1.1 | Environme | ental Laws20 | |
| | 6.1.2 | Hazardous | Substance | |
| | 6.1.3 | Environme | ental Audit20 | |
| | | 6.1.3.1 | Initial Audit20 | |
| | | 6.1.3.2 | Special Audit20 | |
| | | 6.1.3.3 | Exit Audit20 | |

| 6.2 | Environmental Inspection21 | | |
|-------------------------------|----------------------------|--|--------|
| | 6.2.1 | Environmental Cost | 21 |
| | 6.2.2 | Hazardous Substance Release | .22 |
| | 6.2.3 | Best Management Practices | .22 |
| | 6.2.4 | UST Facility | .22 |
| | 6.2.5 | AST Facility | .22 |
| 6.3 | General I | Environmental Obligations of Lessee | .22 |
| 6.4 | Hazardou | us Substances Use on Premises | .23 |
| | 6.4.1 | Permitted Use of Hazardous Substances | .23 |
| | 6.4.2 | Conditional Hazardous Substances Use with Port Approval | .23 |
| | 6.4.3 | Above Ground and Mobile Hazardous Substances Storage Tanks | .24 |
| | 6.4.4 | Port Review of Lessee's Proposed Hazardous Substance Use | .24 |
| | 6.4.5 | Notice of Port's Review of Hazardous Substance Use | .24 |
| | 6.4.6 | Conditions of Hazardous Substance Use Approval | .24 |
| | 6.4.7 | Discharge and Treatment of Industrial Wastewater | .25 |
| 6.5 | Off-Site | Waste-Like Materials | .25 |
| 6.6 | Lessee's | essee's Liability25 | |
| | 6.6.1 | Hazardous Substance Releases | .25 |
| | 6.6.2 | Limitation of Lessee's Liability | .25 |
| 6.7 Environmental Remediation | | nental Remediation | .26 |
| | 6.7.1 | Immediate Response | .26 |
| | 6.7.2 | Remediation | .26 |
| | 6.7.3 | Natural Resources Damages Assessment and Restoration | .26 |
| | 6.7.4 | Report to the Port | 27 |
| | 6.7.5 | Port's Approval Rights | 27 |
| 6.8 | Notice | | 27 |
| 6.9 | Split San | npling | 27 |
| 6.10 | Port's Ri | ght to Perform on Behalf of Lessee | 27 |
| 7. INDEMNITY, INSURANCE | | | |
| Blue H | Ieron Sep | otic and Drain Services Lease | Page 4 |

| 7.1 | Lessee's General Indemnity; Reimbursement of Damages | | |
|--------|--|--|--|
| 7.2 | Insurance Requirements | | |
| | 7.2.1 | Certificates; Notice of Cancellation | |
| | 7.2.2 | Additional Insured; Separation of Insureds | |
| | 7.2.3 | Primary Coverage | |
| | 7.2.4 | Company Ratings | |
| 7.3 | Required | Insurance | |
| | 7.3.1 | General Liability Insurance | |
| | 7.3.2 | Property Insurance | |
| | 7.3.3 | Automobile Liability Insurance | |
| | 7.3.4 | Workers' Compensation Insurance | |
| | 7.3.5 | Personal Property Insurance | |
| | 7.3.6 | Lessee's Risks | |
| 7.4 | Waiver o | f Subrogation31 | |
| 7.5 | Periodic Review | | |
| 7.6 | Survival | of Indemnities31 | |
| 8. DA | MAGE A | ND DESTRUCTION | |
| 8.1 | General. | | |
| 8.2 | Restoration | | |
| 8.3 | Termination of Lease Following Casualty | | |
| 9. TEH | RMINATI | ON | |
| 9.1 | Duties or | Termination | |
| 10. DE | FAULT | | |
| 10.1 | Event of | Default | |
| | 10.1.1 | Default of Monetary Obligations | |
| | 10.1.2 | Default in Covenants | |
| | 10.1.3 | Bankruptcy/Insolvency | |
| 10.2 | Remedie | s on Event of Default | |

| | 10.2.1 | Termination of Lease | |
|--------|---------------------|-------------------------------|--|
| | 10.2.2 | Reentry | |
| | 10.2.3 | Reletting | |
| | 10.2.4 | Right to Sue More than Once | |
| 10.3 | No Waive | er of Default | |
| 10.4 | Remedies | s Cumulative and Nonexclusive | |
| 10.5 | Curing L | essee's Default | |
| 10.6 | Administ | rative Costs | |
| 10.7 | Default b | y Port | |
| 11. TR | ANSFER | BY THE PORT | |
| 12. AS | SIGNME | NT OF INTEREST OF RIGHTS | |
| 13. PO | RT CON | TROL AND SERVICES | |
| 14. CO | NDEMN | ATION | |
| 14.1 | General | | |
| 14.2 | Definitio | ns37 | |
| | 14.2.1 | Partial Taking | |
| | 14.2.2 | Total Taking | |
| | 14.2.3 | Taking | |
| | 14.2.4 | Date of Taking | |
| | 14.2.5 | Premises Award | |
| | 14.2.6 | Leasehold Award | |
| | 14.2.7 | Port's Reversionary Interest | |
| | 14.2.8 | Distribution of Takings Award | |
| 14.3 | Total Tal | sing | |
| 14.4 | Partial Ta | aking | |
| 14.5 | Claims A | gainst Condemning Authority39 | |
| 14.6 | Adjustment of Award | | |
| 14.7 | Effect of | Termination | |

| 14.8 | Notice of Taking |
|--------|--|
| 15. PO | RT CONDEMNATION RIGHTS |
| 16. GE | NERAL PROVISIONS |
| 16.1 | Covenants, Conditions, and Restrictions |
| 16.2 | Governing Law |
| 16.3 | No Implied Warranty40 |
| 16.4 | Notices |
| 16.5 | Time of the Essence |
| 16.6 | Non-Waiver41 |
| 16.7 | Survival41 |
| 16.8 | Partial Invalidity41 |
| 16.9 | Limitation on Liability42 |
| 16.10 | Headings, Table of Contents and Table of Definitions |
| 16.11 | Exhibits Incorporated by Reference |
| 16.12 | Modification42 |
| 16.13 | Provisions Applicable to Others |
| 16.14 | Interpretation of Lease; Status of Parties |
| 16.15 | Calculation of Time |
| 16.16 | Absence of Brokers |
| 16.17 | Attorney Fees |
| 16.18 | Right of Parties and Successors in Interest |
| 16.19 | Joint and Several Obligations44 |
| 16.20 | Defined Terms |
| 16.21 | Execution of Multiple Counterparts44 |
| 16.22 | Estoppel Certificates |
| 16.23 | Force Majeure Event44 |
| 16.24 | Mediation |
| 16.25 | Entire Agreement |
| 16.26 | Capacity to Execute |
| Blue I | Heron Septic and Drain Services Lease Page 7 |

| EXHIBIT A (Premises) | 7 |
|--|---|
| EXHIBIT B (Agreement to Pay Costs Example)48 | 3 |
| EXHIBIT C (DEQ Record of Decision) |) |

LEASE

This LEASE (the "Lease"), effective the _____ day of ______, 2024 (the "Effective Date") by and between THE PORT OF COLUMBIA COUNTY, a municipal corporation of the State of Oregon (the "Port"), and BLUE HERON SEPTIC AND DRAIN SERVICES, an Oregon corporation. ("Lessee").

1. AGREEMENT TO LEASE; DESCRIPTION OF PREMISES

Subject to the terms and conditions herein, the Port hereby leases to Lessee, and Lessee leases from the Port, real property located at 1550 Railroad Ave, St. Helens, Oregon. The real property will include an approximate 3,500 square feet (SF) office and shop building and 21,500 SF parking courtyard and covered parking structure thereunder and further depicted, all as proposed in Exhibit A (together, the "Premises"), attached hereto and incorporated herein. The Premises are located within the Port owned property known as the Railroad Corridor Industrial Park ("Railroad Corridor"). The Port shall have the right to reduce the Premises and/or substitute all or a portion of the Premises for other property owned by the Port upon sixty (60) days prior written notice to Lessee in the event that such relocation is required in order to comply with Department of Environmental Quality (DEQ) or other government agency requirements as part of the ongoing environmental cleanup on the Premises. Lessee acknowledges it is aware of this ongoing environmental cleanup and consents to all activities which may occur on the Premises during the term of this Lease, as also referenced in Section 5.3 and Exhibit C. The Lessee may terminate this Lease if the Lessee determines that such reduced Premises or substituted property is not suitable. If the Premises are so relocated, this Lease shall terminate and a new agreement for the substituted property shall be negotiated.

1.1 Use of the Premises

Lessee's use of the Premises shall be limited to operating a commercial business engaged in heavy machinery and truck storage, fleet maintenance, office and administrative, and other activities directly related to the conduct of a septic and drainage services business ("**Allowed Use**"). Lessee agrees that no loaded trucks shall be allowed to be stored on the Premises. No use other than the Allowed Use may be made of the Premises without the prior written approval of the Port, which approval shall be given or denied in the sole discretion of the Port. No Hazardous Substances (as that term is defined in Section 0) may be used, stored, or otherwise handled on or near the Premises, except as permitted under Section 0 of this Lease. Lessee will comply strictly with all present and future rules and regulations of all federal, state, and local governmental bodies having jurisdiction over Lessee's activities occurring within the Premises.

1.1.1 Access

Access to the Premises shall be solely via the route as depicted in Exhibit A and by no other route unless directed by the Port. The port makes no representations or warranties as to the condition or suitability of the existing bridge for Lessee's purposes.

1.2 Compliance with the Law

Lessee's use of the Premises must comply with all applicable laws, ordinances, rules and regulations of the State of Oregon and the United States, and all city, county or other public government authorities or agencies, including, but not limited to, building permit requirements, local fire code, zoning, Department of Environmental Quality regulations, and occupancy codes. Lessee shall also comply with all Port Rules, as amended, and updated from time to time. Lessee shall promptly provide to the Port copies of all written communications (including electronic communications) from any such government entities which relate to Lessee's noncompliance or alleged noncompliance with any law or other government requirements. In no event shall Lessee be permitted to seek or obtain approval to rezone, change the use of, or obtain other land use or land division approvals for, the Premises without the Port's prior written approval, which may be granted, conditioned, or denied in the Port's sole discretion.

2. TERM

2.1 Initial Lease Term

The term of this Lease ("**Term**" or "**Lease Term**") shall commence on the Effective Date (the "**Lease Commencement Date**") and shall continue for five (5) years from the Effective Date ("**Expiration Date**") unless sooner terminated or extended pursuant to the terms of this Lease.

2.2 Lease Year

As used in this Lease, "Lease Year" shall mean, in the case of the first Lease Year, the period beginning on the Lease Commencement Date and ending on the following June 30, and thereafter, each successive twelve (12) month period beginning July 1 and ending June 30 during the Term.

2.3 Extension Options

So long as no uncured Event of Default (defined in Section 10.7) exists under this Lease, Lessee shall have options to extend ("**Extension Options**") the Expiration Date of this Lease for two (2) successive terms (each an "**Extension Term**"). Each Extension Term shall be for three (3) years.

2.3.1 Lessee's Notice of Intent to Extend

Lessee must notify the Port in writing of Lessee's intent to exercise any Extension Option, subject to Lessee's acceptance of: (i) the Port's proposed new Basic Rent (as defined in Section 3.1 and pursuant to Section 3.2); (ii) any proposed Extension Amendments (as defined in Section 2.3.2); and (iii) any proposed adjustment to the Security Deposit pursuant to Section 3.7. Such notice by Lessee must be given not more than three hundred sixty (360) days and not less than ninety (90) days prior to expiration of the then-current Lease Term or Extension Term, as applicable (the "Extension Notice"). This notification by Lessee to Port will begin the negotiation between Port and Lessee. Failure of Lessee to deliver the Extension Notice as provided in this Section 2.3.1 shall automatically extinguish any applicable Extension Option.

2.3.2 Conditions of Extension

The terms and conditions of the Lease for each Extension Term shall be the same as those for the initial Lease Term except that: (a) the Security Deposit will be adjusted to reflect adjustment in Basic Rent as provided in Section 3; (b) insurance provisions will be updated, subject to the provisions of Section 7.3; (c) environmental provisions will be updated, as necessary, to comply with then current Environmental Laws, Environmental Audit requirements and Best Management Practices as those terms are defined in Section 60; and (d) the Port will require any modifications or changes legally required to bring the Lease into compliance with then current law, government regulations, government mandates (except that any improvements to the use or design of which are grandfathered or otherwise permitted under then-existing law or regulations need not be modified or upgraded), or Airport Minimum Standards (collectively, items (b) through (d) in this Section 2.3.2 are the "Extension Amendments"). Upon receipt of Lessee's notice of intent to extend pursuant to Section 2.3, the Port shall have sixty (60) days thereafter to notify Lessee, in writing, of the proposed adjustments to the Security Deposit, including the basis for the adjustment, and any Extension Amendments required by the Port as a condition of granting the then applicable Extension Option, together with a proposed Amendment to the Lease ("Amendment Form") (collectively, "Notice of Amendments and Proposed Rent"). Lessee shall have sixty (60) days from receipt of the Notice of Amendments and Proposed Security Deposit to notify the Port in writing of Lessee's disapproval of any term thereof. Thereafter, Lessee and Port shall find a mutually-agreeable compromise to the disputed terms thereof. If Lessee fails to deliver written notice of Lessee's disapproval of any term in the Notice of Amendments and Proposed Rent, then Lessee shall be deemed to have approved the Extension Amendments, Base Rent and adjusted Security Deposit and shall be obligated for the Extension Term. Lessee shall be responsible for completing all general maintenance, repair and clean-up requirements set forth in Section 4.10, on or before the first day of any Extension Term.

2.4 Holdover

If Lessee does not vacate the Premises at the time required (upon expiration or termination of the Lease), the Port shall have the option to treat Lessee as a tenant from month to month, and Lessee shall pay Rent at an increased rate of 150% above the then-applicable Rent as outlined in Section 3 ("Holdover Rent Increase"), subject to all the provisions of this Lease except the provisions for term. The Holdover Rent Increase shall take effect immediately on the first day of the month after the expiration or termination of the Lease. The holdover tenancy shall be terminable at the end of any monthly rental period on written notice from the Port given not less than 30 days prior to the termination date, which shall be specified in the notice. Lessee waives any notice which would otherwise be provided by law with respect to a month-to-month tenancy.

3. RENT

3.1 Basic Rent and Additional Rent

"Basic Rent" will be payable on or before the first of each month calculated as Basic Rent is calculated as follows (the "Rental Rate"): the total square feet (SF) of the office and shop building (3,000 SF) multiplied by a price of \$0.50 per SF per month and the total SF of the parking lot and covered parking (21,500 SF) multiplied by a price of \$0.47 per SF per year, divided by

twelve, for a total rent of Two Thousand Five Hundred and Ninety Two Dollars (\$2,592) per month. The Port reserves the right to remeasure the square footage of the Premises throughout the lease term. Upon the exercise of such right, if Lessee is found to be occupying an area other than the square footage provided in Section 1, Basic Rent for the Lease Term shall be adjusted accordingly. In addition, if at any time during the Lease Term Lessee is found to be occupying more square footage than stated in this Section, Lessee shall pay monthly additional rent for such excess square footage at the Rental Rate. All Basic Rent is subject to Rent Adjustment as described under Section 3.2. Basic Rent for any partial month shall be pro-rated accordingly. All other sums which become payable by Lessee under this Lease shall be considered "Additional Rent".

3.1.1 Attorney Fees and Staff Costs to Be Paid by Lessee During Term of Lease

Lessee hereby acknowledges and agrees that, in the event Lessee requests that the Port incur attorney fees and staff costs for any work which is solely the result of Lessee requesting that such work be completed, then Lessee shall bear the responsibility to pay for any and all Port costs associated with such a request made by Lessee. Prior to the commencement of any work, Lessee agrees to sign an Agreement to Pay Costs (see attached example as **Exhibit B**, incorporated by reference). If the deposited funds are exhausted Port Attorney and Port Staff may cease work until such time as the parties agree on an additional amount and Lessee makes such additional nonrefundable, fixed fee deposit for costs. Any such additional deposit(s) shall be paid immediately by Lessee in order for any work to continue.

3.2 Rent Adjustment

Except as provided below, after the initial year, on the first day of each successive Lease Year, subsequent to the Effective Date, the amount of Basic Rent will be adjusted to reflect the effect which inflation has had on the purchasing power of the dollar, but in no event will said rent be less than the Basic Rent paid per month during the immediately preceding Lease term. This adjustment will be based upon the change, if any, from the CONSUMER PRICE INDEX FOR ALL URBAN CONSUMERS (CPI-U) EXCLUDING FOOD AND ENERGY FOR WEST REGION, as published by the U.S. Bureau of Labor during the December of the preceding year to the next Rent adjustment (the "CPI Rate"). CPI Rate increases shall not exceed five percent (5%) per year over the first five (5) years of the Lease.

Every Five (5) years, and in the event the Port and Lessee successfully negotiate an Extension Term as provided in Section 2.3 above, to be effective as of the July 1st preceding each Extension Term (each such date an "Adjustment Date"), monthly Basic Rent may be adjusted by comparable rates provided by the Port. Should no comparable leases be available, the Port may use Fair Market Value, as defined in Section 3.2.1, multiplied by ten percent (10%) divided by twelve (12). After each Extension Term Basic Rent for each successive one-year period will be adjusted to reflect the effect which inflation has had on the purchasing power of the dollar (each such date an "Adjustment Date"), but in no event will said rent be less than the Basic Rent paid per month during the immediately preceding Lease term. This adjustment will be based upon the change, if any, from the CPI Rate of the date nearest the commencement of this Lease, as compared with the same CPI Rate published as of the period preceding the expiration of the preceding year

of this Lease. However, in no event shall Basic Rent for any Extension Term be less than the Basic Rent in effect immediately prior to the Adjustment Date.

3.2.1 Fair Market Value

As used in this Lease, "Fair Market Value" shall mean the most probable sale or lease price in terms of money which the land and improvements should bring in a competitive and open market under the condition's requisite to a fair sale or lease, the buyer/lessee and seller/Port each acting prudently, knowledgeably and assuming the price is not affected by undue stimulus. Implicit in this definition is the consummation of a sale/lease as of the applicable Adjustment Date and the passing of title from seller/Port to buyer/lessee under the conditions whereby: (a) the buyer/lessee and seller/Port are typically motivated; (b) each party is well informed or advised and is acting in what it considers its own best interests; (c) a reasonable time is allowed for exposure in the open market; (d) payment is made in terms of cash in U.S. dollars or in terms of financial arrangements comparable thereto; (e) the price represents the normal consideration for the property sold or leased unaffected by special or creative financing or sales concessions granted by anyone associated with the sale/lease; and (f) the land and improvements are clean and uncontaminated. Fair Market Value further means the value of the land and improvements based on: (a) its use as an industrial property, or if such land and improvements are no longer being used for an industrial use, the then highest and best legal use of such land and improvements; and (b) the value of the land exclusive of any improvements except those installed and paid for solely by the Port and not reimbursed solely by Lessee. Furthermore, sale/lease transactions used in the determination of Fair Market Value shall be of property with comparable infrastructure improvements, including, but not limited to, road and rail access and utilities, common areas and other comparable site amenities.

If Lessee disagrees with the Port's new Basic Rent for any renewal option period, then the Port and Lessee will each have thirty (30) days from the date the Port receives Lessee's notice objecting to the Port's new Basic Rent to select an independent Oregon licensed appraiser to determine the current Fair Market Value. Notwithstanding the foregoing, in no event shall the appraisal process begin until Lessee has given notice to the Port exercising its option to renew according to Section 2.3. The selected appraisers shall be members of the Appraisal Institute ("MAI"), unless otherwise agreed upon, in writing, by the parties, and shall be experts in the appraisal of industrial property in Oregon, with at least five (5) years of commercial appraisal experience, including experience in appraising industrial properties of the nature of the property. Each party must notify the other, within the thirty (30)-day period, of the name, address, and telephone number of such party's selected appraiser. The selected appraisers, within sixty (60) days of being selected, must complete their final appraisal reports, and deliver copies of the appraisal reports concurrently to the Port and to Lessee ("First Appraisal"). If the two appraisers and the parties cannot come to an agreement concerning the Fair Market Value within fifteen (15) days of issuance of both appraisal reports, the two appraisers will select a third MAI appraiser to determine which appraisal comes closest to the Fair Market Value. The appraisers shall conduct a full and independent appraisal of the Property and deliver the report documenting his/her findings concurrently to the Port and to Lessee ("Final Appraisal"). The Final Appraisal will take into consideration all information contained in both the Port's and Lessee's appraisals and the third appraiser's own independent analysis, upon which the third appraiser will then choose, as the Fair Market Value, either the Port's appraisal or Lessee's appraisal, based on which one comes closest

to the third appraiser's independent appraisal. The third appraiser shall not propose a middle ground or any modification of either of the two proposed determinations of Fair Market Value. The determination of Fair Market Value by the third appraiser shall be final and binding upon the parties. All court costs and appraisal costs incurred pursuant to this Section 3.2.1 shall be shared equally by the Port and Lessee.

3.2.2 Effective Date of Adjustment; Payment of Adjustments

In the event the Port and Lessee successfully negotiate an Extension Term as provided in Section 2.3 above, to be effective as of the July 1st preceding each Extension Term (each such date an "Adjustment Date"), monthly Basic Rent shall be adjusted as described in this section 3.2. Lessee acknowledges that adjustments to the Basic Rent will be effective as of the Adjustment Date, even though the new rate may not be able to be calculated until after the Adjustment Date. Lessee agrees to (a) begin paying the increased Basic Rent upon notification by the Port; and (b) pay any difference between the Basic Rent actually paid to the Port after the Adjustment Date and the adjusted Basic Rent due for such period within thirty (30) days of notification by the Port.

3.3 Rent Payments

3.3.1 Due Dates

Lessee shall make payment of the Basic Rent for the first calendar month of the Lease Term commencing on the Lease Effective Date and on or before the first (1st) day of each calendar month thereafter during the Lease Term and any Extension Terms ("Due Date"). If proration of Basic Rent is required under this Lease, Basic Rent shall be prorated on a thirty (30)-day month. All other Additional Rent shall be paid within thirty (30) days of the issuance by the Port of an invoice submitted to Lessee or within the time period otherwise expressly provided in this Lease. Payment of Rent shall be made without demand, notice, offset, abatement, or deduction of any kind, to the Port at the following address or such other address as the Port may later designate as provided herein, or by direct deposit to an account specified by the Port:

> Port of Columbia County PO Box 190 100 E. Street Columbia City, OR 97018

3.4 Delinquency Charge

All Rent and other amounts, including Additional Rent not paid when due shall bear a "Delinquency Charge" of one- and one-half percent per month (18% per year) if the total balance remains unpaid for five (5) or more days following its due date. Such interest shall be charged from the due date until the Rent or other amount is paid in full. This Delinquency Charge is subject to periodic change, subject to any limitation on the maximum rate of interest allowed by law, at the Port's sole discretion. No change shall occur, however, without at least thirty (30) calendar

days prior written notice to Lessee. Imposition of a Delinquency Charge shall not constitute a waiver of any other remedies available to the Port for failure to timely pay Rent.

3.5 Returned Checks

If Lessee's check for payment of Rent due under this Lease is returned to the Port for any reason, the payment shall be considered not to have been made and shall be delinquent. In addition to the Delinquency Charge set forth in Section 3.5, the Port may charge Lessee a returned check fee of Thirty-Five Dollars (\$35.00), per returned check. The return check fee is subject to periodic change by the Port. The Delinquency Charge shall continue to accrue until the returned check fee is paid, the check can be cashed, and the Port receives all funds due.

3.6 Acceptance of Rent

The Port's acceptance of a late or partial payment of Rent and/or a Delinquency Charge shall not constitute a waiver of any Event of Default, nor shall it prevent the Port from exercising any of its other rights and remedies granted to the Port under this Lease or by law. Any endorsements or statements on checks of waiver, compromise, payment in full or any other similar restrictive endorsement shall have no legal effect. Lessee shall remain in violation of this Lease and obligated to pay all Rent due even if the Port has accepted a partial or late payment of Rent.

3.7 Security Deposit

On or before the Effective Date, Lessee shall deposit with the Port the sum equal to twomonths' worth of Basic Rent, one year of property taxes, and one year of insurance premiums as the "Security Deposit". The Security Deposit shall secure Lessee's full and faithful performance and observance of all of Lessee's obligations under this Lease and under any other written agreement between Lessee and the Port. The Security Deposit shall not be considered to be held in trust by the Port for the benefit of Lessee and shall not be considered an advance payment of Rent or a measure of the Port's damages in the case of an Event of Default by Lessee. The Port may, but shall not be obligated to, draw upon and apply the Security Deposit to: (a) pay any Rent or any other sums due to the Port by Lessee and not paid on or before the date it is due and the Port shall not be required to give notice or opportunity to cure before drawing on the Security Deposit; (b) to pay attorney fees and costs for any legal work which has been requested by Lessee as outlined in Section 3.1.3; or (c) to remedy any other Event of Default of this Lease, after Lessee has received notice and opportunity to cure, if such notice and opportunity to cure is required under this Lease. If the Port applies any of the Security Deposit to any of the above, Lessee shall, immediately upon demand, replenish the Security Deposit to its full amount. If Lessee fully performs all of its obligations under this Lease, the Security Deposit, or any balance remaining, will be released within thirty (30) days from the Expiration Date or termination of this Lease and delivery of the Premises to the Port. However, if any question exists concerning Lessee's full compliance with the Lease or if there is any obligation under this Lease to be performed after the Expiration Date or earlier termination of this Lease, the Port shall be entitled to require that the Security Deposit remain in place until the Port is fully satisfied that there has been no Default of the Lease and all obligations due under this Lease have been fully performed. In addition to any other remedy provided in this Lease or at law, the Port shall have the option but not the obligation to use the Security Deposit or a portion thereof to offset any costs or damages incurred as a result of Lessee's failure to perform its obligations at the termination or expiration of the Lease. The

Blue Heron Septic and Drain Services Lease

Security Deposit will be adjusted at every Extension Term to be equal to two (2) months' worth of the then Basic Rent plus one and (1) year of Property Insurance, but in no event less than the most recent adjusted Security Deposit. Notwithstanding the above provisions of this Section 3.7, the Port shall have the right at any time during the Lease Term or any Extension Term to require Lessee to deposit an additional Security Deposit with the Port and/or provide additional financial assurance reasonably acceptable to the Port, in an amount or amounts reasonably determined by the Port to be commensurate with any increased risk associated with any of the following events: (i) as a condition of Port approval, as provided in Section 6.1.2 of Lessee's or any Sublessee's use, storage, handling, processing, manufacturing or recycling of Hazardous Substances not authorized under the first two sentences of Section 6.4.1; (ii) if the Port has given notice of violation of any provision of this Lease more than three (3) times during any consecutive twelve (12) month period; or (iii) upon Lessee's exercise of any Extension Option, to adjust for added risks such as increases in Basic Rent, property taxes, and other Additional Rent pursuant to Section 3.

3.8 Taxes

The Port shall throughout the Lease Term be billed directly for the Taxes by the tax assessing authority. The Port will provide Lessee with an estimated annual amount for such taxes ("Port Property Tax Estimate") and Lessee will reimburse the Port for the cost of such taxes by paying to the Port a monthly amount which reflects the Port Property Tax Estimate divided by 12 ("Monthly Tax Payment)". The Monthly Tax Payment will be made by Lessee, along with any Basic Rent, thereafter during the term of this Lease and will be Additional Rent. Any unpaid amounts may be paid with the Security Deposit at the Port's sole discretion. If Lessee's Monthly Tax Payments exceed the Port Property Tax Estimate at the end of a one-year cycle, each April the Port will apply the overpayment as a credit toward future Monthly Tax Payments. In the event the Lessee fails to pay the Monthly Tax Payment the Port may utilize all remedies as set forth in Section 10.2. Lessee understands that should this Lease expire or be terminated prior to the end of any given fiscal tax year ("Tax Year") that Lessee will nonetheless be responsible to pay the tax assessed for the entire Tax Year. The Tax Year for Columbia County is currently July 1 through June 30. If the Premises are assessed with other Port property as part of a larger tax parcel, the Taxes payable by Lessee for the Premises shall be allocated proportionately based on the land area that the Premises represent to such larger tax parcel and, as to improvements, shall include assessments only for the improvements located on the Premises.

3.9 Non-Remonstrance

Lessee waives any right to remonstrate against and shall support local improvement districts or similar shared funding mechanisms to fund future utility and roadway improvements pertaining to the Premises, should such mechanisms be used. Lessee and the Port shall cooperate in working with the city, county and/or state on the allocation of costs among the benefited properties and Lessee shall promptly sign the necessary waiver forms requested by the city, county and/or state.

4. LESSEE 'S OTHER OBLIGATIONS

4.1 Construction of Improvements

4.1.1 Port Approval and Ownership

Lessee shall undertake no construction, alteration, or changes on or to the Premises, including the improvements, without the prior written approval of the Port. Any construction work must comply with the local permit requirements and completed by a licensed contractor. Lessee shall provide to the Port copies of all associated permits and documentation within 180 days and further provide any other documents as requested by Port. All such renovation work shall be deemed part of the improvements, and Lessee shall have no right, title, or interest in the improvements located on the Premises shall remain property of the Port, unless otherwise previously authorized by the Port.

4.1.2 Wetlands Prevention

Lessee shall not create any "wetlands" (as defined under any federal, state, regional or local jurisdiction) on the Premises during the term of this Lease or extension thereof, or on any adjacent Port-owned or non-Port owned premises. Lessee shall also manage the Premises so that no wetlands are allowed to form on the Premises and so that Lessee's development and use of the Premises does not cause the formation of wetlands on any adjacent Port owned or non-Port-owned Premises. If the Port believes that wetlands are likely to form on the Premises and Lessee has not taken corrective action, the Port shall have the right, but not the obligation, upon ten (10) days prior written notice to Lessee (except in the case of an emergency), to enter onto the Premises to correct the situation and charge Lessee for such work. Lessee shall reimburse the Port for such work within thirty (30) days of receipt of invoice from the Port. Lessee shall cooperate with and perform any obligations required under any wetlands corrective or remediation plan.

4.2 **Ownership of Improvements**

The Port shall be the sole owner of improvements made or paid for by the Port and Lessee shall not take depreciation or any other form of tax deduction, based on any improvements made or paid for by the Port. In the event that (a) Lessee requests and the Port agrees to construct a specific improvement on the Premises for Lessee's use; (b) Lessee reimburses the Port in full for such improvement; and (c) the parties agree in writing prior to such construction that Lessee shall become the owner of the improvement upon the Port's completion and Lessee's reimbursement to the Port for such improvement, then upon completion such improvement shall be considered "Lessee Improvements" owned by Lessee, subject to the Port's reversionary rights as set out in Section 14.2.7.

4.3 Signs

Lessee shall contact the Port in order to coordinate the placement of any signage on the leasehold. Lessee shall not erect, install, nor permit upon the Premises any sign or other advertising device without first obtaining the Port's written consent, which the Port may withhold or condition in its sole discretion. Lessee shall remove all signs and sign hardware upon termination or expiration of this Lease and restore the sign location to its former state unless the Port elects to retain all or any portion of the signage.

4.4 No Liens

Lessee agrees to pay, when due, all sums for labor, services, materials, supplies, utilities, furnishings, machinery, or equipment which have been provided or ordered with Lessee's consent to the Premises. Lessee shall not suffer or permit any liens to attach to all or any part of the Premises by reason of any work, labor, services, or materials done for, or supplied to, or claimed to have been done for or supplied to, Lessee or anyone occupying or holding an interest of Lessee in all or any part of the improvements through or under Lessee. If any lien is filed against the Premises which Lessee wishes to protest, then Lessee shall promptly deposit cash with the Port, or procure a bond acceptable to the Port, in an amount sufficient to cover the cost of removing the lien from the Premises or file a bond or deposit money and cause the Premises to be freed of the lien pursuant to ORS 87.076 et. seq. Notwithstanding any notice and cure periods provided in Section 10, failure to remove the lien or furnish the cash or bond acceptable to the Port or to cause the Premises to be freed of the lien pursuant to ORS 87.076 et. seq. within thirty (30) days of receipt of notice of the lien shall constitute an Event of Default under this Lease and the Port shall automatically have the right, but not the obligation, to pay the lien off with no further notice to Lessee and Lessee shall immediately reimburse the Port for any sums so paid to remove any such lien. Except as provided in Section 12, Lessee shall not encumber the Premises or any Lessee Improvements thereon without prior written approval of the Port, which may be withheld or conditioned in the Port's sole discretion.

4.5 Utilities and Services

The Port makes no representations or warranties regarding the availability and/or quality of utilities and other services available to the Property. Lessee is responsible for verifying, obtaining, and paying for all utility installations and connections and all utility services necessary for Lessee's operation. Lessee shall be responsible for paying all fees and charges imposed by the various utilities furnishing those services, including those paid for and apportioned by the Port (if any). Fees shall include, but are not limited to, all installation and connection fees and charges and monthly or periodic fees and assessments charged by the utilities and by any governmental agencies having jurisdiction over the Premises. Before Lessee installs any other utility lines on, under, or above the Premises, Lessee must first obtain the Port's written approval to do so. Before approval is granted Lessee must provide the Port with a detailed survey, with depiction, showing the exact location of such lines. In no event shall the Port be obligated to use any of its powers to acquire easement or utility access for Lessee.

4.6 Safety Requirements

Lessee shall conduct its operations, activities, and duties under this Lease in a safe manner and shall comply with all safety standards imposed by applicable Laws. Lessee shall ensure compliance with the same by all subcontractors and all other persons transacting business with or for Lessee in any way connected with the conduct of Lessee pursuant to this Lease. Lessee shall comply with any safety rules adopted by the Port after thirty (30) days' notice of such adopted rules to Lessee unless exigent circumstances require immediate compliance therewith.

4.7 Fire Safety

Lessee shall exercise due and reasonable care and caution to prevent and control fire on the Premises and, to that end, shall fully maintain the existing fire suppression system on the Premises if it exists, and shall provide and maintain any other fire protection equipment as may be required or appropriate for Lessee's use of the Premises pursuant to applicable Laws and the Port's fire insurance carrier, for the purpose of protecting the Premises and restricting the spread of any fire from the Premises. The Port may provide a preventative maintenance schedule for the existing system and suitable training to Lessee's personnel upon request. Lessee shall comply with any fire safety rules adopted by the Port after thirty (30) days' notice of such adopted rules to Lessee unless exigent circumstances require immediate compliance therewith.

4.8 Security

Lessee shall be fully responsible for the security of the Premises and the Port shall have no responsibility or liability under this Lease for the security of the Premises. Lessee may, at its sole expense, install and maintain a security system on the Premises (or enhance an existing system on the Premises) with prior written approval of the Port. Upon expiration or termination of this Lease, such security system shall only be removed in accordance with Section 9.1.

4.9 Waste, Removal and Demolition

Lessee shall not cause or permit any waste or damage, disfigurement or injury to the Premises or the improvements. Subject to Section 8, Lessee shall not remove or demolish, in whole or in part, any improvements on the Premises without the prior written approval of the Port, which the Port may condition upon the obligation of Lessee to replace the same by an equal or better Improvement or other Improvement specified in such approval.

4.10 Lessee's Maintenance Obligations

Lessee shall maintain the Premises (including the building's gutters, heating, ventilating, and air conditioning systems, locks, hardware, interior walls, ceiling, and floor coverings, paint, lights, windows, doors, exterior siding, landscaping, parking courtyard, all improvements, Lessee Improvements, systems, utilities, equipment, septic system and drain field, and all other maintenance not specifically required of the Port per Section 3.15) in good condition and shall carry out preventive maintenance, repairs, replacements, and cleaning necessary to maintain such condition, including any repairs or alterations required under Lessee's obligation to comply with any laws or regulations at the sole expense of the Lessee. Lessee shall provide written report to the Port of all maintenance performed on the Premises annually.

4.11 Port's Maintenance Obligations

Port shall keep and maintain the structural elements of the Premises (meaning the roof, load bearing walls, structural members, foundation, common areas, and access to the Premises) in reasonably good condition except for damage caused by the Lessee, Lessee's employees, agents, invitees, or contractors, which damage will be repaired at the Lessee's expense.

4.12 Conduct of Business

At all times during the Lease Term and any Extension Terms, Lessee shall be registered and in good standing to do business in Oregon. Lessee shall notify the Port of any cessation in operations that are expected to last more than four (4) weeks and must continuously secure and maintain the Premises at all times during the Lease Term and any Extension Terms. In the event Lessee's cessation in operations lasts longer than eight (8) weeks, the Port shall have the right, but not the obligation, to terminate this Lease.

5. PORT AUTHORITY AND OBLIGATIONS

5.1 Quiet Enjoyment

Subject to Lessee performing all of Lessee's obligations under this Lease and subject to the Port's rights under this Lease and its rights of condemnation under Oregon law, Lessee's possession and quiet enjoyment of the Premises will otherwise not be disturbed by the Port or its officers, commissioners, agents, employees, and contractors.

5.2 Delivery of Premises

Lessee shall have the right to possession of the Premises on the Effective Date. The Port shall have no liability to Lessee for any delay in delivering possession; however, all of Lessee's monetary obligations hereunder, including without limitation Lessee's obligation to pay Basic Rent, will be abated, delayed, and forever excused until possession is actually delivered to Lessee. Lessee shall have the right to terminate this Lease if possession is not delivered by the Effective Date.

5.3 Condition of Premises

The Port makes no warranties, guarantees or averments regarding the condition of the Premises and the access bridge, including, without limitation, the suitability of the Premises for Lessee's intended uses or the availability of accessible utilities or roadways needed for Lessee's intended purposes, or zoning, development, or occupancy requirements. Lessee confirms that the Port or any agent of the Port has made no representation or warranty as to the Premises or bridge except as provided in this Lease. Lessee has been given the opportunity to inspect the Premises and accepts the Premises in "AS IS" condition, with all defects and conditions known and unknown, upon taking possession. The Port shall have no liability to Lessee and Lessee shall have no claim against the Port for any damage or injury caused by any condition of the Premises, including the presence of hazardous substances on the Premises as determined by the Department of Environmental Quality ("DEQ") in ongoing environmental cleanup efforts since a Consent Order was issued in 1995 under Case No.WMCSR-NWR-95-05 and resulting Record of Decision ("ROD"), Record Number DECP/23/26086, a copy of which is attached as Exhibit C. The efforts to clean up the Premises will continue throughout the term of this Lease. Lessee acknowledges it is aware of this ongoing environmental cleanup and consents to all cleanup activities which will occur on the Premises during the term of this Lease. Unless otherwise agreed to, in writing, by the Port, the Port shall have no responsibility to bring the Premises into compliance with any laws, including, without limitation, any building or occupancy codes. Lessee shall be solely responsible for thoroughly inspecting the Premises and ensuring that it is in compliance with all laws.

5.4 Port Construction and Repair Obligation on Premises

The Port shall have no responsibility for the repair or maintenance of the Premises or for construction of any roadways, utilities, or any other improvements on or off of the Premises unless otherwise specifically provided in this Lease. Should the Port undertake any repair or maintenance work on the Premises, the Port shall have no liability for interference with Lessee's use of the Premises which might result from the Port's repair and maintenance efforts and no such efforts shall be construed as a constructive eviction or other eviction of Lessee. Notwithstanding the foregoing and to the extent allowed by law, the Port will be responsible to Lessee for any actual damages caused by the gross negligence or willful misconduct of the Port or any agent, employee or contractor of the Port who performs work which damages Lessee's Premises (but in no event for lost profits or consequential damages). Any repair of damage caused by negligence or breach of this Lease by Lessee, employees, agents, contractors, or invitees, shall be Lessee's responsibility and shall be made at Lessee's sole expense.

5.5 Port and Third-Party Access

The Port shall have the right to enter upon the Premises for the purposes of: (a) confirming the performance by Lessee of all obligations under this Lease; and (b) doing any other act which the Port may be obligated or has the right to perform under this Lease, including an annual inspection of the Premises. Such entry shall be made with reasonable advance written notice of at least twenty-four (24) hours and during normal business hours to Lessee, pursuant to the notice provisions of Section 16.4 except in cases of emergency. In an emergency, the Port shall have the right to use any and all means which the Port may deem reasonable to obtain entry onto the Premises or into any building located on the Premises without liability to Lessee. Lessee may have its representative, if such a representative is reasonably available, accompany the Port. If prior notice of an inspection is not provided under the circumstances set forth in this Section 5.5, the Port shall notify Lessee of the scope and findings of the inspection as soon as practicable after it occurs.

5.6 Remediation Access

The Port, its contractors and consultants, DEQ, EPA, and any other government agency shall have the right to enter upon the Premises and perform any remediation activities required by law or any government agency. In such an event, the Port shall make reasonable efforts to avoid interference with Lessee's operations on the Premises.

6. ENVIRONMENTAL MANAGEMENT AND COMPLIANCE

6.1 **Definitions**

For the purposes of this Lease, the following definitions shall apply:

6.1.1 Environmental Laws

"Environmental Laws" shall include any and all federal, State of Oregon, regional and local laws, regulations, rules, permit terms, codes, ordinances and guidance documents now or hereafter in effect, as the same may be amended or recodified from time to time, and applicable decisional law, which govern materials, substances, regulated wastes, emissions, pollutants, water,

storm water, ground water, wellfield and wellhead protection, cultural resources protection, animals or plants, noise, or products and relate to the protection of health, natural resources, safety or the environment.

6.1.2 Hazardous Substance

"Hazardous Substance" shall include any and all substances defined or designated as hazardous, toxic, radioactive, dangerous, or regulated wastes or materials or any other similar term in or under any applicable Environmental Laws. Hazardous Substance shall also include, but not be limited to, fuels, petroleum, and petroleum-derived products.

6.1.3 Environmental Audit

"Environmental Audit" means an environmental site assessment or compliance audit conducted of the Premises consistent with CERCLA Section 9601(35)(B), 42 U.S.C., 40 C.F.R. part 312, ORS 465.255(6), and any other applicable or relevant and appropriate assessment or auditing standard, including ASTM Standard E2107-00 Standard Practice for Environmental Regulatory Compliance Audits, or its successor, as the same may be amended or recodified from time to time, and any other environmental assessment, sampling and testing as may be necessary or desirable.

6.1.3.1 Initial Audit

Exhibit C shall serve as the Initial Audit ("Initial Audit") and documents the onsite environmental condition of the Premises as of the Lease Commencement Date. The Port acknowledges that the Lessee shall not bear responsibility for remediating the current environmental conditions, also referenced in Section 5.3, of the Premises described in Exhibit C or arising between the date of such reports and the Lease Commencement Date and represents and warranties to Lessee that it is not aware of the presence of Hazardous Substances on the Premises, a Hazardous Substance Release on the Premises, or a violation of Environmental Laws between such dates. Lessee acknowledges they had the option of conducting an Environmental Audit assessment and work before the Lease Commencement Date.

6.1.3.2 Special Audit

If the Port, at any time during the term of the Lease or any extension thereof, has reason to suspect that there has been a Hazardous Substance Release or there is an imminent threat of a Hazardous Substance Release caused by Lessee, Lessee's employees, agents, contractors, licensees, or invitees (including occurring in connection with their occupancy, possession or use of the Premises), the Port may, after written communication of those reasons to Lessee, without limiting its other rights and remedies, conduct an Environmental Audit of the Premises ("**Special Audit**"). If the Special Audit confirms a Hazardous Substance Release which Lessee, Lessee's employees, agents, contractors, licensees, or invitees caused (including occurring in connection with their occupancy, possession, or use of the Premises), then Lessee will be required to reimburse the Port for the reasonable costs of the Special Audit as an Environmental Cost. If no Material Violation is found, the Port will pay for the Special Audit.

6.1.3.3 Exit Audit

At the expiration or termination of this Lease, the Lessee may conduct, at Lessee's expense, an Environmental Audit ("Exit Audit") of the Premises and related property to determine: (i) the environmental condition of the Premises; (ii) whether any Hazardous Substance Release which Lessee, Lessee's employees, agents, contractors, licensees, or invitees caused (including occurring in connection with their occupancy, possession or use of the Premises) has occurred or exists on or about the Premises; and (iii) whether there is evidence of any violation of Environmental Laws or the environmental provisions of this Lease which Lessee, Lessee's employees, agents, contractors, licensees, or invitees caused (including occurring in connection with their occupancy, possession or use of the Premises). The Exit Audit shall be performed not more than sixty (60) days prior to the scheduled Expiration Date of this Lease and a complete copy of the results of the Exit Audit shall be provided to each party, prior to actual termination. In the event this Lease is terminated prior to the scheduled Expiration Date for any reason, Lessee may complete the Exit Audit within sixty (60) days of such actual termination date of this Lease. The Port shall have the right to approve the scope of and consultant for the Exit Audit, though such approval shall not be unreasonably withheld.

6.2 Environmental Inspection

The Port reserves the right, at any time and from time to time, after reasonable advance written notice to Lessee, of two (2) Business Days (as defined in Section 16.4) or more (except that no notice to Lessee shall be required in the event the Port reasonably believes that there exists a condition constituting an imminent and substantial endangerment) to inspect the Premises and Lessee's operations on and use of the Premises: (i) for the presence of or Lessee's management of Hazardous Substances; (ii) for the purpose of sampling Lessee's storm water discharge; (iii) for compliance with Environmental Laws or the environmental provisions of this Lease; and (iv) to facilitate the Port's environmental management, permitting and analysis related to the Premises or any other premises of the Port. Lessee shall use commercially reasonable efforts to arrange these inspections in a manner that will minimize disruption to Lessee's use of the Premises. Lessee may have its representative, if such a representative is reasonably available; accompany the Port during an inspection authorized by this Section 6.2. The Port shall notify Lessee of the scope and findings of the inspection as soon as practical after it occurs.

6.2.1 Environmental Cost

"Environmental Cost: (" shall include, but is not limited to, costs and damages arising from or relating to: (a) any actual or claimed violation of or noncompliance with any applicable Environmental Laws; (b) claims for damages, response costs, any audit costs, fines, fees or other relief relating to matters addressed in any applicable Environmental Laws; (c) injunctive relief relating to matters addressed in any applicable Environmental Laws; (d) Hazardous Substance Releases; and (e) violations of any environmental provisions of this Lease. "Environmental Cost" as used in this Section 6.2.1 and elsewhere in this Lease shall include but not be limited to: (i) costs of preliminary assessment, evaluation, testing, analysis, remedial investigation, feasibility study, removal, remedial action, disposal, monitoring and maintenance, natural resources injury assessment, restoration and compensation; (ii) the cost of decommissioning and removing any underground or aboveground storage tank(s); (iii) Port, Oregon Department of Environmental Quality ("DEQ"), United States Environmental Protection

Blue Heron Septic and Drain Services Lease

Agency ("**EPA**") and Natural Resource Trustees oversight costs; (iv) fees of attorneys, engineers, consultants, experts, , incurred at, before or after trial, on appeal or petition for review, or in any bankruptcy or administrative proceedings; and (v) diminution of value, loss, or restriction on use of Premises, including diminution of value resulting from the residual risk associated with a risk based cleanup.

6.2.2 Hazardous Substance Release

"Hazardous Substance Release" shall include the spilling, discharge, deposit, injection, dumping, emitting, releasing, leaking, or placing of any Hazardous Substance into the air or into or on any land or waters. However, Hazardous Substance Release excludes a release specifically authorized by a then-current and valid permit or authorization issued under applicable Environmental Laws or a de minimus release of Hazardous Substances on an impervious surface that does not and will not likely, either individually or cumulatively with other releases of Hazardous Substances, come in contact with surface water contained in a river water body, or with groundwater, that is appropriately responded to under Environmental Laws, and that is promptly reported to the Port under Section 6.8.

6.2.3 Best Management Practices

"Best Management Practices" shall mean those environmental or operational standards which are either: (a) applicable to a particular business or industry group as a matter of common and accepted practices; (b) adopted or articulated by any of the following: trade associations or professional associations for the particular business or industry group; the business or industry group's own standard operating procedures; or (c) practices specifically defined or identified for a particular business operation or industry group by regulatory agency guidelines.

6.2.4 UST Facility

"UST Facility" shall include underground storage tanks, underground piping, dispensers, related underground and aboveground structures and equipment, including without limitation spill containment features and oil water separators, and the surrounding area used in connection with the operation, activity, or purpose for which the entire system is designed, including without limitation the fueling of motor vehicles and the containment of Hazardous Substances or other materials.6.2.5 AST Facility

"AST Facility" shall include mobile storage tanks and aboveground storage tanks, aboveground piping, dispensers, related underground and aboveground structures and equipment, including without limitation spill containment features and oil water separators, and the surrounding area used in connection with the operation, activity or purpose for which the entire system is designed, including without limitation the fueling of motor vehicles and the containment of Hazardous Substances or other materials.

6.3 General Environmental Obligations of Lessee

Lessee shall manage and conduct all of its activities on or relating to the Premises: (a) in compliance with all applicable Environmental Laws and the environmental provisions of this Lease; (b) in reasonable cooperation with the Port (but at no significant additional cost to Lessee) in the Port's efforts to comply with applicable laws and regulations; and (c) in compliance with

Best Management Practices applicable to Lessee's use of the Premises. Lessee shall manage and, as appropriate, secure the Premises and Lessee's occupation or use of the Premises so as to prevent any violation of law and regulations by any party. Notwithstanding the foregoing, the Port acknowledges and agrees that Lessee shall not be liable for any environmental conditions on the Premises that violate Environmental Laws or cause the Port or Lessee to incur Environmental Costs which are related to the Premises and not created by Lessee and/or existed at, under, on or adjacent to the Premises prior to Lessee's possession or Lessee's receipt of the Premises by assignment or transfer, and the Port agrees that any such conditions shall be the Port's responsibility.

6.4 Hazardous Substances Use on Premises

Lessee shall not use, store, handle, manage, generate, manufacture, dispose of, recycle, or process on the Premises any Hazardous Substances ("Hazardous Substance Use") except as provided in Sections 6.4.1, 6.4.2, and 6.4.3. Lessee shall maintain Material Safety Data Sheets ("MSDS Materials") for each and every Hazardous Substance Use by Lessee, Lessee's agents, employees, contractors, licensees, invitees, to the extent required under Environmental Laws. In order to ensure that the MSDS Materials are available to the Port in the event of a spill or other emergency, the MSDS Materials shall be kept current at all times and a copy of the MSDS Materials shall be kept in a place known to and easily accessible to the Port. Lessee shall dispose of all Hazardous Substances according to applicable Environmental Laws. Except as specifically allowed by state or federal discharge permits, Lessee shall not dispose of any Hazardous Substance, regardless of the quantity or concentration, into any storm or sanitary sewer systems within the Premises, to the ground, into surface water or groundwater, or on other Premises of the Port. Whenever appropriate, Lessee shall strive to minimize Hazardous Substance Use and identify and use non-hazardous alternatives in Lessee's operations.

6.4.1 Permitted Use of Hazardous Substances

Lessee shall be permitted to store, handle, or use limited quantities of certain Hazardous Substances in the following incidental and limited manner, the storage and use of which shall be in compliance with any Environmental Law and shall be in limited, reasonable, and customary quantities (each such use a "**Permitted Use of Hazardous Substances**"). Such Permitted Use of Hazardous Substances shall include and be limited to the following: (a) Hazardous Substances contained in office, janitorial and landscape supplies and cleaning fluids of types and quantities ordinarily used for maintenance of the Premises; (b) petroleum products and lubricants fully contained in equipment, machinery and vehicles on the Premises; and (c) Hazardous Substances contained in equipment, machinery and materials used in the construction and installation of any improvements on the Premises, excluding any stockpiling of Hazardous Substances in connection with such construction or of Off-Site Waste-Like Materials (defined in Section 6.5) containing any Hazardous Substances.

6.4.2 Conditional Hazardous Substances Use with Port Approval

If Lessee's proposed Hazardous Substance Use falls into one or more of the following categories, such use shall be in the Port's sole discretion: (a) a conditionally exempt small quantity generator or small quantity generator of hazardous waste under the Resources Conservation and Recovery Act ("RCRA"); (b) requires an Operating Permit under either federal

or state Clean Air laws; (c) requires a National Pollution Discharge Elimination System (NPDES) permit; (d) requires response planning or reporting under the Emergency Planning and Community Right-To-Know Act, 42 U.S.C. §§ 11001-11050; (e) materials regulated under the Federal Insecticide, Fungicide, Rodenticide Act (FIFRA); and (f) any other use that creates a material risk to human health or the environment, including, but not limited to, uses that may result in explosion, evacuation, or imminent threat to the waterways, groundwater, land, or air (each such use under items (a) through (f) of this Section 6.4.2 is a "Conditional Hazardous Substance Use").

6.4.3 Above ground and Mobile Hazardous Substances Storage Tanks

Lessee may not install, maintain or operate an AST Facility (including fueling trucks), or use fueling entities based offsite for the storage, transportation, transfer and dispensing on the Premises of Hazardous Substances. Any use of above-ground or mobile storage tanks on the Premises shall be reviewed by the Port on a case-by-case basis and subject to the Port's approval, which may be withheld or conditioned at the Port's sole discretion.

6.4.4 Port Review of Lessee's Proposed Hazardous Substance Use

When seeking the Port's prior approval of a Hazardous Substance Use on the Premises, whether by Lessee, or by an existing Lessee shall submit to the Port the following information: (a) quantities and list of any Hazardous Substances proposed to be stored or used; (b) MSDS Materials on all the Hazardous Substances; (c) a description of the intended Hazardous Substance Use; and (d) a copy of any spill and containment plans, and/or management plans for the Hazardous Substance Use required by any local, regional, state or federal agencies under any applicable law ("Lessee's Submittals"). The Port review of the proposed Hazardous Substance Use shall consist of verifying that it is a Permitted Use of Hazardous Substances under Section 6.4.1 or is acceptable to the Port with conditions, as set forth in Section 6.4.2.

6.4.5 Notice of Port's Review of Hazardous Substance Use

The Port shall notify Lessee of the Port's decision within twenty (20) Business Days of receipt of Lessee's Submittals ("Other HSU Notice") and the rest of the provisions in this Section 6.4.5 shall apply. If the Port requests additional information or the proposed Hazardous Substance Use is complex, in the Port's reasonable opinion, the review period shall be extended by an additional ten (10) Business Days after the requested information has been provided. If the Port, in its reasonable discretion, deems it necessary to hire outside consultants to complete the review process due to the complexity of the proposed Hazardous Substance Use or lack of sufficient information for evaluation of the proposed use, Lessee shall bear the expense of such consultants and the time frame for review may be extended as the Port, in its reasonable discretion, deems necessary due to the complexity of the proposed use. Such Other HSU Notice may be given in accordance with Section 16.4. If the Port fails to give Lessee such Other HSU Notice in writing within the time period, together with any applicable extensions, as provided under this Section 6.4.5, Lessee shall notify the Port in writing and the Port shall have an additional seven (7) days to send Lessee the Port's decision on the proposed Hazardous Substance Use. If the Port fails to provide such Other HSU Notice to Lessee within such additional seven (7) day period, the proposed Hazardous Substance Use shall be deemed acceptable to the Port, provided, however, in no event shall the Port's failure to respond constitute an approval of any term that violates Environmental Laws or other applicable laws and regulations or any provision of this Lease.

Blue Heron Septic and Drain Services Lease
6.4.6 Conditions of Hazardous Substance Use Approval

If Port approval is required under this Lease for a proposed Hazardous Substance Use, the conditions of Port approval for such proposed Hazardous Substance Use may include, but not be limited to, requiring Lessee to do one or more of the following: (a) secure additional or increased pollution control insurance or meet other insurance requirements; (b) accept limits or restrictions on the Hazardous Substance Use to assure compatibility with area uses or with any AST or MST Tank Agreement; (c) prepare and submit for Port approval a Spill Prevention, Control and Countermeasures Plan for the Hazardous Substance Use; and (d) submit to the Port copies of any reports or notifications to any regulatory agency relating to the Hazardous Substance Use prior to or at the beginning of the Hazardous Substance Use and periodically throughout the Lease Term.

6.4.7 Discharge and Treatment of Industrial Wastewater

No industrial wastewater discharge shall be made by Lessee into the ground, any surface waters, other body of water, or any Port-owned conveyance or storage system unless such discharge meets the requirements of all applicable laws and regulations. The Port shall have the right, but not the duty, in its sole discretion, to review and approve or disapprove any industrial wastewater management, treatment or discharge system constructed or modified by on behalf of Lessee on the Premises during the term of this Lease.

6.5 Off-Site Waste-Like Materials

Lessee shall not allow any person to store, manage, treat, deposit, place or dispose of slag, debris, recycled concrete, recycled asphalt, construction or demolition debris, wood waste, dredged material, Hazardous Substance-contaminated material, or other wastes from off of the Premises ("**Off-Site Waste-Like Materials**") on the Premises without the prior written approval of the Port, which approval may be granted, conditioned or denied in the Port's sole discretion In the event that the Port in its sole discretion is not satisfied with the origin and likely contents of such Off-Site Waste-Like Materials, the Port may require Lessee, at Lessee's cost, to test such materials to ensure that, in the Port's sole opinion, such materials contain no unacceptable levels of Hazardous Substances.

6.6 Lessee's Liability

6.6.1 Hazardous Substance Releases

Except as provided in Section 6.6.2, Lessee shall be responsible for any Hazardous Substance Release occurring on the Premises or emanating and originating from the Premises onto or into other properties, or in the air or in adjacent or nearby waterways (including ground water) and any Environmental Cost resulting therefrom, which results from or occurs in connection with Lessee's occupancy, use or possession of the Premises (including acts of uninvited third parties who cause a Hazardous Substance Release to occur on the Premises) (a) occurring at any time during the Lease Term or extension thereof; (b) occurring at any time during any holdover period; or (c) occurring during the Lease Term or extension thereof and continuing after the Lease Term or extension thereof and continuing after the Lease Term or extension thereof and continuing after the Lease Term or extension thereof and continuing after the Lease Term or extension thereof and continuing after the Lease Term or extension thereof and continuing after the Lease Term or extension thereof and continuing after the Lease Term or extension thereof.

6.6.2 Limitation of Lessee's Liability

Notwithstanding anything to the contrary provided in Section 6.6 of this Lease, Lessee shall have no responsibility or liability for any Triggering Environmental Event that Lessee can demonstrate: (a) existed on the Premises prior to the Effective Date of this Lease (except if caused by Lessee or Lessee's agents, employees or contractors); (b) is caused by the Port or its agents, employees or contractors after the Effective Date of this Lease; (c) results exclusively from a Hazardous Substance Release that constitutes a contiguous properties situation, as defined in 42 U.S.C.A. § 9607(q) involving a release that occurred from an off-site location not owned, operated, leased or otherwise controlled by Lessee, and migrated onto the Premises; or (d) first occurred after Lessee's vacation of the Premises. The Port shall be responsible, to the extent required by law, for any Triggering Environmental Event caused by the Port, its agents, employees, or contractors, whether occurring before or after the Effective Date of this Lease.

6.7 Environmental Remediation

6.7.1 Immediate Response

In the event of a violation of applicable Environmental Laws for which Lessee is responsible under this Lease, a violation of an environmental provision of this Lease for which Lessee is responsible under this Lease, a Hazardous Substance Release for which Lessee is responsible under this Lease, or the threat of or reasonable suspicion of a Hazardous Substance Release for which Lessee is responsible under this Lease, Lessee shall immediately undertake and diligently pursue all acts necessary or appropriate to do the following: (a) eliminate the threat or reasonable suspicion of such Hazardous Substance Release; (b) correct the violation of Environmental Laws or of environmental provisions of this Lease; and/or (c) investigate, contain, stop, and accomplish source control (if the source is on the Premises) for, Hazardous Substance Releases to the extent required by Environmental Laws and consistent with the permanent remediation requirements of Section 6.7.2. If the source is off the Premises, Lessee will immediately notify the Port upon becoming aware of the Hazardous Substance Release.

6.7.2 Remediation

Lessee shall promptly undertake all actions necessary or appropriate to ensure that any Hazardous Substance Release for which Lessee is responsible under this Lease is permanently remediated and that any violation of any applicable Environmental Laws for which Lessee is responsible under this Lease or any violation by Lessee of an environmental provision of this Lease is corrected. In addition, with respect to any Hazardous Substance Release for which Lessee is responsible under this Lease, Lessee shall restore the Premises or other affected Premises or water to such condition that: (a) to the extent that the Hazardous Substance Release is reportable to the DEQ, a determination of "No Further Action" has been obtained from the DEQ; (b) no ongoing monitoring of Hazardous Substances on the Premises is required under applicable Environmental Laws; (c) no Hazardous Substances are left in place and capped with either soil or other material (except in compliance with subsection (d)); and (d) there are no restrictions or institutional controls under applicable Environmental Laws on the use or development of the Premises other than those consistent with current and reasonably anticipated future land uses at the Premises and surrounding properties. Nothing in this Section 6.7.2 shall require Lessee to remediate any Hazardous

Blue Heron Septic and Drain Services Lease

Page 28

Substance Release for which Lessee is not responsible under Section 6.6.1 or for which Lessee's liability is limited as provided in Section 6.6.2.

6.7.3 Natural Resources Damages Assessment and Restoration

Lessee shall promptly undertake, at Lessee's sole expense, all actions necessary to ensure that any natural resources damages associated with the Premises for which Lessee is responsible under this Lease, and the violation of Environmental Laws, the environmental provisions of this Lease or any Hazardous Substance Release by Lessee, contractors, invitees, agents, or employees, is investigated, determined, quantified, assessed, and permanently restored and compensated for, such that a "No Further Action" determination is obtained from all natural resources trustees with jurisdiction over the Hazardous Substance Release.

6.7.4 Report to the Port

Within thirty (30) days following completion of any assessment, investigation, feasibility study, containment, removal, or remedial action required by this Lease, Lessee shall provide the Port with a written report outlining, in detail, what has been done and the results thereof.

6.7.5 Port's Approval Rights

Except in the case of an emergency or an agency order requiring immediate action, Lessee shall give the Port advance notice before beginning any investigatory, remediation or removal procedures. The Port shall have the right to approve or disapprove (which approval shall not be unreasonably withheld, conditioned, or delayed) the proposed investigatory, remediation and removal procedures and the company(ies) and/or individuals conducting such procedures which are required by this Lease or by applicable Environmental Laws, whether on the Premises or on any affected Premises or water. The Port will have the right to require Lessee to request oversight from the DEQ or to require Lessee to participate under an agreement in DEQ's voluntary cleanup program regarding any investigatory, containment, remediation, and removal activities and/or require Lessee to seek a formal determination from DEQ of No Further Action.

6.8 Notice

Lessee shall promptly notify the Port upon becoming aware of: (a) a violation or alleged violation of any applicable Environmental Laws related to the Premises or to Lessee's occupation or use of the Premises or any environmental provision of this Lease; and (b) any Hazardous Substance Release on, under or adjacent to the Premises or threat of or reasonable suspicion of any of the same. If Lessee fails to notify the Port of a matter as required in this Section 6.8, and if the Port does not otherwise acquire knowledge of the matter, Lessee shall be liable for any aggravation of a Hazardous Substance Release that could reasonably have been avoided if such notification by Lessee had been provided.

6.9 Split Sampling

Except in the case of an emergency or an agency order requiring immediate action, Lessee shall notify the Port at least forty-eight (48) hours in advance of any proposed sampling associated with a Hazardous Substance Release in order to allow the Port to be present or to collect duplicate (or "split") samples so that the Port may conduct its own analysis. Lessee shall provide the Port

with copies of any sampling results and associated chain-of-custody and quality assurance and quality control information within ten (10) days of request by the Port.

6.10 Port's Right to Perform on Behalf of Lessee

In the event Lessee is in default of any of its obligations under this Section 6 or any Environmental Laws, the Port shall have the right, after written notice to Lessee and a period of thirty (30) Business Days provided for Lessee to cure, to perform such obligations and charge Lessee all resulting Environmental Cost, together with interest at the Delinquency Rate, from the date the expense is incurred by the Port until the date paid. The Port may not commence performance on behalf of Lessee under this Section 6.10 if, during the thirty (30) day period, Lessee promptly begins and diligently pursues to completion the performance of the obligations set forth in the Port's notice. Notwithstanding the foregoing, in the event the Port determines that an emergency exists, and Lessee is either unwilling or unavailable to take immediate and appropriate action, the Port may take commercially reasonable action, and charge Lessee all resulting Environmental Cost from the date the expense is incurred by the Port until the date paid. If such Environmental Cost is not reimbursed within thirty (30) days of written demand, the Delinquency Charge will be imposed retroactive to the date the Environmental Cost was incurred. The Port agrees to make a good faith effort to attempt to notify Lessee before beginning any cleanup operation.

7. INDEMNITY, INSURANCE

7.1 Lessee's General Indemnity; Reimbursement of Damages

Lessee agrees to defend (using legal counsel reasonably acceptable to the Port), indemnify, and hold harmless the Port from and against and reimburse the Port for any and all actual or alleged claims, damages, expenses, costs, fees (including, but not limited to, attorney, accountant, paralegal, expert, and escrow fees), fines, Environmental Cost and/or penalties (collectively "Costs") which may be imposed upon, claimed against or incurred or suffered by the Port and which, in whole or in part, directly or indirectly, arise from the following, unless exclusively resulting from the Port's gross negligence or willful misconduct: (a) any act, omission or negligence of Lessee or its agents, employees, contractors or invitees; (b) any use, occupation, management or control of the Premises by Lessee, or its agents, volunteers, employees, contractors or invitees, whether or not due to Lessee's own act or omission and whether or not the resulting damage occurs on the Premises; (c) any condition created in or on the Premises by Lessee, or its agents, employees, contractors or invitees (other than the Port), including without limitation any accident, injury or damage occurring in or on the Premises after the Lease Commencement Date; and (d) any Triggering Environmental Event for which Lessee is responsible under this Lease, including without limitation any injury or damage occurring therefrom; and (e) any breach, violation or nonperformance of any of Lessee's obligations under this Lease. Lessee shall be responsible for payment of all deductibles or self-insured retentions for all claims against Lessee's policies related to this contract including those made directly by an Additional Insured. For purposes of this Section 7.1, Lessee shall be deemed to include Lessee and Lessee's partners, officers, directors, volunteers, employees, agents, invitees, and contractors, and the Port shall mean the Port, its commissioners, directors, volunteers, agents, and employees. Neither Lessee's partners, officers, directors, employees, and agents nor the Port's commissioners, directors,

Blue Heron Septic and Drain Services Lease

Page 30

volunteers, agents, and employees shall be held personally liable under the provisions of this Section 7.1.

7.2 Insurance Requirements

Insurance requirements set forth below do not in any way limit the amount or scope of liability of the Lessee under this Lease. The amounts listed indicate only the minimum amounts of insurance coverage the Port is willing to accept to help insure full performance of all terms and conditions of this Lease. The Port requires and shall be entitled to any broader coverage and/or higher policy limits maintained by the Lessee. Any and all available insurance proceeds in excess of the specified minimum limits of insurance and coverage shall be made available to the Port. All insurance required by Lessee under this Lease shall meet the following minimum requirements.

7.2.1 Certificates; Notice of Cancellation

On or before the Effective Date and thereafter during the Lease Term (and any extension thereof), Lessee shall provide the Port with current certificates of insurance, including a copy of the additional insured endorsement required in Section 7.2.2, executed by a duly authorized representative of each insurer, evidencing the existence of all insurance policies required under this Section 7. The Port shall receive at least thirty (30) days' written notice prior to cancellation, non-renewal, or material change in any policy required under this Section 7. Insurance must be maintained without any lapse in coverage during the Lease Term. The Port shall also be given copies of Lessee's policies of insurance, upon request. Failure of the Port to demand such certificates or other evidence of full compliance with these insurance requirements or failure of the Port to identify any deficiency or noncompliance with coverage requirements shall not be construed as a waiver of Lessee's obligation to maintain the insurance required by this Lease.

7.2.2 Additional Insured; Separation of Insureds

The Port shall be named as an additional insured in each general liability policy, other than employer's liability. Such insurance shall provide cross liability coverage equivalent to the standard Separation of Insureds clause published by the Insurance Services Offices ("**ISO**") or a successor organization.

7.2.3 Primary Coverage

The required policies shall provide that the coverage is primary and will not seek any contribution from any insurance or self-insurance carried by the Port.

7.2.4 Company Ratings

All policies of insurance must be written by companies having an A.M. Best rating of "A" or better, or equivalent. The Port may, upon thirty (30) days' written notice to Lessee, require Lessee to change any carrier whose rating drops below an "A" rating.

7.3 Required Insurance

At all times during this Lease, Lessee shall provide and maintain the following types of coverage.

7.3.1 General Liability Insurance

Lessee shall maintain an occurrence form commercial general liability policy or policies insuring against liability arising from premises operations, independent contractors, products completed operations, personal injury and advertising injury, and liability assumed under an insured contract (including the tort liability of another assumed in a business contract) occurring on or in any way related to the Premises or occasioned by reason of the operations of Lessee in an amount of not less than TWO MILLION DOLLARS (\$2,000,000) or the Lessee's policy limits, whichever is greater, per occurrence. The Port requires and shall be entitled to any broader coverage and/or higher policy limits maintained by the Licensee. Any and all available insurance proceeds in excess of the specified minimum limits of insurance and coverage shall be made available to the Port.

7.3.2 Property Insurance

The Port shall throughout the Lease Term keep the Premises and all other buildings which are owned by the Port on the Premises insured against loss by fire and other hazards by such policies as determined by the Port. The amount of the insurance shall be equal to the full insurable replacement cost of the building and all other Port Improvements located on the Premises. The policy or policies must insure against all risks of direct physical loss or damage (except the perils of flood and/or earthquake unless required by a Lender), including coverage for debris removal and the enforcement of any applicable requirements requiring the upgrading, demolition, reconstruction, or replacement of any portion of the Property as the result of a covered loss. The Port will provide Lessee with an estimated annual amount for such insurance ("Port Insurance Estimate") and Lessee will reimburse the Port for the cost of such insurance by paying to the Port a monthly amount which reflects the Port Insurance Estimate divided by 12 ("Monthly Insurance Payment)". The Monthly Insurance Payment will be made by Lessee, along with any Basic Rent, thereafter during the term of this Lease and will be Additional Rent. Any unpaid amounts may be paid with the Security Deposit at the Port's sole discretion. If Lessee's Monthly Insurance Payments exceed the Port Insurance Estimate at the end of a one-year cycle, each April the Port will apply the overpayment as a credit toward future Monthly Insurance Payments. Lessee shall bear the expense of any insurance insuring the property of Lessee on the Premises against such risks but shall not be required to insure such Lessee property under this Lease.

7.3.3 Automobile Liability Insurance

In the event that automobiles are used in connection with Lessee's business or operations at the Premises, Lessee shall maintain an automobile liability policy or policies insuring against liability for bodily injury, death, or damage to Premises, including loss of use thereof, and occurring in any way related to the use, loading, or unloading of any of Lessee's automobiles (including owned, hired and nonowner vehicles) on and around the Premises. Coverage shall be in an amount of not less than TWO MILLION DOLLARS (\$2,000,000) or the Lessee's policy limits, whichever is greater, for each accident.

7.3.4 Workers' Compensation Insurance

Lessee shall maintain Workers' Compensation insurance for all of Lessee's employees in accordance with all requirements of Oregon law. Lessee shall also maintain employer's liability coverage in an amount of not less than ONE MILLION DOLLARS (\$1,000,000) or the Lessee's policy limits, whichever is greater, per accident and ONE MILLION DOLLARS (\$1,000,000) or the Lessee's policy limits, whichever is greater, per employee for disease. In lieu of such insurance, Lessee may maintain a self-insurance program meeting the requirements of the State of Oregon and a policy of excess workers' compensation and employer's liability insurance.

7.3.5 Personal Property Insurance

Lessee will be responsible to ensure all Lessee's own Personal Property such as removable decorations, detached floor coverings, signs, blinds, furnishings, Lessee Improvements, betterments, and trade fixtures, which items will not be covered by Port's insurance and for which Port and its insurance carriers will have no liability.

7.3.6 Lessee's Risks

Lessee shall be responsible for obtaining any insurance it deems necessary to cover its own risks, including without limitation: (a) personal property, and/or (b) automobile physical damage and/or theft. In no event shall the Port be liable for any: (i) business interruption or other consequential loss sustained by Lessee; (ii) damage to, or loss of, personal property; or (iii) damage to, or loss of, an automobile, whether or not such loss is insured, even if such loss is caused by the negligence of the Port.

7.4 Waiver of Subrogation

Lessee waives any right of action that it and/or its insurance carrier(s) might have against the Port (including the Port's commissioners, employees, volunteers, and agents) for any loss, cost, damage, or expense (collectively "**Loss**") covered by any property insurance policy or policies maintained or required to be maintained pursuant to this Lease. Each party also waives any right of action it and/or its insurance carrier(s) might have against the other party (including its commissioners, employees, and agents) for any Loss to the extent such Loss is a property loss covered under any applicable automobile liability policy or policies required by this Lease. If any party's applicable insurance policies do not allow the insured to waive the insurer's rights of recovery prior to a Loss, such party shall cause such policies to be endorsed to allow the waivers of subrogation required by this Section 7.4.

7.5 Periodic Review

The Port shall have the right to periodically review, including at the beginning of each Extension Term, the types, minimum coverage, limits, and terms of insurance coverage for consistency with then current types, minimum coverage, limits, and terms of insurance coverage for similar operations. If the Port determines that certain types of insurance are not generally available in the marketplace at reasonable terms and pricing for similar operations, Lessee shall not be required to carry such insurance until such insurance becomes available for similar operations. In the event the Port reasonably determines that such types, minimum coverage, limits,

and/or terms should be changed to be consistent with then current types, minimum coverage, limits, and terms of insurance coverage generally available in the marketplace at reasonable terms and pricing for similar operations, the Port will give Lessee a minimum of thirty (30) days' notice of such determination and Lessee shall modify its coverage to comply with the new insurance requirements of the Port. Lessee shall also provide the Port with proof of such compliance by giving the Port an updated certificate of insurance within fifteen (15) calendar days of Lessee's receipt of such updates. Lessee shall have the right to periodically request the Port to conduct a review of the then current types, minimum coverage, limits, and terms of insurance coverage generally available in the marketplace at reasonable terms and pricing for similar operations have changed. The Port shall reasonably consider such request to determine if Lessee's current coverage comports with what is generally available in the marketplace at reasonable terms and pricing for similar operations have cimports with what is generally available in the marketplace at reasonable terms and pricing for similar operations have comports with what is generally available in the marketplace at reasonable terms and pricing for similar operations have comports with what is generally available in the marketplace at reasonable terms and pricing for similar operations have comports with what is generally available in the marketplace at reasonable terms and pricing for similar operations have comports with what is generally available in the marketplace at reasonable terms and pricing for similar operations have comports with what is generally available in the marketplace at reasonable terms and pricing for similar operations and make changes to the insurance requirements under this Lease accordingly.

7.6 Survival of Indemnities

The indemnity agreements set forth in this Section 7 shall survive the expiration or earlier termination of this Lease and be fully enforceable thereafter.

8. DAMAGE AND DESTRUCTION

8.1 General

Lessee shall promptly notify the Port of damage or destruction of any improvements located on the Premises ("Casualty"). In the event (i) the Premises are made wholly untenantable by fire or other casualty and the Port shall decide not to restore or repair same, (ii) 50% or more of the Building is so damaged by fire or other casualty that the Port shall decide to demolish or not rebuild the same, (iii) the Premises are made partially untenantable during the last year of the Term, or (iv) any mortgagee applies the insurance proceeds to reduce its loan balance and the remaining proceeds, if any, are insufficient to pay for the repair or restoration of the Premises and Building, then, in any of such events, the Port shall have the right to terminate this Lease by notice to Lessee within sixty (60) days after the date of such fire or other casualty and Rent shall be apportioned on a per diem basis and paid to the date of such fire or other casualty. Notwithstanding anything to the contrary herein set forth, the Port shall not be obligated to repair or restore the Premises or the Building (and in such event shall exercise its termination right set forth above) if the damage or destruction is due to an uninsurable casualty, or insurance proceeds are insufficient to pay for such repair or restoration, or if any mortgagee applies proceeds of insurance to reduce its loan balance and the remaining proceeds available to the Port are not sufficient to pay for such repair or restoration. Lessee shall fully cooperate with the Port and the Port's insurance carrier in connection with any Casualty.

a) In the event the Premises are made wholly or partially untenantable, and this Lease is not terminated by the Port under this Section 8, then this Lease shall continue in effect and the Port shall proceed diligently to repair and restore the Premises, subject, however, to (i) reasonable delays for insurance adjustments, and (ii) delays caused by forces beyond the Port's reasonable control. In such an event, the rent shall abate in proportion to the non-usability of the Premises during the period while repairs are in progress. b) The Port shall have no liability for any loss, cost, expense, damage, or compensation whatsoever (including any claim for inconvenience, loss of business or annoyance) by reason of any restoration of the Premises or the Building under this Section 8.

c) If the Port does not elect to terminate this Lease pursuant to this Article, the damage is not the result of Lessee's gross negligence or willful misconduct, the damage substantially interferes with Lessee's access to or usage of a material portion of the Premises, and the Port's restoration work cannot be or is not completed within three hundred sixty-five days (365) days after the date of the damage, then Lessee may elect to terminate this Lease by notifying the Port in writing of such termination upon thirty (30) days' notice.

d) In the event of a casualty in which the Lessee Improvements are damaged or destroyed, Lessee shall proceed diligently to repair and restore such Lessee Improvements to the extent insurance proceeds are available or would have been available had Lessee carried all insurance required to be carried by Lessee under this Lease but did not so carry.

8.2 Restoration

In the event of any Casualty, Lessee shall cooperate with the Port in the restoration and/or replacement of the damaged improvements and damage to the Premises. If Lessee intends not to operate on the Premises for any period of time following a Casualty, Lessee must take appropriate measures to secure the Premises during any period of such non-use and to prevent unauthorized parking or other unauthorized uses on the Premises.

8.3 Termination of Lease Following Casualty

If the building should be totally destroyed by fire, tornado, or other casualty, or if it should be so damaged thereby that rebuilding or repairs cannot in the Port's estimation be completed within three hundred and sixty five (365) days after the date upon which the Port is notified by Lessee of such damage, this Lease shall terminate and the Basic Rent shall be abated during the unexpired portion of this Lease, effective upon the date of the occurrence of such damage. The Port shall give notice to Lessee in writing of its determination to terminate this Lease within ninety (90) days following the date of the occurrence of such damage. If the building should be damaged by any peril covered by the insurance to be provided by the Port (but only to such extent the rebuilding or repairs can in the Port's estimation be completed within two hundred (200) days after the date upon which the Port is notified by Lessee of such damage), this Lease shall not terminate, and the Port shall at its sole cost and expense thereupon proceed with reasonable diligence to rebuild and repair such building to substantially the condition in which it existed prior to such damage to the extent that insurance proceeds are available to rebuild, except that the Port shall not be required to rebuild, repair or replace any part of the partition, fixtures, additions and other improvements which may have been placed in, or about the Premises by Lessee. If the Premises are untenantable in whole or in part following such damage, the Basic Rent payable hereunder during the period in which they are untenantable shall be reduced to such extent as may be fair and reasonable under all of the circumstances. In the event that the Port shall fail to complete such repairs and rebuilding within two hundred (200) days after the date upon which the Port is notified by Lessee of such damage, Lessee may at its option terminate this Lease by delivering written notice of termination to the Port as Lessee's exclusive remedy.

9. TERMINATION

9.1 **Duties on Termination**

Upon the Expiration Date or earlier termination of this Lease for any reason, Lessee shall complete all of Lessee's obligations under this Lease, including, but not limited to, meeting all of Lessee's environmental requirements under Section 6 (including without limitation the payment of all Environmental Costs for which Lessee is responsible under this Lease), delivering to the Port all keys and all manuals, warranties and other information in Lessee's possession relating to any of the improvements and surrendering the Premises and all improvements in good clean condition, reasonable wear and tear excepted (subject to the requirement to remove the improvements as set forth in this Lease). All Lessee Improvements shall, at the Port's sole option, be deemed Port Premises (including any buildings constructed on the Premises) and shall not be removed unless the Port directs Lessee to remove such Lessee Improvements, in which case Lessee shall remove the Lessee Improvements and then restore the Premises. All repairs or removals for which Lessee is responsible shall be completed at Lessee's cost prior to termination of this Lease and surrender of the Premises. Except in the event of early termination pursuant to the terms of this Lease, the Port shall give Lessee written notice not later than sixty (60) days prior to the end of the initial Lease Term or the then current Extension Term ("Removal Notice") of the Port's election to have Lessee remove the Lessee Improvements. If the Port fails to timely provide such Removal Notice, such failure shall not be deemed a Default under this Lease and Lessee's obligation to remove the Lessee Improvements, if desired by the Port, shall not be extinguished; however, the date upon which the Lessee Improvements must be removed and any damage to the Premises repaired by Lessee shall be extended by the number of days the Removal Notice was late. If the Lessee Improvements are damaged after the Port has indicated in its Removal Notice that the improvements could remain, the Port may elect, in its sole discretion, to require Lessee to remove the Lessee Improvements and restore the Premises.

10. DEFAULT

10.1 Event of Default

The occurrence of any of the following shall constitute an "**Event of Default**" (also referred to as a "**Default**").

10.1.1 Default of Monetary Obligations

An Event of Default shall occur if Lessee fails to pay any monetary obligation required under this Lease and such failure to pay is not cured within ten (10) days after written notice to Lessee by the Port, except that a failure to pay Basic Rent must be cured within ten (10) days of its due date without notice from the Port.

10.1.2 Default in Covenants

An Event of Default shall occur if Lessee violates any term, covenant, or condition of this Lease and such violation is not cured within thirty (30) days after written notice to Lessee by the Port describing the nature of the violation. If the violation is of such a nature that it cannot be completely cured within the thirty (30) day period, this provision shall be complied with if Lessee begins diligent correction of the violation within such thirty (30) day period and, thereafter,

Blue Heron Septic and Drain Services Lease

Page 36

proceeds in good faith and with all due diligence to effect the cure as soon as reasonably possible. If Lessee violates the same provision of this Lease more than three (3) times in any calendar year, then the Port may declare an immediate Event of Default for which no opportunity to cure needs be given, except as provided in Section 0.

10.1.3 Bankruptcy/Insolvency

The Insolvency of Lessee shall be an immediate Event of Default for which no notice or opportunity to cure need be given. For the purposes of this Lease, and to the extent permitted by the United States Bankruptcy Code, "**Insolvency**" shall mean: (a) an assignment by Lessee for the benefit of creditors; (b) the filing by Lessee of a voluntary petition in bankruptcy; (c) dissolution of Lessee; (d) the appointment of a receiver of the properties of Lessee and failure of Lessee to secure discharge of the receiver within ninety (90) days; (e) the filing of an involuntary petition of bankruptcy and failure of Lessee to secure a dismissal of the petition within ninety (90) days after filing; and (f) attachment of, or the levying of execution on, the leasehold interest and failure of Lessee to secure discharge of the attachment, or release of the levy of execution, within ninety (90) days.

10.2 Remedies on Event of Default

Immediately upon the occurrence of an Event of Default, the Port may, at its option, exercise any of the following rights and remedies, in addition to any other rights and remedies provided elsewhere in this Lease or otherwise at law or in equity.

10.2.1 Termination of Lease

The Port may terminate this Lease and Lessee's right to possession of the Premises. However, the Port shall provide Lessee with an additional thirty (30) days' prior written notice of the Port's intention to terminate this Lease for an Event of Default, during which time period Lessee shall have another opportunity to fully cure the Default and avoid termination even though an Event of Default has already occurred. If Lessee fails to fully cure the Default within the thirty (30) day notice period, provided under this Section 0, the Port may automatically terminate this Lease and Lessee's right to possession of the Premises with no additional notice. No other extensions of time will be granted, and no partial cure shall be accepted.

10.2.2 Reentry

Either with or without terminating this Lease or accepting surrender, the Port may reenter the Premises, or any part thereof, by suitable action or proceeding at law, or as otherwise permitted by applicable law, without being liable for indictment, prosecution, or damages therefor, and may repossess the Premises and remove any person or Premises therefrom, to the end that the Port may have, hold, and enjoy the Premises.

10.2.3 Reletting

Following such reentry either with or without acceptance of surrender, the Port may relet the whole or any part of the Premises from time to time, either in the name of the Port or otherwise, to such Lessees, for such terms ending before, on or after the Expiration Date of this Lease and upon such conditions as the Port may determine to be appropriate. Acts of maintenance, preservation, or efforts to relet the Premises, or the appointment of a receiver upon initiative of the Port to protect the Port's interest under this Lease, shall not constitute a termination of the Lease or an acceptance of surrender of the Lease. The Port, at its option, may make such physical changes to the Premises as the Port, in its discretion, considers advisable and necessary in connection with any such reletting, without relieving Lessee of any liability under this Lease or otherwise affecting Lessee's liability.

10.2.4 Right to Sue More than Once

The Port may sue periodically to recover damages and no action for damages shall bar a later action for damages subsequently accruing.

10.3 No Waiver of Default

No failure by the Port to insist on the strict performance of any agreement, term, covenant, or condition of this Lease or to exercise any right or remedy consequent upon a breach, and no acceptance of partial Rent during the continuance of any such breach, constitutes a waiver of any such breach or of such agreement, term, covenant, or condition. No agreement, term, covenant, or condition to be performed or complied with by Lessee, and no breach by Lessee, shall be waived, altered, or modified, except by a written instrument executed by the Port. No waiver of any breach shall affect or alter this Lease, but each and every agreement, term, covenant, and condition of this Lease shall continue in full force and effect with respect to any other then existing or subsequent breach.

10.4 Remedies Cumulative and Nonexclusive

Each right and remedy in this Lease will be cumulative and will be in addition to every other right or remedy in this Lease, or existing at law or in equity, including, without limitation, suits for injunctive relief and specific performance. The exercise or beginning of the exercise by the Port of any such rights or remedies will not preclude the simultaneous or later exercise by the Port of any other such rights or remedies. All such rights and remedies are nonexclusive.

10.5 Curing Lessee's Default

If Lessee fails to perform any of Lessee's obligations under this Lease, and such failure becomes an Event of Default, the Port, without waiving such failure, may (but shall not be obligated to) perform the same for the account of and at the expense of Lessee, with whatever notice is reasonably possible, if any, in a case of emergency, and in any other cases, only if such failure to begin cure or diligently pursue cure continues after the expiration of thirty (30) days from the date the Port gives Lessee notice of the failure. The Port shall not be liable to Lessee for any claim for damages resulting from such action by the Port except to the extent caused by the gross negligence or willful misconduct of the Port, its agents, employees, and contractors (but in no event for lost profits or consequential damages). Lessee agrees to reimburse the Port, upon demand, any amounts the Port spends in curing Lessee's Default.

10.6 Administrative Costs

If the Port gives Lessee one (1) written notice of a violation of a specific provision of this Lease and Lessee violates the same provision again during any twelve (12) month period of the Lease Term, in addition to all other rights and remedies set forth herein, Lessee agrees to reimburse

the Port for its reasonable administrative costs incurred as a result of any notice of any subsequent violation. Failure by Lessee to pay such costs shall be deemed an Event of Default.

10.7 Default by Port

In the event of any Default by the Port, prior to being entitled to maintain any action, Lessee shall give the Port written notice specifying such Default with particularity, and the Port shall have thirty (30) days within which to cure any such Default, or if such Default is nonmonetary and cannot reasonably be cured within thirty (30) days, the Port shall then be deemed in compliance with this provision so long as it begins the cure within the thirty (30) day period and diligently pursues the cure to completion. Unless and until the Port fails to so cure such Default after such notice, Lessee shall not have any remedy or cause of action by reason thereof. All obligations of the Port hereunder shall be construed as covenants, not conditions, and all such obligations shall be binding upon the Port only during the period of its ownership of the Premises and not thereafter.

11. TRANSFER BY THE PORT

At any time after the Effective Date of this Lease, the Port shall have the right to transfer its interest in the Premises or in this Lease. In the event of such a transfer, Lessee shall attorn to said transferee and recognize transferee as the new lessor under the Lease. Thereafter, the Port shall be relieved, upon notification to Lessee of the name and address of the Port's successor, of any obligations accruing from and after the date of the transfer so long as the transferee agrees to assume all obligations of the Port under this Lease.

12. ASSIGNMENT OF INTEREST OF RIGHTS

Lessee shall not sell, assign, mortgage or otherwise transfer its interest in this Lease, the improvements, the Lessee Improvements, and or/the Premises, either voluntarily or by operation of law, without the prior written consent of the Port, in its sole discretion. Any such transfer or assignment, or attempted assignment, subletting or attempted subletting, without such consent, shall be absolutely null and void and shall, at the option of the Port, terminate all rights of the Lessee under or by virtue of this Lease. Lessee's interests, in whole or in part, cannot be taken by operation of law, nor under or by virtue of any execution or legal process, attachment, or proceeding instituted against the Lessee, or under or by virtue of any bankruptcy or insolvency proceedings had in regard to the Lessee, or in any other manner except as above mentioned. If the Port, in its sole discretion, consents to an assignment or sublease, the following conditions must be satisfied: a) A fee of paid to cover administrative costs; b) Lessee has satisfied all conditions of the Lease precedent to assignment or sublease; c) Lessee and its assignee have completed a standard assignment form provided by the Port and assured the Port that the assignee has the capacity to perform on the Lease; and d) The Port has determined that such assignment or sublease is in the best interest of the Port. Notwithstanding the foregoing, the preceding restrictions on assignments and transfers shall not apply to: (i) the offering, sale or transfer of any of Lessee's shares through or on any public securities market or exchange; (ii) an assignment of this Lease to any entity which is a successor to Lessee either by merger or other consolidation of Lessee; (iii) a pledge by any direct or indirect parent of Lessee of its interests in the Lessee; and (iv) a sale of all or substantially all the assets or shares of Lessee or its direct or indirect parent or affiliate.

13. PORT CONTROL AND SERVICES

It is understood that the Port, as a municipal corporation created as a Port District under the laws of the State of Oregon, has certain legislative control and authority over all of its properties and within its geographical boundaries. None of such legislative authority is hereby ceded, delegated, or diminished. It is expressly recognized by the parties that the Port shall have authority to determine rules regulating the use and conduct upon such Premises, penalties therefore and other matters not involving the direct management and operation of the facility. Nothing herein contained shall be deemed to constitute the Lessee an agent, employee, or other representative of the Port; the status of Lessee with respect to the Port is strictly that of independent contractor and the Port reserves no power to control or direct Lessee in the manner of performance of its duties and obligation, other than through its legislative authority and its right to enforce the provisions of this Lease.

14. CONDEMNATION

14.1 General

In the event any public or private entity having the power of eminent domain exercises its right or power of eminent domain, the parties' rights and obligations shall be governed by the common law of the State of Oregon, the applicable Oregon statutes, and this Lease. To the extent allowed by law, in the event of conflict among the statute, common law and this Lease, this Lease shall prevail. If the condemning authority is the Port, Lessee agrees that it will not contest the Port's right to condemn the Premises, or any portion thereof, for whatever purpose the Port deems necessary, but Lessee may contest the value payable to Lessee due to such condemnation. In cases where the condemning authority is an entity other than the Port, Lessee and the Port agree to cooperate, in good faith, to resolve any issues relating to value and/or authority to condemn.

14.2 Definitions

As used in this Lease, the following definitions shall apply:

14.2.1 Partial Taking

"**Partial Taking**" means the taking by right of eminent domain or other authority of law, or a voluntary transfer under the threat of the exercise of the right of eminent domain or other authority, of any portion of the Premises which does not constitute a Total Taking and which would allow Lessee to continue the intended use of the Premises.

14.2.2 Total Taking

"Total Taking" means the taking by right of eminent domain or other authority of law, or a voluntary transfer under the threat of the exercise of the right of eminent domain or other authority, of so much of the Premises as is necessary for Lessee's occupancy that the Premises, after the taking, is no longer suitable for Lessee's intended use.

14.2.3 Taking

"Taking" refers to either a Partial Taking or a Total Taking.

14.2.4 Date of Taking

"Date of Taking" means: (i) the date on which the condemning authority takes legal ownership or actual physical possession of the Premises being condemned; (ii) the date given in a written notice from the condemning authority as the date that it is deemed to have taken possession or is granted possession by a court; or (iii) such date as is agreed to, in writing, in the event of a sale in lieu of condemnation.

14.2.5 Premises Award

The "**Premises Award**" shall mean the fair market value of the Premises (as defined in Section 14.2.5), including the Premises and all infrastructure improvements made to the Premises by any party and including the Port's Reversionary Interest in Lessee Improvements (defined in Section 14.2.7, but excluding the Leasehold Award, described in Section 14.2.6, which Lessee shall be entitled to claim. The Premises Award shall also include any consequential damage to any part of the Premises that may not be taken. The Premises Award is payable solely to the Port.

14.2.6 Leasehold Award

The "Leasehold Award" shall mean the sum of: (i) the present value, determined as of the Date of Taking, of Lessee's Improvements; (ii) the present value, determined as of the Date of Taking, of Lessee's interest in the leasehold estate, including any claim for reimbursement of any prepaid Rent including prepaid Basic Rent; and (iii) provided the Lease is not terminated, any consequential damages such as the cost of any alterations, modifications, or repairs which may be reasonably required by Lessee in order to place the remaining portion of the Premises not taken in a suitable condition for the continuance of Lessee's occupancy. The Leasehold Award is payable solely to Lessee. No portion of the value attributable to the Premises or improvements owned, made by, or paid for by the Port or other government authority, or any infrastructure improvements, or the Port's residual interest in other improvements, shall be considered a part of the Leasehold Award.

14.2.7 Port's Reversionary Interest

The "**Port's Reversionary Interest**" shall mean the present value, determined as of the Date of Taking, of any right of the Port under this Lease to receive the taken Lessee Improvements at the end of the Lease Term.

14.2.8 Distribution of Takings Award

The Port shall have the right to and shall be entitled to receive directly from the condemning authority, in its entirety and not subject to any trust, the entire Premises Award. Lessee shall have the right to and shall be entitled to receive directly from the condemning authority, in its entirety and not subject to any trust, the entire Leasehold Award. It is the intent of the parties that the Premises Award and the Leasehold Award will equal the total amount of the award paid as a result of a Taking of the Premises.

14.3 Total Taking

If a Total Taking occurs during the Lease Term, this Lease will terminate as of the Date of Taking.

14.4 Partial Taking

If a Partial Taking occurs during the Lease Term, this Lease shall terminate as of the Date of Taking as to the portion of the Premises taken. The Lease will continue in full force and effect as to the remainder of the Premises unless Lessee reasonably determines that the remaining portion of the Premises is inadequate for Lessee's purposes, in which event Lessee may elect to terminate the Lease within thirty (30) days of the date of the Taking by providing the Port written notice of termination, to be given in accordance with Section 16.4. If notice of termination is not received by the Port within thirty (30) days of the Partial Taking, the Lease will continue in full force and affect as to the remainder of the Premises. Unless the Taking is by the Port, there shall be no refund of any prepaid Basic Rent by the Port to Lessee, but Lessee may make a claim for such prepaid Rent against the condemning authority, as part of the Leasehold Award. Lessee shall promptly make, at Lessee's sole expense, all necessary repairs, or alterations to restore the remaining Premises after a Partial Taking.

14.5 Claims Against Condemning Authority

In any case where the Port is not the condemning authority, Lessee and the Port agree to work together, in good faith, in making their respective claims against the condemning authority, in accordance with the provisions of this Section 14. The Port also agrees to work together, in good faith, with any of Lessee's Permitted Leasehold Mortgagees, in making their respective claims against the condemning authority, in accordance with the provisions of this Section 14. Each party shall be responsible for making its own claim for court costs and attorney fees incurred in the condemnation proceedings.

14.6 Adjustment of Award

To the extent that the court does not distribute the Takings award in accordance with the distribution rights set forth in this Section 14, the parties agree, upon receipt of the award, to promptly pay to the other any amount of the award belonging to the other in accordance with the distribution set forth in this Section 14. If the parties cannot agree upon the distribution within twenty (20) days of the date the judgment or decree is entered in the condemnation proceedings, the controversy shall be resolved in the same court as the condemnation action was brought. The cost of resolving any such controversy, including the prevailing party's attorney fees, shall be paid by the losing party, as determined by the court. The award allocation may also be resolved by mediation or arbitration if the Port and Lessee both agree to submit the issue to mediation or arbitration.

14.7 Effect of Termination

If this Lease is terminated pursuant to the provisions of this Section 14, then all charges payable by Lessee to the Port under this Lease, if applicable, will be paid up to the Date of Taking. In the event of termination as a result of a Total Taking, the Port and Lessee will then be released from all further liability under this Lease or, in the event of a Partial Taking, from all further liability under this Lease with respect to the portion of the Premises so taken, except such liability which survives termination.

14.8 Notice of Taking

Either party receiving any notice of intended taking, any service of legal process relating to condemnation, or any other notification in connection with any taking, condemnation, or sale or transfer in lieu of condemnation, shall promptly give the other party notice of such receipt. No sale, transfer, agreement, or settlement with the condemning authority shall be made without the written consent of the Port and Lessee.

15. PORT CONDEMNATION RIGHTS

Nothing in this Lease shall in any way limit the powers and rights of the Port to exercise its governmental rights and powers, including its powers of condemnation and eminent domain. Lessee hereby agrees to waive any right it may have to contest the Port's right or authority to condemn, or its rights of condemnation based on a qualifying public purpose. Lessee agrees that it shall limit any contest with the Port relating to the Port's condemnation/eminent domain rights or authority only to the issue of Lessee's interest in the value of the Premises being condemned. Lessee hereby agrees that the Port will have the full right and authority to condemn this Leasehold interest as long as Lessee is paid for its value as provided in Section 14.

16. GENERAL PROVISIONS

16.1 Covenants, Conditions, and Restrictions

This Lease is subject and subordinate to the effect of any covenants, conditions, restrictions, easements, mortgages, deeds of trust, rights of way, and any other matters of record imposed upon the Premises and to any applicable land use or zoning laws or regulations. This Lease is also subject to the Port's right to grant, in the future, underground utility easements through the portions of the Premises not covered by buildings or other structures. Lessee shall, upon request of the Port, execute and deliver agreements of subordination in the form requested by the Port respecting matters covered by this Section 16.1. The Port agrees to make reasonable attempts to minimize any disturbance to Lessee that may result from installation of any underground utility lines and will be responsible for prompt restoration of the Premises.

16.2 Governing Law

This Lease shall be governed and construed according to the laws of the State of Oregon, without regard to its choice of law provisions. Venue shall be in Columbia County.

16.3 No Implied Warranty

In no event shall any consent, approval, acquiescence, or authorization by the Port be deemed a warranty, representation, or covenant by the Port that the matter approved, consented to, acquiesced in, or authorized is appropriate, suitable, practical, safe or in compliance with any applicable law or this Lease. In no event shall the Port be deemed liable, therefore. Lessee shall be solely responsible for such matters.

16.4 Notices

All notices required or desired to be given under this Lease shall be in writing and may be delivered by email, hand delivery, by facsimile but only in those instances specifically called out

in this Lease where notice by facsimile is allowed, or by placement in the U.S. mail, postage prepaid, as certified mail, return receipt requested, addressed to the Port at:

The Port of Columbia County if by hand delivery: 100 E Street, Columbia City, OR 97018 if by US (United States) mail: P.O. Box 190, Columbia City, OR 97018

and to Lessee at:

Blue Heron Septic & Drain Services 293 Sunset Blvd, St Helens, OR 97051

Any notice delivered by hand delivery shall be conclusively deemed received by the addressee upon actual delivery; any notice delivered by facsimile shall be conclusively deemed received by the addressee upon receipt of a confirmation of facsimile by the party sending the notice; any notice delivered by certified mail as set forth herein shall be conclusively deemed received by the addressee on the third Business Day after deposit in the United States mail; any notice delivered by email shall be conclusively deemed received by the addressee upon receipt of a confirmation email by the party sending the notice. The addresses and facsimile numbers to which notices are to be delivered may be changed by giving notice of such change in accordance with this notice provision.

16.5 Time of the Essence

Time is of the essence in the performance of and adherence to each and every covenant and condition of this Lease.

16.6 Non-Waiver

Waiver by the Port or Lessee of strict performance of any provision of this Lease shall not be deemed a waiver of or prejudice the Port's or Lessee's right to require strict performance of the same provision in the future or of any other provision.

16.7 Survival

Any covenant or condition (including, but not limited to, indemnification agreements), set forth in this Lease, the full performance of which is not specifically required prior to the expiration or earlier termination of this Lease, and any covenant or condition which by its terms is to survive, shall survive the expiration or earlier termination of this Lease and shall remain fully enforceable thereafter.

16.8 Partial Invalidity

If any provision of this Lease is held to be invalid or unenforceable, the remainder of this Lease, or the application of such provision to persons or circumstances other than those to which it is held invalid or unenforceable, shall not be affected thereby, and each provision of this Lease shall be valid and enforceable to the fullest extent permitted by law.

16.9 Limitation on Liability

Notwithstanding anything to the contrary contained herein, the parties acknowledge that it is in their mutual interests to limit their liability to each other, and expressly bargain for and agree to the following:

(a) The Port shall have no liability to Lessee for loss, damage or injury suffered by Lessee on account of theft or any act or omission of any third party (other than Port commissioners, employees, agents, and contractors), including other lessees. References in this Section 16.9 (b) to the Port's agents shall be limited solely to those agents acting within the authority of their agency relationship with the Port, and references in this Section 16.9 (c) to Port contractors shall be limited solely to those contractors acting on the Premises and within the express scope of their contract with the Port.

(b) Although this Lease gives the Port certain rights of inspection, such rights shall impose no obligation on the Port to make any inspections, nor impose liability on the Port if the Port fails to make such inspections, or makes inspection, but fails to disclose or require correction of any defect.

(c) With respect to any liability of the Port under this Lease, the Port shall only be liable for any injury or damage to Lessee to the extent of actual damages and only to the extent allowed by law and shall not be liable for consequential damages (including without limitation lost profits).

(d) With respect to any liability of Lessee under this Lease, Lessee shall only be liable for any injury or damage to the Port to the extent of actual damages and only to the extent allowed by law and shall not be liable for consequential damages (including without limitation lost profits).

(e) With respect to claims made by entities or persons who are not parties to this Lease, the Port and Lessee, as between them, shall retain all rights to common law indemnity and statutory contribution available under applicable Oregon law. This Section 16.9 is not intended to confer any benefit or right to any entity or person who is not a party to this Lease.

16.10 Headings, Table of Contents and Table of Definitions

The article and section headings, table of contents and table of definitions contained in this Lease are for convenience in reference and are not intended to define or limit the scope of any provision of this Lease.

16.11 Exhibits Incorporated by Reference

All Exhibits attached to this Lease are incorporated by reference in this Lease for all purposes.

16.12 Modification

This Lease may not be modified except by a writing signed by the Port and Lessee.

16.13 Provisions Applicable to Others

All provisions of this Lease governing Lessee's use of the Premises and Lessee's activities and conduct on, about or from the Premises shall apply to Lessee's officers, agents, employees, invitees, contractors, and assignees.

16.14 Interpretation of Lease; Status of Parties

This Lease is the result of arms-length negotiations between the Port and Lessee and shall not be construed against either the Port or Lessee by reason of such party's preparation of or revisions to this Lease. Nothing contained in this Lease, including the construction of improvements on the Premises, shall be deemed or construed as creating the relationship of principal and agent, partners, joint venturers, or any other similar such relationship, between the parties hereto. The Port and Lessee shall continue an open dialogue concerning any major issues regarding the terms of this Lease.

16.15 Calculation of Time

All periods of time referred to in this Lease shall include Saturdays, Sundays, and Legal Holidays. However, if the last day of any period falls on a Saturday, Sunday, or Legal Holiday, then the period shall be extended to include the next day which is not a Saturday, Sunday, or Legal Holiday. "Legal Holiday" shall mean any holiday observed by the Federal Government. As used in this Lease, "Business Days" shall exclude Saturdays, Sundays, Legal Holidays and the week between December 25 and January 1. Unless specifically identified as Business Days, all references to days shall be calendar days.

16.16 Absence of Brokers

Lessee and the Port each represent to one another that no commission is owed to any agent or broker in connection with this Lease. If a commission is claimed, the party who the broker claims requested the representation shall pay any commission owed and shall defend, indemnify, and hold the other party harmless from any such claim.

16.17 Attorney Fees

If a suit, action, or other proceeding of any nature whatsoever (including any proceeding under the U.S. Bankruptcy Code), is instituted in connection with any controversy arising out of this Lease or to interpret or enforce any rights or obligations hereunder, the prevailing party shall be entitled to recover attorney, paralegal, accountant, and other expert fees and all other fees, costs, and expenses actually incurred and reasonably necessary in connection therewith, as determined by the court at trial or on any appeal or review, in addition to all other amounts provided by law. Payment of all such fees shall also apply to any administrative proceeding, trial, and/or any appeal or petition for review. Whenever this Lease requires Lessee to defend the Port, it is agreed that such defense shall be by legal counsel acceptable to the Port.

16.18 Right of Parties and Successors in Interest

The rights, liabilities and remedies provided for herein shall extend to the heirs, legal representatives, successors and, as far as the terms of this Lease permit, successors and assigns of the parties hereto. The words 'Port' and 'Lessee' and their accompanying verbs or pronouns,

wherever used in this Lease, shall apply equally to all persons, firms, or corporations which may be or become such parties hereto.

16.19 Joint and Several Obligations

The named Lessee consists of one entity. If, in the future, Lessee consists of more than one individual or entity, the obligations of all such individuals and entities shall be joint and several.

16.20 Defined Terms

Capitalized terms shall have the meanings given them in the text of this Lease.

16.21 Execution of Multiple Counterparts

This Lease may be executed in two or more counterparts, each of which shall be an original, but all of which shall constitute one instrument.

16.22 Estoppel Certificates

Each party (the "**Certifying Party**") agrees to execute and deliver to the other (the "**Requesting Party**"), at any time within thirty (30) days after written request, a statement certifying, among other things: (a) that this Lease is unmodified and is in full force and effect (or, if there have been modifications, stating the modifications); (b) the dates to which Rent has been paid; (c) to the best of the Certifying Party's knowledge, whether or not the Requesting Party is in Default in performance of any of its obligations under this Lease and, if so, specifying the nature of each such Default; and (d) to the best of the Certifying Party's knowledge, whether or not any event has occurred which, with the giving of notice, the passage of time, or both, would constitute a Default by the Requesting Party and, if so, specifying the nature of each such event (referred to herein as an "**Estoppel Certificate**"). The parties agree that any statement delivered pursuant to this Section 0 may be relied upon by the Requesting Party and by potential or actual purchasers and/or lenders with whom the parties may be dealing, regardless of independent investigation. The Requesting Party agrees to pay to the Certifying Party any administrative and legal costs related to providing such Estoppel Certificate.

16.23 Force Majeure Event

If the performance by Lessee of its obligations under Section 8.2 of this Lease (excluding monetary obligations, including, but not limited to taxes and insurance) is delayed or prevented by any extraordinary acts of nature (including floods, explosions, earthquakes), or fires, epidemic, war, terrorism, labor strikes, material shortages, riots, orders of restraint by governmental authority which prevent Lessee from meeting such obligations under this Lease, or other extraordinary casualty which is not reasonably within Lessee's control, and is not due to the fault or negligence of Lessee, one or more of which is defined as a "Force Majeure Event", then Lessee shall be temporarily excused from performance to the extent that performance is delayed by any of the foregoing, without liability under this Lease. Lessee agrees, however, to proceed with all reasonable dispatch to perform such obligations under this Lease after the Force Majeure Event preventing Lessee from carrying out such obligations under this Lease ceases to exist.

16.24 Mediation

Should any dispute arise between the parties to this Lease related to matters set forth in Section 6.10 of this Lease, then it is agreed that such dispute will be submitted to non-binding mediation prior to any litigation. The provisions of this Section 0 shall also apply to any nonbinding mediation with respect to matters covered under Section 14.6 if the parties have agreed to such mediation as provided in Section 14.6. The parties shall exercise good faith efforts to select a mediator who shall be compensated equally by both parties. Mediation will be conducted in Columbia County, Oregon, unless both parties agree otherwise. Both parties agree to exercise good faith efforts to resolve disputes covered by this Section 16.24 through the mediation process. If a party requests mediation and the other party fails to respond within ten (10) days, or if the parties fail to agree on a mediator within ten (10) days, a mediator shall be appointed by the presiding judge of the Columbia County Circuit Court upon the request of either party. The parties agree that mediation shall occur on an expedited schedule and that they will not attempt to delay mediation, the scheduling of the mediation or continue the mediation once it has been set. The parties shall retain all rights at law or in equity with respect to any dispute not covered by this Section 0 and also with respect to those disputes covered by this Section 0 after mediation has been completed. In no event shall the provisions of this Section 0 limit the Port's right to seek immediate injunction relief following an Event of Default to stop an activity or use on the Premises prohibited under this Lease. Nothing in this Section 0 shall be deemed to prevent the Port from taking immediate injunctive or other action in response to an emergency or in response to third-party acts or omissions that pose an imminent threat to the environment or to the health, safety, or security of Port Lessees or of the public.

16.25 Entire Agreement

This Lease represents the entire agreement between the Port and Lessee relating to Lessee's leasing of the Premises. It is understood and agreed by Lessee that neither the Port nor the Port's agents or employees has made any representations or promises with respect to this Lease or the making or entry into this Lease, except as expressly set forth in this Lease. No claim for liability or cause for termination shall be asserted by Lessee against the Port for, and the Port shall not be liable by reason of, any claimed breach of any representations or promises not expressly set forth in this Lease. All oral agreements with the Port are expressly waived by Lessee. This Lease has been negotiated at arm's length between the parties, with both parties being represented by legal counsel. Therefore, no alleged ambiguity or other drafting issues of the terms of this Lease shall be construed, by nature of the drafting, against either party.

16.26 Capacity to Execute

The Port and Lessee each warrant and represent to one another that this Lease constitutes a legal, valid, and binding obligation of that party. The individuals executing this Lease personally warrant that they have full authority to execute this Lease on behalf of the entity for whom they purport to be acting.

IN WITNESS HEREOF, the parties have subscribed their names hereto effective as of the year and date first written above.

| BLUE HERON SEPTIC & DRAIN | THE PORT OF COLUMBIA COUNTY |
|---------------------------|------------------------------------|
| SERVICES | |
| By: prome 1 1 Delannes | By: |
| Print Name: Spencer Desch | Print Name: Sean P. Clark |
| Title: President | Title: _ <u>Executive Director</u> |
| Date: | Date: |

EXHIBIT A

(Premises Depiction)







Shop & Office Building (3,500 SF)

O Premises Access

EXHIBIT B

AGREEMENT TO PAY COSTS

Blue Heron Septic and Drain Services Lease

Page 50

This Agreement to Pay Costs ("Agreement") is made and entered into this _____ day of , 2024, by and between the Port of Columbia County, a municipal

corporation of the State of Oregon ("the <u>Port</u>"), and Blue Heron Septic & Drain Services (the "<u>Lessee</u>") (collectively, "the Parties").

RECITALS

WHEREAS, Lessee approached the Port and requested that the Port incur costs, including staff and attorney time, for the Parties to engage in discussions and negotiations with staff of the Port and have Port General Counsel draft

(work specifically requested by Lessee), and

WHEREAS Lessee's request for Port staff and Port General Counsel to do work on Lessee's behalf has resulted in additional time and costs for which it is appropriate that Lessee compensate the Port ("Port Work"), and

WHEREAS the Parties intend to carry out their rights, duties, and obligations in such a manner so as to avoid action that would increase, directly or indirectly, the Parties exposure to liability,

NOW, THEREFORE, for good, fair, and valuable consideration, the receipt and sufficiency of which is acknowledged, and in consideration of the mutual covenants and agreements set forth herein, the Parties hereby agree as follows:

1. **INCORPORATION OF RECITALS**. The above-stated Recitals are true and correct to the best of their knowledge and are incorporated into this Agreement as though fully set forth herein.

2. NONREFUNDABLE PAYMENT. Lessee agrees to pay a <u>TBD</u> estimated nonrefundable, fixed fee payment ("the Nonrefundable Payment") for Port Work. The Nonrefundable Payment is due in full immediately and must be received by the Port prior to the start of any Port Work.

3. INDEMNIFICATION. Lessee agrees to defend (using legal counsel acceptable to the Port), indemnify and hold harmless the Port, including the Port's Commissioners, officers and employees, from and against any and all actual or alleged claims, damages, expenses, costs, fees (including, but not limited to, attorney, accountant, paralegal, expert, and escrow fees), fines, and/or penalties (collectively "Costs") which may be imposed upon or claimed against or incurred by the Port and which arise from or are in any way connected with the Activities or this Agreement.

4. **LIMITATION OF LIABILITY**. The Port makes no representation or warranty, express or implied, as to the nature, quantity, or quality of any Port Work. In no event shall the Port be liable to Lessee for any special, indirect, incidental, and consequential or punitive damages (including any claim for lost profits or revenues).

5. COUNTERPARTS. This Agreement may be executed in identical counterparts, all of which shall be considered one and the same agreement and shall become effective when counterparts have been signed by each party and delivered to the other party. In the event that any signature is delivered by facsimile transmission or by an e-mail which contains an electronic file of an executed signature page, such signature page shall be deemed to constitute an original instrument, with the same force and effect as execution and delivery of an original and shall create a valid and binding obligation of the party executing the Agreement.

IT IS SO AGREED by the Parties as of the date set forth above.

LESSEE

PORT OF COLUMBIA COUNTY

By:_____

By: ______ Sean P. Clark Executive Director



RECORD OF DECISON

For

AREA 1 UPLAND AND IN-WATER AREAS FORMER POPE AND TALBOT WOOD-TREATING SITE ST. HELENS, OREGON

September 1, 2023





This document was prepared by Oregon Department of Environmental Quality Cleanup Program, Northwest Region 700 NE Multnomah Street, Suite 600 Portland Oregon, 97232

> Contact: Katie Daugherty Phone: 503-860-3943 www.oregon.gov/deq



| Translatio | on or other fo | ormats | | | | <i>.</i> |
|------------------------|--------------------------------------|-------------------------------|-----------------|----------------------------------|------------|---|
| Español | <u>한국어</u> | 繁體中文 | | Русский | | عربية Tiềng Việt |
| 800-452-40 | 11 TTY: 71 | 1 deginfo@ | deq. | oregon.gov | | |
| | - | | | | | |
| Non-disc | rimination | statement | | | | |
| | miniation | Statement | | | | |
| | not diagriminat | o on the heald | ∖ of r | aaa aalar n | 0 ti 0 | nol origin dipobility ago o |
| DEQ does | not discriminat | e on the basis | s of r | ace, color, n | atio | onal origin, disability, age o |
| DEQ does sex in adm | not discriminat nistration of its | e on the basis programs or | s of r activ | ace, color, n /ities. Visit D | atio EQ | nal origin, disability, age o 's <u>Civil Rights and</u> |

TABLE OF CONTENTS

| 1. II | NTROD | UCTION | |
|----------|-------------------|--|----|
| 1.1 | INTR | ODUCTION | |
| 1.2 | SCOF | PE AND ROLE OF THE SELECTED REMEDIAL ACTION | 1 |
| | | | |
| 2. S. | | FIORY AND DESCRIPTION | |
| 2.1 | SITE | LOCATION AND LANDUSE | 4 |
| 2. | 1.1 Are | eas of the Site | |
| | 2.1.1.1 | Area 1 Upland Priority Action Area | 4 |
| | 2.1.1.2 | Area 1 Dock Priority Action Area | 5 |
| | 2.1.1.3 | Area 2 Dock Priority Action Area | 5 |
| | 2.1.1.4 | Cove Priority Action Area. | |
| | 2.1.1.5 | Lower Milton Creek Sediment Area | |
| 2 | 12.1.1.0 | rrent and Reasonably Anticipated Future Land Uses | |
| 2. 2. | 1.3 Adi | iacent Cleanun Sites | 7 |
| 2.2 | PHYS | SICAL SETTING | |
| 2. | 2.1 Cli | mate | |
| 2. | 2.2 Ge | ology | |
| 2. | 2.3 Hy | drogeology | |
| 2. | 2.4 Sur | face Water and Stormwater | 9 |
| | 2.2.4.1 | Area 1 Dock | 9 |
| | 2.2.4.2 | Area 2 Dock | |
| | 2.2.4.3 | Cove Area | |
| | 2.2.4.4 | Upper Milton Creek | |
| 22 | 2.2.4.3 DI AN | | |
| 2.5 | 21 Dh | vsical Plant | |
| 2. | 2311 | Wood-Treating Facility | |
| | 2.3.1.2 | Post-Peeling Facility | |
| 2. | 3.2 Ch | emical Use and Waste Generation and Management | |
| | 2.3.2.1 | Wood-Treating Facility | |
| 2.4 | REGU | JLATORY HISTORY | |
| 2. | 4.1 Pre | eliminary Assessment (1988) and Site Inspection (1990) | |
| 2. | 4.2 Site | e Investigation (1993) | |
| 2. | 4.3 199 | 95 Consent Order & 2008 Orphan Declaration | 14 |
| 2. | 4.4 200 | 00 Remedial Investigation | 14 |
| | 2.4.4.1 | Phase 1 Investigation (1996 to 1997) | |
| | 2.4.4.2 | Phase 2 Investigation (1998) | |
| 2 | 2.4.4.5 15 Int | ring Remedial Investigations (1000-2006) | |
| 2. | 4.5 m 4.6 Sur | nnlemental Remedial Investigation (2010 – 2018) | 10 |
| 2. | 2.4.6.1 | Investigations Performed | |
| | 2.4.6.2 | Investigation Results | |
| 3. R | ESULT | S OF INVESTIGATION(S) | |
| 2 1 | | | 10 |
| 3.1 2 | | THE AND EATENT OF CONTAININATION | |
| 5. | 3111 | Area 1 Unland Non-Aqueous Phase Liquid | |
| | 3.1.1.2 | Area 1 Dock Non-Aqueous Phase Liquid | |

TABLE OF CONTENTS (Continued)

| 3.1.1. | 3 Area 2 Dock NAPL | 21 |
|----------|---|----|
| 3.1.1. | 4 Cove Area NAPL | 21 |
| 3.1.1. | 5 Upper Milton Creek Area NAPL | 22 |
| 3.1.2 | Groundwater | |
| 3.1.3 | Soil | 23 |
| 3.1.4 | Sediment | 23 |
| 3.1.5 | Sediment Porewater | 24 |
| 3.1.6 | Surface Water | |
| 3.1.7 | Air | |
| 3.2 RI | SK ASSESSEMENT | 25 |
| 3.2.1 | Conceptual Site Model | |
| 3.2.2 | Human Health Risk Screening | |
| 3.2.3 | Human Health Risk Assessment | |
| 3.2.3. | 1 Cumulative Human Health Risk Estimates | |
| 3.2.4 | Ecological Risk Assessment | 33 |
| 3.2.4. | Ecological Risk Assessment Lines of Evidence | |
| 3.2.4. | 2 Ecological Risk Assessment Conclusions | |
| 3.2.5 | Human Health and Ecological Risk Summary and RAOs to Address Risk | 37 |
| 3.2.5. | 1 Human Health Risk Summary | |
| 3.2.5. | 2 RAOs to Address Human Health Risk | |
| 3.2.5. | 3 Ecological Risk Summary | |
| 3.2.5. | 4 RAOs to Address Ecological Risk | 40 |
| 3.3 BE | ENEFICIAL USE AND HOT SPOT DETERMINATION | 40 |
| 3.3.1 | Groundwater Beneficial Use Determination | 41 |
| 3.3.2 | Surface Water Beneficial Use Determination | 41 |
| 3.3.3 | Hot Spots | 42 |
| 3.3.3. | 1 Non-Aqueous Phase Liquid | 42 |
| 3.3.3. | 2 Impacts to Beneficial Water Uses | 42 |
| 3.3.3. | 3 Soil/Sediment | 42 |
| 3.3.3. | 4 Contaminated Debris | 42 |
| 3.4 ES | STIMATE OF CONTAMINATION VOLUME | 43 |
| 4. PEER | REVIEW SUMMARY | 44 |
| | | |
| 5. DESCI | RIPTION OF REMEDIAL ACTION ALTERNATIVES | 45 |
| 5.1 RE | EMEDIAL ACTION OBJECTIVES & PRELIMINARY REMEDIATION GOALS | 45 |
| 5.1.1 | Acceptable Risk Levels | 45 |
| 5.1.2 | Site-Specific Remedial Action Objectives | 46 |
| 5.1.3 | Preliminary Remediation Goals | 46 |
| 5.1.4 | Remedial Action Levels and Cleanup Levels | 47 |
| 5.1.5 | Priority Action Areas | 47 |
| 5.2 RH | EMEDIAL ACTION ALTERNATIVES FOR SEDIMENT | |
| 5.2.1 | General Response Actions and Applicable Technologies | 48 |
| 5.2.2 | Estimate of Contaminated Media | 49 |
| 5.2.2. | 1 Estimate of Upland Contamination | |
| 5.2.2. | 2 Estimate of In-water Contamination | 49 |
| 5.2.3 | Common Elements and Assumptions | 50 |
| 5.2.4 | RAA-1: No Action | 52 |
| 5.2.5 | Area 1 Upland Priority Action Area Remedial Action Alternatives | 52 |
| 5.2.6 | In-Water Priority Action Areas Remedial Action Alternatives | 53 |
| | | |

TABLE OF CONTENTS (Continued)

| | 5.2.6.1 | Area 1 Dock Priority Action Area | 53 |
|---|--|---|--------------|
| | 5.2.6.2 | Area 2 Dock Priority Action Area | 53 |
| | 5.2.6.3 | Cove Area Priority Action Area | 54 |
| | 5.2.6.4 U | Upper Milton Creek Priority Action Area | 54 |
| 5 | 5.2.7 Monit | oring, Review, and Contingency Plan | |
| 6. I | EVALUATI | ON OF REMEDIAL ACTION ALTERNATIVES | 56 |
| 6.1 | EVALU | ATION CRITERIA | |
| 6.2 | PROTEC | CTIVENESS | |
| 6.3 | BALAN | CING FACTORS | |
| 6.4 | EVALU | ATION OF ALTERNATIVES AGAINST CRITERIA | |
| 7. (| COMPARA | TIVE ANALYSIS OF REMEDIAL ACTION ALTERNATIVES | 61 |
| 7.1 | AREA 1 | UPLAND PRIORITY ACTION AREA | 61 |
| 7.2 | AREA 1 | DOCK PRIORITY ACTION AREA | 61 |
| 7.3 | AREA 2 | DOCK PRIORITY ACTION AREA | |
| 7.4 | COVE A | REA PRIORITY ACTION AREA | 63 |
| 7.5 | UPPER 1 | MILTON CREEK PRIORITY ACTION AREA | |
| 8 1 | RECOMME | NDED REMEDIAL ACTION ALTERNATIVES IN STAFF REP(| ORT 65 |
| | | | |
| | | | |
| 9. 1 | PUBLIC NO | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE | 66 |
| 9. I 10. | PUBLIC NO | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION | 66 67 |
| 9. 1 10. | PUBLIC NO SELECTI 1 DESCRI | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION | 66 67 |
| 9. 1 10. | PUBLIC NO SELECTI 1 DESCRI 20.1.1 Are | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION ea 1 Upland Priority Action Area | 66 67 |
| 9. 1 10. | PUBLIC NO SELECTI DESCRI 0.1.1 Are 0.1.2 Are | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION ea 1 Upland Priority Action Area ea 1 Dock Priority Action Area | |
| 9. 1 10. | PUBLIC NO SELECTI 1 DESCRI 20.1.1 Are 20.1.2 Are 20.1.3 Are | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION Priority Action Area Priority Action Area Priority Action Area Priority Action Area | |
| 9. 1 10. 10. 1 | PUBLIC NO SELECTI 1 DESCRI 0.1.1 Are 0.1.2 Are 0.1.3 Are 0.1.4 Cor | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION ea 1 Upland Priority Action Area ea 1 Dock Priority Action Area ea 2 Dock Priority Action Area ea 2 Dock Priority Action Area ea 2 Dock Priority Action Area | |
| 9. 1 10. 10. | PUBLIC NO SELECTI 1 DESCRI 0.1.1 Are 0.1.2 Are 0.1.3 Are 0.1.4 Cov 0.1.5 Up | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION va 1 Upland Priority Action Area va 1 Dock Priority Action Area va 2 Dock Priority Action Area ve Area Priority Action Area ver Area Priority Action Area ver Milton Creek Priority Action Area | |
| 9. 1 10. 10. | PUBLIC NO SELECTI 1 DESCRI 0.1.1 Are 0.1.2 Are 0.1.3 Are 0.1.4 Cov 0.1.5 Up 0.1.6 Englishing | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION va 1 Upland Priority Action Area va 2 Dock Priority Action Area va 2 Dock Priority Action Area ve Area Priority Action Area ver Milton Creek Priority Action Area gineering and Institutional Controls | |
| 9. I 10. 10. | PUBLIC NO SELECTI 1 DESCRI 0.1.1 Are 0.1.2 Are 0.1.3 Are 0.1.4 Cov 0.1.5 Up 0.1.6 Eng 0.1.7 Per | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION ea 1 Upland Priority Action Area to 2 Dock Priority Action Area ea 2 Dock Priority Action Area ea 2 Dock Priority Action Area per Milton Creek Priority Action Area gineering and Institutional Controls formance Monitoring | |
| 9. 1 10. 10. 11. 11. 11. 11. 11. 11. 11. 11 | PUBLIC NO SELECTI 1 DESCRI 0.1.1 Are 0.1.2 Are 0.1.3 Are 0.1.4 Cor 0.1.5 Up 0.1.6 Eng 0.1.7 Per 0.1.8 Cor | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION ea 1 Upland Priority Action Area ea 1 Dock Priority Action Area ea 2 Dock Priority Action Area formance Monitoring entingency Measures | |
| 9. 1 10. 10. 10. 10. 10. 10. 10. 10. 10. 10 | PUBLIC NO SELECTI 1 DESCRI 0.1.1 Are 0.1.2 Are 0.1.3 Are 0.1.4 Con 0.1.5 Up 0.1.6 Eng 0.1.7 Pen 0.1.8 Con 2 RESIDU | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION <i>va 1 Upland Priority Action Area va 1 Dock Priority Action Area va 2 Dock Priority Action Area ve Area Priority Action Area ve Area Priority Action Area ve Area Priority Action Area gineering and Institutional Controls formance Monitoring Mathematica Measures Val RISK ASSESSMENT</i> | |
| 9. 1 10. 10. 10. 10. 11. | PUBLIC NO SELECTI 1 DESCRI 0.1.1 Are 0.1.2 Are 0.1.3 Are 0.1.4 Cor 0.1.5 Up 0.1.6 Eng 0.1.7 Per 0.1.8 Cor 2 RESIDU STATUTO | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION Pa 1 Upland Priority Action Area Priority Action Area | |
| 9. 1 10. 10. 10. 10. 10. 10. 11. 12. | PUBLIC NO SELECTI 1 DESCRI 10.1.1 Are 10.1.2 Are 10.1.3 Are 10.1.4 Cor 10.1.4 Cor 10.1.5 Up 10.1.6 Ens 10.1.7 Per 10.1.8 Cor 2 RESIDU STATUTO SIGNATU | DTICE, PUBLIC COMMENT, AND AGENCY RESPONSE ED REMEDIAL ACTION PTION OF THE SELECTED REMEDIAL ACTION ea 1 Upland Priority Action Area ea 1 Dock Priority Action Area ea 2 Dock Priority Action Area formance Milton Creek Priority Action Area formance Monitoring htingency Measures AL RISK ASSESSMENT DRY DETERMINATIONS | |

TABLES

Table 1 – Summary of RI Data Collection

Table 2 - Summary of Risks

- Table 3 Upland and Riverbank Surface Soil Technology Screening
- Table 4 Upland Subsurface NAPL, Soil, and Groundwater Technology Screening
- Table 5 In-Water Sediment and Porewater Technology Screening
- Table 6 Estimated Quantities of Contaminated Media

TABLES (Continued)

- Table 7 Area 1 Upland PAA Remedial Technologies Descriptions and Assemblage of RAAs
- Table 8 Area 1 Dock PAA Remedial Technologies Descriptions and Assemblage of RAAs
- Table 9 Area 2 Dock PAA Remedial Technologies Descriptions and Assemblage of RAAs
- Table 10 Cove Area PAA Remedial Technologies Descriptions and Assemblage of RAAs
- Table 11 Upper Milton Creek PAA Remedial Technologies Descriptions and Assemblage of RAAs
- Table 12 Area 1 Upland PAA Remedial Action Alternatives Rankings and Rationale for Evaluation Criteria
- Table 13 Area 1 Dock PAA Remedial Action Alternatives Rankings and Rationale for Evaluation Criteria
- Table 14 Area 2 Dock PAA Remedial Action Alternatives Rankings and Rationale for Evaluation Criteria
- Table 15 Cove Area PAA Remedial Action Alternatives Rankings and Rationale for Evaluation Criteria
- Table 16 Upper Milton Creek PAA Remedial Action Alternatives Rankings and Rationale for Evaluation Criteria
- Table 17 Summary of Evaluation Criteria Scoring for Each Remedial Action Alternatives for all PAAs
- Table 18 Public Comments Received and Agency Responses

FIGURES

- Figure 1 Site Location Map
- Figure 2 Upland and Priority Sediment Areas Map
- Figure 3A Site Plan
- Figure 3B Key Site Features
- Figure 4 1919 Site Layout Diagram
- Figure 5 1929
- Figure 6 1938 Site Layout Diagram
- Figure 7 USACE 1939
- Figure 8 USACE 1948
- Figure 9 UO 1953
- Figure 10 1959 Site Layout Diagram
- Figure 11 USACE 1966
- Figure 12 UO 1970
- Figure 13 1974 Site Layout Diagram
- Figure 14 EDR 1977
- Figure 15 EDR 1990
- Figure 16 2013
- Figure 17 Generalized Area 1 Cross Section
- Figure 18 Vicinity Cleanup Sites
- Figure 19 Cross Section Locations
- Figure 20 Cross Section A-A'
- Figure 21 Cross Section B-B'
- Figure 22 Cross Section C-C'

FIGURES (Continued)

- Figure 23 Cross Section D-D'
- Figure 24 Cross Section E-E'
- Figure 25 Cross Section F-F'
- Figure 26 Cross Section G-G'
- Figure 27 Cross Section H-H'
- Figure 28 Inferred Basalt Elevation Contours
- Figure 29 Groundwater Elevation Contours, 2012 High and Low Water Events
- Figure 30 Site Topography and Bathymetry (2010)
- Figure 31 Area 1 Site Topography and Bathymetry (2017)
- Figure 32 RI Sampling Locations Before Initiating the Supplemental RI in 2010
- Figure 33 Supplemental RI Sampling Locations
- Figure 34 Schematic Conceptual Site Model
- Figure 35 Estimated Extent of NAPL
- Figure 36 Surface Sediment Exhibiting Creosote Sheen
- Figure 37 Subsurface Sediment Exhibiting Creosote Sheen
- Figure 38 Total PAHs in Groundwater and 2011-2012 of Groundwater Toxic Units ΣGWTU Results
- Figure 39 Total Petroleum Hydrocarbons in Groundwater 2011-2012
- Figure 40 Human Health Risk Assessment Conceptual Site Model
- Figure 41 Ecological Risk Assessment Conceptual Site Model
- Figure 42 Surface Sediment Exhibiting Creosote Sheen
- Figure 43 Upland PAA Alternative 4 Impervious Surface Cap & Permeable Reactive Barrier
- Figure 44 In-Water PAAs and Shoreline Cross-Section Locations
- Figure 45A Cross Section F-F' Area 1 Dock Alternative 2 Installation of Armored Reactive Cap
- Figure 45B Cross Section G-G' Area 1 Dock Alternative 2 Installation of Armored Reactive Cap
- Figure 46 Cross Section H-H' Area 2 Dock Alternative 4 Nearshore Removal, Cap Residuals
- Figure 47A Cross-Section C-C' Cove Area Alternative 4 Riverbank Restoration, Armored Reactive Capping of Residuals
- Figure 47B Cross-Section D-D' Cove Area Alternative 4 Riverbank Restoration, Armored Reactive Capping of Residuals
- Figure 47C Cross-Section E-E' Cove Area Alternative 4 Riverbank Restoration, Armored Reactive Capping of Residuals
- Figure 48 Cross-Sections A-A' & B-B' Upper Milton Creek -Alternative 4 Regrade Bank, Cap Residuals

| AST | above ground storage tank | OAR | Oregon Administrative Rules |
|--------|--|--------|-------------------------------------|
| bgs | below ground surface | O&M | operations and maintenance |
| bml | below mudline | OHA | Oregon Health Authority |
| BMP | best management practice | OHWM | Ordinary High-Water Mark |
| BTEX | benzene, toluene, ethylbenzene, and | OPAHs | Oxygenated PAHs |
| | xylenes | ORS | Oregon Revised Statutes |
| BWUD | beneficial water use determination | OSU | Oregon State University |
| CEPCs | contaminants of potential ecological | PA | Preliminary Assessment |
| | concern | PAA | Priority Action Area |
| Cfree | freely dissolved water concentrations | PAHs | polycyclic aromatic hydrocarbons |
| CMMP | contaminated media management plan | PCBs | polychlorinated biphenyls |
| COC | chemical of concern | PCP | pentachlorophenol |
| COPC | contaminants of potential concern | PDI | pre-design investigation |
| CSM | conceptual site model | PEC | probable effects concentration |
| CTE | Central Tendency Exposure | PHSS | Portland Harbor Superfund Site |
| CULs | cleanup levels | PRB | permeable reactive barrier |
| DNAPL | dense non-aqueous phase liquid | PRGs | Preliminary Remediation Goals |
| DEQ | Oregon Department of Environmental | PVC | polyvinyl chloride |
| | Quality | RAA | remedial action alternative |
| DSL | Division of State Lands | RALs | remedial action levels |
| E&E | Ecology and Environment, Inc. | RAOs | remedial action objectives |
| EES | easement and equitable servitudes | RBCs | risk-based screening concentrations |
| ECSI | environmental cleanup site information | RD/RA | Remedial Design/Remedial Action |
| ENR | enhanced natural recovery | RME | Reasonable Maximum Exposure |
| EPA | Environmental Protection Agency | ROD | Record of Decision |
| EPH | extractable petroleum hydrocarbon | SLVs | screening levels values |
| ERA | ecological risk assessment | SPME | solid-phase microextraction |
| Etc. | etcetera | SQGs | sediment quality guidelines |
| FCV | final chronic value | ΣΙΨΤυ | sum of IW TU |
| HI | Heavy Industrial | ΣGWTU | sum of groundwater TUs |
| HLA | Harding Lawson Associates | ΣSWTU | sum of surface water TU |
| HHRA | human health risk assessment | SVOCs | semi-volatile organic compounds |
| IC | institutional control | TEC | threshold effects concentration |
| in/yr | inches per year | TOC | total organic carbon |
| IŴ | interstitial water | TPH | total petroleum hydrocarbons |
| LDPE | low density polyethylene | TU | toxic unit |
| LOEs | lines of evidence | μg/L | micrograms per liter |
| LOF | Locality of Facility | U.S. | United States |
| mg/kg | milligrams per kilogram | USACE | U.S. Army Corps of Engineers |
| mg/L | milligrams per liter | UST | underground storage tank |
| MNA | monitored natural attenuation | VOCs | volatile organic compounds |
| MNR | monitored natural recovery | VPH | volatile petroleum hydrocarbon |
| NAPL | non-aqueous phase liquid | °F | degrees Fahrenheit |
| NAVD88 | North American Vertical Datum of | % | percent |
| | 1988 | \geq | greater than or equal to |
| NFA | no further action | > | greater than |
| No. | number | < | less than |
| NPDES | National Pollutant Discharge | | |
| | Elimination System | | |

vi

1. INTRODUCTION

1.1 INTRODUCTION

This Record of Decision (ROD) presents the Oregon Department of Environmental Quality's selected cleanup remedy (remedial action) for Area 1 Upland and contaminated sediments located in Upper Milton Creek and Scappoose Bay adjacent to the former Pope and Talbot wood-treating facility located at 1550 Railroad Avenue in St. Helens, Oregon (Site) (Figure 1). The Site has been divided into several areas. DEQ issued a no further action determination for Area 2 Upland (tax lot 302) on April 16, 2012. This ROD is applicable to the Area 1 Upland (tax lot 300) and four priority sediment areas (Area 1 Dock, Area 2 Dock, Cove Area, and Upper Milton Creek) which are defined in Section 2.1.1 and shown on Figure 2. The selected remedial action includes riverbanks located within the priority sediment areas. The selected remedial action was developed in accordance with Oregon Revised Statutes (ORS) 465.200 *et seq.* and Oregon Administrative Rules (OAR) Chapter 340, Division 122, Sections 0010 through 0115, and is based on the administrative record for this Site.

This ROD summarizes information contained in the DEQ Staff Report (DEQ, 2023), as well as the Remedial Investigation, Updated Supplemental Risk Assessment, Human Health Risk Assessment, Ecological Risk Assessment, and Remedial Investigation/Feasibility Study reports completed under DEQ Order of Consent Number (No.) WMCSR-NWR-95-05 signed April 13, 1995, by Port of St. Helens (now Port of Columbia County) and DEQ. DEQ's environmental cleanup site information (ECSI) system designates the Site as No. 0959. A copy of the Administrative Record Index is attached as Section 13.

In selecting the remedial action presented in this ROD, DEQ considered public input following a comment period on the Staff Report.

1.2 SCOPE AND ROLE OF THE SELECTED REMEDIAL ACTION

The selected remedial action addresses both upland and in-water contamination. In the upland, the selected remedial action addresses the presence of creosote non-aqueous liquid (NAPL) and dissolved phase chemical of concerns (COCs) in groundwater and soil in the Area 1 Upland priority action area (PAA). In the in-water areas, the selected remedial action addresses sediment, porewater and surface water impacted by polycyclic aromatic hydrocarbons (PAHs), metals, creosote sheen, and contaminated woody debris in surface and subsurface sediments within the sediment PAAs and adjacent riverbanks where contaminated soils pose a future recontamination risk. The areas with the greatest potential for impacts to human health and the environment are the focus of the feasibility study and a key focus on the selected remedial action. Additional in-water

areas (e.g., with light petroleum sheen) may need further investigation and cleanup in the future (e.g., Lower Milton Creek).

The selected remedial action for Area 1 Upland includes placement of an impervious surface cap over the entire Area 1 Upland PAA and a permeable reactive barrier (PRB) at the top of the riverbank, adjacent to the Cove Area PAA. The selected upland remedial action is intended to address risks associated with direct contact to soil in the upland, limit stormwater infiltration into the NPAL source area in Area 1 Upland, and limit contaminant migration from the upland source area into the in-water areas via groundwater seeps. The long-term effectiveness of the selected remedial action for Area 1 Upland is contingent upon Institutional Controls (ICs) to mitigate risks to potential future industrial, excavation, and construction workers in the upland.

The selected in-water remedial action is intended to address areas with the highest levels of sediment contamination (hot spots, specifically areas with creosote NAPL and heavy to moderate petroleum sheen). Remedial action objectives (RAOs) were developed to focus treatment in areas with the highest contamination levels posing unacceptable risks to humans and ecological receptors. DEQ anticipates further investigations to support remedial design will be performed.

Contamination in upland soil and groundwater pose unacceptable direct contact and vapor inhalation risks to human health (specifically to potential future industrial, excavation, and construction workers). Sediment contamination within the in-water PAAs exceeds acceptable risk levels for both human health (specifically to subsistence fishers) and for aquatic ecological receptors. In addition to risks associated with dissolved phase COCs, the presence of creosote NAPL and moderate to heavy petroleum sheen presents unacceptable risk to human health and aquatic ecological receptors. "Hot spots" (elevated levels of contamination) are present in all of the PAAs.

The following general actions will be components of the selected remedial action:

- Placement of an impervious surface cap over the entire Area 1 Upland PAA and a PRB at the top of the riverbank, adjacent to the Cove Area PAA.
- Establishment of an easement and equitable servitudes (EES) in the upland to maintain the cap in perpetuity, and implementation of ICs to prevent potential risks to future industrial, excavation, and construction workers associated with groundwater and subsurface soil contamination in Area 1 Upland.
- Treatment, containment, or removal of sediment that contains NAPL and/or emanates moderate to heavy petroleum sheen.
- Treatment, containment, or removal of erodible riverbank soils considered to pose a risk of recontamination to sediments.
- Monitored natural recovery (MNR), which consists of the natural burial of surface sediment contamination posing a lower risk through deposition of suspended sediment from Milton Creek, Columbia, and Multnomah Channel watersheds.
- An Oregon Health Authority (OHA) advisory regarding consumption of fish, shellfish, and crayfish in Multnomah Channel and Scappoose Bay to minimize potential risk to human health until contamination levels protective of human health and fish are achieved.
- Long-term monitoring to assess and document progress of active remedial measures and MNR in achieving RAOs and cleanup levels (CULs) long-term and to address any residual risk.
- Periodic land and water use review.

The selected remedial action consists of the following PAA-specific elements:

- In the Area 1 Upland PAA, placement of an impervious surface cap with an engineered stormwater management system over the entire Area 1 Upland to reduce infiltration in combination with an organoclay PRB to intercept and sequester contaminants in the groundwater: preventing recontamination of the in-water remedy.
- Within the Area 1 Dock PAA, timber pilings and impacted surficial woody debris will be removed and an amended isolation cap will be placed across the sediment containing the highest levels of contamination posing unacceptable risks to human health and aquatic ecological receptors.
- In the Area 2 Dock PAA, impacted surficial woody debris will be removed, shallow surface sediment will be removed near the shoreline, and a sand cap will be placed over residual contamination for enhanced natural recovery (ENR).
- In the Cove Area and Upper Milton Creek PAAs, the shallow sediment with the highest concentrations (creosote NAPL and/or moderate to heavy sheen) will be excavated for off-site disposal and an engineered, amended isolation cap will be placed over the sediment within the PAAs along with protective armoring. The riverbank adjacent to these PAAs will be regraded to remove impacted soil and further reduce the potential for recontamination of the in-water remedy.

The selected remedial action will restore the Site to conditions protective of risks to upland Site workers, people who consume fish and shellfish for recreational and subsistence purposes or are directly exposed to contaminated sediment, and aquatic ecological receptors that consume prey from the Site or are directly exposed to contaminated sediment. The selected remedial action is protective of beneficial water uses from releases of petroleum sheen and prevents recontamination of the in-water remedy from the adjacent groundwater seeps and erodible riverbank soils.

2.1 SITE LOCATION AND LANDUSE

The Site consists of approximately 42 acres of industrial zoned land situated within the southern limits of the City of St. Helens. DEQ approved subdividing the upland portion of the Site into two areas (Area 1 and Area 2) in 2008. Area 1 is 25.15 acres and Area 2 is 17.32 acres. The Site also includes adjacent riverbanks and sediment areas. The United States (U.S.) Army Corps of Engineers (USACE) Ordinary High-Water Mark (OHWM) divides the upland and in-water areas. Land above the OHWM is considered upland and the portions of riverbanks and sediments below the OHWM are considered to be in-water. The selected remedial action is specific to Area 1 Upland PAA and four in-water PAAs discussed further in Section 2.1.1. DEQ's ECSI system designates the Site as No. 0959.

The Site lies on the northwestern bank of Scappoose Bay, near the confluence of Scappoose Bay and the Multnomah Channel. The Multnomah Channel is a 21.5-mile distributary, or branch, of the Willamette River. The channel flows to the north adjacent to the west side of Sauvie Island until it meets the Columbia River near St. Helens. Scappoose Bay is a low-energy surface water embayment just west of Multnomah Channel that includes former and current industrial sites. The confluence of the Multnomah Channel and the Columbia River is approximately 1.3 miles downriver from the Site. Milton Creek is a perennial stream located on the Site's western boundary. Milton Creek discharges to Scappoose Bay at the southwest corner of the Site.

2.1.1 Areas of the Site

The upland ground surface across most of Area 1 is generally between 22 to 25 feet North American Vertical Datum of 1988 (NAVD88). The Site's top of bank elevation with Milton Creek and Scappoose Bay are between 19 and 21 feet NAVD88. The OHWM at the Site generally corresponds to a vertical elevation of approximately 14 feet relative to the NAVD88. As previously discussed, riverbanks above the OHWM are included in Area 1 and riverbanks below OHWM are included with adjacent in-water areas. The riverbanks along Scappoose Bay and Milton Creek have steep to vertical banks. Portions of the riverbank, especially along Milton Creek, have potentially erodible soils that pose a recontamination risk to in-water remedies.

2.1.1.1 Area 1 Upland Priority Action Area

Geographically, Area 1 Upland is located approximately one mile east of Highway 30. Area 1 has a street address of 1550 Railroad Avenue in St. Helens, Oregon, and is tax lot 300 located in Section 9, Township 4 North, Range 1 West of the Willamette Baseline and Meridian in Columbia County (Figure 1). The latitude is 45.8413 degrees North, longitude 122.8094 degrees West. Area 1 Upland is accessed by and located at the eastern end of Railroad Avenue. Area 1 Upland covers 25.15 acres of the southwest portion of the Site and is generally flat with a peak elevation of approximately 27 feet above mean sea level. Nearly all of Area 1 Upland is located within the 100-year floodplain. Area 1 Upland is primarily vacant industrial land with a tenant occupying the former office/shop building. The Area 1 Upland boundary is shown on Figure 2.

The former wood-treating plant facilities, above ground storage tanks (ASTs), transfer tables, loading platforms, and underground portion of the creosote pipeline were located within Area 1 Upland (see Figures 3A and 3B). All facilities except for the former office/shop and a storage shed were removed from Area 1 in the early 1960s. Dredge material generated during the deepening of the Columbia River was placed on Area 1 Upland in the early 1970s. The fill material averages 7 to 8 feet thick over the native soil. Creosote NAPL observed in Area 1 Upland riverbanks adjacent to Milton Creek and Scappoose Bay indicates that some lateral movement of NAPL occurs beneath the Site. The progression of Site features from 1919 through 2013 are shown on Figures 4 through 16. A generalized cross-section showing the relationship between the dredge fill material and native soil and basalt bedrock is shown on Figure 17.

2.1.1.2 Area 1 Dock Priority Action Area

The Area 1 Dock PAA consists of an approximately 700-foot section of shoreline adjacent to Area 1 Upland and in-water area located between the former hog fuel loading hopper and downriver terminus of the former transfer table dock (see Figure 2). The wood decking and lateral bracing was removed from both dock structures in 2013. Over 435 the creosote-treated timber pilings remain, and the wood debris is present in the upper 1 to 2 feet of sediment and as deep as 11 feet below the mudline (bml; "mudline" generally refers to the surface water/sediment interface). The source of surface sediment exhibiting moderate to heavy sheen in the Area 1 Dock PAA is creosote-contaminated and/or treated wood debris associated with historical overwater activities, which has been encountered within 1.85 acres of surface sediment underneath and surrounding the former dock structures.

The Area 1 Dock PAA is closer to the main channel of Scappoose Bay and is subjected to stronger river current, as evidenced by the coarser sandy sediments encountered within the intertidal zone. The area's shoreline appears to be relatively stable to eroding. This indicates the environment is generally preventing the deposition of newer clean sediments and the burial of historical contamination.

2.1.1.3 Area 2 Dock Priority Action Area

The Area 2 Dock PAA consists of an approximately 600-foot section of shoreline adjacent to Area 2 Upland and in-water area located near the downriver property boundary and terminus of the former creosote pipeline, AST, and historical off-loading dock (see Figure 2). The wood decking and lateral bracing was removed from both dock structures. Hundreds of closely spaced creosote-treated timber pilings remain. Moderate to heavy petroleum sheen has been observed near timber pilings within the upper 12 inches of sediment. Residual creosote contaminated wood debris has been encountered within 0.3 acres of surface sediment in the former dock structure area.

The main stem of the Multnomah Channel is reportedly 100 to 150 feet offshore with a bottom elevation of approximately -15 feet NAVD88 immediately downstream of the Site. Based on the presence of contamination in surface sediment only (i.e., upper 12 inches) and the hard compact nature of the mudflats with relatively steep slopes towards the channel, the Area 2 Dock PAA appears to be relatively stable to eroding.

2.1.1.4 Cove Priority Action Area

The Cove PAA consists of approximately 300 feet of shoreline adjacent to Area 1 Upland, the island, and around the peninsula (see Figure 2). The Cove PAA is immediately downgradient of the former wood-treating plant operations. The cove is a manmade area created between 1948 and 1953 by extensive dredging and filling of lowlands. The Cove PAA consisted of a partially vegetated slough prior to dredging.

The elevation of the river bottom within the manmade cove and shallow inlet immediately downriver of the peninsula ranges between approximately 0 and 10 feet NAVD88 within approximately 100 to 300 feet of the shoreline before sloping steeply towards the main channel of Scappoose Bay. The main channel of Scappoose Bay adjacent to the Cove PAA widens to approximately 150 feet with bottom elevations between -7 and -5 feet NAVD88.

The Cove PAA is protected from the erosional effects of large storm and tidal surge and is a sediment depositional area. In 1960, wood treating plant operations ceased which has led to the deposition and burial of the old operational sediment surface throughout the Cove PAA. A sediment deposition rate of 0.2 to 1.4 inches per year (in/yr) is implied based on the presence of 1 to 7 feet of relatively clean fine textured sediment above the old contaminated operational sediment surface.

Creosote-impacted groundwater seeps exhibiting moderate to heavy sheen have been intermittently observed along the Cove PAA riverbank, located west of the peninsula. Some sediments within certain areas of the Cove PAA have been observed to exhibit moderate to heavy or slight petroleum sheen. NAPL within the Cove PAA varies vertically from a few inches to 12 feet bml, with much of the historical creosote contamination located 2 to 3 feet below deposited relatively clean sediment.

2.1.1.5 Upper Milton Creek Priority Action Area

Milton Creek discharges to Scappoose Bay at the southwest corner of the Site (see Figure 2). The Upper Milton Creek PAA consists of a 200-foot section of the east bank of Milton Creek where groundwater seeps exhibiting moderate to heavy creosote sheen are present within a layer of sand near the base of the streambank.

The upper portion of Milton Creek adjacent to the Site is characterized by relatively steep embankments (e.g., up to 1.5H:1V with "H" the cutback distance and "V" the depth). The elevation difference between the top and bottom of the streambank is approximately 15 feet along this portion of Milton Creek. Scouring of the creek channel by seasonal runoff and diurnal tidal cycles appears to limit the accumulation of sediment in this portion of the creek. The sediment bed profile generally consists of 0 to 1 foot of silty organic-rich fluff overlying a hardpan layer of clayey silt with gravel and cobbles.

2.1.1.6 Lower Milton Creek Sediment Area

The lower portion of Milton Creek generally includes the confluence of Milton Creek with Scappoose Bay and is characterized by a depositional sediment environment, with 3 to 10 feet of soft silt and varying amounts of sand overlying Columbia River Basalt Bedrock. Because this area

generally has not had observations of creosote NAPL in sediments and is depositional, the area is not considered a PAA. Additional sampling in this and other areas where petroleum sheen has been observed outside of the PAAs may be needed in the future.

2.1.2 Current and Reasonably Anticipated Future Land Uses

The Upland Area is currently zoned for Heavy Industrial (HI) use as part of the Railroad Avenue Industrial Park. The current and reasonably anticipated future uses of the Upland Area remain Heavy Industrial.

Adjacent land uses include railroad tracks and undeveloped land to the north and City-owned industrial development (former Boise White Paper Mill) to the northeast. Northwest of the Site and north of the railroad tracks are several single-family homes. Milton Creek forms the western Site boundary with the former St. Helens Fiberboard Plant and the Columbia County transfer station located on the west side of Milton Creek. Scappoose Bay and the Multnomah Channel of the Willamette River form the southern and eastern Site boundaries.

2.1.3 Adjacent Cleanup Sites

The Site is bordered to the east and northeast by the more than 200-acre former Boise Cascade Mill/Boise White Paper Mill site (ECSI No. 0014) located at 1300 Kaster Road in St. Helens (Figure 18). The former Boise White Paper Mill property is current and former industrial land with various buildings that were previously the location of pulping, bleaching, and milling operations. The former Boise White Paper Mill property issued a Record of Decision in 2023 for the 15-acre In-Water Sediment area. Additional investigation is anticipated to be performed during Remedial Design.

The former St. Helens Fiberboard Plant site (ECSI No.0091) is located at 1645 Railroad Avenue, west of the Site (Figure 18). The inactive former St. Helens Fiberboard Plant site has been used to manufacture a variety of mineral fiber and wood fiber building products. The site is divided into two operable units: Upland area and Lowland/In-water area. Remedial action in the upland area is complete. An interim remedial action measure is underway at the St. Helens Fiberboard Plant site in the Lowland/In-water area.

2.2 PHYSICAL SETTING

2.2.1 Climate

St. Helens has a temperate marine climate characterized by short, dry summers and wet winters. Between 1981 and 2010,¹ the average annual precipitation in St. Helens was 46.6 inches. Most precipitation falls between November and May, with average monthly totals ranging from 0.72 to 7.22 inches and the highest in December. The mean annual temperature ranges from approximately 37.1 to 67.2 degrees Fahrenheit (°F).

¹ National Oceanic & Atmospheric Administration, St. Helens Station.

2.2.2 Geology

Regionally, the Site lies in the Puget-Willamette Lowland, a broad structural and topographic depression located between the Coast Range to the west and the Cascade Range to the east. St. Helens lies in the northern portion of the Portland Basin, one of several complexly faulted basins comprising the Willamette Valley segment of the Puget-Willamette Lowland. The Portland Basin is interpreted to be a pull-apart basin between two *en echelon* right-lateral strike-slip faults (Evarts, 2004) caused by regional compression and shear associated with oblique subduction of the Juan de Fuca Plate beneath the North American Plate off the Oregon coast. In the deepest part of the Portland Basin, up to 1,800 feet of sediment has accumulated since the late Miocene.

The Site is on the west side of Multnomah Channel. The topographic highlands to the north expose Miocene flood basalts of the Grand Ronde Formation, deeply weathered in places. Chemistry and magnetostratigraphy data indicate the basalt is equivalent to the Member of Sentinel Bluffs, which is dated to approximately 15.6 million years. In the area of the Site, the Multnomah Channel is separated from the Columbia River by a peninsula marking the northernmost extent of Sauvie Island. The islands, floodplains and point bars of Multnomah Channel and the Columbia River east of the Site are comprised of Quaternary alluvium (i.e., primarily fine sand and silt).

Soil encountered at the Site consists of fill underlain by native soil, which in turn is underlain by basalt bedrock. The fill ranges from 4 to 12 feet below ground surface (bgs) and consists predominantly of dredge sand with localized areas of silt, coarse gravel, cobbles, wood chips, and metal and brick debris. The native soil, beneath the fill, ranges from 4 to 24 feet bgs and consists primarily of silt with varying amounts of clay and organic matter, and minor amounts of sand and gravel. Basalt bedrock was generally encountered at depths between 6 and 35 feet bgs in the upland area and at depths of 2 to 26 feet beneath river sediment in the offshore area. Bedrock beneath the Site is part of the Grande Ronde flows of the Columbia River Basalt. Detailed cross-sections were prepared for transects through the upland and in-water areas shown on Figure 19. The cross-sections are presented as Figures 20 through 27. The top of basalt surface elevation contours for the upland and in-water area are shown on Figure 28.

2.2.3 Hydrogeology

Regionally, shallow groundwater occurs in the surface fill materials, alluvium, and where the basalt bedrock is near the surface, in the weathered upper flow surfaces. Deeper groundwater zones are located in the more permeable interflow zones between the unweathered basalt flows and are not hydraulically connected to shallow groundwater.

In 1996, 17 monitoring wells, 13 located in Area 1, were installed as part of the risk investigation (Figure 29). Five monitoring wells were installed as pairs (depicted by A/B) with one deeper, though located in same water bearing zone, than the other for a total of 22 monitoring wells. One additional monitoring well was installed in 1998. The monitoring wells were constructed to depths ranging from 9 to 20 feet bgs depending on when bedrock was encountered. Well screens within the 2-inch diameter polyvinyl chloride (PVC) well casings ranged from 5 feet in length to 10 feet.

Following installation and development, depth to water measurements were regularly collected from the monitoring wells from August 1996 to March 1999. Groundwater was measured at depths

ranging from 2 to 8.5 feet bgs and fluctuates seasonally. Groundwater generally flows towards the nearest water body. Groundwater discharge to surface water is strongly influenced by seasonal rainfall and fluctuations in river stage. Groundwater contours for high and low water conditions are shown on Figure 29.

2.2.4 Surface Water and Stormwater

The Site lies on the northwestern bank of Scappoose Bay near the confluence of Scappoose Bay and the Multnomah Channel of the Willamette River. The confluence of the Multnomah Channel and the Columbia River is approximately 1.3 miles downriver from the Site. Milton Creek is a perennial stream located on the Site's western boundary and discharges to Scappoose Bay at the southwest corner of the Site. The Upper Milton Creek PAA is located in and adjacent to Milton Creek. The main channel of Scappoose Bay is offshore of Area 1 Upland. The Cove Area and Area 1 Dock PAAs are located in Scappoose Bay. Area 2 Dock PAA is located in the Multnomah Channel, downstream of Scappoose Bay. The locations of the PAAs and surface water bodies are shown on Figure 2. Bathymetry contours from 2010 and 2017 surveys are shown on Figures 30 and 31.

Based on a review of bathymetric maps generated in 2010 and 2017, the main channel of Scappoose Bay is a stable feature in the vicinity of Area 1. Tidal fluctuations of water levels occur daily in Scappoose Bay and Multnomah Channel, in addition to seasonal fluctuations in water levels as a result of runoff events in the watershed. The Site's shoreline is routinely inundated above the OHWM (14 feet NAVD88) during high tides (typically between April and July). Mudflats are commonly visible along the Site's shoreline between approximately 5 to 10 feet NAVD88. Depending on shoreline conditions, basalt topography, and channel dimensions, portions of the Site's shoreline experience sediment deposition (e.g., flat, soft bottom cove areas), while riverbank areas closer to the main channel appear to exhibit localized erosion of sediment from large storm events and/or tidal cycles. Specific conditions of each in-water PAA are described below.

2.2.4.1 Area 1 Dock

The 700-foot-long Area 1 Dock shoreline and in-water area is closer to the main channel of Scappoose Bay and is subject to stronger river current. The main channel of Scappoose Bay widens and deepens adjacent to and within the Area 1 Dock. Specifically, the bottom of the main channel ranges between 200 and 250 feet NAVD88 wide and between -10 and -5 feet NAVD88 deep. The river bottom slopes steeply downward within 50 to 100 feet of the shoreline. The Area 1 Dock shoreline is relatively stable to eroding, which generally prevents the deposition of newer clean sediments and the burial of historical contamination.

Deteriorating creosote-treated timber pilings, remnants of dock structures, are present throughout this area. Finer textured wood debris, described as "pulverized wood" in sampling logs and is interpreted to be hog fuel (e.g., wood shavings/chips) likely spilled from the hog fuel conveyor, is also present in sediment. The distribution and thickness of the woody debris is highly variable in this area, ranging from completely absent to more than 11 feet thick.

Contaminated sediments in the Area 1 Dock have been characterized to an approximate depth of -25 NAVD88 or about 25 feet bml. Sediments in the western portion of the area, in the vicinity of the hog fuel conveyor and loading hopper, were characterized as sand to sandy silt. Sediment in the eastern portion of the area, in the vicinity of the transfer table and dock, were characterized as silt. Basalt bedrock was encountered at various depths throughout the Area 1 Dock. Crosssections along transects B-B', G-G', and H-H' (see Figure 19 for cross-section locations) are included as Figures 21, 26, and 27.

2.2.4.2 Area 2 Dock

The 600-foot long and 150-foot-wide Area 2 Dock shoreline and in-water area is in the Multnomah Channel. Bathymetry data is not available for this area. Only a limited bml investigation of the Area 2 Dock has been performed. Surface sediments generally consist of silty sand with varying amounts of creosote wood debris.

Based on the presence of contamination in surface sediment and the hard compact nature of the mudflats with relatively steep slopes towards the channel, the Area 2 Dock shoreline appears to be relatively stable to eroding. Creosote-treated timber pilings, remnants of dock structures, define this area. The main stem of the Multnomah Channel is 100 to 150 feet offshore with a bottom elevation of approximately -15 feet NAVD88 immediately downstream.

2.2.4.3 Cove Area

Prior to development, the Cove Area consisted of a partially vegetated slough. The Cove Area was created between 1948 and 1953 through extensive dredging and filling of historical lowlands (see Figures 7, 8, and 9). The Cove Area is a depositional area, protected from erosional effects of large storm and tidal surges.

The elevation of the river bottom within the cove and shallow inlet immediately downriver of the peninsula ranges between approximately 0 and 10 feet NAVD88 within approximately 100 to 300 feet of the shoreline before sloping steeply towards the main channel of Scappoose Bay. The main channel of Scappoose Bay adjacent to the Cove Area widens to approximately 150 feet with bottom elevations between -7 and -5 feet NAVD88.

After operations ceased in 1960, deposition and burial of the old operational sediment surface occurred throughout the Cove Area. A sediment deposition rate of 0.2 to 1.4 in/yr is implied based on the presence of 1 to 7 feet of fine textured sediment above the old contaminated operational sediment surface characterized by creosote wood debris.

The buried layer of creosote wood debris is interpreted to be the former in-water surface during plant operation. The creosote wood debris ranges from 1 to 7 feet bml beneath much of the Cove Area. The thickness of the creosote wood debris progressively thins and diminishes in magnitude further from the shoreline.

Contaminated sediments in the Cove Area have been characterized to an approximate depth of -25 NAVD88 or about 25 feet bml. Sediments were characterized as silt. Basalt bedrock was encountered at various depths through the Cove Area. Cross-sections along transects A-A', D-D',

F-F', and G-G' (see Figure 19 for cross-section locations) are included as Figures 20, 23, 25, and 26.

2.2.4.4 Upper Milton Creek

Upper Milton Creek's sediment profile consists of up to 1 foot of silty organic-rich material overlying a 1- to 6-foot layer of gray, stiff, clayey-sandy silt with gravel and cobbles. Scouring of the creek channel by seasonal runoff and tidal cycles limits accumulation of sediment in this portion of the creek. The shoreline is generally a relatively steep embankment covered with vegetation.

2.2.4.5 Stormwater

No stormwater management features were identified during a review of historical facility maps.

2.3 PLANT OPERATIONS

The Site was first developed for industrial purposes in 1912 when St. Helens Creosoting Company established a wood treating facility. A sawmill operated in the northeast portion of the Site from 1915 to the mid-1930s (see Figure 3B). Companies acquired by Pope and Talbot, Inc. purchased the property in 1938 and continued to operate the wood treating facility until 1960. The plant was dismantled in the fall of 1960. The Port of St. Helens (now the Port of Columbia County) purchased the property in 1963.

The Site was vacant between 1960 and 1974. Dredge material generated from deepening of the Columbia River was placed on the property in the early 1970s in an effort to level and raise the property's surface grade by several feet (see Figure 12). Since 1974, the Site has either been vacant or leased for the following industrial uses:

- Pole peeling and pole storage facility (1974 to 1991).
- Storage yard for a marine construction/dredging company (1993 to 1998).
- Small private wood-working business (2000 to 2005).
- Marine log salvage and sawmill (2006 to 2008).
- Log storage, pole peeling, and sawmill (2009 to 2012).
- Miscellaneous small-scale private businesses (2014 to present).

Figures showing facility features during different periods are included as Figures 4 through 16.

2.3.1 Physical Plant

While multiple companies have operated at the Site, there have been two primary physical plant layouts.

2.3.1.1 Wood-Treating Facility

The original wood-treating facility constructed in 1912 consisted of a creosote AST, pipeline, a retort/pump house, creosote retorts, transfer table, tram tracks, and dock. The mill consisted of a mill building, power house, and vent burner. By 1929 the wood-treating facility had expanded to include three large volume creosote ASTs, with removal of the original creosote AST, two fuel oil ASTs, a crane fuel AST, blacksmith/machine shop, boiler house, experimental kilns, hog fuel bin, conveyor and hopper, a barge house, and additional docks. An above and below ground pipeline conveyed creosote from the largest creosote AST to the eastern dock (Area 2 Dock). This facility configuration essentially remained until the plant was dismantled in 1960. Figures 3A and 3B shows historical features.

Creosote was delivered to the Site via ships which transferred their contents at the Area 2 dock. An aboveground pipeline pumped the creosote to a large AST from which creosote was pumped to several smaller ASTs. Open "process recovery tanks" collected condensed water from the Boulton treatment process. This process consists of enclosing wood in a treatment cylinder, introducing hot creosote or pentachlorophenol (PCP) solution and establishing a vacuum at 220°F in the cylinder. This process removes water from the wood as water vapor that is condensed outside the cylinder. The collected water then passed through a series of chambers with skimmers to recover any treatment solution with filtered process water discharging to Scappoose Bay (See Figure 3A). All other waste materials were recycled.

2.3.1.2 Post-Peeling Facility

An office and maintenance shop were built in 1974 or 1975 for the pole-peeling operation. Two underground storage tanks (USTs) and two waste oil ASTs were present. The 10,000-gallon diesel fuel UST and the 1,000-gallon gasoline UST were installed in 1975 adjacent to the maintenance shop building. Both USTs were decommissioned in-place in 1990. DEQ assigned UST Facility No. 5196 to the USTs. Figures 3A and 3B shows historical features.

2.3.2 Chemical Use and Waste Generation and Management

The follow section discusses the available information for the two primary operations that were present on the Site.

2.3.2.1 Wood-Treating Facility

Products that were treated included pilings, poles, cross arms, paving blocks, pipe and tank staves, structural timbers, and railroad cross ties. The primary wood-treating formula used was creosote. PCP was reportedly used in 1953 and 1954. Chromated copper arsenate, a wolmanizing salt, was

reportedly used in 1953. No other information on chemical use and waste generation and management from the wood-treating facility has been located.

2.4 REGULATORY HISTORY

Investigations have been performed at the Site and adjacent in-water areas since 1988. The investigations are documented in the administrative file (ECSI No. 959). Information is also available online through DEQ's ECSI database.

2.4.1 Preliminary Assessment (1988) and Site Inspection (1990)

DEQ performed a Preliminary Assessment (PA) of the Site in 1988. The PA identified unanswered questions related to past site uses and disposal practices that may have impacts on sensitive environments and requested the U.S. Environmental Protection Agency (EPA) perform a Site Inspection.

E&E, on behalf of the EPA, performed a Site Inspection from July 1989 to May 1990. The Site Inspection was performed to further assess potential source areas, evaluate contaminant migration pathways and identify areas warranting additional investigation. E&E collected eight subsurface soil samples, one surface soil sample, three seep solid samples, four surface water samples, and five sediment samples. Additional samples were collected off-Site for background purposes. Sample locations are shown on Figure 32. Samples were analyzed for one or more of the following: volatile organic compounds (VOCs), semi-volatile organic compounds (SVOCs), pesticides, polychlorinated biphenyls (PCBs), phenols, and metals.

PAHs associated with creosote were detected in subsurface soil, seep solid, sediment, and surface water samples collected at or immediately downgradient of the former wood treating facilities. PCP was detected in one subsurface soil sample collected near the former PCP and wolmanizing salts retorts.

2.4.2 Site Investigation (1993)

Harding Lawson Associates (HLA) performed a site investigation above the groundwater table from March to April 1993. The site investigation was performed to further assess potential source areas. HLA advanced 32 soil borings, the majority located in Area 1. A total of 51 subsurface soil samples, five surface soil sample, and one seep solid samples were analyzed. Sample locations are shown on Figure 32.

Samples were analyzed for contaminants including gasoline, diesel, heavy oil, VOCs, PAHs, and metals. Gasoline, diesel, heavy oil, VOCs, PAHs (mainly PAHs and PCP), and/or metals were detected in subsurface soil samples. The site investigation concluded the majority of contaminated soil is present in the vicinity of the former wood treating facilities and in the native soil located below the dredge fill material. Concentrations of arsenic were determined to be typical of background with the possible exception of arsenic detected in immediate vicinity of the former wood treatment building.

2.4.3 1995 Consent Order & 2008 Orphan Declaration

DEQ issued an Order on Consent (Consent Order) to Pope and Talbot, Inc. and the Port of St. Helens (now Port of Columbia County). The Consent Order required the parties to conduct a Remedial Investigation and prepare a Feasibility Study.

Pope and Talbot, Inc. declared bankruptcy in November 2007 and ceased performing work under the Consent Order. The Port of St. Helens ceased work shortly after due to financial constraints. DEQ declared the Site an Orphan on April 28, 2008. The Port of St. Helens later identified insurance policies and tendered claims to the insurance companies. The Port of St. Helens then continued the work required under the Consent Order. The insurance policies are expected to cover a portion of the selected remedial action but not the entire cost.

2.4.4 2000 Remedial Investigation

Investigations were performed at the Site in two phases between 1996 and 1998. This investigation included both Area 1 and Area 2. The results of the investigations are included in the April 7, 2000 Remedial Investigation report prepared by GeoEngineers, Inc. The investigations were performed to evaluate the nature, extent, and risk associated with historic releases of hazardous substances at the Site. A summary of samples collected, analysis performed, and matrices sampled throughout the remedial investigation period (1996 through 2017) is provided on Table 1.

2.4.4.1 Phase 1 Investigation (1996 to 1997)

The initial phase of investigation was performed between July 1996 and November 1997. The investigation consisted of upland soil and groundwater and in-water sediment and surface water immediately adjacent to the Site's shorelines. A total of 31 soil borings were advanced, generally until basalt bedrock was encountered, at the Site in July 1996. Monitoring wells were constructed in 22 of the soil borings. Boring and well locations are shown on Figure 32.

Groundwater elevation measurements were collected on a monthly basis with groundwater samples collected quarterly. Six groundwater monitoring events were performed during Phase 1.

Surface water (SW1 through SW4) and sediment samples (SHI01S, SHI01B through SHI12B) were collected adjacent to the Site in Milton Creek, Scappoose Bay, and Multnomah Channel in October and November 1996. Surface water and sediment sample locations are shown on Figure 32.

2.4.4.2 Phase 2 Investigation (1998)

Additional investigation of surface soil, subsurface soil, groundwater, riverbank seeps, surface water, and bedrock was performed in 1998. A total of 26 soil borings (GP1 through GP26) were advanced at the Site in September 1998. One additional monitoring well (MW-18) was constructed. Three borings (RCB-1, RCB-2, RCB-3) were advanced into the basalt to evaluate hydraulic characteristics of the basalt formation. Boring and well locations are shown on Figure 33.

Groundwater from monitoring wells, surface water (SW-MC1, SW-MC2, SW-SB1 through SW-SB7), and a surface soil sample (Bank1) near a seep were collected in September 1998. Seep water samples (Seep1, Seep2, Seep3) were collected in October 1998. Surface samples (SS1 through SS24) were also collected to evaluate the gravel roadways in October 1998. Soil sample and seep locations are shown on Figure 33.

2.4.4.3 Phase 1 & 2 Investigation Results

A visible creosote-petroleum product (NAPL) was observed in subsurface soil samples and in groundwater monitoring wells in Area 1 Upland. In borings completed within close proximity of the former wood-treating operations, creosote impacts were observed throughout the entire thickness of the native soil unit (depths ranging between about 5 to 18 feet bgs). The vertical thickness of creosote-impacted soil appears to decrease with distance from the wood treatment plant and ASTs.

All media samples were submitted for chemical analysis for one or more of the following contaminants of potential concern (COPC): total petroleum hydrocarbons (TPH); VOCs; SVOCs, chlorinated phenols; dioxins and furans; and select metals (arsenic, cadmium, chromium, copper, lead, and zinc). Sediment samples were also analyzed for grain size, sulfides, and total organic carbon (TOC). Bioassay tests were performed on two sediment samples.

In general, TPH, VOCs, and SVOCs associated with creosote were detected in groundwater samples obtained from monitoring wells and soil borings completed in the vicinity of the former wood treating operations. Benzene, toluene, ethylbenzene, and xylenes (BTEX) represents the majority of VOCs detected in groundwater samples. PAHs represent the majority of the SVOCs detected in groundwater samples. Relatively low levels of metals and chlorinated phenols were detected in Area 1 Upland groundwater samples.

During periods of low surface water, small, localized groundwater seeps have been observed along the steeper sections of the Scappoose Bay and Milton Creek shoreline adjacent to Area 1 Upland. Many of these seeps appear to daylight near the fill and native soil interface. Creosote-impacted seeps occasionally are accompanied by a sheen or visible NAPL, although at times the sheen has been a natural (organic) sheen not related to creosote. BTEX and PAHs were detected in October 1998 seep water samples obtained from the Scappoose Bay and Milton Creek shorelines.

Shallow sediment conditions (upper 10 centimeters) along the Area 1 Upland shoreline at the Site were explored by obtaining a series of discrete and composite sediment samples along eight transects and background locations. PAHs were detected in the sediment samples.

Subsequent investigations were conducted to further evaluate the nature and extent of contaminants in sediments in preparation for completion of human health and ecological risk assessments.

2.4.5 Interim Remedial Investigations (1999-2006)

This section discusses actions conducted after the data set collected for the 2000 Remedial Investigation report and before the start of the supplementation remedial investigation data collection.

Annual groundwater monitoring was conducted in 1999, 2000, 2001, 2003, 2004, 2005, and 2006. Groundwater samples were analyzed for TPH, BTEX, PAHs, and dissolved metals (arsenic, cadmium, chromium, copper, lead, and zinc).

Sediment sampling was completed in 2003 (SD-100 through SD-120 and SD-BG-01 through SD-BG05), 2004 (SD-103, SD-104, SD-105, SD-112, SD-117, SD-119, SD-121 to SD-124, SD-MC-A through SD-MC-E), and 2005 (SD-125 through SD-141) as part of an ecological risk assessment (ERA). Offshore surface sediment samples obtained adjacent to the Site and from background locations were analyzed for one or more of the following constituents: PAHs, arsenic, dioxin and furans, TOC, total sulfides, ammonia, grain size distribution, and bioassay testing. Sample locations are shown on Figure 32.

A baseline human health risk assessment (HHRA) and ERA was completed in 2006. The results of the HHRA and ERA are discussed in Sections 3.2.3 and 3.2.4, respectively.

2.4.6 Supplemental Remedial Investigation (2010 – 2018)

The 2000 Remedial Investigation Report presented data collected at the Site through 1998. The initial phases of the remedial investigation were focused on the upland portions of the Site, with only limited characterization of surface water and surface sediment along its 4,500 linear feet of shoreline of Milton Creek, Scappoose Bay, and the Multnomah Channel. Unlike the site-wide assessment described in the 2000 Remedial Investigation Report, the supplemental remedial investigation findings are focused on the Area 1 Upland, offshore areas adjacent to Area 1 within Milton Creek and Scappoose Bay, and the offshore area adjacent to Area 2 within Multnomah Channel near the terminus of the former creosote pipeline and residual pilings associated with a historical dock. Multiple investigations were performed between 2010 through 2018 as part of the supplemental remedial investigation. The actions performed are summarized below. Sample locations are shown on Figure 33.

2.4.6.1 Investigations Performed

Quarterly groundwater monitoring of Area 1 wells was performed between September 2010 and August 2012. Groundwater samples were analyzed for TPH, BTEX, PAHs, and dissolved metals.

Multiple phases of shoreline inspections were conducted during low water conditions (generally in late September and early October) in 2010, 2011, 2012, 2017, and 2018 to designate specific areas where NAPL (i.e., creosote sheen, product, and/or tar body) was observed in near-surface sediment, and to refine the locations and boundaries of visible creosote-impacted groundwater seeps in the bank. Surface sediment samples were collected at areas of obvious or suspected creosote impact and were analyzed for TOC, TPH, PAHs, VOCs, extractable petroleum

hydrocarbon (EPH) and volatile petroleum hydrocarbon (VPH) constituents, and total metals (arsenic, cadmium, chromium, copper, lead, and zinc).

Two phases of upland and offshore NAPL delineation were conducted in 2010 and 2011. The initial delineation in 2010, generally consisted of soil boring equipment equipped with TarGOST® laser-induced fluorescence technology. The second delineation effort consisted of direct observations of sampling cores and sediment, soil, and basalt sample collection for analysis from borings. Soil samples were submitted for TPH and PAHs analysis. Sediment samples were analyzed for TPH, PAHs, TOC, and grain size.

Additional monitoring wells (MW-2A, MW-19, MW-20, MW-21, MW-22, MW-23, MW-24, and MW-25) were installed in 2011 within Area 1 Upland to better delineate the upland extent of creosote NAPL. Groundwater monitoring was performed in July 2011.

In 2012, sediment and porewater sampling was conducted adjacent to Area 1 in Scappoose Bay and Milton Creek to assess the bioavailable or bioaccessible concentrations of PAHs associated with creosote within and near the base of the benthic (biologically active) sediment environment.

To evaluate potential temporal variability resulting from changes in river stage and groundwater elevation, porewater sampling was conducted in July 2012 (high river stage and high groundwater elevation) and October 2012 (low river stage and low groundwater elevation). Whole or bulk sediment samples were analyzed for 18 parent PAH compounds and 16 groups of alkylated PAH homologues, TOC, and black carbon. Sediment porewater samples were analyzed for 34 PAHs (parent and alkylated PAHs)

In August 2013, the dilapidated wooden deck of the former transfer table dock was removed to facilitate the October 2013 delineation of creosote-related contamination within the underlying nearshore soil and sediment. Soil and sediment samples from varying depths (dependent upon field observations) were visually screened for creosote NAPL and submitted for TPH and PAHs analysis.

A phased offshore data gap investigation consisting of a high-accuracy bathymetric survey, offshore groundwater discharge survey, sheen mapping in surface sediment, and in situ porewater and surface water passive sampling was completed between April 2017 and November 2018, to further assess the offshore extent of creosote NAPL and the risks that bioavailable fractions of PAHs and TPH pose to aquatic invertebrate receptors outside the inferred NAPL areas. Surface sediment samples were analyzed for 34 PAHs, VPH, EPH, VOCs, TOC, and black carbon. Time integrated (20 days) surface water (1 foot above mudline) and sediment porewater (two depth intervals of 3 to 8 inches and 24 to 29 inches bml) were sampled using passive sampling devices and analyzed for freely dissolved PAHs (34 and 63 compounds), VPH, oxygenated PAHs (OPAHs), and hopane and cholestane compounds.

2.4.6.2 Investigation Results

The conceptual site model (CSM) was refined to better account for offshore sediment contamination (particularly NAPL along the Site's shorelines), as well as upland NAPL migration and dissolved-phase contaminant transport. A schematic CSM is included as Figure 34.

The lateral and vertical extent of NAPL in offshore sediment adjacent to the Site was defined. Wood debris in surface sediment along 700 feet of the Area 1 shoreline is a source of NAPL beneath the former hog fuel dock and transfer table dock structures and extends approximately 75 feet offshore. The creosote contaminated wood debris beneath the former transfer table dock extends approximately 200 feet offshore.

Although upland groundwater discharges to sediment porewater and surface water within Milton Creek and Scappoose Bay, the risk to human health and ecological receptors from dissolved-phase constituents transported via groundwater appears to be low outside the observed areas of direct NAPL seepage.

The updated supplemental HHRA indicated low potential risk to human receptors other than direct contact with NAPL and Area 1 groundwater by excavation workers and direct contact with NAPL by shoreline anglers and/or transients.

The updated supplemental ERA indicates that the presence of creosote NAPL (inclusive of moderate to heavy petroleum sheen) presents unacceptable ecological risk to benthic invertebrates and fish where it is observed in surface sediment. In general, the 2017 surface sediment porewater sampling indicated dissolved-phase constituents in deeper sediment and/or upland groundwater do not adversely impact benthic invertebrates outside the prior remedial action areas.

The hot spot evaluation indicated that groundwater and sediment within the delineated extent of NAPL and NAPL-affected groundwater seeps are hot spots in accordance with DEQ guidance. Dissolved-phase creosote constituents in porewater and surface water that cause unacceptable risk to benthos and fish are also potential hot spots.

3.1 NATURE AND EXTENT OF CONTAMINATION

As summarized previously, investigations of the Site commenced in 1988 and continued through 2018. While the early investigations (e.g., 1988 PA, 1990 Site Inspection, 1993 Site Investigation, 2000 Remedial Investigation) included sampling in the entire Upland Area, the Supplemental Remedial Investigation focused primarily on Area 1 Upland, offshore areas adjacent to Area 1 within Milton Creek and Scappoose Bay, and the offshore area adjacent to Area 2 within Multnomah Channel near the terminus of the former creosote pipeline and residual pilings associated with a historical dock. The results of the Supplemental Remedial Investigation included the identification of five PAAs by contaminants including creosote (TPH) and PAHs. The nature and extent of contaminants are summarized below.

The Locality of Facility (LOF) is defined as any point where a human or ecologic receptor contacts or is reasonably likely to come into contact with contaminants from the Site. The LOF takes into account the likelihood of the contaminants migrating over time. For the purpose of this ROD, the LOF is the soil and shallow water-bearing zone above Basalt Bedrock within Area 1 Upland and portions of sediment and surface water in Milton Creek and Scappoose Bay. The extent of the LOF is shown on Figure 2.

3.1.1 Non-Aqueous Phase Liquid

Creosote NAPL has been encountered in soil, groundwater, groundwater seeps, and sediment at the Site. The NAPL composition is dominated by diesel- and oil-range hydrocarbons (although chromatograms do not match a diesel or oil fingerprint) and PAHs (naphthalene, phenanthrene, 2-methylnaphthalene, and others). The majority of creosote NAPL appears to have been trapped by capillary forces within the native soil before reaching the basalt bedrock. The total thickness of creosote impact varies from 1 to 8 feet, and the impacted intervals typically occur at depths greater than 10 feet bgs (below the Fill Zone). The estimated lateral extent of NAPL and creosote sheens are shown on Figures 35, 36, and 37.

3.1.1.1 Area 1 Upland Non-Aqueous Phase Liquid

The Area 1 Upland extent of NAPL, although not continuous throughout, has been generally defined within an approximately 4-acre area. The distribution of NAPL beneath the former operations area appears to be a function of subsurface migration from the former creosote storage tanks, plant, and retorts as well as direct releases from other ancillary operations (e.g., the transfer table).

Potentially mobile creosote is present in the native soil at various depths below the fill and above the basalt bedrock and within distinct stratigraphic layers that are less than 18 inches thick. The total thickness of creosote varies from 1 to 8 feet. In general, this vertical layering or profile of NAPL becomes thinner and deeper within the native soil farther from the former operations area at depths greater than 10 feet bgs. The inferred extent of NAPL indicates that topographical highs on the basalt surface, such as the ridge to the east of the former operations area along the Scappoose Bay shoreline, appear to influence the subsurface distribution of NAPL. The bedrock appears to be a barrier to downward vertical migration of the NAPL.

3.1.1.2 Area 1 Dock Non-Aqueous Phase Liquid

Between the former hog fuel dock and northeastern (downriver) terminus of the former transfer table dock (Area 1 Dock), discontinuous, isolated pockets of creosote NAPL have been observed in many locations at or near the sediment surface.

The source of sediment contamination is creosote contaminated and/or treated wood debris. This debris differs in character from debris found in the Cove Area. Much of the wood debris is finer textured and is described as "pulverized wood" in sediment sampling logs. The pulverized wood is interpreted to be hog fuel (i.e., wood shavings/chips) likely spilled from the hog fuel conveyor and during over-water transfers from ships/barges to the hog fuel hopper and conveyor system. Some of this material appears to have become saturated with creosote and accumulated along the shoreline and in sediment surrounding, and downstream of, the former hog fuel dock.

Additionally, some of the contaminated wood debris appears to be related to deterioration of the creosote-treated dock structures (i.e., timber piles, lateral bracing, decking) that existed along this portion of the Area 1 shoreline. The distribution and thickness of the woody debris is highly variable in this portion of the Site, ranging from completely absent to more than 11 feet thick (sediment boring SB-16). The lateral extent of surface sediment and/or wood debris capable of producing a creosote-related sheen on the overlying water surface when disturbed extends approximately 75 feet offshore between the former hog fuel dock and transfer table dock structure.

Within the western half of the primary wood dock structure of the former transfer table, thick sequences of contaminated wood debris are intermingled with dimensional lumber, logs, metal ties, and other debris. Heavy sheens are easily generated by disturbing the surface sediment around many of the remaining log pilings.

Creosote NAPL associated with wood debris was primarily observed in the upper 24 inches of sediment within the central and western portions of the former transfer table dock area. Deposited wood and creosote-related contamination were observed from the surface to approximately 9 feet bml in the far western portion of the former transfer table dock. In general, the thickness of the wood mass decreased to the northeast away from the southwest area of the former transfer table dock and is not present in significant thickness to the north of boring DAI-HA-02 or to the east of location DAI-PH-11. The lateral extent of surface sediment creosote contaminated wood debris extends approximately 200 feet offshore within the footprint of the former transfer table dock.

An upland and slightly in-water area of subsurface creosote staining and isolated blebs of NAPL were encountered in the native soil beneath the northwest corner of the former transfer table dock. Creosote impacted soil with residual NAPL decreases in thickness and is encountered at deeper intervals within the native soil as distance increases in all directions from the northwest corner of the former transfer table dock. This corner of the former transfer table dock is covered by recently

deposited sand (natural deposition) and there is no visible surface evidence of impacted groundwater seeps from this upland area of NAPL.

3.1.1.3 Area 2 Dock NAPL

In October 2017, DEQ staff inspected the Area 2 shoreline and observed creosote-impacted surface sediment near the terminus of the former creosote pipeline and residual piling associated with a historical dock (Area 2 Dock). Creosote sheen occurrence was mapped in surface sediment along the approximately 900-foot-long portion of the Scappoose Bay shoreline. Moderate to heavy petroleum sheens were noted only in borings amongst the pilings near the shoreline that encountered creosote-contaminated wood debris within the upper 12 inches of sediment. Sediment poling beyond 50 feet of the shoreline (even amongst the pilings), did not produce a visible sheen. The lateral extent of creosote contaminated wood debris covers an approximately 200-foot-long by 50-foot-wide section (0.2 acre) of the Area 2 shoreline beneath and immediately downriver of the former dock and creosote AST.

3.1.1.4 Cove Area NAPL

Creosote and fuel oil ASTs were immediately upland of the Cove Area and a hog fuel boiler and pole peeling station operated along the shoreline. Based on historical photographs and information on plant operations and its subsequent demolition, significant amounts of creosote contaminated hog fuel, tree bark, and dimensional lumber appear to have accumulated in the Cove Area. NAPL has migrated from these upland source areas into approximately 0.8 acre of the Cove Area.

Heavy petroleum sheens are visible in seasonal groundwater seeps along a 300-foot section of bank seepage west of the peninsula. Near-shore Cove Area borings typically encountered 6 to 8 feet of NAPL impacted sediment, with a maximum thickness of 12 feet observed in SB-11. The near-shore presence of moderate to heavy petroleum sheens in surface sediment appears to be related to the NAPL/groundwater seeps, the presence of creosote-treated wood debris, and gas ebullition resulting from biodegradation of organic matter in deeper sediments.

Moderate to heavy petroleum sheens have been detected in subsurface sediment up to 220 feet offshore beneath the Cove Area (borings SB-04 and SB-09). In general, creosote-impacted sediment decreases in thickness and is encountered at deeper intervals within the Cove Area as distance increases from the shoreline. Outside the inferred extent of upland NAPL, the vertical mobilization of sheen via ebullition has not been observed in the Cove Area.

Riverward of the solid red NAPL boundary depicted on Figure 37, the source of subsurface sediment contamination transitions to creosote contaminated wood debris. This buried layer of creosote wood debris is interpreted to be the old in-water operational surface while the plant was active and during its demolition. Creosote-impacted wood debris ranges from 1 to 7 feet bml beneath much of the Cove Area, and progressively thins and diminishes in magnitude outward from the shoreline. Vertically, contaminant levels attenuate rapidly within the sediment stratigraphic layers. Sediment within the western portion of the Cove Area is generally unimpacted by upland NAPL migration and/or creosote contaminated wood debris.

Light or slight petroleum sheens have been intermittently observed in the relatively clean, more recently deposited surface sediment beyond the solid NAPL boundary depicted on Figure 37. Based on the results of bulk sediment and depth-discrete porewater sampling, these slight surficial sheens appear to be attributed to localized particles of creosote-treated wood debris in the surface sediment and do not appear to be attributed to NAPL migration from the underlying sediment.

3.1.1.5 Upper Milton Creek Area NAPL

Groundwater seepage with a creosote-related sheen has been observed along a 200-foot section of the east bank of the Upper Milton Creek and analytical results of bank samples indicate contact with NAPL. The Upper Milton Creek in-water area is located over 350 feet west of any known historical wood-treating features and/or operations. Field evidence of NAPL was generally not observed in borings.

The primary sources of contamination to the Upper Milton Creek in-water area are heavy petroleum sheens seeping from a relatively thin (1-foot-thick or less) sand layer situated at an approximate depth of 19 to 20 feet below the top of the bank. Biogenic sheens also seep from the eastern bank of Milton Creek and are intermingled with petroleum sheens. The platy relatively odorless biogenic sheens, which are easily distinguished from the fluid, odoriferous creosote related petroleum sheens likely emanate from areas where upland creosote contamination and wood debris is more completely degraded.

3.1.2 Groundwater

Groundwater monitoring has been performed frequently enough to provide understanding of groundwater conditions. Residual NAPL continues to be a source for dissolved-phase creosote-related constituents in shallow groundwater beneath the Site. The extent of PAH and TPH groundwater contamination is shown on Figures 38 and 39, respectively.

The highest concentrations of PAHs and TPH in groundwater were generally detected in monitoring wells MW 22, MW-2A, and MW-3A, within the inferred NAPL area, followed by detections in wells at the southeastern and northeastern edges (MW-23, MW-24, and MW-25) of the former wood treating operations area. Detections of PAHs were generally lower in wells to the west of and outside of the inferred upland NAPL area and in wells at the far eastern edge of Area 1. Analytical results from groundwater samples collected during the supplemental remedial investigation are representative of current Site conditions and are summarized below.

Total PAHs were detected in groundwater samples collected within or near the former operations area, with a similar distribution of relative concentrations as TPH. Total PAHs detected in groundwater ranged in concentration from 0.35 micrograms per liter (μ g/L) at well MW-18 to a maximum of approximately 26,800 μ g/L at well MW-3A in 2010; NAPL was also detected at MW-3A during the 2010 monitoring event. The maximum total PAHs in groundwater in 2011/2012 was approximately 22,500 μ g/L in well MW-22.

Petroleum hydrocarbons indicative of creosote were detected in groundwater samples collected within or near the former operations area. TPH detections were primarily in the diesel range. The maximum diesel-range hydrocarbon concentration was detected in well MW-3A at 374 milligrams

per liter (mg/L) in 2010. NAPL was also detected at MW-3A and sampled during the 2010 monitoring event. The maximum diesel-range hydrocarbon concentration in 2011/2012 was 116 mg/L at MW-3A. The maximum concentration of TPH in the oil range was 103 mg/L at MW-3A in 2010; the maximum oil-range hydrocarbon concentration in 2011/2012 was 3.81 mg/L at MW-2A.

Metals detected in groundwater were arsenic, chromium, copper, lead, and zinc. The distribution of metal detections does not generally correlate with NAPL or the former wood treating operations area. Arsenic was detected in every well sampled at concentrations up to 17.5 μ g/L at MW-23. Chromium was detected infrequently just above reporting limits in MW-3B and MW-5, and at slightly higher concentrations (maximum of 5.83 μ g/L) at MW-2A. Copper was detected at MW-5 at a maximum concentration of 18.5 μ g/L; the only other detection of copper was just above the reporting limit at MW-4B. Lead was detected infrequently in wells MW-3B, MW-5, and MW-9, at levels up to 3.48 μ g/L. The maximum zinc detection, 683 μ g/L, was at MW-20 in 2011; other detections at MW-20 were below 10 μ g/L. Well MW-5, located near the northern property boundary, has had consistent detections above 200 μ g/L. All other zinc concentrations were below 200 μ g/L, with most below 10 μ g/L.

3.1.3 Soil

The fill and native soil contact generally represents the original ground surface at the time of woodtreating facility operation. This results in the creosote-impacted soil in the vicinity of the former wood-treating facilities being generally limited to the native soil unit, located beneath 5 to 10 feet of dredge sand fill. In borings completed in close proximity to the former wood-treating plant and former creosote ASTs, varying degrees of creosote impact have been observed throughout the entire thickness of the native soil unit with depths ranging between about 5 feet to approximately 23 feet bgs at the basalt bedrock contact. In general, creosote-impacted soil decreases in vertical thickness and is encountered at deeper intervals within the native soil stratum as distance increases from the former wood-treating operations.

Creosote-related hydrocarbons (diesel and oil ranges) were detected in soil samples collected from 19 of 28 explorations completed in Area 1, with a maximum concentration of 6,320 mg/kg (sum of diesel- and oil-range hydrocarbons) in the MW-24 boring. Concentrations of total PAHs in soil ranged from not detected to a maximum of 3,410 milligrams per kilogram (mg/kg) in boring MW-24.

3.1.4 Sediment

The in-water areas were divided into multiple areas during the remedial investigations. However, this ROD focuses on the four in-water PAAs (Area 1 Dock, Area 2 Dock, Cove Area, and Upper Milton Creek). Since the 2000 remedial investigation report, over three hundred additional sediment samples from the Scappoose Bay and Milton Creek shorelines along Area 1 have been analyzed for PAHs, TPH, VPH/EPH, VOCs, TOC, and/or black carbon. Additionally, subsurface information was obtained via the 67 borings and depth discrete sampling, various sediment sampling events, and multiple shoreline inspections. Analytical results from sediments samples

collected during the supplemental remedial investigation are representative of current Site conditions and are summarized below. Sample locations are shown on Figures 32 and 33.

Petroleum hydrocarbons were detected in sediment samples collected along the Scappoose Bay shoreline. These were primarily diesel-range hydrocarbons with some oil-range hydrocarbon detections. A maximum of 14,000 mg/kg of diesel-range hydrocarbons was detected in sample SB-11 at 4 feet bml. Sample SB-11 was located within the estimated NAPL extent just west of the peninsula.

Total PAHs in the 2011 to 2013 surface sediment samples ranged from not detected to a maximum of approximately 4,800 mg/kg in SSED-6 located east of the former hog fuel dock within an area of creosote wood debris. Total PAHs in the 2017 surface sediment samples, all obtained outside inferred NAPL areas, ranged from 1.58 mg/kg (PWS-090517-6) to 53.54 mg/kg (PWS-090517-9). As stated above, the highest relative total PAH concentrations were found within the old buried operational surface and/or within the inferred areas of offshore NAPL.

TOC concentrations in surface sediment samples ranged from 520 mg/kg at SB-23 to 150,000 mg/kg at SSED-6. These locations were east of the former hog fuel dock and within 25 feet of each other. The average TOC concentration was approximately 21,000 mg/kg. TOC content varied across the investigated area, and there does not appear to be a discernable pattern of high or low concentrations by locality. In general, samples with higher TOC were observed to contain wood debris, including wood chips, sticks, and roots.

The percentage of black carbon in surface sediment samples ranged from 0.10 to 0.34 (with the exception of one location at which black carbon was not detected), with an average of 0.18 percent (%) black carbon. Similar to TOC, black carbon content varied across the investigated area with no discernable pattern of high or low concentrations by locality.

As expected, the highest concentrations of PAHs in sediment occurred within the area of the inferred extent of NAPL, although NAPL was not observed in all samples collected within this area. Although TPH was not tested in sediment as frequently as PAHs, the highest TPH detections were also observed within the inferred extent of offshore NAPL.

3.1.5 Sediment Porewater

Sediment porewater samples were only collected during the supplemental remedial investigation. The majority of porewater samples were collected outside of the offshore inferred NAPL areas or outside of the in-water PAAs.

All sediment porewater concentrations reported from the ex-situ Solid-Phase Microextraction (SPME) and in-situ low density polyethylene (LDPE) passive samplers, are freely dissolved water concentrations (C_{free}). In accordance with EPA guidelines, the direct measurements of the freely dissolved 34 PAH concentrations in sediment porewater or interstitial water (IW) are used along with their expected final chronic value (FCV) water/lipid partitioning behavior to calculate a hazard quotient, referred to as a toxic unit (TU). The narcosis based Tus are considered additive, and evaluation of the sum of the Tus requires that all 34 PAH analytes be included in the sum of IW TU (Σ IWTU). Calculations of Σ IWTU have been used on this project as a benchmark for

predicting the toxicity of PAHs to benthic invertebrates with a potential for unacceptable risk when the Σ IWTU is greater than or equal to (\geq) 1.

The results of depth-discrete porewater sampling were used to evaluate dissolved phased mass flux from deeper impacted sediment. In general, detected concentrations of dissolved PAHs in "deep" sediment porewater beneath the biologically active zone (i.e., 24 to 29 inches bml) were higher than those measured in the overlying "shallow" sediment and surface water. Porewater sampling conducted beyond the delineated areas of NAPL indicate that creosote constituents attenuate rapidly in the sediment pore space between 2 and 1 foot bml.

3.1.6 Surface Water

Surface water analytical results from 2017 are representative of current Site conditions and were used to determine nature and extent. Low levels of C_{free} PAHs were detected in all eight 2017 surface water samples. Total PAHs measured on the in-situ LDPE passive samples after 20 days in surface water ranged from 0.0192 µg/L (surface water sample co-located with PWS-090517-3) to 0.0797 µg/L (surface water sample co-located with PWS-090517-10).

The 18 parent PAH and 16 alkylated PAH homologue compounds (34 PAHs) results in surface water were compared to the FCVs and their corresponding Tus summed (Σ SWTU). Surface water samples co-located with PWS-090917-9, PWS-090517-10, PWS-090617-12, and PWS-090617-13 produced a sum of surface water toxic unit (Σ SWTU) \geq 1.

In general, 34 PAH levels in the 2017 surface water samples were either consistent or slightly higher than detected concentrations in co-located shallow sediment porewater. The two notable exceptions to this data trend are PWS-090617-2 and PWS-090617-12 (located in Lower Milton Creek and the Area 1 Dock, respectively), where detected concentrations of PAHs in shallow porewater are higher than levels measured in the overlying surface water samples.

3.1.7 Air

Airborne dust transport is not a likely significant contaminant migration pathway. The original ground surface at the time the former wood-treating facility operated is currently located beneath 5 to 10 feet of non-contaminated dredge sand fill.

Gravel roads built atop the clean fill (at current ground surface) reportedly had oil applied to them between 1974 and 1991 for dust suppression. PCBs were detected in surface soil samples collected from former oiled gravel roadbeds at the Site.

3.2 RISK ASSESSEMENT

The standards for a protective cleanup are defined in the ORS and OAR. ORS 465.315 sets standards for degree of cleanup required, risk protocol, hot spots of contamination, etcetera (etc.). OAR 340-122-0084 describes the requirements for risk assessments while OAR 340-122-0115 provides additional definition of protectiveness.

Baseline risk assessments were initially conducted in 2006 to evaluate risks to human health and ecological receptors. The results of the baseline risk assessments are summarized in the Baseline Human Health Risk Assessment (Bridgewater Group and Kennedy/Jenks Consultants, 2006a) and the Baseline Ecological Risk Assessment Summary Report (Bridgewater Group and Kennedy/Jenks Consultants, 2006b). Updated supplemental HHRA and ERAs, incorporating new data as well as additional exposure scenarios, were completed in 2020 and are summarized in Appendices F and G of the Updated Supplemental Remedial Investigation Report, respectively (Cascadia Associates, 2020a; Cascadia Associates, 2020b).

The results of the risk assessments that have been conducted for human health and ecological receptors at the former Pope & Talbot site are summarized below. The residual risk associated with the selected remedy is summarized in Section 10.2.

3.2.1 Conceptual Site Model

This ROD addresses contamination in Area 1 Upland as well as within the nearshore sediment areas adjacent to the Site within Scappoose Bay and Milton Creek. The primary source of contamination at the Site is releases from historical industrial activities during plant operation (see Section 2.3). As discussed in Section 3.1, creosote is the primary contaminant in soil, groundwater, and sediment samples collected in the vicinity of the former wood-treating facilities. During plant operations, leaks and spills of creosote NAPL seeped into the ground and collected in the native soil above the basalt bedrock within an approximately 4-acre portion of the Area 1 Upland. A schematic of the CSM is shown on Figure 39.

Creosote is a multi-component NAPL that contains many hydrocarbons, primarily PAHs, phenolic compounds, and carrier fluids such as diesel. In the approximately 60 years since the plant closure, subsurface creosote (either as free phase NAPL, sorbed onto solids, or dissolved in water) appears to have undergone compositional changes due to weathering (e.g., loss of more soluble or biodegradable components). Subsequently, residual Site contamination consists predominantly of creosote-derived PAHs (i.e., no light NAPL or evidence of carrier fluids).

Another significant source of in-water impacts is residual creosote-contaminated wood debris from historical operations and historical infrastructure (timber piling and dock structures). Much of this wood debris appears to have been shed from the upland, over-water operations, and/or from log rafts historically moored along the shoreline. The woody material, much of it saturated with creosote, settled out over time and eventually was incorporated into the offshore sediment profile. In general, Site use and tenant activities since the wood treating operations ceased in 1959 do not appear to have significantly caused or contributed to the contamination levels encountered during recent investigations.

Current contamination sources in the upland include contaminated groundwater and soil. Volatile compounds in groundwater have the potential to volatilize into indoor and outdoor air. Contaminated groundwater has the potential to migrate to surface water, sediment, sediment porewater, and surface water in the in-water area. Current contamination sources in the in-water area include sediment/beach soil and contaminated woody debris.

The **CSM for evaluation of human health risks** is summarized in Figure 40. The following human health exposure scenarios are considered potentially complete for the Site:

- Industrial workers through ingesting and coming into direct contact with upland surface soil; inhaling contaminants volatilized from upland surface and subsurface soil; ingesting upland surface soil; and inhaling contaminants volatilized from upland groundwater.
- Excavation workers through ingesting and coming into direct contact with upland surface and subsurface soil; inhaling contaminants volatilized from upland surface and subsurface soil; ingesting upland surface and subsurface soil; and direct contact with upland groundwater.
- Construction workers through ingesting and coming into direct contact with upland surface and subsurface soil; inhaling contaminants volatilized from upland surface and subsurface soil; ingesting upland surface and subsurface soil; and direct contact with upland groundwater.
- Sport fishers through ingesting fish and shellfish impacted by contaminants bioaccumulated from surface water; ingesting and coming into direct contact with beach soil/sediment; inhaling contaminants volatilized from beach soil/sediment; and ingesting surface water.
- Subsistence fishers through ingesting fish and shellfish impacted by contaminants bioaccumulated from surface water; ingesting and coming into direct contact with beach soil/sediment; inhaling contaminants volatilized from beach soil/sediment; and ingesting surface water.
- Recreational trespassers, transients, and adult boaters through ingesting and coming into direct contact with beach soil/sediment; inhaling contaminants volatilized from beach soil/sediment; coming into direct contact with surface water; ingesting surface water; and coming into direct contact with sediment/sediment porewater.

The CSM for evaluation of ecological risks is summarized in Figure 41.

The following **terrestrial** ecological exposure scenarios are considered complete or potentially complete for the Site:

- Upland plants through direct contact with upland surface and subsurface soil.
- Soil invertebrates through ingestion of and direct contact with upland surface and subsurface soil.
- Herbivorous birds and mammals through ingestion of and direct contact with upland surface and subsurface soil; ingestion of upland biota; ingestion of and direct contact with beach soil/near-shore sediment; and ingestion of and direct contact with surface water.

- Insectivorous birds and mammals through ingestion of and direct contact with upland surface and subsurface soil; ingestion of upland biota; ingestion of and direct contact with beach soil/near-shore sediment; and ingestion of and direct contact with surface water.
- Carnivorous birds and mammals through ingestion of and direct contact with upland surface and subsurface soil; ingestion of upland biota; ingestion of and direct contact with beach soil/near-shore sediment; and ingestion of and direct contact with surface water.

The following **aquatic** ecological exposure scenarios are considered complete or potentially complete for the Site:

- Wetland and aquatic plants through direct contact with beach soil/near-shore sediment, offshore sediment, sediment porewater, and surface water.
- Benthic invertebrates through ingestion of and direct contact with beach soil/near-shore sediment, offshore sediment, and surface water; direct contact with sediment porewater; and ingestion of aquatic biota.
- Fish, through ingestion of and direct contact with beach soil/near-shore sediment, offshore sediment, and surface water; direct contact with sediment porewater; and ingestion of aquatic biota.
- Insectivorous birds, through ingestion of and direct contact with upland surface soil, beach soil/near-shore sediment, and surface water; and ingestion of upland and aquatic biota.
- Piscivorous mammals and birds through ingestion of and direct contact with upland surface and subsurface soil; ingestion of upland biota; ingestion of and direct contact with beach soil/near-shore sediment, ingestion of and direct contact with surface water; and ingestion of aquatic biota.

3.2.2 Human Health Risk Screening

The HHRA evaluated PAHs, TPH, select VOCs, select metals, PCBs in soil, and dioxins and furans in sediment, as requested by DEQ (Cascadia Associates, 2020a). The constituent classes evaluated in the human health risk screening for each environmental medium are summarized below:

Air. No air data have been evaluated from the Site.

Sediment inside the area of inferred NAPL. The Updated Supplemental HHRA does not evaluate data from sediment samples collected within areas of inferred NAPL because exposure to NAPL is considered to have unacceptable risk and therefore does not require further evaluation.

Sediment outside the area of inferred NAPL. PAHs, TPH, and dioxins and furans were evaluated. The 2006 HHRA evaluated discrete and composite sample results collected in 1996, as well as a small number of samples collected in 2005, but it excluded multiple

sediment samples collected in 2003 and 2004. Therefore, DEQ determined that a comprehensive quantitative assessment of all available sediment data was appropriate in the supplemental HHRA (DEQ, 2014a).

Soil. To adequately reflect current risk assessment guidance, risk from PCBs in upland soil was re-evaluated in the supplemental HHRA using current risk-based criteria (DEQ, 2010) for all Site receptors (industrial worker, excavation worker, and construction worker). Per DEQ's request, risk from select COPCs to construction workers was also reassessed.

Sediment Porewater. Human receptors are not exposed to sediment porewater as a separate media and a separate evaluation of porewater was not conducted (DEQ, 2014b), except to compare porewater concentrations to screening levels values (SLVs) protective of people consuming fish that have accumulated Site-related chemicals.

Surface Water. Eight surface water samples were collected concurrently with porewater samples in 2017 and analyzed for 34 PAHs. These concentrations were screened against Preliminary Remediation Goals (PRGs), as shown in Table F-2E of the Supplemental HHRA (Cascadia Associates, 2020a). Surface water samples were also evaluated for protection of people consuming fish from carcinogenic PAHs by comparing them to the same Site-specific SLVs as for porewater.

Groundwater. The 2006 HHRA evaluated groundwater data collected between 1996 and 2004 and concluded that the potential exposure scenario of groundwater exposure by an excavation worker resulted in risk estimates exceeding DEQ acceptable risk levels. PAHs were therefore retained as a COC in groundwater based on the findings of the 2006 HHRA (Bridgewater Group and Kennedy/Jenks Consultants, 2006a). The Updated Supplemental HHRA evaluated all groundwater data collected from Area 1 Upland between 2005 and 2012, including those collected from monitoring wells within the inferred extent of subsurface NAPL. Monitoring well locations and inferred extent of subsurface NAPL are shown on Figure 39. Groundwater from Area 2 Upland was not evaluated because this portion of the Site received an NFA determination from DEQ in 2008.

The 2020 Supplemental HHRA evaluated risks associated with PAHs, TPH, BTEX, and select metals (arsenic, cadmium, chromium, lead, and zinc) in groundwater.

3.2.3 Human Health Risk Assessment

The 2006 Baseline HHRA and 2020 Supplemental HHRA (Bridgewater Group and Kennedy/Jenks Consultants, 2006a; Cascadia Associates, 2020a) describe in detail the procedures used to evaluate the potential risks to human health associated with the chemicals in the media described above. The findings of the HHRAs are summarized below.

Chemicals of Potential Concern.

The HHRAs identified the following COPCs for each of the contaminated media on the Site:

Sediment. All PAHs, TPH, and dioxins and furans detected in sediment were retained as sediment COPCs.

Soil. COPCs from the 2006 HHRA were retained as COPCs in the Updated HHRA (acenaphthene, methylene chloride, benzo(a)anthracene, benzo(a)pyrene, benzo(b)fluoranthene, benzo(k)fluoranthene, chrysene, indeno(1,2,3-cd)pyrene, 2-methylnaphthalene, fluoranthene, fluorene, naphthalene, phenanthrene, pyrene, PCP, arsenic, and lead). A full list of selection criteria for COPCs is provided in Table 1 of the Baseline HHRA (Bridgewater Group and Kennedy/Jenks Consultants, 2006a). PCBs were also retained as COPCs in soil in the Supplemental HHRA.

Surface Water. PAHs were retained as surface water COPCs.

Groundwater. The selection of groundwater COPCs from the supplemental HHRA is illustrated in Table F-1 of the Supplemental HHRA (Cascadia Associates, 2020a). Eleven PAHs and TPH were retained as groundwater COPCs. No metals or VOCs were retained as groundwater COPCs.

Pathway Analysis. The Baseline HHRA identified the following primary human health exposure pathway of concern: potential future on-site excavation workers exposed to PAHs in contaminated groundwater in Area 1 Upland of the Site. The Supplemental HHRA further evaluated exposures to recreational and subsistence anglers consuming fish from Scappoose Bay, in addition to the rest of the exposure scenarios listed in Section 3.2.1, incorporating more recent data and subsequent input from DEQ.

Cumulative Risk. The Supplemental HHRA evaluated cancer risk using the Central Tendency Exposure (CTE) and Reasonable Maximum Exposure (RME) methods to evaluate risks to human health associated with the Site for on-site industrial workers, on-site excavation workers, on-site construction workers, recreational trespassers in Scappoose Bay, recreational trespasser in Milton Creek, sport and subsistence fishers in Scappoose Bay, and sport and subsistence fishers in Milton Creek (see Table 2).

3.2.3.1 Cumulative Human Health Risk Estimates

To evaluate cumulative cancer risk, the Total Lifetime Excess Cancer Risk was calculated for each receptor. To evaluate non-cancer risk, the total hazard indices were calculated for each receptor. Resulting cumulative cancer and non-cancer risk estimates for each receptor are discussed in detail in Section 6 of the Supplemental HHRA and are presented in Table 2.

Excess lifetime cancer risk estimates did not exceed Oregon's environmental cleanup standard of 10⁻⁶ for exposure to individual carcinogenic substances or 10⁻⁵ for multiple carcinogenic substances for the following human health exposure scenarios: industrial workers, excavation workers, construction workers, recreational trespassers in Scappoose Bay, recreational trespassers in Milton Creek, or sport fishers in Scappoose Bay. Excess cancer risk estimates were not derived for sport or subsistence fishers in Milton Creek due to insufficient sediment data.

Non-cancer hazard indices did not exceed Oregon's environmental cleanup standard of one for the following exposure scenarios: recreational trespassers in Scappoose Bay, recreational trespassers

in Milton Creek, sport fishers in Scappoose Bay, subsistence fishers in Scappoose Bay, sport fishers in Milton Creek, or subsistence fishers in Milton Creek.

The following section summarizes risk estimates exceeding Oregon's environmental cleanup standard for exposure to carcinogenic and non-carcinogenic substances for each receptor exposure scenario.

Cumulative Cancer Risk Estimate Exceedances.

Adult/child subsistence fishers in Scappoose Bay. For adult/child subsistence fishers in Scappoose Bay, the excess cancer risk estimate exceeded Oregon's environmental cleanup standard of 10⁻⁶ for exposure to individual carcinogenic substances or 10⁻⁵ for multiple carcinogenic substances. The total excess cancer risk estimate was $6x10^{-5}$ and $3x10^{-5}$ for RME and CTE (see Table 2). The HHRA recognized that both recreational and subsistence anglers could provide fish to children so both adult and child receptor scenarios were considered. These risk estimates are based on bioaccumulation of COPCs from sediment, surface water, and/or porewater. There is unacceptable risk for the subsistence fisher from the consumption of fish exposed to Site-related dioxins and furans in the Scappoose Bay area; exceedances were mostly limited to detected concentrations in samples collected from within the extent of known petroleum sheen and are therefore anticipated to be co-located with hydrocarbons in the areas of petroleum sheen (Cascadia Associates, 2020a, Section 6.6; Table F-2C). Exposure scenarios for bioaccumulation from surface water and sediment indicate a low level of concern for bioaccumulation of carcinogenic PAHs. Bioaccumulation from porewater is considered a secondary line of evidence (LOE) and was used as a supplement to evaluate risk.

Cumulative Non-Cancer Risk Estimate Exceedances.

Industrial workers. For industrial workers, the total hazard index exceeded Oregon's environmental cleanup standard of one. The total hazard indices were 770 and 54 for RME and CTE, respectively (see Table 2).

Elevated risk estimates are dominantly associated with industrial workers inhaling contaminants volatilizing from groundwater into indoor and outdoor air in the upland. The risks associated with this pathway are based on exposure to volatilization of petroleum hydrocarbon concentrations in localized groundwater samples (MW-3A and MW-22) within the extent of inferred subsurface NAPL in Area 1 Upland (Cascadia Associates, 2020a, Tables F-6A, F-6B, F-6C, F-6D). The inhalation risk associated with volatilization to outdoor and indoor air contributes nearly 100% of the potential risk to the total non-cancer risk estimates for industrial workers.

Direct contact with surface soil also poses potential risk to industrial workers. This risk is based on exposure to PCBs in a limited number of soil samples collected in the upland (SS-08, SS-10, SS-19, SS-22, and SS-24). The direct contact with soil pathway contributes relatively little to the cumulative estimated non-cancer risk to industrial workers. Specifically, hazard indices for this pathway were 2 and 0.5 for RME and CTE, respectively (Cascadia Associates, 2020a, Tables F-13E and F-13F).

Excavation workers. For excavation workers, the total hazard index exceeded Oregon's environmental cleanup standard of one. The total hazard indices were 390 and 28 for RME and CTE, respectively (see Table 2).

Estimated non-cancer risks to the excavation worker are dominated by direct contact with and inhalation of petroleum hydrocarbons in groundwater (Cascadia Associates, 2020a, Tables F-7A, F-7B). Direct contact with surface soil, and direct contact, ingestion, and inhalation of subsurface soil also contribute to the cumulative estimated risk to potential excavation workers (Cascadia Associates, 2020a, Tables F-13C, F-13D).

No unacceptable risks were estimated for direct exposure to soil, or for indirect exposure to contaminants in soil from inhalation outside the inferred NAPL areas, with the exception of an exceedance of risk calculated due to the localized high concentration of arsenic in one deep soil sample (MW-10). The potential risks from direct exposure to groundwater in an excavation were not quantified because the COPCs are semi-volatile. DEQ does not recommend a quantitative assessment of dermal exposures to SVOCs because their dermal permeability coefficients are outside of the effective predictive domain used to model this exposure factor. Instead, a qualitative discussion was provided in the uncertainty section of the Supplemental HHRA and concluded that:

- It is reasonable to assume there are unacceptable risks or hazards from dermal contact with groundwater where NAPL is present.
- It is not possible to accurately quantify the magnitude of the risk from dermal contact with water containing appreciable levels of PAHs or other semi-volatile compounds (such as TPH consisting of weathered creosote) in areas where NAPL is not present.
- Thus, an administrative or engineering control should be applied to the Site to control future exposures to groundwater. Administrative and engineering controls have already been established for Area 2 and similar controls could easily be applied to Area 1 of the Site.

Construction workers. For construction workers, the total hazard index exceeded Oregon's environmental cleanup standard of one. The total hazard indices were 17 and 0.6 for RME and CTE, respectively (see Table 2).

Estimated non-cancer risks to the construction worker are dominated by direct contact, ingestion, and inhalation of subsurface soil (Cascadia Associates, 2020a, Tables F-13A, F-13B). Direct contact with surface soil also contributes to the cumulative estimated risk to potential excavation workers (Cascadia Associates, 2020a, Tables F-13A, F-13B). This risk estimate is based on exposure to the localized high concentration of arsenic in one deep soil sample (MW-10). With the exception of this one sample from MW-10, no unacceptable risks were estimated for a potential construction worker's direct exposure to soil, or from indirect exposure to contaminants in soil, outside the inferred NAPL areas.

3.2.4 Ecological Risk Assessment

The ERA reports (Bridgewater Group and Kennedy/Jenks Consultants, 2006b; Cascadia Associates, 2020b) describe in detail the procedures used to evaluate the potential risks to ecological receptors at the Site. A Level II ERA was conducted to evaluate terrestrial and aquatic ecological exposures associated with Site-related contamination (Bridgewater Group and Hart Crowser, 2003). The 2003 Level II ERA identified contaminants of potential ecological concern (CEPCs) and appropriate ecological assessment endpoints. The Level II ERA concluded that no CPECs are present in surface soil and therefore that no further ERA activities for terrestrial receptors are necessary at the Site. The Level II ERA identified potential unacceptable risk to ecological receptors in the in-water areas of Scappoose Bay and Milton Creek adjacent to the Site (Bridgewater Group and Hart Crowser, 2003; Bridgewater Group and Kennedy/Jenks Consultants, 2006b, Section 2).

Based on the conclusions of the Level II ERA, the 2006 ERA Report evaluated risks to aquatic ecological receptors associated with contaminant data for Scappoose Bay sediment, surface water, and seeps. This assessment concluded that contaminant concentrations in Scappoose Bay seeps may pose unacceptable risk to ecological receptors and identified that preventing ecological receptors from direct contact with Scappoose Bay seeps should be a preliminary remedial action goal for the Site (Bridgewater Group and Kennedy/Jenks Consultants, 2006b, Section 3).

At DEQ's request, the 2020 Supplemental ERA re-evaluated the CSM for the entire site; evaluated ecological risks using sediment, porewater, and surface water data collected since 2005; and applied new methods and research to the data evaluation as additional LOEs supporting ecological risk conclusions. The presence of NAPL is presumed to present an unacceptable risk for ecological receptors. Therefore, ecological risks associated with media within the estimated extent of NAPL were not quantitatively evaluated in the Supplemental ERA (Cascadia Associates, 2020b).

3.2.4.1 Ecological Risk Assessment Lines of Evidence

The Supplemental ERA used multiple LOEs to evaluate ecological risk. The first and primary LOE was the presence or absence of a surface sheen. Additional LOEs included comparisons of environmental concentrations in various environmental media to SLVs based on toxic responses of aquatic organisms (e.g., benthic organisms and fish) to individual chemical constituents. The LOEs used in the Supplemental ERA are discussed below:

Presence or absence of sheen: The presence of NAPL and/or a petroleum sheen was used as a primary LOE in evaluating risk to aquatic receptors. The presence of NAPL or a sheen is a violation of narrative water quality criteria and is considered an unacceptable risk to ecological receptors due to the potential for fouling of membranes and gills or ingestion of accumulated product on feathers or fur during preening. Multiple field events (2010, 2011, 2012, 2013, 2017, and 2018) were conducted to properly delineate the boundary of NAPL/sheen, and the results are shown on Figure 42.

Sediment toxicity:

Total PAHs: Bulk sediment concentrations of total PAHs were compared with the threshold effects concentration (TEC) and probable effects concentration (PEC) to evaluate

risk to benthic organisms associated with direct contact with sediment (Macdonald et al., 2000). TECs and PECs were developed from a number of published sediment quality guidelines (SQGs). TECs reflect lower limits of sediment toxicity where toxicity is not predicted to occur; PECs reflect upper limits, above which adverse effects are predicted to have a high probability of occurring. Evaluating bulk sediment concentrations against both TECs and PECs provides a bracket for expected toxicity and provides a secondary LOE in assessing risk of sediment toxicity to the benthos. Concentrations of total PAHs and a comparison to PEC and TEC values are presented in Table G-10 of the Supplemental ERA (Cascadia Associates, 2020a).

Total petroleum hydrocarbons: PECs and TECs do not exist for TPH. TPH fractions were screened against criteria developed as part of the Portland Harbor Superfund Site's (PHSS's) ERA (Windward, 2013), with values updated to correct a calculation error. Use of PHSS's SLVs are considered a secondary LOE in risk evaluation. Concentrations and SLVs are presented in Table G-10 of the Supplemental ERA (Cascadia Associates, 2020a).

Porewater toxicity:

PAHs: Concentrations of total PAHs in porewater samples were converted into total TU values, specifically the Σ IWTU, to evaluate the potential for narcotic risk to benthic invertebrates. Each individual PAH congener concentration was divided by its unique final chronic value (based on narcotic effects to benthic organisms) to calculate the TU. The sum of all Tus for a sample was calculated as the Σ IWTU. Σ IWTU values ≥ 1 indicate potential risk to benthic organisms. Σ IWTU values are presented in Table G-6a and Table G-11a of the Supplemental ERA (Cascadia Associates, 2020a).

VPH: Concentrations of VPH in 2017 porewater samples were used as a secondary LOE for determining risk at the Site. Concentrations were compared to PHSS's ERA SLVs and presented in Table G-11b of the Supplemental ERA (Cascadia Associates, 2020a).

OPAHs: OPAHs were measured in 2017 porewater samples to provide a secondary LOE for determining risk at the Site. The presence of oxygen atoms in OPAH molecules makes OPAHs more polar and hydrophilic than PAHs, often resulting in higher freely dissolved concentrations of OPAHs than PAHs in sediment porewater. No SLVs exist for OPAHs, so there was no quantitative assessment of risk based on OPAH concentrations. Concentrations of OPAHs in surface water samples are presented in Table G-12a of the Supplemental ERA (Cascadia Associates, 2020a).

Surface water toxicity:

PAHs: As with porewater, concentrations of PAHs in surface water samples were converted to Tus. The sum of all Tus for a given sample were calculated as the sum of surface water TU (Σ SWTU). A Σ SWTU greater than 1 indicates the potential risk to benthic organisms, and these calculations were used as a primary LOE in risk evaluation. Σ SWTUs are presented in Table G-6b and G-12a of the Supplemental ERA (Cascadia Associates, 2020a). Concentrations of C_{free} PAHs in surface water were also compared to published SLVs for individual PAHs where available and presented in Table G-12b of the Supplemental ERA (Cascadia Associates, 2020a).

OPAHs: As with porewater, OPAHs were measured in a number of samples to further define risk from PAHs. Concentrations of OPAHs in surface water were not screened against any SLVs, as no risk-based screening criteria exists. Total OPAH concentrations are presented in Table G-12a of the Supplemental ERA (Cascadia Associates, 2020a).

Groundwater toxicity: Groundwater was evaluated to determine if upland dissolved phase constituents have the potential to adversely affect Scappoose Bay. Dissolved phase concentrations in groundwater were evaluated based on the same approach used for surface water and porewater. This approach was completed for nearshore groundwater wells and did not account for any potential change in concentrations during migration to Scappoose Bay.

PAHs: As with porewater and surface water, concentrations of PAHs in groundwater were converted to Tus. The sum of all Tus in a given sample were calculated as the Σ GWTU. A Σ GWTU greater than 1 indicates the potential risk to benthic organisms. As mentioned above, benthic organisms are not directly exposed to groundwater; thus, this analysis is strictly used to examine the potential for risk to benthic organisms following upland groundwater migration to Scappoose Bay. PAH concentrations and Σ GWTUs are presented in Table G-13 of the Supplemental ERA (Cascadia Associates, 2020a). Concentrations of individual PAHs in groundwater were compared to DEQ Level II SLVs (DEQ, 2001b) to evaluate potential risk after upland groundwater migration to Scappoose Bay. Concentrations and associated SLVs are presented in Table G-13 of the Supplemental ERA (Cascadia Associates, 2020a).

TPH: Concentrations of diesel- and gasoline-range hydrocarbons were compared to Freshwater Ecotox Aquatic Habitat Goal Levels from San Francisco Bay Regional Water Quality Control Board (SFBRWQCB, 2019). As described above, this approach was used as a conservative analysis of potential risk to Scappoose Bay from upland groundwater migration. Concentrations and comparisons to SLVs are presented in Table G-13 of the Supplemental ERA (Cascadia Associates, 2020a).

3.2.4.2 Ecological Risk Assessment Conclusions

The following conclusions were drawn from the Supplemental ERA:

- The presence of creosote NAPL and moderate to heavy petroleum sheen present unacceptable risk to ecological receptors where they are observed in surface sediment, porewater, and surface water. Areas exhibiting moderate to heavy petroleum sheen in surface sediment are shown on Figure 42.
- Issues with the 2012 measurement of PAHs C_{free} in sediment porewater centered around elevated detection limits, particularly high molecular weight PAHs, and a perceived low bias for detecting select low molecular weight PAHs (e.g., naphthalene and its alkylated isomers). This uncertainty to confidently assess porewater toxicity led DEQ to consider other LOEs. Subsequently, passive samplers were deployed in 2017 within offshore areas where prior efforts to assess porewater toxicity and habitat impairment posed by Site -related contamination had been inconclusive. The quantitation of chemicals in

sediment porewater using LDPE passive sampling devices developed by the Food Safety and Environmental Stewardship program at Oregon State University (OSU) is more robust and precise than the 2012 direct measurements of freely dissolved PAHs in sediment porewater. For example, naphthalene, a key component of PAH contamination in upland groundwater, was detected in 26 of 30 porewater samples collected using passive samplers in 2017, compared to only 4 out of 59 porewater samples analyzed by the ASTM International SPME approach in 2012.

- The 2017 porewater data demonstrate low or nonexistent potential narcotic risk to benthic organisms outside of delineated offshore NAPL areas. Specifically, the ΣIWTU exceeded in only one of the 15 Phase 3 locations, PWS-090617-2, based on porewater measurements obtained within the biologically active zone (i.e., upper 12 inches of sediment). Sediment porewater sample PWS-090617-2 was obtained in Lower Milton Creek near the confluence with Scappoose Bay. This area is not currently included in the in-water PAAs and therefore may need to be addressed via additional sampling and action moving forward.
- Like porewater, surface water samples with a Σ SWTU ≥ 1 are considered unacceptable. Four of the eight 2017 surface water passive sampling sites (PWS-090917-9, PWS-090517-10, PWS-090617-12, and PWS-090617-13) resulted in a Σ SWTU slightly above 1 (1.07). Detected concentrations of alkyl PAHs C4-naphthalene appear to be the primary ecological risk driver in surface water.
- Risk to fish is negligible for the Scappoose Bay exposure area from either direct exposure to surface water or from the bioaccumulation of fluoranthene or pyrene.
- A screening of monitoring wells along Scappoose Bay indicates dissolved phase constituents may pose a risk to aquatic receptors from discharge of upland groundwater to surface water based on exceedances of select SLVs for PAHs and TPH and Σ GWTU values greater than 1. There is still a large degree of uncertainty associated with this pathway; further evaluation is needed, including additional sampling at the Site of groundwater discharge.

Taken together, along with the results of earlier ERAs, bioassay results, and the multiple lines of evidence considered over more than two decades of investigation, the 2020 ERA demonstrates:

- The presence of NAPL and/or petroleum sheen present unacceptable risk for ecological receptors.
- Dissolved phase groundwater constituents may pose risk to aquatic receptors from discharge of upland groundwater to the aquatic environment; consistent with the findings presented in the 2006 ERA, unacceptable risk to ecological receptors from direct contact with groundwater migrating to Scappoose Bay via seeps should be a preliminary remedial action goal for the Site.
- There is low or nonexistent potential risk to ecological receptors at the Site outside of delineated areas of NAPL and petroleum sheen.

3.2.5 Human Health and Ecological Risk Summary and RAOs to Address Risk

This section connects the unacceptable risks identified in the HHRAs and ERAs with RAOs developed to focus the remediation on addressing those risks. In this section the RAOs are referenced in context of the risks they aim to mitigate; the RAOs are described in more detail in Section 5.1.2.

3.2.5.1 Human Health Risk Summary

The HHRA identified unacceptable risks associated with Area 1 Upland related contamination in the following human health exposure scenarios:

- Adult/child subsistence fishers in Scappoose Bay.
- Industrial workers working in the upland area of the Site.
- Excavation workers working in the upland area of the Site.
- Construction workers working in the upland area of the Site.

3.2.5.2 RAOs to Address Human Health Risk

Below is a summary of RAOs selected for the remedy to address unacceptable risks to human health:

Adult/child subsistence fishers in Scappoose Bay.

A core focus of the remediation is on limiting bioaccumulation of COPCs in sediment that drive the excess lifetime cancer risk estimate for subsistence fishers, and on limiting bioaccumulation of carcinogenic PAHs in surface water and sediment that contribute risk to subsistence fishers. RAO 1a, RAO 1c, and RAO 2 all directly address this goal by: preventing releases of creosote NAPL (defined as moderate to heavy petroleum sheen) to the aquatic environment; preventing unacceptable risk to aquatic receptors associated with direct contact with NAPL in riverbank seeps, surface sediment, and surface water; and protecting aquatic receptors from exposure to contaminants in surface sediment (as defined as the top 12 inches of sediment and associated porewater) and surface water, respectively. RAO 4 also addresses risk associated with in-water receptors bioaccumulating contaminants migrating from the groundwater in the upland to the aquatic environment by preventing recontamination of the in-water remedy from the defined NAPL riverbank seep areas and contaminated groundwater discharge. RAO 1b also addresses potential risk to adult/child fishers and other recreational users by preventing unacceptable risk to recreational users associated with direct contact to NAPL seeps in surface sediment. The selected remedial alternative for the Area 1 Upland, Area 1 Dock PAA, Area 2 Dock PAA, and Cove Area PAA address this risk by limiting contaminants migrating from the upland into the in-water areas via seeps and soil erosion, and through a combination of removing and capping hot spots of contamination in the inwater PAAs.

Industrial workers working in the upland area of the Site.

RAO 8 directly addresses risk to industrial workers by protecting Site workers from vapor intrusion into buildings that causes unacceptable risk. The selected remedy for Area 1 Upland will cap upland soil contamination and address contaminant migration into the aquatic environment. The remedy does not, however, actively remove contamination in upland groundwater or soil in the source area. The long-term effectiveness of the selected Area 1 Upland remedy is therefore dependent on institutional and engineering controls to mitigate risks to potential industrial workers in the upland. The selected remedy for the upland assumes that an EES would be made between the Port of Columbia County and DEQ. The EES would include requirements to follow a contaminated media management plan (CMMP), require installation of engineering controls to prevent vapor intrusion into any future structures, land use restrictions, water use restrictions, identification and maintenance of engineered barriers, and long-term monitoring requirements.

RAO 6 directly addresses risk to industrial workers in areas of unacceptable risk related to PCB contamination in soil. The selected upland remedy will cap surface soil contamination in the Upland PAA, mitigating this risk to potential industrial workers.

Excavation workers working in the upland area of the Site.

RAOs 5, 6, and 7 directly address risk to excavation workers by protecting construction/excavation workers when working in upland area(s) of inferred NAPL and/or near monitoring well boring MW-10; preventing industrial workers in areas of unacceptable risk related to PCB contamination in gravel roadways; and preventing exposure to groundwater in Area 1 Upland by Site workers, respectively. As described above, the selected remedy for Area 1 Upland will cap soil contamination and address contaminant migration into the aquatic environment but will not actively remove contamination in upland groundwater or subsurface soil in the source areas. The long-term effectiveness of the selected remedial action is therefore dependent on institutional and engineering controls to mitigate risks to potential excavation workers in the upland. To be protective of a potential future excavation worker, this EES would need to mitigate or prevent: direct contact with and inhalation of contaminants from upland groundwater; and direct contact, ingestion, and inhalation of subsurface soil.

Construction workers working in the upland area of the Site.

RAO 5 directly addresses potential risk to construction workers by protecting construction/excavation workers when working in upland area(s) of inferred NAPL and/or near monitoring well boring MW-10. As described above, the selected remedy for the Area 1 Upland will cap upland soil contamination and address contaminant migration into the aquatic environment but will not actively remove contamination in upland subsurface soil. The long-term effectiveness of the Area 1 Upland selected remedial action is therefore dependent on institutional and engineering controls to mitigate risks to potential construction workers in the upland. To be protective of a potential future construction worker, this EES would need to mitigate or prevent direct contact, ingestion, and inhalation of subsurface soil.
3.2.5.3 Ecological Risk Summary

Based on the conclusions of the 2020 Supplemental ERA and an evaluation of all available LOEs, the following Site conditions were organized into primary and secondary indicators of aquatic ecological risk in the Feasibility Study (GeoEngineers, 2022):

Primary LOEs:

- Observations of a moderate to heavy petroleum sheen (i.e., creosote NAPL) in surface sediment (1 foot bml).
- Surface water concentrations representing a Σ SWTU ≥ 1 based on the direct measurement of 34 PAHs.
- Surface sediment porewater concentrations representing a $\Sigma IWTU \ge 1$ based on the direct measurement of 34 PAHs.

Secondary LOEs

- Observations of a slight petroleum sheen in surface sediment and subsurface sediment porewater (greater than [>]1 foot bml) concentrations representing a ΣIWTU ≥ 1.
- Bulk sediment total 34 PAH concentrations \geq PEC of 22.8 mg/kg.
- Significant concentrations of total TPH in bulk sediment relative to fraction-specific SLVs.
- Bulk sediment total 34 PAH concentrations >10 mg/kg, representing the mid-point between the TEC of 1.6 mg/kg and the PEC of 22.8 mg/kg.
- Total TPH porewater concentrations greater than fraction-specific SLVs
- Porewater concentrations in surface sediment representing one-half of a TU.
- Relative magnitude of C_{free} 63 PAHs > 0.5 μ g/L (not risk-based).
- Relative magnitude of C_{free} total OPAHs concentrations > 0.5 µg/L (not risk-based).

Based on these primary and secondary LOEs, the following conclusions were drawn in the Feasibility Study concerning ecological risk (GeoEngineers, 2022):

• Presence of creosote NAPL and moderate to heavy petroleum sheen presents unacceptable risk to ecological receptors where it is observed in surface sediment and surface water.

- Risk to ecological receptors in sediment porewater and surface water are largely driven by lighter end and more soluble PAHs, including naphthalene and methylnaphthalenes, C-3 and C-4 naphthalenes, C-2 and C-3 fluorenes, and C-2, C-3 and C-4 phenanthrenes/anthracenes. This conclusion is consistent for multiple pathways, including direct toxicity (e.g., comparison with Tier II final chronic values), narcosis (i.e., measured by toxic units), and bioaccumulation (i.e., using food chain multipliers) (DEQ, 2020).
- The presence of slight petroleum sheens in surface sediment and $\Sigma IWTU \ge 1$ in subsurface sediment has the potential to result in unacceptable risk due to transformation of PAHs to more biologically active metabolites, and variability in the depth of the biologically active zone (DEQ, 2020).
- Total dissolved parent and alkylated PAHs in groundwater should be used to determine levels discharging to offshore sediment and surface water via colloidal groundwater transport, in addition to freely dissolved concentrations and NAPL migration.

3.2.5.4 RAOs to Address Ecological Risk

Based on the findings of the ERAs, the primary focus of the in-water remedy is on limiting aquatic ecological receptors' exposure to NAPL and petroleum sheen. **RAO 1a** directly addresses estimated ecological risk by preventing releases of creosote NAPL, defined as moderate to heavy petroleum sheen, to the aquatic environment. **RAO 1c** directly addresses this risk by preventing unacceptable risk to aquatic receptors associated with direct contact with NAPL in riverbank seeps, surface sediment, and surface water. **RAO 2** addresses this risk by protecting aquatic receptors from exposure to contaminants in surface sediment (as defined as the top 12 inches of sediment and associated porewater) and surface water that result in toxic effects, respectively. **RAO 3** addresses this risk by preventing recontamination of the in-water remedy from adjacent contaminants from the groundwater in the upland by preventing recontamination of the in-water discharge.

3.3 BENEFICIAL USE AND HOT SPOT DETERMINATION

The criteria used to evaluate Remedial Action Alternatives (RAAs) for groundwater and surface water depend on whether a "hot spot" is present or not, as determined by a loss of "current or reasonably likely future" beneficial use of the water resource.

OAR 3401-122-0115(9) defines beneficial uses of water as any current or reasonably likely future beneficial use of groundwater or surface water by humans or ecological receptors.

OAR 340-122-0115(32) defines hot spot of contamination as:

(a) For groundwater or surface water, hazardous substances having a significant adverse effect on beneficial uses of water or waters to which the hazardous substances would be reasonably likely to migrate and for which treatment is reasonably likely to restore or protect such beneficial uses within a reasonable time, as determined in the feasibility

study; and (b) For media other than groundwater or surface water, (e.g., contaminated soil, debris, sediments, and sludges; drummed wastes; "pools" of dense, non-aqueous phase liquids submerged beneath groundwater or in fractured bedrock; and non-aqueous phase liquids floating on groundwater), if hazardous substances present a risk to human health or the environment exceeding the acceptable risk level, the extent to which the hazardous substances:

(A) Are present in concentrations exceeding: (i) 100 times the acceptable risk level for human exposure to each individual carcinogen; (ii) 10 times the acceptable risk level for human exposure to each individual noncarcinogen; or (iii) 10 times the acceptable risk level for exposure of individual ecological receptors or populations of ecological receptors to each individual hazardous substance. (B) Are reasonably likely to migrate to such an extent that the conditions specified in subsection (a) or paragraphs (b)(A) or (b)(C) would be created; or (C) Are not reliably containable, as determined in the feasibility study.

3.3.1 Groundwater Beneficial Use Determination

A beneficial water use determination (BWUD) was prepared as part of the 2000 remedial investigation and updated as part of the 2018 supplemental remedial investigation. The findings of the 2000 BWUD indicated that beneficial water uses were limited to aquatic habitat, recreation, and aesthetic quality. Surface water could be used to provide irrigation and/or industrial process/cooling water in the future.

No direct beneficial uses of groundwater occurred at or adjacent to the Site. Potable water at the Site is provided by the City of St. Helens' municipal system. According to the Oregon Water Resource Department database, some domestic wells are present within 1 mile of the Site, none of which are located within the LOF. The nearest wells are located approximately 0.5 mile cross-gradient from the Site.

The BWUD concluded that based on the poor yield of the shallow aquifer (less than 0.5 gallon per minute) and the availability of municipal water supply, future direct beneficial uses of groundwater within the LOF are not expected.

Regarding the cleanup of sediments in Milton Creek, Scappoose Bay, and the Multnomah Channel, the reasonably likely future beneficial use of groundwater is limited to discharge through riverbank seeps and/or sediments which provides habitat for plants, mammals, birds, fish and/or benthic organisms.

3.3.2 Surface Water Beneficial Use Determination

Surface water in the vicinity of the Site includes Milton Creek, Scappoose Bay, and the Multnomah Channel. Groundwater likely discharges to surface water in the area of the Site. Surface water within the LOF is currently used for recreation and aquatic and terrestrial habitat. The Boise Cascade (ECSI No. 0014) facility downstream of the Site was identified as using surface water for facility processes. Plant operations have significantly decreased since 2015 with only tissue manufacture remaining. The use of surface water in current operations is unknown but would be

significantly less than when pulp and paper operations were active. Reasonably likely future beneficial water uses were determined to include recreational, habitat, and limited industrial use.

3.3.3 Hot Spots

This ROD evaluates soil, groundwater, sediment, porewater, and surface water potential hot spots in the Area 1 Upland and the four in-water PAAs.

3.3.3.1 Non-Aqueous Phase Liquid

Creosote NAPL is present in the subsurface soil beneath the former wood treatment operations area and represents a hot spot. Creosote NAPL at the Site has migrated from the historical ground surface in the former wood-treating operations area to deeper soil and groundwater, affecting near-shore sediment, porewater, and surface water adjacent to Area 1 Upland.

The presence of NAPL in monitoring well MW-3A and downgradient riverbank seeps represents a mobile hot spot to the Cove Area. Riverbank seeps exhibiting a moderate to heavy petroleum sheen are considered mobile hot spots that serve as an ongoing source of contamination to sediment, porewater, and surface water in Upper Milton Creek and Cove Area.

3.3.3.2 Impacts to Beneficial Water Uses

Riverbank groundwater seeps exhibiting a visible creosote sheen is an indication of unacceptable direct contact risk to human health and ecological receptors and is considered a mobile hot spot that serves as an ongoing source of contamination to sediment, porewater, and surface water. Dissolved phase creosote constituents in porewater and surface water that cause unacceptable risk to benthos and fish are considered hot spots. Further assessment during remedial design will be needed to determine if total dissolved creosote constituents in groundwater is a hot spot.

3.3.3.3 Soil/Sediment

The inferred extent of creosote NAPL in upland subsurface soil, riverbank seeps, and near-shore surface sediment represent a soil/sediment hot spot. Subsurface sediment exhibiting a moderate to heavy petroleum sheen may be considered a hot spot if it is not reliably containable. Subsurface sediment exhibiting a creosote sheen outside the in-water PAAs boundaries appears to be reliably contained and is not currently considered a sediment hot spot.

3.3.3.4 Contaminated Debris

Creosote contaminated wood debris is present in surface sediment near and beneath the former overwater operations. Moderate to heavy petroleum sheens are easily generated by disturbing the surface sediment containing this wood debris. Therefore, shoreline and offshore areas of surficial creosote contaminated wood debris is considered a sediment hot spot. Subsurface creosote contaminated wood debris appears to be reliably contained and is not considered a sediment hot spot.

3.4 ESTIMATE OF CONTAMINATION VOLUME

Estimated volumes of contaminated media including soil, sediment, and wood debris are presented on Table 6.

This ROD is based on technical documents that have been reviewed by a technical team at DEQ. The team consists of the project manager, a hydrogeologist, environmental engineer, and human health and environmental toxicologist. The team unanimously supports the selected remedial action.

5.1 REMEDIAL ACTION OBJECTIVES & PRELIMINARY REMEDIATION GOALS

The process for selection of a remedial action by DEQ is outlined in OAR 340-122-0090, focusing on selection of an action that: a) is protective of present and future public health, safety, and welfare of human health and the environment; b) is based on balancing of remedy selection factors; and c) satisfies requirements for hot spots of contamination. DEQ's Guidance for Conducting Feasibility Studies (1998, updated 2006 and 2017) provides more detailed guidance on the remedy selection process, including the development of RAOs, identification of general response actions, identification and screening of remedial technologies, and assembly of RAOs for evaluation. Each of these steps were considered by DEQ, and discussed below, in the selection of a remedial action for the Area 1 Upland and four sediment in-water PAAs.

RAOs and acceptable risk levels, as defined in OAR 340-122-0115(1) through (6), were developed based on the identified beneficial uses, exposure pathways, and the findings of the risk assessments. RAOs are media-specific goals for protecting human health, safety, and the environment and were developed to address the standards established in OAR 340-122-0040. Specifically, the remedial action must achieve the numeric standards for protectiveness that correspond to acceptable risk levels; treat or remove hot spots to the extent feasible; prevent or minimize future releases and migration of hazardous substances in the environment; and provide long-term care or management as necessary and appropriate.

RAOs provide the framework for developing and evaluating RAAs, as any remedy DEQ selects or approves must achieve these Site-specific goals. The RAOs are listed in Section 5.1.2.

5.1.1 Acceptable Risk Levels

Risk-based concentrations (RBCs) and screening levels used to draw conclusions about potential risks to human and ecological health are described in detail in the HHRA and ERA reports. Protectiveness levels were developed for different media at the Site to guide remedy selection during the Feasibility Study (GeoEngineers 2022, Feasibility Study, Section 2.8.2). These protectiveness levels were established to ensure humans and ecological receptors were protected from the potential risks identified in the risk assessments that are summarized in Section 3.2.5 above. Specifically, the Feasibility Study established acceptable risk levels for: PCBs and arsenic in upland soil to protect construction workers and industrial workers working in the upland; naphthalene in groundwater to protect industrial and/or occupational workers from vapor intrusion into buildings; carcinogenic PAHs and TPH in sediment to protect subsistence fishers; and PAHs in porewater and surface water to protect ecological receptors. The Feasibility Study also establishes that the presence of NAPL and/or moderate to heavy petroleum sheen present unacceptable risk to humans and ecological receptors in upland soil, groundwater, sediment, porewater, and surface water (GeoEngineers 2022, Feasibility Study, Section 2.8.2). CULs to inform decision-making during and after remedial implementation will be developed during the remedial design process.

5.1.2 Site-Specific Remedial Action Objectives

Site-specific RAOs were developed for soil and groundwater in Area 1 Upland and sediment, porewater, and surface water in the four sediment in-water PAAs. The RAOs describe what the remedial action is expected to achieve to protect human health, ecological receptors, and beneficial uses, as required by OAR 340-122-0040. The RAOs for the Site are as follows:

Area 1 Upland PAA RAOs:

- RAO 3: Prevent recontamination of in-water remedy from adjacent contaminated riverbank soils.
- RAO 4: Prevent recontamination of in-water remedy from the defined NAPL riverbank seep areas and contaminated groundwater discharge.
- RAO 5: Protect construction and/or excavation worker receptors when working in upland area(s) of inferred NAPL and/or near monitoring well boring MW-10.
- RAO 6: Protect industrial workers in areas of unacceptable risk related to PCB contamination found in gravel roadways.
- RAO 7: Prevent direct exposure to groundwater at Area 1 by Site workers.
- RAO 8: Protect Site workers from vapor intrusion into buildings that cause unacceptable risk.

In-Water Sediment PAA RAOs:

- RAO 1a: Prevent releases of creosote NAPL, defined as moderate to heavy petroleum sheen, to aquatic environment.
- RAO 1b: Prevent unacceptable risk to recreational users associated with direct contact with NAPL in riverbank seeps and surface sediment.
- RAO 1c: Prevent unacceptable risk to aquatic receptors associated with direct contact with NAPL in riverbank seeps, surface sediment, and surface water.
- RAO 2: Protect aquatic receptors from exposure to contaminants in surface sediment (as defined as the top 12 inches of sediment and associated porewater) and surface water that result in toxic effects.

5.1.3 Preliminary Remediation Goals

PRGs were developed for Area 1 Upland and the four sediment in-water PAAs using available Site information to identify endpoint concentrations or risk levels that are believed to be sufficiently protective of human health and the environment. The PRGs describe Site conditions or concentrations in specific media that are designed to be protective of certain receptors from exposure to COCs through particular pathways within a reasonable amount of time. The PRGs for the Site are as follows:

Area 1 Upland PRGs

- Protect construction and excavation workers from direct exposure to NAPL and contaminated groundwater.
- Remove, contain, or treat surface (0 to 3 feet bgs) soil with total PCBs greater than 0.74 mg/kg.
- Protect building occupants from vapor intrusion that results in unacceptable risk.

In-Water Sediment PAAs PRGs

- Remove, contain, or treat NAPL, including heavy to moderate sheen, in surface sediment.
- If not contained, remove, contain, or treat NAPL, including heavy to moderate sheen, in subsurface sediment.
- Remove, contain, or treat NAPL, including heavy to moderate sheen, in shoreline seeps.
- Achieve ΣIWTU less than (<) 1 in surface sediment porewater within 10 years of removal, containment, or treatment of NAPL in surface sediment and riverbank seeps.
- Achieve all applicable water quality criteria in surface water within 10 years of removal, containment, or treatment of NAPL in surface sediment and riverbank seeps.

5.1.4 Remedial Action Levels and Cleanup Levels

RAOs and PRGs were developed to inform technology evaluations for Area 1 Upland and the four Sediment PAAs. Remedial action levels (RALs) are contaminant-specific concentrations used to identify where an active remedial technology should be applied to reduce risks more effectively than MNR alone. CULs are contaminant-specific concentrations that should not be exceeded following remedial action. RALs and CULs will be developed during remedial design, if necessary.

5.1.5 **Priority Action Areas**

Area 1 Upland and four sediment in-water PAAs were established for the Site (see Figure 2). PAAs were identified as areas that are highly likely to require active remediation and are characterized by:

• NAPL and/or moderate to heavy petroleum sheen in surface sediment and riverbank seeps.

• Contaminant concentrations that result in unacceptable human and ecological risk with a high degree of certainty.

The Area 1 Upland and four sediment in-water PAAs are discussed in more detail in Section 2.1.1.

5.2 REMEDIAL ACTION ALTERNATIVES FOR SEDIMENT

RAAs were developed and evaluated during the Feasibility Study, and presented in DEQ's Staff Report along with the recommended remedial action. The RAAs are based on general response actions that include: 1) no action, 2) engineering controls and/or ICs, 3) treatment, 4) excavation and offsite disposal, 5) excavation and on-site disposal, 6) and any combination of the general response actions, as appropriate. Several remedial action technologies were evaluated for each general response action. The technologies were screened in accordance with OAR 340-122-0085(4), which requires meeting the threshold criterion of protecting human health and the environment and considering their relative merits/drawbacks with respect to the remedy selection factors. Tables 3, 4, and 5 provide the rationale for eliminating or carrying forward general response actions and technologies based on Site characteristics, environmental media conditions, and contaminant type. Viable response actions and technologies that can meet the threshold criterion were assembled into RAAs. Given the Site contains hot spots of soil and sediment contamination in Area 1 Upland and in-water areas, the Feasibility Study included an evaluation of a treatment based alternative and/or an excavation and offsite disposal alternative per OAR 340-122-0085.

5.2.1 General Response Actions and Applicable Technologies

Technologies that were carried forward after the initial screening and combined to develop comprehensive RAAs are summarized below:

Engineering and/or Institutional Controls. Engineering controls are physical measures to prevent or minimize exposure to hazardous substances or reduce the mobility or migration of hazardous substances. ICs include legal or administrative actions to reduce exposure to hazardous substances. IC examples include land use restrictions, long-term Site management plans (cap inspection/maintenance and contingency plans), and public access restrictions.

Natural Recovery. Natural recovery relies on ongoing, naturally occurring processes such as sedimentation, biodegradation, and dispersion to contain, reduce, or destroy the toxicity or availability of the contaminants. Monitored natural attenuation (MNA) includes monitoring of these natural processes to assess the rate at which the contaminant concentrations are being reduced. MNA does not include active remedial measures. ENR is the process of accelerating MNA, typically through addition of a thin layer of clean material (e.g., sand). However, natural recovery should not be considered as a viable treatment option for hot spots.

Treatment. Treatment is the permanent and substantial elimination or reduction in the toxicity, mobility, or volume of hazardous substances with the use of in-situ or ex-situ remedial technologies. In-situ treatment for contaminated soils may include solidification/stabilization,

enhanced bioremediation, or phytoremediation. In-situ treatment for contaminated sediments may include the addition of reactive materials to conventional sediment caps, such as organoclay, activated carbon, biochar, or an oleophobic biobarrier. Ex-situ technologies for contaminated soils or sediments may include solidification/stabilization or thermal treatment.

Containment. Containment of soils and sediments includes capping, an engineering control that involves the placement of material over the contaminated area. Cap material can be tailored to Site-specific needs. Common cap types include engineered, armored, and reactive caps. The primary functions of a cap are: 1) physical isolation of the contaminated sediment from human and ecological receptors; 2) stabilization of contaminated soil or sediment; and 3) reduction of the flux of dissolved contaminants into the water column (sediments). Capped areas can be engineered in a manner to achieve long-term stability, which may require additional reinforcement. Containment of groundwater can include hydraulic containment using pumping systems, impermeable barrier walls, or permeable reactive barriers that removes contaminants as water flows through.

Removal and Disposal. This technology involves the physical removal (full or partial) of contaminated soil and sediment by excavation or dredging. Material may be disposed offsite to a permitted landfill that is authorized for such disposal under state and federal law. Disposing material offsite would prevent mobility and minimize risk to receptors. Material may also be disposed on-site in a pre-determined upland consolidation disposal facility and managed according to local, state, and federal law. For sediments, some form of dewatering is typically required prior to disposal.

5.2.2 Estimate of Contaminated Media

The estimates for the quantities of contaminated media considered for each of the general response actions above is summarized in Table 6.

5.2.2.1 Estimate of Upland Contamination

The upland source area is approximately 3.45 acres (150,370 square feet) and is currently covered by an average depth of 8 feet of clean fill. Below the clean fill, creosote NAPL in present in the subsurface soil between depths below 8 ft bgs and 25 feet bgs, which results in an estimated total volume of 95,000 cubic yards of creosote-impacted soil.

5.2.2.2 Estimate of In-water Contamination

<u>Area 1 Dock PAA</u>: Wood-related creosote NAPL in surface sediment underneath and surrounding the previous dock structures has been defined by an area approximately 1.85 acres (80,586 square feet). The area has an estimated 435 timber piles. The assumption is these piles are an average of 25 feet long, resulting in a total estimate of 163 tons of wood waste. The estimated depth of the area where hot-spot material is present in the shallow surface sediment with the impacted surficial woody debris is approximately 3,000 cubic yards. In general, the concentrations increase with depth in this area, with elevated concentration reaching depths of 11 feet bml in some area, resulting in a total volume of impacted sediment of 32,800 cubic yards of sediment.

<u>Area 2 Dock PAA</u>: The creosote-impacted wood debris in surface sediments in the Area 2 Dock PAA is approximately 0.34 acres (15,000 square feet) and the total volume of the shallow (mudline to 1 foot bml) impacted sediment is approximately 560 cubic yards. The area has an estimated 235 piles. The assumption is these piles are approximately 25 feet in length and result in approximately 88 tons of wood waste. The combined total for the estimated sediment and wood waste is 519 tons of hazardous waste and 222 tons of non-haz (including the wood piles). In general, the elevated sediment concentrations and presence of moderate to heavy sheen coincides with the surficial debris.

<u>Cove Area PAA</u>: The extent of nearshore surface sediment (up to 1 foot bml), which are largely soft and fine-grained, impacted by creosote NAPL in the Cove Area covers an area of approximately 1.2 acres (52,275 square feet) and is estimated to be approximately 1,950 cubic yards. Moderate to heavy sheen has been identified along approximately 300 linear feet of the shoreline, with steep slopes across approximately 500 linear feet of the riverbank. The sediment concentrations decrease with depth into the subsurface, with depths of contamination identified up to 12 feet bml, with an average depth of 5 feet bml resulting in an overall volume of impacted sediment of 17,200 cubic yards of sediment.

<u>Upper Milton Creek PAA</u>: The area of Upper Milton Creek where moderate to heavy sheen has been identified in the near surface (up to 1 foot bml) sediment and adjacent riverbank, extends approximately 200 linear feet and within an area of less than 0.05 acres (2,200 square feet) The estimated volume of soil and sediment that would be removed to regrade the bank along the creek from approximately 2H:1V to 4H:1V is approximately 1,020 cubic yards.

5.2.3 Common Elements and Assumptions

Area 1 Upland and In-Water Common Elements and Assumptions

The technologies assembled into alternatives represent a spectrum of potential remedial strategies, ranging from capping and natural recovery to complete removal options. Common elements and assumptions to the Area 1 Upland and in-water RAAs are included below and may not be specifically included in descriptions of RAAs.

- <u>Institutional controls.</u> ICs include administrative and legal mechanisms, such as EES, CMMPs, land use restrictions, and water use restrictions to reduce risk to human health, ensure long-term protectiveness of cleanup actions. The purpose of institutional controls is to provide notification regarding the presence of COCs, regulate the disturbance and management of contaminated materials, and aid in the long-term care of cleanup action, including long-term monitoring. If RAOs and PRGs are not completely achieved following remedial action, interim ICs may be necessary, such as a fish advisory.
- <u>Engineering controls.</u> Engineering controls include physical measures to prevent or minimize exposure to hazardous substances in contaminated materials that remain on Site. Examples of engineering controls include measures to prevent access, including fencing, paving pervious surfaces, directing stormwater away from contaminated media, and installing vapor barriers beneath future buildings constructed in upland contaminated areas.

- <u>Remedial design investigations.</u> Additional data collection and evaluations inform remedial design, including but not limited to additional chemical contamination characterization, debris and capping evaluations, measurements of groundwater seepage and river currents, hydraulic and erosion modeling, geotechnical investigations, and seismic design considerations. Preparation and implementation of a comprehensive predesign investigation (PDI) inform Remedial Design/ Remedial Action (RD/RA) as well as support post-performance monitoring activities and post-construction residual risk assessment.
- <u>Recontamination potential.</u> RD/RA addresses recontamination potential of a constructed sediment remedy. Potential source control pathways are further examined during the PDI.
- <u>Post-construction, long-term monitoring.</u> Long-term monitoring is conducted following implementation of remediation actions to ensure remedial actions continue to be protective of human health and the environment and perform as designed and constructed in accordance with RAOs. Long-term monitoring is conducted in perpetuity for isolation walls and caps placed on upland soil and riverbanks and in-water areas. Performance monitoring includes but is not limited to visual inspections for petroleum sheens, cap integrity testing, sediment porewater/surface water sampling, and MNR assessment within and outside the PAAs where applicable to demonstrate sufficient rate of degradation is occurring. Long-term monitoring for natural recovery is conducted until all COC concentrations in sediment/riverbank soil, groundwater/porewater, and surface water are less than established CULs. Updates to appropriate management strategies are based on results of long-term monitoring.
- <u>Permitting</u>. Permitting requirements include endangered species act consultation, biological opinion, 404 USACE/Division of State Lands (DSL) permit, National Pollutant Discharge Elimination System (NPDES) permit, DSL lease negotiations, and applicable floodplain development permits.
- <u>Construction considerations.</u> Remedy construction considerations include evaluated vegetation removal, riverbank stabilization, cap construction, construction management and oversight, and water quality monitoring.
- <u>Achievement of RAOs and PRGs.</u> RAOs and PRGs should be achieved at construction completion, or the shortest reasonable period of time.
- <u>In-water work window.</u> In-water work is performed in the allowable Oregon Department of Fish and Wildlife work window of July 1 to October 31 and December 1 to January 31. The Multnomah Channel is tidally influenced.
- <u>Debris removal.</u> Debris is removed, to the extent necessary or practicable, in areas where applicable, and transported and disposed of at an off-site facility. Similarly, remnant wood pilings are typically cut at the mudline where active remedial technologies are applied. Removed or cut pilings are disposed of at an off-site facility.

In-Water Common Elements and Assumptions

In-water PAAs require active remedy consideration. Multiple objectives were considered during the Feasibility Study for development of in-water alternatives, including integrating chemical isolation and physical protection goals during cap design; habitat enhancements; and slope angle effects on remedy implementation and monitoring, stope stability, wave and wake impacts on the shoreline, and habitat creation.

The following are common elements of all RAAs for the in-water PAAs (except the No Action alternatives) and may not be specifically included in the RAA descriptions.

- Land lease and easement for nearshore contaminated sediment capping.
- Site use restrictions (e.g., limit access to riverbank areas, post signage informing presence of contamination, communication outreach).
- Sediment Management Plan.

In addition to elements listed above, the following are elements of all in-water RAAs that include caps and may not be specifically included in the RAA descriptions:

- The design of all caps need to consider the effectiveness of containing contamination from underlying materials and the potential for recontamination from upland sources.
- Various evaluations are needed to support remedial design, including but not limited to chemical transport modeling, sensitivity analyses, chemical isolation evaluations, erosion protection evaluations, habitat enhancement evaluations, debris and capping evaluations, measurements of groundwater seepage and river currents, hydraulic and erosion modeling, flood impacts, sea level rise resilience, geotechnical investigations, and seismic design considerations.
- The remedial design determines the specific materials, blends, thicknesses, extent, armoring stone sizing, and grading needed to achieve RAOs. Remedial design would result in the preparation of cap design plans and specifications.

5.2.4 RAA-1: No Action

All Upland and In-Water PAAs include RAA-1 (No Action). A "no action alternative" is included for comparative purposes only as stipulated in OAR 340-122-0085(2) and DEQ guidance. Under this RAA, no actions to treat, remove, or monitor COCs would be performed. There would be no reduction in site risk, and thus this RAA is not considered protective by DEQ.

5.2.5 Area 1 Upland Priority Action Area Remedial Action Alternatives

The following summarizes the RAAs considered for the Area 1 Upland PAA, with detailed descriptions of the remedial technologies included in Table 7. Creosote NAPL observed in Area 1 Upland riverbanks adjacent to Milton Creek and Scappoose Bay indicates that some lateral movement of NAPL occurs beneath the Site. As a result, much of the RAAs include source control measures.

RAA-2 fulfills RAOs 5 through 8 by protecting Site workers. Though RAA-2 reduces contaminant mass flux, it does not prevent recontamination of the in-water remedy from adjacent contaminated riverbank soils (RAO 3) or riverbank NAPL seeps (RAO 4). RAA-3 through RAA-6 fulfill RAOs 3 through 8 by protecting Site workers and preventing recontamination of the in-water remedy by controlling contaminant mass flux towards Milton Creek and/or the Cove Area of Scappoose Bay.

- RAA-2: Impervious surface cap and MNA (\$5,565,000).
- RAA-3: Impervious surface cap, hydraulic containment, enhanced bioremediation, and MNA (\$15,097,000).
- RAA-4: Impervious surface cap, permeable reactive barrier, and MNA (\$7,890,000).
- RAA-5: Impervious surface cap, impermeable isolation wall, and MNA (\$8,494,000).
- RAA-6: Excavation and offsite disposal of NAPL area, impervious surface cap, and MNA (\$60,386,000).

5.2.6 In-Water Priority Action Areas Remedial Action Alternatives

5.2.6.1 Area 1 Dock Priority Action Area

The following summarizes the RAAs considered for the Area 1 Dock PAA, with detailed descriptions of the remedial technologies included in Table 8.

The Area 1 Dock RAAs address RAOs 1 and 2 by preventing releases of creosote NAPL to the aquatic environment, preventing risk associated with direct contact by recreational users and aquatic receptors, and protecting aquatic receptors from exposure to contaminants in surface sediment and surface water.

- RAA-2: Armored reactive cap (\$5,382,000).
- RAA-3: Nearshore removal action, upland consolidation, and armored reactive capping (\$8,138,000).
- RAA-4: Nearshore removal action, offsite disposal, and armored reactive capping (\$9,528,000).
- RAA-5: Complete removal and offsite disposal (\$29,181,000).

5.2.6.2 Area 2 Dock Priority Action Area

The following summarizes the RAAs considered for the Area 2 Dock PAA, with detailed descriptions of the remedial technologies included in Table 9.

The following Area 2 Dock RAAs address RAOs 1 and 2 by preventing releases of creosote NAPL to the aquatic environment, preventing risk associated with direct contact by recreational users and

aquatic receptors, and protecting aquatic receptors from exposure to contaminants in surface sediment and surface water.

- RAA-2: Armored reactive cap (\$1,309,000).
- RAA-3: Nearshore removal action, upland consolidation, and an ENR sand cap (\$1,596,000).
- RAA-4: Nearshore removal action, offsite disposal, and an ENR sand cap (\$1,604,000).

5.2.6.3 Cove Area Priority Action Area

The following summarizes the RAAs considered for the Cove Area PAA, with detailed descriptions of the remedial technologies included in Table 10.

The following Cove Area RAAs address RAOs 1 through 4 by preventing releases of creosote NAPL to the aquatic environment; preventing risk associated with direct contact by recreational users and aquatic receptors; protecting aquatic receptors from exposure to contaminants in surface sediment and surface water; and preventing recontamination of the in-water remedy from adjacent riverbank contaminated soils, NAPL riverbank seeps, and contaminated groundwater discharge.

- RAA-2: Armored reactive cap (\$3,551,000).
- RAA-3: Riverbank restoration, nearshore removal action, upland consolidation, and armored reactive capping (\$4,836,000).
- RAA-4: Riverbank restoration, nearshore removal action, offsite disposal, and armored reactive capping (\$5,984,000).
- RAA-5: Complete removal and offsite disposal (\$17,606,000).

5.2.6.4 Upper Milton Creek Priority Action Area

The following summarizes the RAAs considered for the Upper Milton Creek PAA, with detailed descriptions of the remedial technologies included in Table 11.

The Upper Milton Creek RAAs address RAOs 1 through 4 by preventing releases of creosote NAPL to the aquatic environment; preventing risk associated with direct contact by recreational users and aquatic receptors; protecting aquatic receptors from exposure to contaminants in surface sediment and surface water; and preventing recontamination of the in-water remedy from adjacent riverbank contaminated soils, NAPL riverbank seeps, and contaminated groundwater discharge.

- RAA-2: Armored reactive cap (\$898,000).
- RAA-3: Regrade streambank, limited removal action, upland consolidation, and armored reactive capping (\$1,146,000).

• RAA-4: Regrade streambank, limited removal action, offsite disposal, and armored reactive capping (\$1,216,000).

5.2.7 Monitoring, Review, and Contingency Plan

There are numerous sources of uncertainty at the Site that make it difficult to predict the long-term effectiveness of any of the RAAs described above, including:

- Heterogeneity in the subsurface.
- Potential changes in future groundwater or surface water use patterns (i.e., beneficial uses).
- Potential changes in future land use and zoning.
- Changes in community concerns regarding remedial actions at the Site.
- Long-term performance of active treatment and/or cap areas.
- The long-term potential for deposition, erosion, or net-neutral conditions in the Sediment Area.
- Factors related to climate change (i.e., rainfall and sea-level rise).

Because of these uncertainties, a Performance Monitoring, Review, and Contingency Plan will be developed that will evaluate the performance of the remedy, and any changes that may affect the ability of the remedy to meet the RAOs. The objective of the Monitoring, Review, and Contingency Plan will be to maintain the overall protectiveness of the selected remedy by establishing a series of decision criteria and related response actions for each potential area of uncertainty identified above, and the RAOs identified in Section 5.1.2 of this document. Section 10.1.8 provides a description of potential contingency measures that could be implemented in the event the RAOs are not achieved following remedy implementation.

6.1 EVALUATION CRITERIA

The criteria used to evaluate the RAAs described below are defined in OAR 340-122-0090 and establish a two-step approach to evaluate and select an RAA. The first step evaluates whether an RAA is protective; if not, the RAA is unacceptable and the second step of evaluation is not required. The RAAs considered protective are evaluated and compared with each other using five balancing factors. The five balancing factors are 1) effectiveness in achieving protection, 2) long-term reliability, 3) implementability, 4) implementation risk, and 5) reasonableness of cost.

An evaluation of how each alternative achieves the preference for treatment or removal of hot spots is also included. Lastly, consideration is given to how each alternative achieves green remediation, as described in DEQ's Green Remediation Policy. The alternative that compares most favorably against the balancing factors and complies with the hot spot criteria is selected for implementation. A residual risk assessment is then conducted for the selected alternative to document that it is protective of human health and the environment.

6.2 PROTECTIVENESS

The protectiveness of a given remedial action is evaluated by assessing whether an alternative would eliminate, reduce, or control risks to human health and the environment and achieve applicable RAOs and PRGs.

OAR 340-122-0090 states that protectiveness may be achieved by any of the following methods:

- Treatment.
- Excavation and offsite disposal.
- Engineering controls.
- ICs.
- Any other method of protection.
- A combination of the above.

Except for hot spots, there is no preference for any one of the above methods for achieving protectiveness. Where a hot spot has been identified, OAR 340-122-0090(4) establishes a preference for treatment or removal to the extent feasible, including a higher threshold for evaluating the reasonableness of costs for treatment.

Each RAA was screened for whether it is or is not protective. Alternatives that were deemed not to be protective were not evaluated further. Alternatives deemed to be protective were evaluated for the remaining criteria.

6.3 BALANCING FACTORS

The RAAs determined to be protective are evaluated against the following balancing factors defined in OAR 340-122-0090(3) unless otherwise noted:

Effectiveness. Effectiveness is the ability of an alternative to achieve protectiveness. Criteria for evaluating effectiveness include:

- Magnitude of the residual risk from untreated waste or treatment residuals (residual risks), without considering risk reduction achieved through on-site management of exposure pathways (i.e., engineering controls and ICs); the characteristics of the residuals are considered to the degree that they remain hazardous, considering their volume, toxicity, mobility, propensity to bioaccumulate, and propensity to degrade.
- Adequacy of any engineering controls and ICs necessary to manage residual risks.
- The extent to which the remedial action restores or protects existing or reasonably likely future beneficial uses of water.
- Adequacy of treatment technologies in meeting treatment objectives.
- The time until RAOs are achieved.

Each RAA was evaluated and assigned a ranking between 0 and 5 for effectiveness in protecting human health and ecological receptors from contamination risk. A ranking of 0 indicates that an alternative is not effective and a ranking of 5 indicates an alternative provides a high degree of effectiveness.

Long-term reliability. This factor includes evaluation of the ability of an alternative to achieve RAOs over the long-term following remedy implementation. Criteria for evaluating effectiveness include:

- The reliability of treatment technologies in meeting treatment objectives.
- The reliability of engineering controls and ICs needed to manage residual risks, taking into consideration the characteristics of the hazardous substances being managed, the ability to prevent migration and manage risk, and the effectiveness and enforceability over time of the controls.
- The nature and degree of uncertainties associated with any necessary long-term management (e.g., operations, maintenance, monitoring).

Each RAA was evaluated and assigned a ranking between 0 and 5 for long-term reliability in protecting human health and ecological receptors from contamination risk. A ranking of 0 indicates that an alternative has little to no long-term reliability and a ranking of 5 indicates an alternative has a high degree of long-term reliability.

Implementability. This factor includes evaluation of the ease or difficulty in implementing an alternative. Criteria for evaluating implementability include:

- Practical, technical, and legal difficulties and unknowns associated with the construction and implementation of the technologies, engineering controls, and/or ICs, including the potential for scheduling delays.
- The ability to monitor the effectiveness of the remedy.
- Consistency with regulatory requirements, activities needed to coordinate with and obtain necessary approvals and permits from other governmental bodies.
- Availability of necessary services, materials, equipment, and specialists, including the availability of adequate treatment and disposal services.

Each RAA was evaluated and assigned a ranking between 0 and 5 for implementability. A ranking of 0 indicates that an alternative has a significant degree of difficulty in implementing the alternative and a ranking of 5 indicates an alternative has little to no difficulty in implementing the alternative.

Implementation Risk. This factor includes evaluation of the potential risk to human health and the environment associated with remedial action implementation. Criteria for evaluating implementation risk include:

- Potential impacts to the community, workers involved in implementing the remedial action, and the environment and the effectiveness and reliability of protective measures to mitigate these impacts.
- Time until the remedial action is complete.

Each RAA was evaluated and assigned a ranking between 0 and 5 for implementation risk. A ranking of 0 indicates that an alternative has significant implementation risk and a ranking of 5 indicates an alternative has little to no implementation risk.

Reasonableness of Cost. This factor evaluates the reasonableness of the costs associated with the remedial action. Each alternative is assessed for reasonableness of cost by considering:

- Capital, operations and maintenance (O&M), and periodic review costs.
- The net present value of the above.
- The degree to which costs associated with the remedial action are proportionate to the benefits to human health and the environment through risk reduction or management.

- For any hot spots identified, the degree to which costs associated with the remedial action are proportionate to the benefits created through restoration or protection of beneficial uses of water.
- The degree of sensitivity and uncertainty of the costs.

In general, the least expensive remedial action is preferred unless the additional cost of a more expensive corrective action is justified by proportionately greater benefits to one or more of the other balancing factors. A higher threshold is used for evaluating the reasonableness of costs for treatment of hot spots than for remediation of non-hot spot areas.

Each RAA was evaluated and assigned a ranking between 0 and 5 for reasonableness of cost. A ranking of 0 indicates that an alternative has the highest cost and a ranking of 5 indicates an alternative has the lowest estimated cost.

Preference to Treat or Remove Hot Spots. As defined by OAR 340-122-0115(b), nongroundwater or surface water hot spots exist if hazardous substances present an unacceptable risk to human health or the environment and if the contamination is sufficiently concentrated, likely to migrate, or are not reliably containable. Under OAR 340-122-0090(4), a preference is given to alternatives that include treatment or excavation of hot spots to the extent feasible. In addition, a higher cost threshold is applied to sites where hot spots are present.

Each RAA was evaluated and assigned a ranking between 0 and 5 for hot spot treatment or removal. A ranking of 0 indicates that an alternative has no preference to treat or removal hot spots and a ranking of 5 indicates a meaningful reduction in hot spot contamination.

Green Remediation. Green remediation includes practices that lessen the overall environmental impact of remedial actions, such as limiting resources required for implementing the remedy, reducing generation of waste, and reducing greenhouse gas emissions. Alternatives were ranked and evaluated for their inclusion of green remediation technologies or methods.

Each RAA was evaluated and assigned a ranking between 0 and 5 for green remediation. A ranking of 0 indicates that an alternative has no preference for green remediation and a ranking of 5 indicates an alternative has meaningful reduction of resources and greenhouse gas emissions.

6.4 EVALUATION OF ALTERNATIVES AGAINST CRITERIA

Each RAA for each PAA was evaluated for protectiveness and balancing criteria. First, alternatives were screened for whether they achieve or do not achieve protectiveness. Alternatives that do not achieve protectiveness were not further evaluated while those that do achieve protectiveness were evaluated for the remaining criteria. Second, scores and discussion are provided for each balancing factor. Third, initial scores are calculated as the sum of the rankings for effectiveness, long-term reliability, implementability, implementation risk, and cost. Lastly, final scores are calculated as the initial score plus the scores for hot spot treatment and green remediation.

The following sections summarize the outcomes of the scoring and RAA selection for each PAA. Please reference Tables 12 through 16 for a more detailed explanation regarding how each alternative was evaluated against each criterion and Table 17 for a scoring summary for all RAAs considered for each PAA. All alternatives except RAA-1 (No Action) are considered to provide some level of protectiveness and were evaluated for all criteria.

7. COMPARATIVE ANALYSIS OF REMEDIAL ACTION ALTERNATIVES

In this section, each of the RAAs determined to be protective are compared with each other for the remedy evaluation criteria identified in Section 6.1. Because RAA-1 (No Action) is not considered protective and is only used as a baseline for evaluating the other RAAs, RAA-1 is not included in the comparative analysis. Sections 7.1 through 7.5 provide discussion summarizing the major conclusions of the comparative analysis and justification for differentiating issues specific to each of the upland and in-water PAAs. Additional details for each of the PAAs is presented in Tables 12 through 16. Table 17 provides an overall summary of the rankings for each of the PAAs.

7.1 AREA 1 UPLAND PRIORITY ACTION AREA

RAA-4 consisting of an impervious surface cap and permeable reactive barrier had the highest overall score (23) for any of the upland RAAs considered. RAA-6 ranked higher than RAA-4 for effectiveness and long-term reliability because RAA-6 includes removal of the hot spot volume of soil and transporting the material off-site to a permitted facility for disposal. However, during implementation, RAA-6 would have more short-term risks, including limitations related to critical resources, such as dump-trucks that may be limited to meet the demand necessary for completing the project on schedule; higher production of greenhouse gasses; and overall risk to the neighbors as a result of the large increase in traffic. RAA-2 would be easy to implement and therefore scored higher than the other RAAs for implementability and implementation risk. RAA-2 would result in the lowest short-term implementation risks. However, the effectiveness is ranked lower for RAA-2 because it would not meet several of the upland RAOs, including RAO 3 and RAO 4 (prevent recontamination of the in-water remedy). All RAAs would require institutional and engineering controls (e.g., Site use restrictions, CMMP, impervious surface cap, stormwater management, inspections, and maintenance) to achieve RAO 5 through RAO 8.

The cost to complete RAA-2 is the lowest of the RAAs. However, the alternative does not reduce the overall contaminant mass. The cost to complete RAA-4 is slightly less than the cost to complete RAA-5 and both would result in similar removal of upland contaminated soil and be effective in meeting all the upland RAOs (RAO 3 through RAO 8). Although RAA-3 would limit residual mass flux from the upland in groundwater, the cost to implement RAA-3 is higher than RAA-4 and RAA-5 due to the on-going monitoring and maintenance that would be required for the hydraulic containment system. The cost of RAA-6 is an order of magnitude higher than the other RAAs considered for the upland PAA and is considered unreasonable, especially when considering the associated short-term implementation risks.

Additional details and explanation are included in Table 12.

7.2 AREA 1 DOCK PRIORITY ACTION AREA

For the Area 1 Dock PAA, RAA-2 consisting of the installation of an armored reactive isolation cap had the highest initial score (16) considering the five primary evaluation criteria scores and the highest final score (21) when the additional criteria of hot spot treatment and green remedial considerations were also evaluated. RAA-2, RAA-3, and RAA-4 include an armored reactive

isolation cap, which will reduce contaminant mass flux to the river and provide a barrier to direct contact with underlying impacted residual or source sediment. RAA-2 scored lower for effectiveness and long-term reliability than RAA-3 and RAA-4, which included removal of shallow hot spot sediment and creosote-impacted timber piles and surficial woody debris. However, RAA-2 scored highest for implementability compared to RAA-3 and RAA-4 because the significant volume of old, highly weathered wood debris present in the Area 1 Dock PAA is unlikely to be successfully removed, would likely make surface and subsurface sediment removal very difficult, and also likely result in an overall increase in the short-term risks to receptors related to the management of the impacted sediment and debris. RAA-2 has less short-term implementation risks to receptors and the environment based on a more focused scope of work, compared to RAA-3 and RAA-4.

Data gaps exist regarding contamination depth profiles in this PAA, and preliminary data suggest that removal of surface material may expose deeper, more highly contaminated material, which was tied into the implementability scores. Additional characterization and evaluation during remedial design would be needed to determine whether removing surface material is feasible and protective.

RAA-2 (amended isolation cap only), is the easiest to implement, has the lowest short-term risks to receptors, and has the lowest cost compared to other alternatives evaluated. However, RAA-2 removes the lowest volume of creosote-contaminated media compared to the other RAAs. RAA-5 would remove the largest volume of hot spot material, has the highest cost, and has the lowest overall score of all the RAAs evaluated, due to challenges associated with implementation and increased implementation risk.

If woody debris can be successfully removed, and subsurface sediment and debris could be accessed, up to approximately 80% of hot spot material could be removed through implementation of RAA-3 and RAA-4, while nearly all hot spot material could be removed with RAA-5. This would result in improved effectiveness and long-term reliability, but the increased scale and complexity of the required construction would make implementation challenging, implementation risks high, and have notably greater costs, particularly for RAA-5. Additional characterization during remedial design will provide additional information to consider if some areas of shallow sediment can be removed without decreasing the effectiveness and implementability in the Area 1 Dock PAA.

Additional details and explanation are included in Table 13.

7.3 AREA 2 DOCK PRIORITY ACTION AREA

RAA-4 scored highest overall for the Area 2 Dock PAA and consists of nearshore removal, offsite disposal, and placing a sand cap for ENR and to cover remaining residuals. RAA-4 and RAA-2 (armored reactive isolation cap only) both had the highest initial score (16). However, RAA-4 had the highest final score (22) because RAA-4 scored higher for effectiveness and long-term reliability through removal of the majority of the contaminated, near-shore surface hot spot material (up to approximately 84%). Each of the alternatives is easy to implement, with local resources available. Additional resources would be required to complete RAA-4 because of the additional scope of work required to dewater the sediment and transport the material to a permitted

disposal facility, which resulted in a lower score for implementability and an increase in the shortterm risks to receptors related to off-site transport (pollution, truck traffic through residential areas, etc.). RAA-4 effectively meets DEQ's preference to remove hot spot material from the PAA and removes more hot spot material from the PAA compared to RAA-2. RAA-3 would result in the same volume of hot spot media removed. However, the material would remain on-site, which increases the level of uncertainty for the long-term reliability of an on-site landfill.

The cost to complete the preferred alternative, RAA-4 is slightly higher than the cost to implement RAA-2 and RAA-3. However, the RAA-4 includes the removal of hot spot material compared to RAA-2 and the offsite disposal component included for RAA-4 resulted in a higher score and advantage over RAA-3 because RAA-4 would reduce permitting and stakeholder negotiations and eliminate the engineering controls, ICs, and monitoring and maintenance that would be required for on-site consolidations.

Additional details and explanation are included in Table 14.

7.4 COVE AREA PRIORITY ACTION AREA

RAA-4 consists of riverbank restoration, nearshore removal action, offsite disposal, and an armored reactive cap. RAA-2, RAA-3, and RAA-4 scored the same with an overall score of 21 points for the Cove Area PAA. RAA-3 and RAA-4 scored lower than RAA-2 (armored reactive cap only) for implementability and implementation risks due to their increased scopes. However, RAA-3 and RAA-4 include regrading the riverbank to reduce erosional forces imposed on the cap and remove near-shore shallow sediment, where hot spot concentrations are present, resulting in increased effectiveness and long-term reliability compared to RAA-2. RAA-5 would completely remove all surface NAPL source material and provide improved effectiveness and long-term reliability, but the increased scale and complexity of the required construction would make implementation challenging and result in an overall increase in the implementation risks. RAA-3 scored slightly lower than RAA-4 for the long-term reliability because RAA-3 would require long-term monitoring and maintenance and there is additional uncertainty related to maintaining an onsite landfill.

The cost for RAA-2 was the lowest of all the alternatives evaluated. However, RAA-2 also removes the smallest volume of hot spot material than the other RAAs evaluated. Alternatives RAA-3 and RAA-4 were similar in cost. RAA-5 had a notably greater costs than the other alternatives. Although the final score of RAA-4 was tied with RAA-3, RAA-4 would reduce permitting and stakeholder negotiations and eliminate the engineering controls, ICs, and monitoring and maintenance that would be required for on-site consolidation.

Additional details and explanation are included in Table 15.

7.5 UPPER MILTON CREEK PRIORITY ACTION AREA

RAA-4, consisting of streambank regrading, limited removal action, offsite disposal, and an armored reactive cap had the highest initial score (17) considering the five primary evaluation criteria and tied with RAA-3 for the highest final score (21) when the additional criteria of hot spot

treatment and green remedial considerations were also evaluated for the Upper Milton Creek PAA. RAA-3 and RAA-4 both include streambank regrading, which provides many advantages over RAA-2, which does not include regrading. The regrading will remove some hot spot material, facilitate placement of the cap material, improve worker safety, and reduce erosional forces imposed on the cap, thereby resulting in increased effectiveness and long-term reliability compared to RAA-2. The long-term reliability was slightly lower for RAA-3 compared to RAA-4 because there are more future unknowns related to placing material within the 100-year flood plain for RAA-3. The implementation risks for RAA-2 and RAA-3 are less than the short-term risks to complete RAA-4 because RAA-4 would require truck traffic outside of the Site to transport wastes to the landfill. Each of the RAAs are feasible and easy to implement with readily available resources.

The cost to complete each of the RAAs in the Upper Milton Creek PAA are similar, with the cost to complete RAA-2 (armored cap only) slightly less. However, RAA-2 does not meet DEQ's preference for hot spot removal. The cost to complete RAA-4 is the highest of all the alternatives. RAA-4 would reduce permitting and stakeholder negotiations and eliminate the engineering controls, ICs, and monitoring and maintenance that would be required for on-site consolidation.

Additional details and explanation are included in Table 16.

8. RECOMMENDED REMEDIAL ACTION ALTERNATIVES IN STAFF REPORT

After a detailed evaluation of the remedial alternatives for each PAA (see Section 6 and 7), the most feasible, protective alternative was identified and presented to the public for review and comment in the form of a Staff Report. The recommended remedial alternatives are:

- Area 1 Upland PAA (RAA-4): This alternative includes placement of an impervious surface cap over the entire Area 1 Upland PAA and a permeable reactive barrier at the top of the riverbank, adjacent to the Cove PAA.
- Area 1 Dock PAA (RAA-2): This alternative includes removal of piles and creosoteimpacted surficial woody debris and placement of an armored reactive cap.
- Area 2 Dock PAA (RAA-4): This alternative includes removal of surficial woody debris and 235 piles; removal of nearshore contaminated sediment to a depth of 1-foot below the sediment surface, where creosote sheen has been observed; and the placement of a thin, 1foot-thick sand cap for ENR and to cover remaining sediment that may have residual impacts.
- **Cove Area PAA (RAA-4):** This alternative includes excavation and regrading for hot spot sediment and soil removal and armored reactive capping along the riverbank.
- Upper Milton Creek PAA (RAA-4): This alternative includes removal of hot spot soil and sediment, regrading the adjacent bank, installing an armored reactive cap across the Upper Milton Creek PAA, and conducting long-term MNR.

The remedial alternatives for each PAA form an integrated, cost-effective approach that removes and contains contaminated media, including hot spots, through a combination of upland groundwater in-situ treatment using sequestration agents (e.g., organoclays and granular activated carbon amendments); physical isolation of the in-water contamination through a combination of contaminated sediment and debris removal, capping, and contaminant burial (where natural recovery is already occurring); and reduction of risks to receptors through immobilization of bio-available contaminants in the in-water areas.

9. PUBLIC NOTICE, PUBLIC COMMENT, AND AGENCY RESPONSE

Pursuant to ORS 465.320 and OAR 340-122-0100, notice of DEQ's recommended remedial action for the Site was published in the Oregon Secretary of State's *The Oregon Bulletin* June and July issues and bi-weekly in *The Oregonian* and *Columbia County Spotlight* newspapers during June and July. The public notice was also posted on DEQ's public notifications web page. The 60-day comment period commenced on June 1, 2023, and ended on July 31, 2023. Public comments received and DEQ's response are included on Table 18.

DEQ also held several virtual informational sessions with various partners, community members, and stakeholders. Information about the Site, contamination, and recommended remedial action was presented followed by time for questions and answers. One of the informational sessions, minus the questions and answers portion, was recorded and is available to the public.

The Staff Report published May 31, 2023, presented DEQ's recommendation in greater detail based on previous work conducted at the Site, including remedial investigations, risk assessments, and feasibility study activities. These supporting documents are available online through DEQ's ECSI database for the Site (ECSI 0959).

DEQ's selected remedial action is consistent with the recommended remedial action presented in the Staff Report. The selected remedial action for contaminated media in the Area 1 Upland and contaminated sediments located in Upper Milton Creek and Scappoose Bay adjacent to the former Pope and Talbot wood-treating is protective, and reflects the best balance of effectiveness, long-term reliability, implementability, implementation risks, reasonableness of costs, preference to treat or remove hot spots, and green remediation. Long-term monitoring and maintenance will be required to ensure the remedy remains protective over time. The selected action therefore satisfies the requirements of ORS 465.314 and OAR 340-122-0090.

10.1 DESCRIPTION OF THE SELECTED REMEDIAL ACTION

The following paragraphs describe the selected remedial action for each of the PAAs at the Site. There are no significant changes from the remedial action recommended in the Staff Report and the selected remedial action presented in this ROD. In addition to the specific remedial action described for each PAA below, Site-wide ICs will ensure long-term effectiveness of the selected remedial action for the Area 1 Upland PAA, and in-water ICs and MNR will help ensure effectiveness of the selected remedial action for all of the in-water PAAs. The total estimated cost for the selected remedial action is 22.1 million dollars.

10.1.1 Area 1 Upland Priority Action Area

The selected remedial action for Area 1 Upland PAA is RAA-4, which includes placement of an impervious surface cap over the entire Area 1 Upland PAA and a permeable reactive barrier at the top of the riverbank, adjacent to the Cove area, as conceptually shown in Figure 43. When combined with the Site-wide ICs, the selected remedial action protects on-site workers from direct exposure to the impacted media in the upland and achieves RAOs 5 through 8. The selected remedial action provides source control for the upland groundwater and in combination with the Upper Milton Creek and Cove Area RAAs, also achieves prevention of recontamination of inwater remedies, which will achieve RAOs 3 and 4.

The surface cap would be graded to direct stormwater away from the cap and into a stormwater collection and conveyance system that discharges outside of the Area 1 Upland PAA. Routing stormwater away from the Area 1 Upland PAA reduces infiltration into the underlying subsurface soil, where creosote NAPL is present, and associated groundwater flux towards in-water locations. The impermeable cap also provides protection against direct contact risks to receptors due to PCBs in the roadway soil and provides a vapor barrier to mitigate inhalation risks associated with volatilization of contaminants into ambient air. The PRB will be approximately 2 to 4 feet wide, consist of adsorptive and reactive materials (e.g., organoclay or granular activated carbon), and be placed at the top of the riverbank, upland of the Cove Area, where creosote NAPL seeps have been observed. PRBs have been shown to be effective in treating similar contaminants at other wood-treating cleanup sites. The PRB treatment wall will intercept and sequester creosote NAPL and dissolved phase COCs in groundwater before discharging into the Cove Area of Scappoose Bay, thereby providing source control for the groundwater and potentially mobile NAPL present in the

Area 1 Upland PAA. Although hot spots will remain within the Area 1 Upland PAA, the PRB combined with the engineered impervious surface cap and stormwater management system is expected to significantly reduce hot spots mobilizing downgradient of the PRB and stimulate biodegradation of dissolved phase PAHs and TPH in groundwater that migrate towards the Cove Area PAA. The actual footprint of the PRB, type of amendment, and material thickness and depth will be determined during remedial design.

The materials removed during PRB installation will be transported off-site for disposal. The selected remedial action will result in removal of approximately 3,000 tons of hot spot material that will be transported to a subtitle C landfill, and additional approximate 520 tons of impacted material that will be transported for disposal at a subtitle D landfill.

10.1.2 Area 1 Dock Priority Action Area

The selected remedial action for Area 1 Dock PAA is RAA-2, which includes removal of piles and creosote-impacted surficial woody debris and placement of an armored reactive cap. The armored reactive cap will remove direct contact risks to receptors, as identified in RAOs 1a, 1b, and 1c and immobilize bioavailable contaminants in porewater and surface water, which protects aquatic receptors from future exposures of any contaminants remaining below the cap (RAO 2). The Area 1 Dock PAA is shown on Figure 44 and the cross-sections for the selected remedial action are shown on Figures 45A and 45B.

An estimated total of 435 piles will be removed or attempted to be removed from the Area 1 Dock PAA. The piles and surficial debris (approx. 200 tons) will be removed and transported off-site for disposal as non-hazardous waste at a subtitle D landfill. The impacted sediment and debris removed will be transported off-site for disposal. The selected remedial action will result in removal of approximately 160 tons of impacted material that will be transported for disposal at a subtitle D landfill.

Additional sampling will be conducted during RD to inform the actual extent of the cap. Additional sampling will be completed following removal of the debris to verify the leave surface is consistent with the assumptions used for the final amended cap design. DEQ anticipates sampling will also determine if some removal of highly concentrated sediment proximal to and underlying creosote-impacted debris material and in shallow surface sediments is warranted. If found to be necessary, some hot spot surface sediment will be removed for off-site disposal at a permitted landfill, prior to installing the armored reactive cap.

Following removal of the piles and surficial debris, an estimated 1-foot-thick amended isolation cap will be placed across 80,600 square feet (approximately 56% of PAA) of the Area 1 Dock PAA where heavy to moderate sheen is present. The final amendments used in the amended isolation cap will be determined during remedial design. Areas around any remaining piles or dock structures that cannot be removed may require a different formulation of the amendment (e.g., proprietary product), dosage of amendment, or application method to achieve the performance goals of the amended isolation cap. The amended isolation cap will be armored with an estimated 2-foot-thick layer of protective rock with a habitat rock cover.

Surface and subsurface sediment exceeding the hot spot, highly concentrated criteria, will remain in-place, extending in some areas to depths greater than 14 feet bml, unless during remedial design it is determined that removal of this material is feasible and protective. However, this sediment would be isolated from direct contact through installation of the amended isolation cap, and the protective armor cover. Removal of the surficial debris, creosote-impacted timber piles, and installation of the amended isolation cap satisfies the preference for treatment of hot spots to the extent practicable. MNR will be implemented for the Area 1 Dock PAA outside of the limits of the protective isolation cap, with an overall timeline of 10 years to reach the remedial goals for the PAA. If the remedial goals are not met in this time, additional action may be needed.

10.1.3 Area 2 Dock Priority Action Area

The selected remedial action for Area 2 Dock PAA is RAA-4. RAA-4 addresses the in-water RAOs (RAO 1a, 1b, 1c, and 2) and is protective of human health and ecological receptors through a combination of removing surficial woody debris and 235 piles; removing nearshore contaminated sediment to a depth of 1-foot below the sediment surface, where creosote sheen has been observed; and the placement of a thin, 1-foot-thick sand cap over the entire PAA for ENR and to cover remaining sediment that may have residual impacts. The specific cap material, thickness, extent, and sufficiency or insufficiency of using sand only will be determined during remedial design. The Area 2 Dock PAA is shown on Figure 44 and the cross-section for the selected remedial action is shown on Figure 46.

Robust best management practices (BMPs) will be used during the removal of woody debris, piling, and sediment, including the use of a cofferdam and dewatering the area to allow for landbased access to the debris and impacted surface sediment. This management approach during implementation will reduce the short-term risks to in-water receptors and allow for a thorough removal of the sheen-impacted sediment in the Area 2 Dock PAA. The impacted sediment and debris will be transported off-site for disposal. The selected remedial action will result in removal of approximately 500 tons of hot spot material that will be transported to a subtitle C landfill, and additional approximate 200 tons of impacted material that will be transported for disposal at a subtitle D landfill.

10.1.4 Cove Area Priority Action Area

The selected remedial action for the Cove Area PAA is RAA-4, which is a combination excavation and regrading for hot spot sediment and soil removal and armored reactive capping along the riverbank. The amended isolation cap will immobilize bioavailable contaminants in porewater and surface water, which protects aquatic receptors from future exposures of any contaminants remaining below the cap (RAO 2). A depiction of the Cove Area PAA is shown on Figure 44 and the cross-sections for the selected remedial action are shown on Figures 47A, 47B, and 47C.

An estimated total of 139 piles and the hot spot areas of sediment, where heavy sheen has been observed in the Cove Area, will be removed. Removal of the heavy sheens in this area satisfies DEQs hot spot rule and RAOs 1a, 1b, and 1c. Approximately 540 linear feet along the adjacent bank will be regraded to achieve a stable slope of approximately 5H:1V. Removal of the adjacent bank soil will remove additional hot spots and prevent recontamination of in-water remedy from

the adjacent bank area, which satisfies Upland RAO 3. The impacted sediment, bank soil, and debris removed will be transported off-site for disposal. The selected remedial action will result in removal of approximately 3,800 tons of hot spot material that will be transported to a subtitle C landfill, and additional approximate 1,600 tons of impacted material that will be transported for disposal at a subtitle D landfill.

Following removal of the soil and sediment, an estimated 1-foot-thick amended isolation cap will be placed across the Cove Area PAA, estimated to be an area of 50,000 square feet (approximately 33% of PAA) to address the residual soil and sediment concentrations in the areas where the highest concentrations were detected. The final amendments will be determined as part of the remedial design. The amended isolation cap will also be protected by the addition of an estimated 2-foot-thick layer of armor stone with a habitat rock cover. MNR will be implemented for the Cove Area PAA outside of the limits of the cap.

10.1.5 Upper Milton Creek Priority Action Area

The selected remedial action for Upper Milton Creek PAA is RAA-4. RAA-4 includes removal of hot spot soil and sediment, regrading the adjacent bank, installing an amended isolation cap across the Upper Milton Creek PAA, and conducting long-term MNR. The amended isolation cap will immobilize bioavailable contaminants in porewater and surface water, which protects aquatic receptors from future exposures of any contaminants remaining below the cap (RAO 2). Removal of the adjacent bank soil will prevent recontamination of in-water remedy from the adjacent bank area, which satisfies Upland RAO 3. The selected remedial action in Upper Milton Creek PAA also achieves prevention of direct contact with sheen (RAOs 1, 2, and 3). A depiction of the Upper Milton Creek PAA is shown on Figure 44 and the cross-section for the selected remedial action is shown on Figure 48.

Hot spot sediment, including locations where heavy sheen has been observed in the creek, will be removed using land-based equipment. Removal of the heavy sheens in this area satisfies DEQs hot spot rule and RAOs 1a, 1b, and 1c. The streambank adjacent to the creek will be regraded to reduce the current slope (up to 2H:1V) to a more stabilize slope (i.e., 4H:1V). The impacted sediment and impacted bank soil will be analyzed to determine the final disposition of the material. However, it is assumed that up to 1,550 tons of hot spot material will be transported off-site for disposal at a subtitle C landfill and an additional 82 cubic yards of impacted material will be transported for disposal at a subtitle D landfill.

Following removal of the impacted bank soil and creek sediment, an estimated 1-foot-thick isolation cap amended with organoclay will be placed across an approximate 2,000-square-foot area to cover residual soil and sediment. The amended isolation cap will also be protected by the addition of an estimated 2-foot-thick layer of armor stone with a habitat rock cover.

10.1.6 Engineering and Institutional Controls

Until such time that RAOs are achieved, limiting the potential for humans to ingest fish contaminated with primary COCs under recreational angling and subsistence angling scenarios, to the extent practicable, will be necessary. An advisory regarding consumption of fish, shellfish, and crayfish in Multnomah Channel and Scappoose Bay was issued by OHA in December 2020.

Signage communicating the OHA advisory will be installed to dissuade trespassers, recreational anglers, and subsistence anglers from fishing at or near the Site.

Additional ICs will be implemented as needed to manage human health and ecological risks. Other expected ICs that will be considered and more fully addressed during remedial design for the Area 1 Upland PAA include a prohibition on groundwater pumping/use in the upland and a restriction on upland property excavation, new construction, or redevelopment without DEQ approval/authorization. ICs will also be needed for the in-water PAAs, such as restrictions related to dredging within the project limits, vessel anchoring, and limits on navigation over in-water capped areas, etc.

10.1.7 Performance Monitoring

The remedial design work will also be used to inform the requirements for the performance monitoring for the Site, including a determination of points of compliance and timelines for performance monitoring, and an identification of contingency measures that may be implemented in the event that remedial measures prove to be ineffective or do not meet RAOs within specified time frames.

Short-term performance monitoring for the remedial action is anticipated to be completed directly after remedy implementation during Years 0 to 4, and long-term performance monitoring would be performed as determined to be necessary at routine intervals (e.g., Year 5, 10, and 15, etc.) thereafter.

Short- and long-term monitoring activities anticipated for the Site include upland groundwater sampling, in-water multi-beam bathymetric surveys, collection and analysis of surface sediment and porewater sampling, and potentially bioassays and benthic surveys to confirm RAOs are being achieved. Typical performance monitoring programs include a combination of the following monitoring activities:

- Periodic groundwater monitoring to evaluate the concentrations in the vicinity of the Upland PRB are consistent with anticipated design concentrations and the wall is performing as intended (i.e., no excessive biofouling, no excessive movement of the creosote NAPL body is occurring in the vicinity of the PRB, no evidence of active seeps along bank).
- Periodic cap inspections in the upland, along with in-water bathymetric surveys to ensure that the physical integrity of all caps is maintained and not compromised by scour, erosion, or other physical disturbances (e.g., prop-wash, wave-wake effect, etc.).
- Visual observations along Upper Milton Creek and Cove Area PAA banks to ensure the riverbanks remain stable and are not posing a recontamination issue due to excessive erosion and/or chemical breakthrough.
- Sediment, porewater, and/or surface water samples to monitor the overall effectiveness of the in-water remedy and performance of the amended isolation caps over time to ensure that cleanup goals are being achieved within the cleanup time frame (10 years) and

maintained in sediment and porewater at the Site (and if they are not, evaluate whether the Site is being recontaminated by on-site and/or off-site sources).

- Monitor the effectiveness of natural recovery in MNR and ENR areas to ensure that recovery is progressing in areas outside of the in-water PAAs (prioritizing areas where heavy to moderate sheen have been observed) and recovery is progressing in the time frame (10 years) as included in the designed remedial action and identified in the overall PRGs for the Site. This effort should also include further evaluation of areas outside of the in-water PAAs exhibiting light petroleum sheen, especially areas where elevated contaminant levels have been previously observed (e.g., Lower Milton Creek near the location of the 2017 PWS-090617-2 sediment porewater sample).
- Collect data in remedial areas to evaluate contaminant concentration trends in each of the media (groundwater, sediment, and porewater) relative to the final RAOs and PRGs established for the Site during the remedial design effort.

10.1.8 Contingency Measures

In the event that the selected remedial action for each individual PAA is implemented and determined not to meet RAOs, either in the form of upland groundwater sequestration, debris and/or sediment removal, amended isolation capping, or natural recovery, contingency measures may be necessary. If contingency measures are necessary, they are expected to rely on the augmentation of selected remedial action.

Potential contingency measures could include: a) upgrading MNR areas to ENR (i.e., applying sand cover material to MNR and/or ENR areas); b) upgrading ENR areas to in situ treatment (i.e., applying more robust ENR effort in areas by extending the footprint of the ENR area or adding an amendment to the cover); c) upgrading ENR areas to engineered cap; d) increasing cap thickness and/or amendment concentrations; and e) localized hot spot removal with or without capping. Contingency measures will be formally addressed in a Performance Monitoring, Review, and Contingency Plan that will be prepared as part of the forthcoming remedial design document for the property following issuance of this ROD.

10.2 RESIDUAL RISK ASSESSMENT

OAR 340-122-0084(4)(c) requires a residual risk evaluation of the selected remedy to demonstrate that the standards specified in OAR 340-122-0040 will be met, namely:

- Assure protection of present and future public health, safety, and welfare, and the environment.
- Achieve CULs (i.e., the highest of acceptable RBCs or background concentrations).
- For designated hot spots of contamination, evaluate whether treatment or removal is reasonably likely to restore or protect a beneficial use within a reasonable time.

• Prevent or minimize future releases and migration of hazardous substances in the environment.

In the upland, the selected remedy achieves acceptable risk levels through an impervious cap that will limit direct contact with soil. The impervious cap will also limit stormwater infiltration into the upland soil and groundwater source area where hot spot levels of contamination are present, including creosote NAPL. The selected upland remedy also includes a PRB that will limit contaminant migration from the upland source area to the in-water area. The upland remedy does not directly treat source material in upland groundwater or subsurface soil; risks to potential industrial workers, excavation workers, and construction workers in the upland will therefore need to be addressed through ICs (EES and CMMP).

The long-term effectiveness of the selected remedy for Area 1 Upland is dependent on ICs and engineering controls to mitigate risks to potential industrial workers, construction workers, and excavation workers in the upland. The selected remedy for the upland assumes that an EES would be made between the Port of Columbia County and DEQ. ICs and engineering controls, including a CMMP, land use restrictions, water use restrictions, identification of engineered barriers, and long-term monitoring requirements will all need to be incorporated into the EES. To be protective of potential future industrial workers, this EES will need to mitigate or prevent inhalation exposure to contaminants volatilizing to both outdoor and indoor air from groundwater contamination in the source area of the upland. To be protective of potential future excavation workers, this EES will need to mitigate or prevent direct contact with and inhalation of contaminated groundwater in the upland. To be protective of potential future construction and excavation workers, this EES would also need to mitigate or prevent direct contact, ingestion, and inhalation of contaminated subsurface soil in the upland.

In the in-water areas, the selected remedy achieves acceptable risk levels through a combination of removal and capping of contaminated sediment. Concentrations of COCs in surface sediment will be immediately reduced at construction completion. In addition, cleaner sediments from upstream will continue to be deposited, further reducing COC concentrations at the sediment surface. The selected remedy does not, however, result in the immediate removal or destruction of all Site-related NAPL, petroleum sheen, or contaminated woody debris in the in-water areas. Monitoring after remedial implementation will be needed to evaluate any residual risk to subsistence fishers, sport fishers, or ecological receptors associated with residual contamination in sediment, porewater, or surface water in the in-water areas. As described in Section 10.1.8, contingency measures to address any residual risks that are identified will be addressed in a Performance Monitoring, Review and Contingency Plan that will be prepared as part of the forthcoming remedial design document for the Site.

The selected in-water remedy also achieves cleanup risk levels through preventing ongoing releases and direct contact with the NAPL and petroleum sheen at the seep locations and through immobilization of bioavailable contaminants in pore water and surface water in the in-water areas. Although the selected remedy does not result in the actual removal or destruction of all Site-related contaminants, the addition of reactive material to capping material in in-water PAAs is intended to make bioavailable COCs in sediment less bioavailable to humans and ecological receptors. Specifics of cap composition will be identified during remedial design with the intention of reducing bioavailability and mobility of contamination beneath the in-water caps. DEQ anticipates

that porewater data collected post-construction will be used as a LOE to evaluate the efficacy of the remedy in reducing the bioavailability of COCs in impacted in-water areas.

Some potential residual Site risks, primarily associated with light petroleum sheen, are expected for a relatively short period of time over a limited area at the completion of the remedial action. However, the limited residual Site risks are expected to be mitigated through natural recovery processes, and by Year 10, Site-related risks are expected to be within acceptable levels throughout the in-water area. Performance monitoring will track the progress of the preferred remedy in achieving the RAOs following remedy construction in the PAAs. Additional investigation may be needed to evaluate and mitigate any potential residual risks (e.g., associated with light petroleum sheen) in in-water areas outside of the PAAs (e.g., Lower Milton Creek). As discussed in Sections 5.2.7, 10.1.7, and 10.1.8, the performance, monitoring, review, and contingency plans will be developed during remedial design.
The selected remedial action for contaminated media in the Area 1 Upland and contaminated sediments located in Upper Milton Creek and Scappoose Bay adjacent to the former Pope and Talbot wood-treating is protective, and reflects the best balance of tradeoffs considering effectiveness, long-term reliability, implementability, implementation risks, and reasonableness of costs. Long-term monitoring and maintenance will be performed to ensure the remedy remains protective over time. The selected remedial action therefore satisfies the requirements of ORS 465.314 and OAR 340-122-0090.

Kevin Parrett, Manager Northwest Region Cleanup Program Department of Environmental Quality

September 1, 2023 Date

ADMINISTRATIVE RECORD INDEX Former Pope & Talbot Wood-Treating Site St. Helens, Oregon

The Administrative Record consists of the documents on which the selected remedial action for the Site is based. The primary documents used in developing and evaluating RAAs for the former Pope & Talbot Wood-Treating Site are listed below. Additional background and supporting information can be found in the former Pope & Talbot Wood-Treating Site (ECSI No. 0959) project file located at DEQ Northwest Region Office, 700 NE Multnomah Street, Suite 600, Portland, Oregon and online.

SITE-SPECIFIC DOCUMENTS

- Amec Environmental & Infrastructure, Inc. 2014. Draft Supplemental Remedial Investigation Report. May 22, 2014.
- Bridgewater Group and Hart Crowser. 2003. Level II Ecological Risk Assessment. October 17, 2003.
- Bridgewater Group and Kennedy/Jenks Consultants. 2006a. Human Health Risk Assessment, Pope & Talbot, Port St. Helens Site. November 28, 2006.
- Bridgewater Group and Kennedy/Jenks Consultants. 2006b. Ecological Risk Assessment Summary Report. November 27, 2006.
- Cascadia Associates. 2017. Phase 1 and 2 Offshore Data Gap Investigation Map Groundwater Discharge Areas and Sheen Occurrence. August 15, 2017.
- Cascadia Associates. 2018. Progress Report. Results of the Offshore Data Gap Investigation. April 20, 2018.
- Cascadia Associates. 2019. Map Creosote Sheen Occurrence Along Area 2 Shoreline. January 17, 2019.
- Cascadia Associates. 2019. Feasibility Study Work Plan. November 4, 2019.
- Cascadia Associates. 2020a. Updated Supplemental HHRA. January 2020.
- Cascadia Associates. 2020b. Updated Supplemental Ecological Risk Assessment. January 2020.
- Cascadia Associates. 2020c. Updated Supplemental Remediation Investigation Report. January 17, 2020.
- DEQ. 1988. Preliminary Assessment. December 14, 1988.
- DEQ. 1995. Order of Consent No. WMCSR-NWR-95-05, between Port of St. Helens (now Port of Columbia County) and DEQ. Effective April 13, 1995.
- DEQ, 2014a. DEQ Comments on Supplemental Remediation Investigation Report, Proposed Draft Annotated Outline, November 22, 2013, Port of St. Helens/Former Pope & Talbot Wood Treating Site, 1550 Railroad Avenue, St. Helens, Oregon, ECSI #959, January 30, 2014.

- DEQ, 2014b. E-mail from Deborah Bailey to Michelle Peterson re: PSH Evaluation of Porewater and Dermal Risks from Water Contact in the Supplemental HHRA, March 6, 2014.
- DEQ, 2020. Letter to Mr. Craig Allison with the Port, re: Conditional Approval, Updated Supplemental Remedial Investigation Report, Former Pope & Talbot Wood-Treating Site, ECSI No. 959. September 14.
- DEQ, 2023. Staff Report, Recommended Remedial Action, May 31, 2023.
- Ecology And Environment, Inc., 1990. Site Inspection. July 1990.
- Evarts, Russel C, 2004, Geologic Map of the Ariel Quadrangle, Clark and Cowlitz Counties, Washington.
- GeoEngineers. 2000. Remediation Investigation. April 7, 2000.

GeoEngineers. 2022. Revised Feasibility Study. September 22, 2022.

Harding Lawson Associates. 1994. Vadose Zone Contamination Investigation. January 28, 1994

STATE OF OREGON

- Oregon's Environmental Cleanup Laws, Oregon Revised Statutes 465.200-.900, as amended by the Oregon Legislature in 1995.
- Oregon's Groundwater Protection Act, Oregon Revised Statutes, Chapter 468B.
- Oregon's Hazardous Substance Remedial Action Rules, Oregon Administrative Rules, Chapter 340, Division 122, adopted by the Environmental Quality Commission in 1997.
- Oregon's Hazardous Waste Rules, Chapter 340, Divisions 100 120.
- Oregon's Water Quality Criteria, Chapter 340, Division 41, Columbia Basin.

GUIDANCE AND TECHNICAL INFORMATION

- DEQ. 2001a. Cleanup Program Quality Assurance Policy. September 1990, updated April 2001.
- DEQ. Consideration of Land Use in Environmental Remedial Actions. July 1998.
- DEQ. Green Remediation Policy. November 2, 2011.
- DEQ. Guidance for Conducting Beneficial Water Use Determinations at Environmental Cleanup Sites. July 1998.
- DEQ. Guidance for Conduct of Deterministic Human Health Risk Assessment. May 1998; updated May 2000.
- DEQ. Guidance for Conducting Feasibility Studies. July 1998, updated 2006 and 2017.
- DEQ. 2001b. Guidance for Ecological Risk Assessment: Levels I, II, III, IV. April 1998; updated December 2001.
- DEQ. Guidance for Identification of Hot Spots. April 1998.

- DEQ. Guidance for Use of Institutional Controls. April 1998.
- DEQ. Human Health Risk Assessment Guidance, Environmental Cleanup Program, October 18, 2010.
- MacDonald, D., Ingersoll, C. & Berger, T. Development and Evaluation of Consensus-Based Sediment Quality Guidelines for Freshwater Ecosystems. Archives of Environmental Contamination and Toxicology. 39, 20–31 (2000).
- San Francisco Bay Regional Water Quality Control Board (SFBRWQCB), 2019. Environmental Screening Levels, Tier 1 ESLs, 2019 (Rev. 2). San Francisco Bay Regional Water Quality Control Board. Rev. 2.
- Windward. 2013. Portland Harbor RI/FS, Remedial Investigation Report Appendix G, Baseline Ecological Risk Assessment, Final. Prepared for the Lower Willamette Group. Seattle, WA.

Table 1 **Summary of RI Data Collection Updated Supplemental RI Report** Former Pope & Talbot Wood-Treating Site St. Helens, Oregon

Sample Count

| | | | - | | _ | | - | | | | | | | | _ | | |
|-------------------|---------|-------|-------|------|------|------|------|------|------|---------------------------------------|------|------|-------|-------|------|------|----------|
| Matrix | 1996 | 1997 | 1998 | 1999 | 2000 | 2001 | 2003 | 2004 | 2005 | 2006 | 2007 | 2010 | 2011 | 2012 | 2013 | 2017 | Total |
| NAPL | 1 | 1 | | | | | | | | | | 1 | | 1 | | | 4 |
| Groundwater | 57 | 136 | 36 | 29 | 26 | 26 | 22 | 13 | 11 | 13 | 1 | 12 | 20 | 61 | | | 463 |
| Porewater | | | | | | | | | | | | | 11 | 63 | | 30 | 74 |
| Sediment | 24 | | | | | | 26 | 30 | 20 | | | | 120 | 92 | 2 | 15 | 329 |
| Seep Water | 1 | | 3 | | | | | | | | | | | | | | 4 |
| Soil | 165 | | 102 | | | | | | | | | | 12 | | 23 | | 302 |
| Surface Water | 4 | | 10 | | | | | - 14 | | | | | - | | · | 9 | 14 |
| Total | 252 | 137 | 151 | 29 | 26 | 26 | 48 | 43 | 31 | 13 | 1 | 13 | 163 | 217 | 25 | | 1,175 |
| Analytical Test C | ount | | | | | | | | | | | | | | | | <u> </u> |
| Method Group | 1996 | 1997 | 1998 | 1999 | 2000 | 2001 | 2003 | 2004 | 2005 | 2006 | 2007 | 2010 | 2011 | 2012 | 2013 | 2017 | Total |
| Dioxins | 6 | | | | | | | | | | | | | | | | 6 |
| Fuels | 75 | 89 | 66 | 18 | | | 22 | 11 | 11 | 12 | 1 | 25 | 121 | 61 | 25 | | 537 |
| Herbicides | 105 | 90 | 65 | 18 | 17 | 18 | | | | | | | | | | | 313 |
| Metals | 105 | 89 | 67 | 18 | 17 | 18 | 11 | 6 | 3 | 5 | | 12 | 23 | 61 | | | 435 |
| SVOCs | 140 | 99 | 80 | 21 | 18 | 20 | 48 | 41 | 31 | 13 | 0 | 12 | 126 | 193 | 25 | 45 | 867 |
| VOCs | 106 | 89 | 71 | 18 | 17 | 18 | | | | | | 12 | 7 | | | 15 | 338 |
| Other | · | | | | | | | | | | | | 136 | 209 | 25 | 15 | 385 |
| Total | 537 | 456 | 349 | 93 | 69 | 74 | 81 | 58 | 45 | 30 | 1 | 61 | 413 | 524 | 75 | 75 | 2,866 |
| Analytical Result | s Count | | | | | 1 | | | | | 1 | | | | | | |
| Matrix | 1996 | 1997 | 1998 | 1999 | 2000 | 2001 | 2003 | 2004 | 2005 | 2006 | 2007 | 2010 | 2011 | 2012 | 2013 | 2017 | Total |
| NAPL | 69 | 6 | | | | | | | | · · · · · · · · · · · · · · · · · · · | | 3 | | 35 | | | 113 |
| Groundwater | 2,156 | 4,468 | 1,337 | 951 | 762 | 843 | 477 | 238 | 223 | 262 | 4 | 355 | 462 | 1,551 | | | 14,089 |
| Porewater | | | | | | | | | | | | | | 2,226 | | 80 | 2,226 |
| Sediment | 432 | | | | | | 468 | 540 | 360 | | | | 2,571 | 2,548 | 36 | 80 | 7,035 |
| Seep Water | 21 | | 93 | | | | | | | | | | | | | | 114 |

Total Please refer to notes at end of table.

2,962

176

1,983

450

951

762

843

945

778

583

262

4

5,816 4,474 3,863

Soil

Surface Water



221

358 3,254 6,360 457

421

80

240

5,587

626

29,950

Table 1 **Summary of RI Data Collection Updated Supplemental RI Report** Former Pope & Talbot Wood-Treating Site St. Helens, Oregon

Analyte Class NAPL Dioxins Fuels Herbicides Metals SVOCs VOCs Groundwater Fuels Herbicides Metals SVOCs VOCs Porewater PAHs Other Sediment Fuels Metals SVOCs VOCs Other Seep Water SVOCs VOCs Soil Dioxins Fuels Herbicides Metals

Analytical Tests by Matrix Count

Acronyms/Abbreviations

PAHs = polycyclic aromatic hydrocarbons

SVOCs = semivolatile organic compounds

VOCs = volatile organic compounds

NAPL = nonaqueous-phase liquid

RI = Remedial Investigation

Other = Total Organic Carbon, Dissolved Organic Carbon, grain size, and/or percent solids

Notes

SVOCs

VOCs

Other

Fuels

Metals

SVOCs

VOCs

Total

Surface Water

Herbicides

Sample Count = number of samples collected from each medium

Analytical Test Count = number of tests run for each chemical class

Analytical Results Count = number of individual analyte results for each medium

Analytical Tests by Matrix Count = number of tests run for each chemical class and each medium

SVOCs include PAHs by various methods; refer to the 2000 RI Report (GeoEngineers 2000) for a full list of SVOC analytes reported in the RI.

otal

3,299

Table 2 Summary of Risks Former Pope & Talbot Wood Treating Site St. Helens, Oregon

| | Excess I Cance | ifetime er Risk | | Hazard Index | | | _ | | | | |
|---|---------------------|--------------------|---------------------|--------------|---------|-------------|-------------------|--|--|--|--|
| Source / Pathway | RME/Max Detected | СТЕ | RME/Max Detected | RME* | CTE | | Source | | | | |
| Industrial Worker | | | | | | | | | | | |
| Outdoor Air / Inhalation | 1E-06 | 3E-08 | 3.9E+02 | NA | 2.7E+01 | Table F-8 | 2019 Updated HHRA | | | | |
| Indoor Air / Inhalation | 2E-06 | 5E-08 | 3.9E+02 | NA | 2.7E+01 | Table F-8 | 2019 Updated HHRA | | | | |
| Surface soil / direct contact | NA | NA | 2.1E+00 | NA | 4.6E-01 | Table F-13G | 2019 Updated HHRA | | | | |
| Total | 4E-06 | 8E-08 | 7.7E+02 | NA | 5.4E+01 | | | | | | |
| Excavation Worker | | | | | | | | | | | |
| Surface soil / direct contact | NA | NA | 2.1E+00 | NA | 4.6E-01 | Table F-13G | 2019 Updated HHRA | | | | |
| Subsurface soil / direct contact-ingestion-inhalation | 2E-09 | 1E-10 | 1.4E+00 | 6.4E-01 | 1.0E-01 | Table F-13G | 2019 Updated HHRA | | | | |
| Groundwater / direct contact-inhalation | NA | NA | 3.9E+02 | NA | 2.7E+01 | Table F-8 | 2019 Updated HHRA | | | | |
| Total | 2E-09 | 1E-10 | 3.9E+02 | 6.4E-01 | 2.8E+01 | | | | | | |
| | | Constru | ction Worker | | | | | | | | |
| Surface soil / direct contact | NA | NA | 2.9E-01 | NA | 2.4E-02 | Table F-13G | 2019 Updated HHRA | | | | |
| Subsurface soil / direct contact-ingestion-inhalation | 6E-08 | 2E-08 | 1.4E+00 | 6.4E-01 | 5.8E-01 | Table F-13G | 2019 Updated HHRA | | | | |
| Total | 6E-08 | 2E-08 | 1.7E+00 | 6.4E-01 | 6.0E-01 | | | | | | |
| | Recre | ational Tresp | asser in Scapp | oose Bay | | - | | | | | |
| Sediment / direct contact-ingestion-inhalation | 2E-06 | 4E-07 | 3.7E-01 | NA | 5.4E-01 | Table F-11 | 2019 Updated HHRA | | | | |
| | Recr | eational Tres | passer in Milte | on Creek | | | | | | | |
| Sediment / direct contact-ingestion-inhalation | 9E-07 | 2E-07 | 9.7E-05 | NA | 2.5E-05 | Table F-11 | 2019 Updated HHRA | | | | |
| | | Sport Fisher | in Scappoose | Вау | | | | | | | |
| Fish tissue / consumption (general) | 2E-06 | 3E-06 | NA | NA | 7.9E-03 | Table F-11 | 2019 Updated HHRA | | | | |
| Fish tissue / consumption (subsistence) | 6E-05 | 3E-05 | NA | NA | 8.0E-02 | Table F-11 | 2019 Updated HHRA | | | | |
| | | Sport Fishe | r in Milton Cre | ek | | | | | | | |
| Fish tissue / consumption (general) | NA | NA | 2.3E-03 | NA | 1.1E-03 | Table F-11 | 2019 Updated HHRA | | | | |
| Fish tissue / consumption (subsistence) | NA | NA | 1.9E-02 | NA | 1.1E-02 | Table F-11 | 2019 Updated HHRA | | | | |
| DEQ Acceptable Risk Level | 1E-05 | 1E-05 | 1.0E+00 | 1.0E+00 | 1.0E+00 | | | | | | |

Notes:

NA explanations include:

- Surface soil / direct contact (industrial worker / excavation worker - Noncarcinogenic hazards were not evaluated in the 2006 HHRA.

- Groundwater / direct contact-inhalation (excavation worker) - Cancer risks and noncancer hazards from dermal exposures in water to semi-volatile compounds are not quantitatively evaluated; refer to the uncertainty discussion of dermal exposure for the excavation worker.

- Fish tissue / consumption (sport fisher) - The bioaccumulative PAHs are not carcinogenic.

RME - reasonable maximum exposure

CTE - central tendency exposure

HHRA - human health risk assessment

^A The cancer risk and hazard index for surface water include all samples from Scappoose Bay and Milton Creek; the data were not segregated into exposure units.

Shading indicated total acceptable risk level exceeded.

*Two RMEs were calculated for the construction worker and excavation worker scenarios: 1) with all data and 2) without the statistical outlier (Dixon's Test, p<0.05), per EPA guidance (EPA, 2000; EPA, 2013a)



Upland and Riverbank Surface Soil Technology Screening Port of Columbia County - Former Pope & Talbot Wood Treating Site St. Helens, Oregon

| General Response | Pomodial Toobnology | Description | | | Sevening Commonte | |
|---------------------------|---|---|--|--|---|---|
| Action | Remeular recimology | Description | Effectiveness | Implementability | Cost | Screening comments |
| NO ACTION | No Action | No Action | Not effective in achieving remedial action objectives (RAOs). | Easy to Implement | No capital or operation and maintenance (O&M) costs incurred. | Does not meet threshold criteria, but required to be retained for comparison purposes. |
| INSTITUTIONAL CONTROLS | Legal Restrictions, Regulations, and Covenants | May include restrictions such as: deed restrictions, easements, and covenants, attached to property-related documents; legal bans or controls of activities. | Can be effective at controlling human exposures, but less effective (or not effective) at controlling ecological exposures. Is not effective at controlling or reducing contamination migration. Most suitable for use in conjunction with other active technologies. | Likely to require acceptance and cooperation of multiple parties to implement. | Low | Potentially applicable to address human exposure in conjunction with other technologies and/or to address upland creosote contamination beneath existing fill cap. |
| | Contaminated Media Management Plan | Development and publication of protocols for handling and managing contaminated soil/riverbank during future work to protect workers, public health, ecological exposures, and the environment. | Effective for management of contaminated soils during future work. Effective at preventing human exposures. Not effective at preventing ecological exposures without other active technologies. Most suitable for use in conjunction with other active technologies. Is not effective at controlling or reducing contamination migration. | Easy to Implement | Low | Applicable in conjunction with other technologies and/or to protect excavation/construction workers from upland creosote contamination. |
| | Signage / Notifications / Advisories | Posting of signs and/or distribution of notifications regarding health concerns in area of contamination. | Can be effective at reducing human exposures via public education, but not effective at controlling ecological exposures. Is not effective at controlling or reducing contaminant migration. Most suitable for use in conjunction with other active remedial action technologies. | Easy to Implement | Low | Retain as potential technology to limit human health exposure. |
| | Monitoring | Development of sampling and analysis plan (SAP), quality assurance project plan (QAPP), and laboratory analysis of samples collected from soil. | Effective for documenting site conditions and exposure risks, evaluating migration and naturally occurring processes, and effectiveness of remediation actions. Does not address contaminant reduction or receptor exposures. | Currently implemented to the maximum extent feasible. | Low | Retain to monitor the effectiveness of selected remedial action alternative. |
| ENGINEERING CONTROLS | Physical Barriers (e.g., Fencing, Floating Booms) | The upland portions of the Site are currently fenced and access gates locked. Linked floating barriers (log booms) currently limited motorized boat access to a portion of the Site's riverbank (Cove Area). | Effective at controlling trespasser access to upland areas and riverbank in the Cove Area. Not effective in controlling trespassing along the entire length of the Site's shoreline. Not effective at reducing contaminant migration. Does not address or limit ecological exposure. | Upland is currently fenced. Relatively easy to extend log booms where existing offshore piling is present. May be difficult to obtain approval and install piling and barriers along the entire Site shoreline. | Low to Moderate | Existing chain-link fence will continue to mitigate upland trespassing and other barriers could be expanded to reduce boater and transient trespasser access to the Site's shorelines. These engineering controls have been retained and could be used in conjunction with other technologies to achieve RAOs. |
| | Stormwater Management | Regrading shoreline topography to promote infiltration outside inferred extent of NAPL and mitigate overland runoff towards adjacent surface water bodies and erosion of riverbank materials. In addition to regrading Site topography, could include the installation of stormwater collection, conveyance, and retention features. | Effective in mitigating overland flow and riverbank erosion. Except for localized areas of polychlorinated biphenyl's (PCBs), upland surface soil lack contaminants of concern (COC). Does not address contaminant reduction and subsurface mobility. | Easy to implement. Post-construction monitoring required. | Moderate | May be necessary to meet RAOs if used in conjunction with other technologies to prevent runoff from the site or riverbank erosion. |



| General Response | Bemedial Technology | Description | Description Screening Criteria | | | |
|-------------------------|-------------------------|---|---|---|--|---|
| Action | Kenieulai Technology | Description | Effectiveness | Implementability | Cost | Screening comments |
| CONTAINMENT | Physical Barrier / Cap | Involves covering contaminated surface soil and riverbank material with clean material to prevent exposure. Upland surface soil consists of fill (dredged sand and gravel) imported to the Site after the wood treatment operations ceased. Except for localized gravel roadbeds, upland surface soil is expected to be clean. Installation of an engineered cap over impacted riverbank sediments is applicable. Armoring and/or vegetation can be used as a method of preventing riverbank erosion. | Effective at preventing direct contact with contaminated riverbank. Does not address contaminant reduction, but engineered cap could be designed to reduce contaminant mass flux and/or erosion of contaminated riverbank material. Cap design must be compatible with permit requirement and expected future Site use. | Site is largely vacant/undeveloped and the installation of a riverbank cap is feasible using common earth materials and land-based construction equipment. Permitting will be required (grading, stormwater, etc.) and work must be completed within the in-water work window. Cap would need to be compatible with current beneficial uses (e.g., ecological habitat) and its integrity maintained in perpetuity. Would require regrading of the shoreline and balanced cut/fill in flood plain. Would generate waste spoils for off-site disposal or on-site treatment. | Moderate to high planning, permitting and implementation costs associated with riverbank cap. Low to moderate long- term maintenance and performance monitoring costs. | Upland NAPL area is already sufficiently capped. Physical barrier / capping of riverbank in NAPL and contaminated groundwater seep areas is retained as a potential technology alone or in conjunction with other technologies to achieve RAOs. |
| | Adsorptive Cap | Installation of an engineered cap containing amendments (e.g., adsorptive materials such as activated carbon, biochar, oleophilic biobarrier, or organoclay) to promote the immobilization and biodegradation of NAPL and dissolved-phase COCs along the riverbank. These adsorptive amendments are usually emplaced directly on (or mixed into) contaminated media, as components in engineered caps, or within engineered mats that are placed on the contaminated area. The use of amendments in a shoreline cap does not preclude the need for a physical barrier to prevent erosion and direct contact by human and ecological receptors. The use an adsorptive cap is not applicable nor considered for addressing upland surface soil containing PCBs. | If implemented properly, an engineered riverbank cap with adsorptive amendments can be highly effective in mitigating the discharge of creosote NAPL and dissolved-phase COCs at concentration above PRGs. An adsorptive riverbank cap does not address upland contaminant source areas, but rather is designed to sequester and enhance the biodegradation of creosote COCs in groundwater before it discharges into the aquatic environment. Pilot testing of often required to assess effectiveness. | Extensive permitting required and work must be completed within the in water work window below ordinary high water. Cap would need to be compatible with current beneficial uses (e.g., ecological habitat) and its integrity maintained in perpetuity. Would require regrading of the shoreline and balanced cut/fill in flood plain. Would generate waste spoils for off site disposal or on site treatment. | Moderate to high planning, permitting, and implementation costs associated with riverbank cap. Low to moderate long- term maintenance and performance monitoring costs. | Reactive capping technologies have been retained to address the shoreline NAPL seeps and contaminated groundwater discharge. |
| REMOVAL AND DISPOSAL | Excavation | Mechanical removal of contaminated surface soil and riverbank. Excavation of some or all of the soil and riverbank containing COCs (e.g., PCBs, creosote-related constituents) for subsequent treatment and/or disposal. Focused excavation may include only higher concentrations or "hot spot" soil. Site restoration could include backfill with imported soil and regrading disturbed areas to prevent runoff or erosion. | Effective in removing hot spot soil and riverbank materials. Addresses direct exposure pathways (where applicable). Does not address or control ongoing contaminant migration unless combined with containment (e.g., adsorptive cap) or subsurface removal. | Implementation involves conventional land-based construction equipment and methods. Depending on extent of excavation and surface water conditions, may have short-term adverse impact on current land use and existing riparian / aquatic habitat. | High | Applicable for removal of surface hot spot riverbank material. Upland excavation of all upland soil containing COCs (e.g., PCBs, creosote-related constituents) does not appear to be required or feasible, but should be retained for detailed evaluation in the Feasibility Study (FS). |
| | Off-Site Disposal | Soil impacted by historical wood treating operations are considered a listed hazardous waste (F032, F034, and F035). Therefore, off-site disposal would likely have be at Chemical Waste Management in Arlington, Oregon, as a Corrective Action Management Unit (CAMU)-eligible waste. | Landfills are controlled, managed facilities that are effective in preventing future exposures. Removes listed contamination and generated hazardous waste from the flood plain. | Implementation involves CAMU eligibility process, transportation of contaminated media for potentially long distances, and potential macro-encapsulation at the landfill. Off-site transportation requires the elimination of free liquids from saturated excavation spoils. | High | Not a stand alone technology. Applicable for handling of excavated upland and riverbank soils. |
| | On-site Upland Landfill | Construction of a permitted upland landfill facility at the Site for the disposal of the excavated soil and riverbank material. Would require suitable area and acceptance by local, state and federal permitting agencies. | Effective at removing source material from cleanup area and placement in a self-managed waste facility. Addresses direct exposure pathways and migration by removing contaminant mass from cleanup area. Would require placement below clean cap and ongoing maintenance/monitoring of landfilled material. | On-site landfilling of hot spot soil (listed hazardous waste) below a clean cap is not compatible with site conditions (shallow groundwater, within the flood plain) or current/future site use (industrial). | High | Not retained because contaminated soil and sediment is a listed hazardous waste, shallow water table, and the Site is situated within the 100-year flood plain. |

Table 3

Page 154 of 240

| General Response | Bemedial Technology | Description | | Screening Criteria | | Screening Comments | |
|--|--------------------------------|---|--|--|---|---|--|
| Action | Kemediai Technology | Description | Effectiveness | Implementability | Cost | | |
| IN-PLACE PHYSICAL/ CHEMICAL TREATMENT | Chemical Oxidation | Chemically converts hazardous contaminants to less toxic compounds. Effective in destroying organic contaminants and oxidizing inorganic contaminants to less toxic/less mobile forms. Can include oxidant chemicals such as peroxides, permanganates, or ozone. | Can be highly effective at destruction of organic contaminants or oxidation of inorganics. | Equipment and vendors are readily available. Generally not appropriate for surface applications. Would be destructive to existing beneficial organics in soil. | High | Not retained because technology generally not appropriate for surface soil and riverbank applications without high implementation risks. Lower cost options exist. | |
| | Soil Flushing | Circulation of water, steam, or an amended aqueous solution through the contaminated soil and riverbank to detach and collect particle-bound contaminants. The circulated water or steam is then recovered and treated. | Not appropriate or effective for NAPL recovery in the riverbank. | Difficult to maintain control of NAPL migration, amended water, and steam. Inefficient removal within fine-grained soil. | Moderate to High | Technology not retained as implementation risks are high and other more suitable technologies are available. | |
| | Solidification / Stabilization | Contaminants are physically bound or enclosed in a stabilized mass (solidification) or chemical reactions are induced between the stabilizing agent and contaminants to reduce the contaminant mobility (stabilization). Methods may include the addition of Portland cement, lime, kiln dust, pozzolan, sorbent clay (i.e., bentonite), and proprietary reagents. | Potentially suitable at reducing mobility of and accessibility to site contaminants. Difficult to ensure complete enclosure of upland soil and riverbank with <i>in-situ</i> implementation. Reduction of bioavailability of organic contaminants could be effective with use of carbon (or similar) addition to upland soil and riverbank. | Solidifying and/or stabilizing upland areas (e.g., paving) could be compatible for industrial land use. However, solidification of contaminated riverbank is not compatible with efforts to enhance riparian and shallow water habitat. Stabilization of the contaminated riverbank would likely require regrading and the use of adsorptive amendments. | High implementation cost, except that incorporation of additives into cap, which can be relatively inexpensive. | Stabilization (immobilization) of creosote-impacted riverbank material retained as potentially useful technology when combined with containment and/or capping. Other process options (e.g., vitrification, soil freezing) not retained because less suitable to Site conditions and high cost. | |
| IN-PLACE BIOLOGICAL TREATMENT | Enhanced Bioremediation | Add amendments (e.g., nutrients, electron acceptors) to stimulate the natural degradation or use engineered capping materials to facilitate/enhance the bioremediation of creosote constituents in riverbank material. | Not effective or considered for upland PCB-impacted soil. Creosote NAPL is expected to severely limit microorganism activity / performance. Can be difficult to achieve full coverage and have meaningful contaminant mass reduction within a reasonable amount of time. Only effective if coupled with containment technologies. | Would require regrading of the riverbank and must be able to withstand seasonal flooding and hydrodynamics of the river system. Would need to be coupled with either an upland containment wall or passive adsorptive barrier. | Moderate | Not retained as standalone technology in treating creosote NAPL without soil flushing (which was not retained) and high implementation risks. Could be effective on dissolved PAHs, with low implementation risks, if used in conjunction with NAPL containment technologies (e.g., barrier wall and/or reactive permeable barrier). | |
| | Phytoremediation | The process of using plants to remove, transfer, stabilize, and/or destroy contaminants in upland soil or riverbank. | Can be effective at removing a variety of shallow dissolved- phase organic compounds from soil / riverbank through plant uptake in the plant rhizosphere. Upland contamination is buried beneath 5 to 8 feet of clean fill. Not effective in removing creosote NAPL. | Would require planting of suitable plants for site conditions, or changing of site conditions to accommodate more trees along the Site's shoreline. May not be compatible with current and future industrial site use. | Low | Not as standalone technology for addressing NAPL. However, is retained for addressing dissolved-phase mass flux. | |
| NATURAL PROCESSES | Monitored Natural Recovery | Naturally occurring physical processes (advection, desorption, dispersion, diffusion, dilution, resuspension, and volatilization), and biological processes (biodegradation) reduce contaminant concentrations. Process is monitored to verify exposures. | Not effective in reducing NAPL mass/mobility or achieving RAOs within a reasonable amount of time within the priority action areas (PAAs). | Easy to implement. Must be combined with long-term monitoring of COC in sediment porewater and surface water. Must be used in conjunction with NAPL removal and/or containment. | Low | Natural recovery has been ongoing since the wood treating operations ceased in 1960. Within the PAAs, natural recovery alone (i.e., in the absence of other technologies) is not effective and will not achieve RAOs within an reasonable amount of time. | |

Table 3



Page 155 of 240

| General Response | Romodial Technology | Description | | Screening Criteria | | Screening Comments | |
|--|--------------------------------|--|--|---|-----------------|--|--|
| Action | Remedial Technology | Description | Effectiveness | Implementability | Cost | Screening comments | |
| <i>EX-SITU</i> PHYSICAL/ CHEMICAL TREATMENT | Separation | Use of physical means to separate coarse-grained material (which would have less contamination) from Native Soil for beneficial reuse or separation of debris prior to further treatment or disposal. | Generally not effective for upland soil and riverbank material t encountered at the Site. | Commercial equipment is available for separation (i.e., sieves). Separated sand may be available for potential beneficial use (would require verification testing and identification of potential use). Bench scale testing may be needed to define specific operating parameters. | Low to Moderate | Sediments previously identified to have high organic content. Not retained because the impacted material removed would primarily consist of fine-grained material. | |
| | Soil Washing | Contaminants are separated from the excavated soil with wash water augmented with additives to help remove creosote. | Most suitable for highly refined petroleum products (e.g., gasoline, diesel) in coarse-grained material. Less effective on viscous contamination like creosote in fine-grained material. | Elutriate would require treatment and disposal, which could significantly increase the overall cost of treatment. Bench-scale testing would be required during design. Requires staging area for treatment or transport to approved off-site facility. Air quality standards may be affected by open-air treatment methods. | High | Not retained because not compatible with creosote- impacted fine-grained soil and more cost-effective options are available. | |
| | Chemical Oxidation | Includes the application of chemical oxidants for the purpose of remediating excavated soils. Generally involves reduction / oxidation (redox) reactions that chemically convert hazardous contaminants to less toxic or less mobile forms. | Can be effective in reducing particle-bound PAHs in well mixed media slurry. May not be cost effective on organic-rich media with creosote NAPL due to the large amounts of oxidizing agent required. | Risks associated with handling of oxidant in above- ground application. Bench-scale testing would be required during design. Requires staging area for treatment or transport to off-site facility. Air quality standards for site workers may be affected by open-air treatment methods. | High | Not retained because technology has relatively high implementation risks to workers and less costly options are equally protective and available. | |
| | Chemical Extraction | Excavated upland surface and riverbank soil is mixed with an extractant (e.g., acid or solvent), which dissolves the contaminants. The solution is then placed in a separator to remove the contaminant/extractant mixture for treatment. | Most suitable to semi-volatile or inorganic contamination. Less effective in fine-grained soil/sediment. | Difficult to remove all contaminant/extractant from organic-rich sediment and would likely require finish treatment. Elutriate would require treatment and disposal, which could significantly increase the overall cost of treatment. Bench-scale testing would be required during design. Requires staging area for treatment or transport to off-site facility. Air quality standards may be affected by open-air treatment methods. | High | Not retained because not compatible with sediment grain size and more cost-effective options available. | |
| | Solidification / Stabilization | Contaminants are physically bound or enclosed in a stabilized mass (solidification) or chemical reactions are induced between the stabilizing agent and contaminants to reduce the contaminant mobility (stabilization). Methods may include the addition of Portland cement, lime, kiln dust, pozzolan, sorbent clay (i.e., bentonite), and proprietary reagents. PCB impacted gravel could potentially be used for asphalt aggregate for onsite paving. | Can be effective at reducing mobility of contaminants or solidifying for disposal. | Requires staging area for treatment or transport to off- site facility. Dewatering of riverbank soil may be necessary. Air quality standards for site/occupational workers may be affected by open-air dewatering and stabilizing mixing methods. | Moderate | Retained as a potential component off-site disposal or onsite treatment. | |

Table 3



Page 156 of 240

| General Response | Remedial Technology | Description | | | Screening Comments | |
|--|--|---|---|--|--------------------|--|
| Action | Kemediai Technology | | Effectiveness | Implementability | Cost | Screening comments |
| EX-SITU PHYSICAL/ CHEMICAL TREATMENT (CONTINUED) | Solar Detoxification | Contaminants are destroyed by photochemical and thermal reactions using ultraviolet energy in sunlight or artificial UV light. Usually involves application of catalyst agent. | Can be effective at treating a variety of organic compounds. Most effective when used with a catalyst agent (i.e., titanium dioxide). Does not address inorganic contaminants such as metals. | Implementation with sunlight limited by availability (not effective during nighttime and limited effectiveness during cloudy/wet seasons). Requires staging area for treatment or transport to off-site facility. Air quality standards for site/occupational workers may be affected by open-air treatment methods. | Moderate | Not retained because adequate space is not available at the site and limited usefulness for creosote. |
| <i>EX-SITU</i> BIOLOGICAL TREATMENT | Land Treatment/ Landfarming | Land treatment reduces contaminant concentrations through biological processes. Excavated soil is placed in lined and g bermed treatment cells and manipulated as necessary to improve biological conditions (such as tilling material to aerate and mix in nutrient amendments). | Can be effective at removing PAHs in soil. Most effective with control of moisture, heat, nutrients, oxygen, and pH to enhance biodegradation. Effectiveness is reduced by the presence of NAPL and low ambient temperatures during 8 months of the year. | Requires large upland area for soil treatment. Requires the dewatering and removal of debris from soil. Erosion and stormwater controls need to address ponding and/or contaminated runoff. Bench-scale testing would be required to define operating parameters, particularly during wet weather. Long anticipated implementation period with extensive performance and confirmation monitoring. Air quality standards may be affected by open-air treatment methods. | Moderate | Not retained because adequate space is not available at the site and limited usefulness for creosote. |
| | Biopiles | Soil is mixed with amendments, placed in aboveground enclosures, and aerated with blowers or vacuum pumps. Microorganisms present degrade the contaminants present. | Effective at removing volatile organic contamination from soil. Most effective with control of moisture, heat, nutrients, oxygen, and pH to enhance biodegradation. Effectiveness is reduced by the presence of NAPL. | Requires area for soil treatment or transport to an off- site facility. Requires initial dewatering of saturated soil and ongoing moisture control. Requires stormwater and leachate management and control. Bench-scale testing would be required to define operating parameters. Air quality standards may be affected by aeration treatment methods. | Moderate to High | Not retained because adequate space is not available at the site and limited usefulness for creosote. |
| | Slurry-phase Biological Treatment | A slurry of soil with water and other additives is mixed to keep solids suspended and microorganisms present in the slurry in contact with the soil contaminants. When complete, the slurry is dewatered and the treated soil/sediment is disposed on or offsite. | Can be effective at treating a variety of organic compounds and diluting NAPL into manageable concentrations. | Requires staging area for treatment cell or transport to an off-site facility. Slurry dewatering generates liquid waste stream that will require treatment or disposal. Bench-scale testing would be required to define operating parameters. Air quality standards may be affected by open-air treatment methods. | High | Not retained because other more cost-effective options available. |
| | Off-site Incineration | High temperatures are used to combust (in the presence of oxygen) organic constituents in hazardous wastes. | High temperatures result in generally complete decomposition of organic chemicals. Effective across wide range of soil characteristics. | Requires air pollution control device. Listed hazardous waste designation will likely rule out acceptance into most treatment facilities. Involves high energy consumption. | High | Unlikely that a permitted facility will accept this listed hazardous waste stream for treatment. If found, anticipate significant cost for transportation and treatment. Other less expensive technologies available. |
| TREATMENT | On-Site Thermal Desorption / Pyrolysis / Hot Gas Decontamination | Waste soils are heated to either volatilize (desorption and hot gas) or to anaerobically decompose (pyrolysis) organic contaminants. Off-gas is collected and treated. | Effective at removing organic materials from excavated sediment/soil (particularly volatile organics). Pyrolysis generally used for semi-volatiles. Efficiency of PAH thermal desorption will be approximately 90%. | Requires the on-site mobilization of thermal desorption units and permitting with local land use and air quality agencies. Requires dewatering of excavated media, off- gas treatment, and extensive confirmation sampling. | High | Not retained because other more cost-effective options available. |

Note:

Shading indicates technologies that have been eliminated from consideration.

Table 3



Page 157 of 240

Upland Subsurface NAPL, Soil, and Groundwater Technology Screening

Port of Columbia County - Former Pope & Talbot Wood Treating Site

St. Helens, Oregon

| General Response | | | | | | |
|---------------------------|--|--|---|---|---|--|
| Action | Remedial Technology | Description | Effectiveness | Implementability | Cost | Screening Comments |
| NO ACTION | No Action | No Action | Not effective in achieving remedial action objectives (RAOs) | Easy to Implement | No capital or operations and maintenance (O&M) costs incurred | Does not meet threshold criteria, but required to be retained for comparison purposes. |
| | Legal Restrictions, Regulations, and Covenants | May include restrictions such as: deed restrictions, easements, and covenants, attached to property- related documents; legal bans or controls of activities. | Can be effective at controlling human exposures, but less effective (or not effective) at controlling ecological exposures. Is not effective at controlling or reducing contamination migration. Most suitable for use in conjunction with other active technologies. | Likely to require acceptance and cooperation of multiple parties to implement. | Low | Potentially applicable to address human exposure in conjunction with other technologies and/or to address upland creosote contamination beneath existing fill cap. |
| INSTITUTIONAL CONTROLS | Groundwater Use Restrictions | Restrict use of groundwater within the locality of facility (LOF). | Effective at preventing direct contact, but use restrictions are not effective at controlling or reducing contaminant migration. Most suitable for use in conjunction with other active technologies. | As there is no planned future use of on-site groundwater, or off-site property owners to coordinate with, this can be easily implemented. | Low costs associated with implementing the use restrictions | Applicable for precluding the future use of groundwater for industrial, irrigation, or domestic use. Not applicable at addressing migration to Milton Creek or Scappoose Bay. |
| | Monitoring | Laboratory analysis of groundwater samples for assessing the effectiveness of other remedial technologies. | Effective for documenting site conditions and exposure risks, evaluating migration and naturally occurring processes, and effectiveness of remediation actions. Does not address contaminant reduction or receptor exposures. | Easy to Implement | Low | Applicable to confirm effectiveness of other technologies. |
| | Control of Building heating ventilation and air conditioning (HVAC) System | Use HVAC system to maintain positive pressure in buildings overlying creosote nonaqueous phase liquid (NAPL). | Effective in mitigating potential risk of vapor intrusion in buildings overlying highly concentrated hot spots. Does not address contaminant reduction. Generally used in conjunction with other engineering controls. | Easy to implement. Post-construction monitoring required. | Low | May be necessary to meet remedial action objectives (RAOs) if used in conjunction with NAPL containment technologies. |
| | Vapor Barriers | Installation of low-permeable barriers beneath buildings overlying groundwater containing creosote NAPL to prevent vapor intrusion. | Effective in mitigating potential risk of vapor intrusion in buildings overlying highly concentrated hot spots. Does not address contaminant reduction. | Difficult to implement on existing structures. Easy to implement during future construction. Post-construction monitoring required. | Moderate | May be necessary to meet RAOs if used in conjunction with NAPL containment technologies. |
| CONTROLS | Alternative Water Supply | Site is currently serviced by municipal water supply. An easement and equitable servitudes (EES) precluding the use of groundwater and surface water within the LOF will likely be a part of the Site remedy. | Effective in preventing the use of contaminated groundwater. No current or likely future use of contaminated groundwater. Does not address NAPL migration to surface water. | Easy to expand municipal water service. Use restrictions requires local and State of Oregon Water Resources Department (OWRD) approvals. | High capital costs, low to moderate O&M costs. | Not retained as the Site is currently serviced by municipal supply and the beneficial use of shallow groundwater limited to discharge to surface water. |
| | Wellhead Construction or Treatment | Site-specific construction techniques to avoid pumping from contaminated zones or treatment at individual impacted water supply wells with the use of <i>Ex-Situ</i> Physical/Chemical/Thermal treatment. | Effective in reducing contaminant concentrations in groundwater prior to use either through use of a well design that does not pump impacted groundwater or wellhead treatment. | Easy to implement. Treatment units are readily available. Requires ongoing testing and system maintenance to remain effective. | Low to high capital costs and O&M costs, depending on treatment technology. | Not retained as the Site is currently serviced by municipal supply and the beneficial use of groundwater is limited to discharge to surface water. |



Page 158 of 240

| General Response | | | | Screening Criteria | | |
|------------------|---|--|--|---|---|---|
| Action | Remedial Technology | Description | Effectiveness | Implementability | Cost | Screening Comments |
| CONTAINMENT | Physical Barrier - Cutoff Wall | Installation of vertical barriers (e.g., sheet piling, soil- bentonite slurry wall, grout, etc.) to prevent migration of groundwater containing creosote NAPL. | Effective at preventing lateral migration. Requires keying into underlying Basalt Bedrock. Surface cap and/or hydraulic control often necessary as supplemental measures to achieve containment. | Moderately easy to implement using conventional construction equipment. Requires pre-construction bench-scale testing and post-construction monitoring. May be difficult to achieve seal in Basalt Bedrock. | High implementation costs. Low to moderate O&M and long-term monitoring costs. | Potentially applicable technology for Site conditions. |
| | Riverbank - Engineered Cap | Installation of an engineered cap over impacted riverbank soil is applicable. Riverbank caps may include placement of clean soil, gravel, armoring, and/or vegetation as a method of preventing riverbank erosion. | Effective at preventing direct contact with contaminated riverbank soil. Does not address contaminant reduction, but engineered cap could be designed to reduce contaminant mass flux and/or erosion of contaminated riverbank material. Cap design can also be compatible with expected future site use. | Site is largely vacant/undeveloped and the installation of a riverbank is cap feasible using standard construction equipment. Extensive permitting required and work must be completed within the in-water work window below ordinary high water. Cap would need to be compatible with current beneficial uses (e.g., ecological habitat) and its integrity maintained in perpetuity. Would require regrading of the shoreline and balanced cut/fill in flood plain. Generate waste spoils for off-site disposal or on-site treatment. | Moderate to high planning, permitting and implementation costs associated with riverbank cap. Low to moderate long-term maintenance and performance monitoring costs. | Upland NAPL area is already sufficiently capped. Physical barrier / capping of riverbank in NAPL seep areas is retained as a potential technology alone or in conjunction with other technologies to achieve RAOs. |
| | Riverbank Permeable Reactive Barrier | Installation permeable reactive cap to prevent migration of NAPL from upland areas to Scappoose Bay. A number of reactive materials are available for consideration including activated carbon, biochar, organoclay, and oleophilic biobarriers. | Effective at prevent lateral migration of erodible soil along the riverbank or NAPL/groundwater at seeps. A permeable reactive cap is designed to reduce contaminant mass flux and/or erosion of contaminated riverbank material. | Moderately easy to implement using conventional construction equipment. Requires pre-construction bench-scale testing and post-construction monitoring. Extensive permitting required and work must be completed within the in-water work window. Cap would need to be compatible with current beneficial uses (e.g., ecological habitat) and its integrity maintained in perpetuity. Would require regrading of the shoreline and balanced cut/fill in flood plain. Would generate waste spoils for off-site disposal or on-site treatment. | High implementation costs, low to moderate O&M and long-term monitoring costs. | Potentially applicable technology for Site conditions. |
| | Pumping / Hydraulic Containment | Extraction well(s) with submersible pumps to lower the water table and create hydraulic gradients that direct contaminant migration into the extraction wells situated along the Site's shoreline. Extracted groundwater would require treatment before discharge (see <i>Ex-Situ</i> Physical/Chemical/ Thermal Treatment). | Effective in porous soils for preventing further contaminant migration. May also be used in conjunction with other technologies. Not efficient for complete removal of contaminant mass. | Existing monitoring wells could be utilized, although new and/or larger wells will likely be needed to capture full length of area of concern. Extraction of large volumes of water may be required to contain the discharge of groundwater to the river. Discharge of treated water would need to be permitted. | Moderate to high capital costs due to anticipated high pumping rates and anticipated treatment required prior to disposal. New extraction wells would likely be required. Moderate to high O&M costs. | Potentially applicable technology for Site conditions. |
| | Pump and Treat Using Vertical Extraction Wells | Extraction wells with submersible pumps and/or belt skimmers within the upland NAPL areas to reduce the volume and mobility of groundwater containing NAPL. Oil-water separation and treatment of extracted groundwater would be required before discharge (see <i>Ex-Situ</i> Physical / Chemical / Thermal Treatment). | Effective in porous soils for reducing the mobility of NAPL. Less effective on NAPL in fine-grained soil. Generally not effective in significantly reducing contaminant mass unless combined with other technologies. | Existing monitoring wells could be utilized, although new and/or larger wells likely needed to capture full length of area of concern. Extraction of large quantities of water may be required to contain discharge near shoreline with limited NAPL recovery / mass reduction. Discharge of treated water to surface water would likely be difficult to permit and routinely meet permit conditions. | Moderate to high capital costs due to installation of extraction wells and treatment system, disposal. New extraction wells likely required. Moderate to high O&M costs. | Not retained as a standalone technology. Potentially applicable if used sparingly in conjunction with containment technologies. |



Page 159 of 240

| General Response | | | | Screening Criteria | - | |
|--|---|---|---|--|---|---|
| Action | Remedial Technology | Description | Effectiveness | Implementability | Cost | Screening Comments |
| REMOVAL AND DISCHARGE (NAPL AND GROUNDWATER) | Pump and Treat Using Horizontal Cut-off Trench(es) | Trench filled with porous media along the top of riverbank immediately upgradient of NAPL seeps. Trench contains gravity drains to sump/pump. Oil- water separation and treatment of extracted groundwater would be required before discharge (see <i>Ex-Situ</i> Physical / Chemical / Thermal Treatment). | Effective in reducing NAPL migration to riverbank and surface water. May also be used in conjunction with other technologies. | Would likely require trenching down to Basalt Bedrock using long-reach trackhoe and management of contaminated soils. Treatment, management, and discharge of recovered water and creosote would likely require extensive long-term oversight. Discharge of treated water to surface water would likely be difficult to permit and routinely meet permit conditions. | Moderate to High | Potentially applicable technology for Site conditions. |
| | Discharge to Surface Water | Discharge of recovered and treated water into surface water. Municipal sanitary sewer supply is not available. | Effective for disposal of extracted groundwater. Oil-water separation and treatment of water would be necessary prior to disposal. | A discharge permit would be required and would likely be difficult to obtain. Extensive treatment of water would be needed to meet permit criteria and offshore RAOs. Noncompliant discharge has the potential to confound or interfere with in-water RAOs (no sheen and Σ SWTU < 1) | Moderate to high disposal costs depending upon treatment required, permit fees, and monthly usage fees. | Not applicable as there is no suitable location for surface discharge and high implementation risks. |
| | Discharge to Reinjection Wells | Upland discharge of extracted and treated water into the granular Fill Zone outside the inferred NAPL areas. | Moderate effectiveness, depending upon whether injection wells can be adequately located to prevent increasing groundwater gradient. | Underground injection control permit required for reinjection. | Moderate to high capital and O&M costs for reinjection wells. | Potentially applicable discharge option if used in conjunction with limited groundwater extraction. |
| | Reuse | Reuse of treated water for non-potable use such as industrial and irrigation. | Effective for treated, extracted groundwater. | A suitable on-site use would need to be identified that could accommodate a steady flow rate in all seasons. Treatment would be required prior to reuse. | Moderate to high costs depending upon storage, treatment, pumping and conveyance requirements. | No identified potential use suitable for flow rate expected from extraction system. |
| REMOVAL AND DISPOSAL (SOIL) | Excavation | Mechanical removal of contaminated soil and riverbank. Excavation of some or all of the soil and riverbank containing creosote NAPL for subsequent treatment and/or disposal. Focused excavation may include only higher concentrations or "hot spot" soil. Site restoration could include backfill with imported soil and regrading disturbed areas to prevent runoff or erosion. | Effective in removing hot spot soil and riverbank. Addresses direct exposure pathways (where applicable) and migration by reducing or controlling NAPL migration. | Implementation involves conventional construction equipment and methods. Depending on extent of excavation, may have short-term adverse impact on current land use and existing riparian / aquatic habitat. | High | Applicable for removal of hot spot riverbank material. Upland excavation of all upland soil containing creosote NAPL does not appear to be required or feasible, but should be retained for detailed evaluation in the FS. |
| | Off-Site Disposal | Soil impacted by wood treating operations is considered a listed hazardous waste. Therefore, off- site disposal would likely be at Chemical Waste Management in Arlington, Oregon as a Corrective Action Management Unit (CAMU)-eligible waste. | Landfills are controlled, managed facilities that are effective in preventing future exposures. Removes listed contamination and generated hazardous waste from the flood plain. | Implementation involves CAMU eligibility process and transportation of contaminated media on public roads for potentially long distances. Transportation by truck requires elimination of free liquids from saturated excavation spoils. | High | Not a stand alone technology. Applicable for handling of excavated sediment and riverbank. |
| REMOVAL AND DISPOSAL (SOIL) | Onsite Upland Landfill | Construction of a permitted upland landfill facility at the Site for the disposal of the excavated soil. Would require suitable area and acceptance by permitting agencies. | Effective at removing source material from cleanup area and placement in a self-managed waste facility. Addresses direct exposure pathways and migration by removing contaminant mass from cleanup area. Would require placement below clean cap and ongoing maintenance/monitoring of landfilled material. | On-site landfilling of hot spot soil (listed hazardous waste) below a clean cap is not compatible with current and future site use (industrial) and permitting is unlikely within the 100-year flood plain. | High | Not retained because contaminated soil and sediment is a listed hazardous waste and the Site is situated within the 100-year flood plain. |
| File No. 25331-001-01 | Chemical Oxidation | Includes the application of chemical oxidants such as peroxides, permanganates, or ozone for the purpose of remediating contaminated groundwater. Generally involves reduction/oxidation (redox) reactions that chemically convert hazardous contaminants to less toxic or less mobile forms. | Can be effective at destruction of organic contaminants in homogeneous porous soil. Significantly less effective and difficult to achieve uniform treatment in heterogenetic fine- grained soil matrix. Would need to be coupled with groundwater recirculation, which could mobilize residual NAPL. Generally not effective when soil matrix contains significant amounts of wood debris (e.g., buried piling in former operational surface). | Would be difficult to get uniform contact of oxidant with NAPL in thin sand layers and would require multiple injections and/or mixing points. Could be implemented as slurry but would require significant containment effort (such as installation of sheet piling). Care would be needed to prevent secondary impacts (such as from mobilized NAPL or metals) during oxidation. | Moderate to High | Not retained because it would be difficult to ensure adequate coverage and has high implementation risks. |
| Appendix B September 3 | 22, 2022 | | 3 of 6 | | | GEUENGINEEKS |

Page 160 of 240

| General Response | | | | | | |
|---|---|--|--|--|--|---|
| Action | Remedial Technology | Description | Effectiveness | Implementability | Cost | Screening Comments |
| | Air Sparging / Soil Vapor Extraction (SVE) | Injecting air below the water table to treat organics in groundwater, combined with soil vapor extraction in vadose soil. Increasing the contact between water and air to enhance volatilization. | Effective for volatile contamination. Not effective on semi- volatile polycyclic aromatic hydrocarbons (PAHs) and residual NAPL. Would require shallow vapor extraction to prevent uncontrolled vapor migration. Sparging will turn plume aerobic which will interfere with monitored natural attenuation (MNA) processes already in effect. | Radius of influence in fine-grained matrix would require multiple injection and recovery points. Equipment and technology for air sparging and SVE is readily available. Sparging would cause groundwater mounding that could increase NAPL mobility. | Moderate | Not retained because Site conditions would severely limit effectiveness. High implementation risks. |
| IN-SITU PHYSICAL / CHEMICAL / THERMAL TREATMENT | Steam Flushing/Steam Stripping | Steam is injected into the contaminated aquifer to vaporize less volatile organics. | May increase temperature of water discharging into river. Used in conjunction with vapor recovery. May be effective for increasing usability of SVE for low-volatility compounds. | Equipment and technology are readily available. Treatment of recovered vapors would likely be required. Could mobilize residual NAPL and would likely require significant containment effort. | High | Not retained due to high implementation risks. |
| | Passive / Reactive Riverbank Cap | Barriers placed across groundwater movement that allows passage of water while facilitating degradation or removal of contaminants. Reactive materials can be incorporated in riverbank cap to treat groundwater prior to emergence into river (reactive cap). | Can be effective in controlling NAPL migration and doesn't require groundwater extraction and treatment. | Could be implemented using conventional excavation, shoring, and backfill equipment. Extensive long-term monitoring required. Relatively easy to add reactive materials to riverbank caps. | High costs for installation. Low to moderate costs for performance and compliance monitoring, and periodic maintenance. Relatively low incremental cost to add to riverbank cap. | Potentially applicable technology for Site conditions if used in conjunction with containment technologies. |
| | Smoldering | Self-sustaining treatment for active remediation to thermally oxidize NAPL contaminants in the subsurface. After initial ignition, the process is maintained by using adjacent contaminants as the fuel source for continued combustion. | Can be implemented below the water table, although operation below the water table may impact the efficiency of the heating units. Control of the combustion is maintained by adjusting the supply of oxygen. Will require the use of SVE system to capture vapors from smoldering process. Radius of influence (ROI) on the order of 7 feet has been documented in field studies. May result in an increase in the temperature discharging to the River. | Would require remediation equipment and infrastructure similar to air sparge/soil vapor extraction to control air flow and vapor recovery. Limited full-scale examples. | Limited full-scale examples available for estimating purposes. Expect the implementation costs to be high due to drilling and equipment costs. | Potentially applicable technology for further consideration; however it is a relatively new technology and may have significant risk related to effectiveness and Implementability compared to other technologies. |
| IN-SITU PHYSICAL / | Solidification / Stabilization | Contaminants are physically bound or enclosed within a stabilized mass (solidification and vitrification), or chemical reactions are induced between the stabilizing agent and contaminants to reduce their mobility (stabilization), or additives are used to reduce mobility or bioavailability of contaminants (immobilization). Could be directly applied/mixed with soil or applied as part of an active capping approach. | Potentially suitable at reducing mobility of and accessibility to site contaminants. Difficult to ensure complete enclosure of upland soil and riverbank with <i>in-situ</i> process. Reduction of bioavailability of organic contaminants could be effective with use of (for example) carbon addition to soil and riverbank. | Difficult to obtain full stabilization in-situ in heterogeneous subsurface. Technologies such as vitrification to solidify mass would require significant power supply. Incorporation of additives into cap in order to stabilize mass is relatively simple. | High implementation cost, except that incorporation of additives into cap material relatively inexpensive. | Potentially applicable technology for Site conditions. |
| CHEMICAL / THERMAL TREATMENT (CONTINUED) | EcoSPEARS® | EcoSPEARS® are an emerging proprietary remediation product designed primarily for removal of PCBs (although they may be suitable for other persistent pollutants as well) from sediments via sorption onto the polymer material that is pushed into the surface sediment. | EcoSPEARS® are not suitable for upland or subsurface media (groundwater). | It is not expected that the EcoSPEARS® could be installed in the upland or bank areas. | Limited full-scale examples available for estimating purposes. Expect implementation costs to be moderate to High due to the expected treatment area. | Not retained because the technology is not suitable for upland or subsurface implementation. |
| | Multi-Phase Extraction (MPE) | MPE provides simultaneous extraction of soil vapor, contaminated groundwater, and separate phase liquid using single vacuum pump, multiple in-well pumps, or bioslurping. | Generally effective for NAPL removal in low to moderate soil permeability. May require vapor effluent treatment. | Equipment and technology for MPE are readily available. Treatment of recovered soil vapors and groundwater would be required prior to discharge. | Moderate to high capital and O&M costs. Higher costs if vapor treatment needed. | Not retained because effective less costly passive containment technologies exist. |

Page 161 of 240

| General Response | | | | | | |
|--|---|--|--|--|--|--|
| Action | Remedial Technology | Description | Effectiveness | Implementability | Cost | Screening Comments |
| | Enhanced Bioremediation | Addition of nutrients, electron acceptors, or other amendments to groundwater to enhance bioremediation. | Not likely to be effective in reducing NAPL mass and mobility without groundwater recirculation. Could be effective in cut off trench downgradient of primary NAPL zones. | Injecting amendments below the water table could mobilize NAPL. Uniform treatment performance would require extensive monitoring and oversight. | Moderate to high implementation and O&M costs. Hight cost if NAPL containment is needed. | Not retained as standalone technology. Potentially applicable if used in conjunction with containment or NAPL cut off technology. |
| TREATMENT | Phytoremediation | The process of using plants to remove, transfer, stabilize, and/or destroy contaminants in groundwater. | Can be effective at removing a variety of organic compounds in shallow groundwater through plant uptake in the plant rhizosphere. Not effective in limiting NAPL mobility, particularly in deeper sand layers. | Would require planting of suitable plants for Site conditions or changing of Site conditions to accommodate plants (such as regrading the riverbanks). | Moderate | Not retained because incompatible with Site conditions. Not effective as depth to NAPL is expected to be deeper than plant rhizosphere. |
| IN-SITU NATURAL PROCESSES | Monitored Natural Recovery | Naturally occurring physical processes (advection, desorption, dispersion, diffusion, dilution, resuspension, and volatilization), and biological processes (biodegradation) reduce contaminant concentrations. Process is monitored to verify exposures. | Not effective in reducing NAPL mass and mobility within a reasonable time frame. Anerobic degradation of PAHs is slow and NAPL will continue to be long-term source of toxicity to aquatic receptors if not combined with containment technology. | Easy to implement. Requires long-term monitoring of NAPL and COCs in groundwater. | Low | Natural recovery alone is not effective in achieving RAOs. Must be used in conjunction with NAPL containment. |
| EX-SITU PHYSICAL / CHEMICAL / THERMAL TREATMENT (SOIL) | Solidification / Stabilization | Contaminants are physically bound or enclosed in a stabilized mass (solidification) or chemical reactions are induced between the stabilizing agent and contaminants to reduce the contaminant mobility (stabilization). Methods may include the addition of Portland cement, lime, kiln dust, pozzolan, sorbent clay (i.e., bentonite), and proprietary reagents. | Potentially suitable at reducing mobility of and accessibility to contaminants in excavated material prior to disposal. | Easy to implement. Amendments would be mixed into excavated material using common construction equipment. | High to very high implementation cost. | Solidification and stabilization are potentially applicable if used in conjunction with excavation/ removal and either on-site or off-site disposal. |
| | Dewatering | Management and treatment of water as part of a saturated soil removal action. Methods may include passive dewatering in lined/bermed stockpiles, on barges, in geotextile tubes (filters), filter presses or other mechanical dewatering methods, or dewatering by adding chemical reagents or adsorptive materials. | Various methods can be effectively used (selected based on site conditions and degree of dewatering needed) to remove water from removed saturated media. Debris may need to be removed from soil prior to dewatering. Resultant water may need to be treated prior to disposal. | Fine-grained nature of soil may require long drying times and significant operational effort to meet landfill free liquid standards. Water removed from the contaminated media would require either treatment, evaporation, or absorptive onto a solid material (e.g., perlite). | Low to Moderate | Not a stand-alone technology. Retained as potentially applicable for use in conjunction with other technologies such as preparation of excavated saturated media for off site disposal. |
| | Air Stripping | Air is pumped through a water column of extracted groundwater designed to increase exposed surface area (such as a packed column, shallow tray), allowing transfer of contaminant mass from the aqueous phase to the vapor phase. | Very effective at removing many volatile organic compounds from extracted water stream. It may be necessary to treat treatment of vapor effluent with carbon or other technology. | Applicable for treatment of site contaminants in extracted water. Treatment equipment is readily available. Requires air emission testing and modeling to determine if off gas treatment is required. Disposal of water would be required. | Moderate to High | Not practical due to anticipated high pumping rates required to achieve cleanup in a reasonable timeframe. |
| | Sprinkler Irrigation | Extracted groundwater is pumped through a sprinkler irrigation system to increase the surface area of the water that is exposed to air, allowing transfer from the aqueous phase to the vapor phase. Typically done over a porous media. | Effective at removing many organic contaminants from the extracted water stream. Simpler system than more aggressive treatment technologies (such as air stripping). | Applicable for treatment of site contaminants in extracted water, but requires significant treatment system area. | Low to Moderate | Not retained since land use not compatible with Site conditions. |
| EX-SITU PHYSICAL / CHEMICAL / THERMAL TREATMENT (GROUNDWATER) | Precipitation / Coagulation / Flocculation | Use of chemical additives to aqueous solution containing dissolved contaminants to transform the dissolved contaminants into an insoluble solid, allowing removal of the contaminant from the liquid phase by sedimentation or filtration. The use of coagulant compounds can increase particle size and aid sedimentation. | Generally only effective for treating inorganic contaminants. | Treatment equipment is readily available. Disposal of water would be required. | Moderate to High | Not compatible with Site contaminants. |

Page 162 of 240

| General Response | | | | Screening Criteria | | |
|------------------|----------------------|--|--|---|------------------|--|
| Action | Remedial Technology | Description | Effectiveness | Implementability | Cost | Screening Comments |
| | Constructed wetlands | Utilizes natural geochemical and biological processes inherent in an artificial wetland ecosystem to remove contaminants from extracted groundwater. | Effective at removing organic and inorganic contaminants from the extracted groundwater. | Requires large land area to implement. May introduce attractive nuisance hazard for local wildlife. Low expected O&M costs. | Moderate to High | The site is within a flood plain area. Not retained because the required space may not be suitable with Site usage. |
| | Trickling Filter | Contaminants in extracted groundwater are put into contact with microorganisms in attached or suspended growth biological reactors. | Effective at removing organic contaminants from the extracted groundwater. Efficiency may be impacted by temperature and variations. May not be efficient enough to reach treatment goals | Difficult to maintain effectiveness with variable groundwater concentrations. Reactors would require significant area. Routine maintenance may be required. | Moderate to High | Not retained because the required space may not be suitable with Site usage. |
| | Adsorption | Collecting/concentrating constituents on surface of a sorbent material such as activated carbon to remove the constituent from the bulk liquid. | Effective at removing organic compounds from extracted groundwater. | Disposal of water would be required. Treatment equipment is readily available. O&M costs would be expected to be high. | Moderate to High | Retained for use in support of treating groundwater removed as part of excavation dewatering or hydraulic containment. |
| | lon Exchange | lon exchange removes ions from the liquid phase by exchange with a counter ion on the exchange media. | Effective at removing inorganic contaminants from groundwater. | Easy to implement and treatment technology is readily available. O&M costs would be expected to be high. | Moderate to High | No compatible with site contaminants. |

Note:

Shading indicates technologies that have been eliminated from consideration.

Table 4



Page 163 of 240

In-Water Sediment and Porewater Technology Screening Port of Columbia County - Former Pope & Talbot Wood Treating Site St. Helens, Oregon

| General Response | Remedial Technology | Description | | Screening Criteria | Screening Comments | |
|---------------------------|--|--|---|--|--|--|
| Action | Remeular recimology | Description | Effectiveness | Implementability | Cost | Screening comments |
| NO ACTION | No Action | No Action | Not effective in achieving remedial action objectives (RAOs). | Easy to Implement | No capital or operation and maintenance (O&M) costs incurred. | Does not meet threshold criteria, but required to be retained for comparison purposes. |
| | Legal Restrictions, Regulations, and Covenants | May include restrictions such as: deed restrictions, easements, and covenants, attached to property-related documents; legal bans or controls of activities. | Can be effective at controlling human exposures, but less effective (or not effective) at controlling ecological exposures. Is not effective at controlling or reducing migration. Most suitable for use in conjunction with other active technologies. | Likely to require acceptance and cooperation of multiple parties to implement. | Low | Potentially applicable to address human exposure in conjunction with other technologies and/or to address residual contamination. |
| | Sediment / Soil Management Plan | Development and publication of protocols for handling and managing contaminated sediments/soil during future work to protect workers, public health, ecological exposures, and the environment. | Effective for management of contaminated sediments / soils. Not effective at preventing human or ecological exposures without other active technologies. | Easy to implement | Low | Potentially applicable in conjunction with other technologies and/or to address residual contamination. |
| INSTITUTIONAL CONTROLS | Signage / Notifications / Advisories | Posting of signs and/or distribution of notifications regarding health concerns, trespassing, or no-wake zones in areas of contamination. Can be effective at reducing human exposures via pub education, but not effective at controlling ecological ex- A no-wake zone may be effective at reducing migration erodible riverbank soils, but will not control migration. suitable for use in conjunction with other active technol | Can be effective at reducing human exposures via public education, but not effective at controlling ecological exposures. A no-wake zone may be effective at reducing migration of erodible riverbank soils, but will not control migration. Most suitable for use in conjunction with other active technologies. | Easy to implement | Low | Potentially applicable to address human exposure in conjunction with other technologies. |
| | MonitoringLaboratory analysis of samples collected from sediment, porewater and surface water.Effective for documenting site conditions and exposure risks, evaluating migration and naturally occurring processes, and effectiveness of remediation actions. Does not address contaminant reduction or receptor exposures. | | Easy to implement | Low | Potentially applicable to confirm effectiveness of other technologies. | |
| ENGINEERING CONTROLS | Physical Barriers (e.g., Floating Booms) Linked floating barriers (log booms) currently limited motorized boat access to a portion of the Site's riverbank (Cove Area). Effective at controlling trespasser access to riverbank in the Cove Area. Not effective in controlling trespassing along the entire length of the Site's shoreline. Not effective at reduce contaminant migration. Does not address or limit ecologic exposure. | | Effective at controlling trespasser access to riverbank in the Cove Area. Not effective in controlling trespassing along the entire length of the Site's shoreline. Not effective at reducing contaminant migration. Does not address or limit ecological exposure. | Relatively easy to extend log booms where existing offshore piling is present. May be difficult to obtain approval and install piling and barriers along the entire Site shoreline. | Low to Moderate | Existing barriers could be expanded to reduce boater and transient trespasser access to the Site's shorelines. These engineering controls have been retained and could be used in conjunction with other technologies to achieve RAOs. |
| | Physical Barriers (Log Boom) | Log boom placed off the shoreline of Areas 1 and 2. Maintenance of the log boom in place to prevent access to the Cove Area. | Effective at preventing beach access to recreational boat users and can be effective at dissipating wakes from passing boats. Does not address or limit ecological exposure. | Likely to require acceptance and permitting from multiple parties to implement. Relatively simple technology to implement. | Moderate | Potentially applicable to address human exposure in conjunction with other technologies and/or to address residual contamination / recontamination of in-water sediment. |



| General Response | Domodial Toohnology | Description | Screening Criteria | | Description Screening Criteria | | | Screening Comments | |
|-----------------------|-------------------------------------|--|--|---|--------------------------------|---|--|--------------------|--|
| Action | Remedial recimology | Description | Effectiveness | Implementability | Cost | Screening Comments | | | |
| | Сар | Installation of an engineered cap over impacted sediment. Armoring of riverbank below the ordinary high water mark (OHWM) and sediment within the intertidal zone may be necessary along the Area 1 and 2 shorelines. | Should be very effective for Site contaminants that have low- solubility and high sorption (i.e., chemicals likely to remain bound to sediment). | Generally uses proven technologies. Partial dredging followed by capping could be implemented. | Moderate to High | Retained as applicable technology. | | | |
| CONTAINMENT | Adsorptive Cap | Installation of an engineered cap containing adsorptive additives (e.g., activated carbon or organoclay) over impacted sediment. Adsorptive layers would be placed in the lower portion of the cap to treat groundwater passing through the cap. The upper portion would consist of habitat material or armoring, as appropriate. | May be effective in reducing recontamination due to nonaqueous phase liquid (NAPL) migration and seeps. It is not clear at this time if an adsorptive amended cap would be more effective than an passive isolation cap (e.g., traditional sand capping). | It is difficult to estimate the useful life of amendments used to sequester and degrade contaminants without pilot testing. | Moderate to High | Retained as applicable technology. | | | |
| | Dredging/Excavation | Mechanical removal of contaminated sediment. Could include: mechanical dredging, hydraulic dredging, or land- based excavation of exposed near-shore material during seasonal low water and low tide. | Effective in removing impacted sediment and debris. | Dredging/excavation equipment is readily available. | High | Applicable for removal of impacted sediment/debris. | | | |
| | Off-Site Disposal | Sediment impacted by wood treating operations are considered a listed hazardous waste. Therefore, off-site disposal would likely be at Chemical Waste Management in Arlington, Oregon, as a Corrective Action Management Unit (CAMU)-eligible waste. | Landfills are controlled, managed facilities that are effective in preventing future exposures. Addresses direct exposure pathways and migration by removing contaminant mass from Site. | Implementation involves transportation of contaminated sediment on barge or trucks. Transportation by truck requires elimination of free liquids from sediment. | High | Not a stand alone technology. Potentially applicable for handling of dredged/excavated sediment. | | | |
| REMOVAL AND DISPOSAL | Onsite Upland Landfill | Construction of a permitted upland landfill facility at the Site for the disposal of the dredged/excavated sediments. Would require suitable area and acceptance by permitting agencies. | Effective at removing source material from cleanup area and placement in a self-managed waste facility. Addresses direct exposure pathways and migration by removing contaminant mass from cleanup area. Would require ongoing maintenance of landfill. | On-site landfilling of hot spot sediment (listed hazardous waste) below a clean cap is not compatible with current and future Site use (industrial) and permitting is unlikely within the 100-year flood plain. | High | On-site landfilling of the hot spot sediment (listed hazardous waste) below a clean cap is not compatible with the current and future Site use (industrial) and permitting is unlikely within the 100-year floodplain. | | | |
| | Upland Re-Use | Re-use of dredged/excavated sediment as fill at an upland site. Would require material characterization. Sediment to be re-used would need to be below applicable risk-based screening levels and/or appropriate exposure protection measures would required at the re- use site. | Effective at removing source material from cleanup area and placement at an upland site for re-use. Not effective if material exceeds applicable screening levels. May require additional measures (e.g., capping) to reduce exposure. | Easy to difficult depending on location of re-use site and any additional measures that might be required. Typically would require beneficial re-use approval from DEQ. Suitable re-use site needed. | Moderate | Not retained as no beneficial re-use site has been identified. | | | |
| REMOVAL AND DISPOSAL | Confined Aquatic Disposal (CAD) | Disposal area is excavated in open water or utilizes existing low spots in the water body The disposal cell is then filled with the dredged / excavated sediment and covered with clean material (i.e., capped). | Effective at removing source material from cleanup area and placement in a self-managed waste facility. Addresses direct exposure pathways and migration by removing contaminant mass from cleanup area. Placement and design of CAD facility must account for potential soluble contamination migration. Would require ongoing maintenance of disposal facility. | Potential for increased releases during disposal. Mitigation would be required. Unlikely to find location suitable for disposal cell. Significant permitting effort would be required and acceptance by regional stakeholders. Would require long-term monitoring and maintenance. | High | Not retained because, among other issues, no suitable site is available. | | | |
| | Confined Disposal Facility (CDF) | A disposal facility built on-site within land or reclaimed land. The sediment is deposited in the lower, saturated portion of the cell. Sediment in the CDF is physically separated from waterway by clean soil. | Effective at removing source material from cleanup area and placement in a self-managed waste facility. Addresses direct exposure pathways and migration by removing contaminant mass from cleanup area. Placement and design of CDF facility must account for potential soluble contamination migration. Would require ongoing maintenance of disposal facility. | Potential for increased releases during disposal. Mitigation would be required. Significant permitting effort would be required and acceptance by regional stakeholders. Would require long-term monitoring and maintenance. | High | Not a stand alone technology. Potentially applicable for handling of dredged/excavated sediment. | | | |
| File No. 25331-001-01 | 2022 | I | 2 of 6 | I | | GeoEngineers 🖊 | | | |

Page 165 of 240

| General Response | Bemedial Technology | Description | Screening Criteria | | | Screening Comments | |
|---|--------------------------------|--|--|---|------------------|---|--|
| Action | Kemediai Technology | Description | Effectiveness | Implementability | Cost | | |
| | Chemical Oxidation | Involves application of oxidizing agents that result in reduction / oxidation (redox) reactions that chemically convert hazardous contaminants to less toxic or less mobile forms. | Can be highly effective at destruction of organic contaminants. However, would be difficult to achieve full coverage (contact between oxidant and COC). Is not effective in sediment matrix containing significant amounts of creosote-treated wood debris. | Would be difficult to get obtain full coverage of oxidant in sediment. Could be implemented as slurry but would require removal of wood debris and significant containment effort (such as installation of sheet piling). Less suitable for shallow sediments as application difficult to separate from free water zone and multiple injection/mixing points would be needed. Care would be needed to prevent secondary impacts (such as from mobilized NAPL) during oxidation. | Moderate to High | Not retained because technology is incompatible with Site conditions. | |
| IN-PLACE PHYSICAL / CHEMICAL TREATMENT | Sediment Flushing | Water or some other aqueous solution is circulated through the contaminated sediment to desorb contaminants. The circulated water is then recovered and treated. | Bench scale tests at other sites have shown to be effective. Less effective for organic contaminants bound on organic matter (wood debris) and would require an upland treatment operation. | Would be difficult to get full coverage of solution through sediment. Less suitable for shallow sediments as application difficult to separate from free water zone and multiple injection/mixing points would be needed. | Moderate to High | Not retained as implementation risks are high and other more suitable technologies are available. | |
| | Solidification / Stabilization | Contaminants are physically bound or enclosed in a stabilized mass (solidification) or chemical reactions are induced between the reagent and contaminants to reduce their mobility (stabilization). | Most suitable to inorganic contaminants. Resultant sediments may not provide suitable ecological habitat. | High-energy solidification would be inefficient (or impractical) with saturated sediments. | Moderate to High | Not retained because technologies incompatible with Site conditions. | |
| | Electrokinetic Separation | Application of a low-intensity direct current through the sediment between electrodes (cathode array and anode array). This mobilizes charged ion species causing movement toward the electrodes. | Effective at removing inorganic ions and some polar organics from saturated soil. No demonstrated application to sediment treatment. | Requires significant electrical power and would have high implementation risks in standing water. Would be difficult to control in shallow sediments. | Moderate to High | Not retained because technology incompatible with Site contaminants and has high implementation risks. | |
| | Electrochemical Oxidation | Technology for degrading organic contaminants <i>in situ</i> by applying an electrical current across electrodes placed in the subsurface to ionize oxidizing species (i.e., metal ions) and cause oxidation of the COC. | Laboratory bench scale tests suggest technology could be effective for organics. Application in sediments is untested and would be experimental. | Requires significant electrical power and would have high implementation risks in standing water. Would be difficult to control in shallow sediments. | Moderate to High | Not retained because it is an unproven technology and has high implementation risks. | |
| IN-PLACE PHYSICAL / | Organoclay Sediment Cap | A semi-permeable layer of organoclay is placed over impacted sediment to adsorb contaminants as impacted groundwater or porewater moves through the material. | May be effective in reducing recontamination due to NAPL migration and seeps. It is not clear at this time if a reactive cap would be more effective than a traditional cap. Would require long-term monitoring to verify effectiveness of the cap. | There are full-scale examples of organoclay sediment caps. It may be difficult to estimate the useful life of a reactive cap. Required construction equipment is expected to be relatively easy to obtain for implementation. | Low to Moderate | Retained as applicable technology. | |
| CHEMICAL TREATMENT | Activated Carbon Amendment | Activated carbon is blended into sediments to increase sorptive capacity of sediment and reduce bioavailability of organic contaminants. | Could effectively control residual concentrations after dredging or to enhance natural recovery. | Several pilot projects have demonstrated effectiveness for organic constituents. Relatively simple technologies required for implementation. | Low to Moderate | Potentially applicable for management of residual concentrations of PAHs. | |
| | Biochar Amendment | Biochar is a sorbing agent made by pyrolysis of biomass that is similar to activated carbon. The biochar is spread or blended into sediments to increase sorptive capacity of sediment and reduce bioavailability of organic contaminants. | Could effectively control residual concentrations after dredging or to enhance natural recovery. Available information suggests that biochar is generally less effective as a adsorptive agent than activated carbon. Would require long-term monitoring to verify effectiveness of the cap. | There is limited experience in full-scale implementation of biochar amendment of sediment. Required construction equipment is expected to be relatively easy to obtain for implementation. | Low to Moderate | Potentially applicable for management of residual concentrations of PAHs. | |



Page 166 of 240

| General Response | Pemedial Technology | Description | | Screening Criteria | | Screening Comments | |
|----------------------------------|--|---|---|---|------------------|---|--|
| Action | Action | | Effectiveness | Implementability | Cost | Screening comments | |
| IN-PLACE PHYSICAL / | Ecospears | A relatively new remediation product designed primarily for removal of PCBs from sediments via sorption onto the polymer material that is pushed into the surface sediment. | Laboratory bench scale tests suggest technology could be effective for some Site contaminants. Application in sediments is generally untested and would be experimental. Little to no application literature is available to assess effectiveness. Primarily developed for removal of PCBs, although it may be suitable for other persistent pollutants. | There is limited experience in full-scale implementation of biochar amendment of sediment. Required construction equipment is expected to be relatively easy to obtain for implementation. | Moderate to High | Not retained because it is an unproven technology and has high implementation risks. | |
| CHEMICAL TREATMENT (cont.) | Image: Second system Image: Second system A relatively new technology, oleophilic biobarriers utilize a oleophilic (strong affinity for oils) plastic geocomposite U Image: Oleophilic Biobarrier A relatively new technology, oleophilic biobarriers utilize a oleophilic (strong affinity for oils) plastic geocomposite U Image: Oleophilic Biobarrier Image: Oleophilic Biobarrier Image: Oleophilic Biobarrier Image: Oleophilic Biobarrier Image: Oleophilic Biobarrier Image: Oleophilic Biobarrier Image: Oleophilic Biobarrier Image: Oleophilic Biobarrier Image: Oleophilic Biobarrier Image: Oleophilic Biobarrier Image: Oleophilic Bio | | Unlike other adsorptive barriers which are designed to reduce the bioavailability of NAPL, oleophilic biobarriers are designed to increase aerobic conditions within the mat to promote biodegradation by naturally occurring microorganisms. | There is limited experience in full-scale implementation of oleophilic biobarriers. Required construction equipment is expected to be relatively easy to obtain for implementation. | Moderate | Retained and potentially applicable for managing petroleum sheens within the waterway. | |
| | Enhanced Bioremediation | Addition of nutrients, electron acceptors, or other amendments to sediment to enhance bioremediation. | Not likely to be effective for many of the contaminants of concern. | Blending of amendments into sediment would require disturbance of sediment and would potentially cause significant resuspension. | Moderate | Not retained because not effective on NAPL and many of the high molecular weight PAHs. | |
| IN-PLACE BIOLOGICAL TREATMENT | PLACE BIOLOGICAL TREATMENT Phytoremediation The process of using plants to remove, transfer, stabilize, and/or destroy contaminants in soil or sediment. Can be effective at rem compounds from soil/se plant rhizosphere. | | Can be effective at removing a variety of organic and inorganic compounds from soil/sediment through plant uptake in the plant rhizosphere. | Would require planting of suitable plants for Site conditions, or changing of Site conditions to accommodate plants (such as the construction of engineered wetlands). May not be compatible with current industrial Site use. | Moderate | Not retained because incompatible with Site conditions. | |
| NATURAL PROCESSES | Monitored Natural Recovery | Naturally occurring physical processes (advection, desorption, dispersion, diffusion, dilution, resuspension, sedimentation, and volatilization), and biological processes (biodegradation) reduce contaminant concentrations. Process is monitored to verify exposures. In areas of low natural sedimentation processes, natural recovery can be enhanced by placement of a thin-layer cap. | Effectiveness for sediments is primarily related to sedimentation. Would require long time-frame for degradation of residual organics. Natural sedimentation likely will not be effective in the Upper Milton Creek, Area 1 Dock, Area 2 Dock offshore areas due to the higher energy of the creek and channel. | Easy to implement. Monitoring of COC concentrations in sediment porewater and surface water would require long-term passive sampling and sheen monitoring. May require significant timeframe to reach cleanup goals. For enhanced recovery, thin-layer caps use conventional technologies. | Low | Natural recovery is applicable in areas of lower energy such as the Cove Area. Monitored natural recovery is not suitable in areas where COC concentrations exceed preliminary remediation goals (PRGs). | |
| | Enhanced Monitored Natural Recovery | Enhancement of natural attenuation consisting of a thin- layer cap (i.e., 4 to 6 inches to 1 foot of sand) that provides a surface layer of cleaner sediment accelerating physical isolation that occurs from the natural sedimentation process. | Would be effective at providing a surface layer to provide separation and attenuation of dissolved-phase contaminant mass flux. Would require long timeframe for degradation of residual organics. Enhanced recovery through placement of a thin layer cap could be very effective, especially in areas with relatively low exceedance factors. | Easy to implement. Monitoring of PAHs in sediment porewater and surface water would require long-term monitoring. May require significant timeframe to reach cleanup goals. For enhanced recovery, thin-layer caps use conventional technologies. | Moderate | Enhanced recovery is potentially applicable, especially in areas with relatively lower exceedance ratios. | |



Page 167 of 240

| General Response | Bomodial Tashnalagy | Description | | Screening Criteria | Screening Comments | |
|---|--------------------------------|--|---|---|--------------------|--|
| Action | Rememai recimology | Description | Effectiveness | Implementability | Cost | - Screening comments |
| | Dewatering | Removal of water from dredged/excavated sediment (such as to facilitate disposal). Methods may include passive dewatering on barges, in an upland lined and bermed stockpile, geotextile tubes (filters), filter presses, other mechanical dewatering methods, or dewatering by adding chemical reagents or adsorptive materials. | Various methods can be effectively used (selected based on site conditions and degree of dewatering needed) to remove water from dredged/excavated sediment. Debris may need to be removed from sediment prior to dewatering. Resultant water may need to be treated prior to disposal. | Fine-grained nature of sediment may require long drying times and significant operational effort to meet landfill free liquid standards. Water removed from the contaminated media would require either treatment, evaporation, or absorptive onto a solid material (e.g., perlite). | Low to Moderate | Not a stand-alone technology. Retained as potentially applicable for use in conjunction with other technologies such as preparation of excavated saturated media for off-site disposal. |
| | Separation | Use of physical means to separate sandier sediments (which would have less contamination) for beneficial reuse or separation of debris prior to further treatment or disposal. | Generally not effective for site sediment conditions. | Commercial equipment is available for separation (i.e., sieves). Separated sand may be available for potential beneficial use (would require verification testing and identification of potential use). Bench scale testing may be needed to define specific operating parameters. | Low to Moderate | Sediments previously identified to have high organic content. Not retained because the impacted material removed would primarily consist of finer sediments. |
| | Sediment Washing | Contaminants are separated from the dredged sediment with wash water augmented with additives to help remove contamination. | Most suitable for highly refined petroleum products (e.g., gasoline, diesel) in coarse-grained material. Less effective on viscous contamination like creosote in fine-grained material. | Elutriate would require treatment and disposal, which could significantly increase the overall cost of treatment. Bench-scale testing would be required during design. Requires staging area for treatment or transport to approved off-site facility. Air quality standards may be affected by open-air treatment methods. | High | Not retained because not compatible with creosote- impacted fine-grained sediment and more cost- effective options are available. |
| EX-SITU PHYSICAL/ CHEMICAL TREATMENT | Chemical Oxidation | Includes the application of chemical oxidants for the purpose of remediating contaminated sediments/soils. Generally involves reduction/oxidation (redox) reactions that chemically convert hazardous contaminants to less toxic or less mobile forms. | Can be highly effective at destruction of organic contaminants. May not be cost effective for high contaminant concentrations or high organic sediments due to large amounts of oxidizing agent required. Less efficient for low concentrations compared to other technologies. | Risks associated with handling of oxidant in above- ground application. Bench-scale testing would be required during design. Requires staging area for treatment or transport to off-site facility. Air quality standards for site workers may be affected by open-air treatment methods. | High | Not retained because technology has relatively high implementation risk to workers and equally effective lower cost technologies are available. |
| | Chemical Extraction | Dredged / excavated sediment is mixed with an extractant (e.g., acid or solvent), which dissolves the contaminants. The resultant solution is placed in a separator to remove the contaminant / extractant mixture for treatment. | Most suitable to semi-volatile or inorganic contamination. Less effective in fine-grained sediment. | Difficult to remove all contaminant / extractant from sediment - would likely require finish treatment. Elutriate would require treatment and disposal, which could significantly increase the overall cost of treatment. Bench-scale testing would be required during design. Requires staging area for treatment or transport to off-site facility. Air quality standards may be affected by open-air treatment methods. | High | Not retained because not compatible with sediment grain size and more cost-effective options available. |
| | Solidification / Stabilization | Contaminants are physically bound or enclosed in a stabilized mass (solidification) or chemical reactions are induced between the stabilizing agent and contaminants to reduce their mobility (stabilization). Methods may include the addition of Portland cement, lime, kiln dust, pozzolan, sorbent clay (i.e., bentonite), and proprietary reagents. | Can be effective at reducing mobility of contaminants (most suitable to inorganics) or solidifying for disposal. | Would need to be significantly dewatered prior to solidification. Requires staging area for treatment or transport to off-site facility. Air quality standards for Site occupational workers may be affected by open-air dewatering methods. | Moderate | Would likely still require landfill disposal and dewatering technologies already retained above. |



Page 168 of 240

| General Response | Pemedial Technology | Description | | Screening Criteria | Screening Comments | |
|--|--|--|---|--|--------------------|--|
| Action | Rememai recimology | Description | Effectiveness | Implementability | Cost | Screening comments |
| EX-SITU PHYSICAL/ CHEMICAL TREATMENT (cont.) | Solar Detoxification | Contaminants are destroyed by photochemical and thermal reactions using ultraviolet energy in sunlight or artificial UV light. Usually involves application of catalyst agent. | Can be effective at treating a variety of organic compounds. Most effective when used with a catalyst agent (i.e., titanium dioxide). | Implementation with sunlight limited by availability (not effective during nighttime and limited effectiveness during cloudy/wet seasons). Requires staging area for treatment or transport to off-site facility. Air quality standards for Site/occupational workers may be affected by open-air treatment methods. | Moderate | Not retained because space requirement and has limited usefulness for Site contaminants. |
| | Land Treatment/ Landfarming | Land treatment reduces contaminant concentrations through biological processes. Dredged / excavated sediment is placed in lined and bermed treatment cells and manipulated as necessary to improve biological conditions (such as tilling material to aerate and mix in nutrient amendments). | Can be effective at removing PAHs in sediment. Most effective with control of moisture, heat, nutrients, oxygen, and pH to enhance biodegradation. Effectiveness is reduced by the presence of NAPL, wood debris, and low ambient temperatures during 8 months of the year. | Requires large upland area for sediment treatment. Requires the dewatering and removal of debris from sediment. Erosion and stormwater controls need to address ponding and/or contaminated runoff. Bench- scale testing would be required to define operating parameters, particularly during wet weather. Long anticipated implementation period with extensive performance and confirmation monitoring. Air quality standards may be affected by open-air treatment methods. | Moderate | Not retained because space requirement and has limited usefulness for Site contaminants. |
| EX-SITU BIOLOGICAL TREATMENT | Biopiles | Sediment is mixed with amendments, placed in aboveground enclosures, and aerated with blowers or vacuum pumps. Microorganisms present degrade the contaminants present. | Effective at removing volatile organic contamination from sediment. Most effective with control of moisture, heat, nutrients, oxygen, and pH to enhance biodegradation. Effectiveness is reduced by the presence of NAPL. | Requires area for sediment treatment or transport to an off-site facility. Requires dewatering of sediment, and controls likely to be needed for contaminant migration from runoff and leachate. Bench-scale testing would be required to define operating parameters. Air quality standards may be affected by open-air treatment methods. | Moderate | Not retained because effectiveness is severely limited by the presence of NAPL and wood debris. |
| | Slurry-phase Biological Treatment | An aqueous slurry of sediment with water and other additives is mixed to keep solids suspended and microorganisms in contact with the particle-bound contaminants. When complete, the slurry is dewatered and the treated sediment is disposed of. | Can be effective at treating a variety of organic compounds and diluting NAPL into manageable concentrations. | Requires area for treatment cell or transport to an off- site facility. Slurry dewatering generates liquid waste stream that will require treatment or disposal. Bench- scale testing would be required to define operating parameters. Air quality standards may be affected by open-air treatment methods. | High | Not retained because other more cost-effective options available. |
| | Off-Site Incineration High temperatures are used to combust (in of oxygen) organic constituents in hazardous | | High temperatures result in generally complete decomposition of organic chemicals. Effective across wide range of sediment characteristics. | Requires air pollution control device. Listed hazardous waste designation will likely rule out acceptance into most treatment facilities. Involves high energy consumption. | High | Unlikely that a permitted facility will accept this hazardous waste stream for treatment. If found, anticipate significant cost for transportation and treatment. Other less expensive technologies available. |
| TREATMENT | On-Site Thermal Desorption/ Pyrolysis/ Hot Gas Decontamination | Dredged spoils are heated to either volatilize (desorption and hot gas) or to anaerobically decompose (pyrolysis) organic contaminants. Off-gas is collected and treated. | Effective at removing organic materials from excavated sediment/soil (particularly volatile organics). Pyrolysis generally used for semi-volatiles. Efficiency of PAH thermal desorption will be approximately 90%. | Requires the on-site mobilization of thermal desorption units and permitting with local land use and air quality agencies. Requires dewatering of excavated media, off- gas treatment, and extensive confirmation sampling. | High | Not retained because other more cost-effective options available. |

Note:

Shading indicates technologies that have been eliminated from consideration.

Table 5



Page 169 of 240

| Priority Action Area | Timber Piling Area (: | | Volume (approx. cubic | Depth | Length | Offsite-Disposal (Preferred RAA) ¹ | |
|--|--------------------------|----------------|--------------------------|--|----------------------------------|--|--------------------------------|
| · | Each | (approx. acre) | yards) | (feet) | (approx. LF) | Haz (tons) ² | Non-Haz (tons) ² |
| Upland Area | | | | | | | |
| Soil hot-spot (Includes NAPL) | | 3.45 | 95,000 | 8 feet bgs to approx. 25 feet bgs. (upper 8-feet, clean cap overburden) | | 2,947 | 520 |
| Area 1 Dock | | <u>.</u> | | | | | <u> </u> |
| Wood Debris/ Surface Sediments (NAPL/moderate to heavy sheen) | 435 | 1.85 | 3,000 | mudline to 1 foot bml (biologically active zone) | - | | 163 |
| Subsurface Sediments | | 1.85 | 29,800 | Varies; >1 foot bml up to 11 feet bml | | | |
| Area 2 Dock | | | | | | | |
| Wood Debris/ Surface Sediments (Creosote Impacted) | 235 | 0.34 | 560 | mudline to 1 foot bml (biologically active zone) | | 519 | 222 |
| Cove Area | | | | | | | |
| Riverbank hotspots (NAPL seeps)/Riverbank Regrading from 2H:1V to 4H:1V | | | | 11 to 25 feet bgs in 1-foot-thick or less sand layers | 300 LF seeps/540 LF regrading | 3,787 | 1,623 |
| Wood Debris/ Surface Sediments (NAPL/moderate to heavy sheen) | 139 | 1.2 | 1,950 | mudline to 1 foot bml (biologically active zone) | | | |
| Subsurface Sediments | | 0.85 | 15,250 | Varies; >1 foot bml up to 12 feet bml, average 5 feet bml | | | |
| Upper Milton Creek | | <u>I</u> | | | | | <u></u> |
| Riverbank hotspots (NAPL seeps)/Riverbank Regrading from 2H:1V to 4H:1V | | 0.05 | 1,020 | 19 feet bgs in 1-foot- thick or less sand layer | 200 LF seeps/ regrading | 1,550 | 82 |

Notes:

¹ See Section 8 of Staff Report for description of preferred RAAs.

 2 Estimates assume 1.6 tons/CY for impacted soil and sediment.

| > | greater than | LF | linear feet |
|---------|----------------------|------|-----------------------------|
| approx. | approximately | NAPL | non-aqueous phase liquid |
| bgs | below ground surface | RAA | Remedial Action Alternative |
| bml | below mud line | V | vertical |
| Н | horizontal | | |

Page 170 of 240

| Table 7. Area 1 Upland PAA remedial technologies descriptions and assembla | ze of RAAs. |
|--|-------------|
|--|-------------|

| | Impervious Surface Cap & Stormwater System | Hydraulic Containment & Enhanced Bioremediation | Permeable Reactive Barrier | Impermeable Isolation Wall | Complete Excavation and Offsite Disposal | Monitored Natural Attenuation |
|---|---|--|---|---|--|---|
| Remedial Action Alternative ¹ | Cap constructed across the entire PAA, consisting of low-permeability (1 x 10-6 cm/s or less) material (e.g., concrete or bituminous asphalt pavement). This PAA was previously covered with 7 to 8 feet of imported fill material and a low permeability surface cap would further isolate contamination from human and ecological receptors. However, the main purpose of the cap would be to reduce stormwater infiltration through contaminated soil and reduce groundwater flux towards in-water locations. The cap will be graded to collect and convey stormwater to a bioswale for infiltration outside the groundwater plume. Stormwater management reduces erosion caused by stormwater runoff. | Groundwater pumping, treatment, and recirculation system. The groundwater extraction system will consist of wells or interceptor trenches constructed between the upland NAPL area and the Cove Area PAA. Extracted groundwater will include NAPL separation, treatment with adsorptive media, and amended with electron acceptors and nutrients to aid in enhanced natural biodegradation. Treated and amended groundwater will be infiltrated back to areas with residual contamination. A pilot study is needed to determine groundwater extraction and infiltration rates, influence of fine-grained native soil, radius of influence, infiltration gallery sizing, and groundwater treatment methodologies. Microbial profiling and bench- scale testing of total PAH degradation under different conditions is needed to determine the appropriate electron acceptors and nutrients for groundwater amendment. Regular upkeep and maintenance required over the long term. | PRB placed along top of riverbank, upland of the Cove Area NAPL seeps, to capture mobile creosote NAPL and dissolved phase COCs in groundwater prior to discharging to the Cove Area. The PRB will consist of adsorptive (e.g., organoclay) and reactive materials to (1) remove contaminants from groundwater and (2) stimulate treatment of contaminants (e.g., adding electron acceptors to enhance biodegradation). The PRB will be 2 to 4 feet wide and extend to the top of the Basalt Bedrock, approximately 25 feet bgs but deeper in some areas. A total of 3,467 tons of creosote-impacted soil would be removed and transported for off- site disposal. A pilot study and modeling are needed to select the PRB media, amendments, installation methods, thickness, depth, length, and placement. Media fouling and reduced permeability impact long-term maintenance needs. | Isolation wall surrounding the entire length of the inferred upland NAPL area and extending to the underlying bedrock to physically isolate the contamination in the upland NAPL area from groundwater flow and reduce contaminant transport. Wall would be constructed with low permeability (1 x 10-7 cm/s or less) material (e.g., bentonite slurry). Construction by excavating a 3-foot-wide trench, blending excavated soil with bentonite, and replacing the material to the trench. Wall keyed into bedrock, with bedrock locations greater than 25 feet bgs requiring specialized equipment. Significant excavation required. Assumption that excess material or material deemed unsuitable for backfill will be disposed offsite at a Subtitle C landfill. Additional investigation needed to delineate lateral and vertical upland NAPL extent. During remedial design the isolation wall type and blend of material and placement would be determined. | Remove all NAPL-impacted soil and sediment within the inferred upland NAPL area to an approximate depth of 25 feet bgs. Contaminated soil transported offsite for disposal at a Subtitle C landfill. Dewatering, treatment of extracted groundwater, and discharge to the Scappoose Bay under permit. Shoring will be required near the Scappoose Bay shoreline. Confirmation soil samples will be collected from the sidewalls and base of the excavated area to ensure complete removal. Clean fill will be used to backfill the entire excavated area followed by construction of an impervious surface cap. No long-term ICs required. | Creosote in subsurface soil and groundwater is expected to slowly reduce over time through natural processes, including diffusion, dispersion, hydrolysis, sorption, and biodegradation. However, MNA alone is not expected to reduce COC concentrations in source areas within a meaningful timeframe and is only considered in combination with other remedial actions. |
| RAA-1: No Action | | | | | | |
| RAA-2: Impervious Surface Cap, MNA | Х | | | | | x |
| RAA-3: Impervious Surface Cap, Hydraulic Containment, Enhanced Bioremediation | Х | Х | | | | x |
| RAA-4: Impervious Surface Cap, MNA, Permeable Reactive Barrier | X | | Х | | | X |
| RAA-5: Impervious Surface Cap, MNA, Impermeable Isolation Wall (NAPL Area) | Х | | | x | | X |
| RAA-6: Excavation and Offsite Disposal, Impervious Surface Cap (over residuals), MNA | X | | | | X | X |

Notes:

| See Staff Report Section 5 |
|----------------------------------|
| below ground surface |
| centimeters per second |
| Constituents of Concern |
| Cubic Yards |
| Institutional Control |
| Monitored Natural Attenuation |
| Non-Aqueous Phase Liquid |
| Priority Action Area |
| Polycyclic Aromatic Hydrocarbons |
| Permeable Reactive Barrier |
| Remedial Action Alternative |
| Square Feet |
| percent |
| Selected Remedial Action |
| |

| | Armored Reactive Cap | Nearshore Removal Action | Upland Consolidation | Offsite Disposal | Complete Excavation |
|--|--|---|---|---|--|
| Remedial Action Alternative ¹ | An engineered reactive cap will be constructed over the entire Area 1 Dock PAA (1.85 acres). Debris, including timber piling and other remnants of dock structures, and vegetation will be removed prior to cap placement. The cap will include a reactive chemical isolation layer designed to contain and isolate contaminants. Cap armoring will be needed to ensure cap stability and protection from erosion and scouring, particularly given the stronger river currents in this area, and will include aquatic supporting material. During remedial design, the cap materials, amendments, rock armoring size, layer thicknesses, and extent will be determined. Evaluations of how remaining debris may impact long-term cap performance will be conducted. | Upper nearshore sediment exhibiting moderate to heavy petroleum sheen (up to 80%) will be excavated using land-based equipment. Prior to excavation, debris, including timber piling and other remnants of dock structures, and vegetation will be removed. A temporary cofferdam system will be installed surrounding the PAA to extend the reach of the land-based removal equipment. The area behind the cofferdam will be dewatered by pumping, with pumped water treated prior to discharge. During remedial design the methods for removal, depth and extent of excavation, locations suitable for a cofferdam area, methods and criteria for treating pumped water, and appropriate treated water discharge areas will be determined. A flood rise analysis will also be conducted and may impact regrading needs. Wood debris has been observed as deep as 11 feet bml and an estimated 435 creosote-treated piles will need to be removed. Geotechnical investigations will be conducted to address soft sediment and the feasibility of removing wood debris due to the significant volume and age of the debris. If removing full pilings is not feasible, pilings will be cut at the surface to facilitate cap placement. | Excavated material (7,744 tons) and 435 timber piles will be dewatered and stabilized as needed (e.g., drying agents) and placed in a lined upland containment area, covered with clean overburden material, and covered with a low-permeability cap as described for the Upland PAA. During remedial design the dewatering and stabilization methods, exact dimensions of the containment area, lining material, composition and depth of clean overburden material, and surface cap specifications will be determined. | Removed material will be dewatered and stabilized as needed (e.g., drying agents) and transported to an offsite disposal facility. The proper facility for disposal (i.e., Subtitle C or Subtitle D landfill) will be determined by material profiling. A total of 7,744 tons of creosote-impacted soil would be removed and transported for off-site disposal, along with 435 piles. Trucks, train, and/or barge may be used depending on the volume of removed material in conjunction with other PAAs. During remedial design the dewatering, stabilization, and transportation methods will be determined. | Remove all creosote-impacted sediment within the PAA to a depth of 12 feet. A steel interlocking sheet pile wall will be installed along the riverward perimeter of the PAA. Sediment and riverbank material will be removed using land-based and barge-based equipment. Clean replacement material will be placed within the excavation area following removal activities. The remedial design will determine methods for implementing the sheet pile wall, equipment and methods for material removal, dewatering and stabilization methods, profiling required for offsite disposal, and composition of clean replacement material. |
| RAA-1: No Action | | | | | |
| RAA-2: Armored Reactive Cap | X | | | | |
| RAA-3: Nearshore Removal Action, Upland Consolidation, and Armored Reactive Capping of Residuals | X | X | X | | |
| RAA-4: Nearshore Removal Action, Offsite Disposal, and Armored Reactive Capping of Residuals | X | X | | X | |
| RAA-5: Complete Removal and Offsite Disposal | | | | | Х |

1See Staff Report Section 5bmlbelow mudlineCYCubic YardsPAAPriority Action AreaRAARemedial Action AlternativeSFSquare Feet%percentSelected Remedial Action

Table 9. Area 2 Dock PAA remedial technologies descriptions and assemblage of RAAs.

| | Armored Reactive Cap | | Sand Cap | | Nearshore Removal Action | | Upland Consolidation | Offsite Disposal |
|--|---|---|---|--|--|--|---|--|
| Remedial Action Alternative ¹ | An engineered reactive cap will be constructed over 600 feet of nearshore sediments exhibiting moderate to heavy petroleum sheen. Debris, including timber piling and other remnants of dock structures, and vegetation would be removed prior to cap placement. The cap will include a reactive chemical isolation layer designed to contain and isolate contaminants. Cap armoring will be needed to ensure cap stability and protection from erosion and scouring, particularly given the stronger river currents in this area, and will include aquatic supporting material. During remedial design, the cap materials, amendments, rock armoring size, layer thicknesses, and extent will be determined. Evaluations of how remaining debris may impact long-term cap performance will be conducted. | Cap area: 0.34 acres (15,182 SF) 235 Piles | A layer of sand will be placed throughout the PAA, including the 3,000 square foot area that is not accessible by land-based equipment. As most of the contamination is in the surface sediments, following removal action there will be limited contamination residuals within excavation areas. During remedial design, the cap material, thickness, extent, and sufficiency or insufficiency of sand only will be determined. Evaluations of how remaining debris may impact long-term cap performance will be conducted. | Cap area: 0.34 acres 15,182 SF) 235 Piles | The upper 12 inches of nearshore sediment exhibiting moderate to heavy petroleum sheen (up to 84%) will be excavated using land-based equipment. Prior to excavation, debris, including timber piling and other remnants of dock structures, and vegetation will be removed. A temporary cofferdam system will be installed surrounding the PAA to extend the reach of the land-based removal equipment. However, approximately 3,000 square feet of the PAA will not be accessible by land-based equipment. The area behind the cofferdam will be dewatered by pumping, with pumped water treated prior to discharge. During remedial design the methods for removal, depth and extent of excavation, locations suitable for a cofferdam system, methods for dewatering the cofferdam area, methods and criteria for treating pumped water, and appropriate treated water discharge areas will be determined. A flood rise analysis will also be conducted and may impact regrading needs. Petroleum sheen near creosote-impacted wood debris has been observed within the upper 12 inches of sediment. An estimated 235 creosote-treated piles will need to be removed. Geotechnical investigations will be conducted to address the feasibility of removing wood debris. | Volume removed: 560 CY (100% hot spot removed) 235 Piles | Excavated material will be dewatered and stabilized as needed (e.g., drying agents) and placed in a lined upland containment area, covered with clean overburden material, and covered with a low-permeability cap as described for the Upland PAA. During remedial design the dewatering and stabilization methods, exact dimensions of the containment area, lining material, composition and depth of clean overburden material, and surface cap specifications will be determined. | Removed material will be dewatered and stabilized as needed (e.g., drying agents) and transported to an offsite disposal facility. The proper facility for disposal (i.e., Subtitle C or Subtitle D landfill) will be determined by material profiling. Trucks, train, and/or barge may be used depending on the volume of removed material in conjunction with other PAAs. During remedial design the dewatering, stabilization, and transportation methods will be determined. |
| RAA-1: No Action | | | | | | | | |
| RAA-2: Armored Reactive Cap | X | | | | | | | |
| RAA-3: Nearshore Removal Action, Upland Consolidation, and Capping of Residuals | | | Х | | Х | | х | |
| RAA-4: Nearshore Removal Action, Offsite Disposal, and Capping of Residuals | | | X | | X | | | Х |
| Notes: 1 | See Staff Report Section 5 | | | | | | | |

CY Cubic Yards PAA Priority Action Area PRB Permeable Reactive Barrier RAA Remedial Action Alternative SF Square Feet % percent Selected Remedial Action Table 10. Cove Area PAA remedial technologies descriptions and assemblage of RAAs.

| | Armored Reactive Cap | Riverbank Restoration & Limited Removal Action | Upland Consolidation | Offsite Disposal | Complete Excavation |
|---|---|--|---|--|--|
| Remedial Action Alternative ¹ | An engineered reactive cap will be constructed over the 300-foot section of streambanks exhibiting moderate to heavy petroleum sheen. Debris, including timber piling and other remnants of dock structures, and vegetation will be removed prior to cap placement. The cap will include a reactive chemical isolation layer to contain and isolate contaminants. Cap armoring will be needed to ensure cap stability and protection from erosion and scouring, though the Cove's off-channel backwater setting provides some protection against wake- and wind-driven waves. The armored surface and slope must be compatible with shallow water habitat. According to preliminary CapSim modeling, the cap will effectively prevent contaminants from entering the river. Potential risks associated with differential settlement below the cap can be mitigated using capping materials that evenly distribute the weight of armoring (e.g., marine mattress). Mitigation will be needed to offset the loss of intertidal shallow water habitat. Sediment porewater and surface water quality outside the cap boundary is expected to continue naturally recovering as new, cleaner sediment is deposited in the Cove (0.2 to 1.4 in/yr). During remedial design, the cap materials, amendments, rock armoring size, layer thicknesses, and extent will be determined. | Approximately 500 feet of riverbank will be regraded as needed to a sufficiently shallow slope (e.g., 4H:1V) to reduce erosion potential and aid in land-based removal and long-term monitoring. This regraded area will include the 300-foot section of riverbank with intermittent NAPL seeps. Regrading facilitates improved access by land-based equipment, hot spot removal, more controlled placement of cap material, reduced erosional forces imposed on the cap, and long-term monitoring. The upper portion of the nearshore sediment exhibiting moderate to heavy sheen (up to 90%) will be excavated using land- based equipment. This approach will remove nearshore hot spot areas to the extent practicable. Prior to excavation, debris, including timber piling and other remnants of dock structures, and vegetation will be removed. A temporary cofferdam system will be installed surrounding the PAA to extend the reach of the land-based removal equipment. The area behind the cofferdam will be dewatered by pumping, with pumped water treated prior to discharge. During remedial design the exact slopes, methods for regrading and removal, depth and extent of excavation, locations suitable for a cofferdam system, methods for dewatering the cofferdam area, methods and criteria for treating pumped water, and appropriate treated water discharge areas will be determined. A flood rise analysis will also be conducted and may impact regrading needs. | Excavated material will be dewatered and stabilized as needed (e.g., drying agents) and placed in a lined upland containment area, covered with clean overburden material, and covered with a low-permeability cap as described for the Upland PAA. During remedial design the dewatering and stabilization methods, exact dimensions of the containment area, lining material, composition and depth of clean overburden material, and surface cap specifications will be determined. | Removed material will be dewatered and stabilized as needed (e.g., drying agents) and transported to a Subtitle C landfill. Trucks, train, and/or barge may be used depending on the volume of removed material in conjunction with other PAAs. During remedial design the dewatering, stabilization, and transportation methods will be determined. | Remove all creosote-impacted sediment within the PAA to a depth of 12 feet. A steel interlocking sheet pile wall will be installed along the riverward perimeter of the PAA. Sediment and riverbank material will be removed using land-based and barge-based equipment. Clean replacement material will be placed within the excavation area following removal activities. The remedial design will determine methods for implementing the sheet pile wall, equipment and methods for material removal, dewatering and stabilization methods, profiling required for offsite disposal, and composition of clean replacement material. |
| RAA-1: No Action | | | | | |
| RAA-2: Armored Reactive Cap | x | | | | |
| RAA-3: Riverbank Restoration, Nearshore Removal Action, Upland Consolidation, and Reactive Capping of Residuals | х | X | Х | | |
| RAA-4: Riverbank Restoration, Nearshore Removal Action, Offsite Disposal, and Reactive Capping of Residuals | x | x x x | | | |
| RAA-5: Complete Removal and Offsite Disposal | | | | X | X |
| Notes: 1 | See Staff Report Section 5 | | | | |

| CY | Cubic Yards |
|-------|--------------------------------------|
| H:V | horizontal distance to vertical rise |
| in/yr | inches per year |
| NAPL | Non-Aqueous Phase Liquid |
| PAA | Priority Action Area |
| RAA | Remedial Action Alternative |
| SF | Square Feet |
| % | percent |
| | Selected Remedial Action |

Table 11. Upper Milton Creek PAA remedial technologies descriptions and assemblage of RAAs.

| | Armored Reactive Cap | | Regrading & Removal Action | ı | Upland Consolidation | | |
|--|---|---------------------------------------|---|--------------------------------|--|---|--|
| Remedial Action Alternative ¹ | An engineered reactive cap will be constructed over the 200-foot section of streambanks exhibiting moderate to heavy petroleum sheen. Debris and vegetation will be removed prior to cap placement. The cap will include a reactive chemical isolation layer to contain and isolate contaminants. Cap armoring and shoreline stabilization will be needed to ensure cap stability and protection from erosion and scouring. During remedial design, the cap materials, amendments, rock armoring size, layer thicknesses, and extent will be determined. The cap will be placed following any regrading and removal activities. | Cap area: 0.05 acres (2,200 SF) | Currently, the upper portion of Milton Creek has relatively steep 15-foot embankments and generally consists of 0 to 1 foot of soft sediment overlying compact clayey silt with gravel and cobbles. The streambank will be regraded as needed to a sufficiently shallow slope (e.g., 3H:1V) to facilitate hot spot removal, more controlled placement of cap material, reduced erosional forces imposed on the cap, and long-term monitoring. The lower portion of the streambank exhibiting moderate to heavy sheen will be excavated using land-based equipment. Contaminated sediments are primarily within the reach that the seeps occur and within the soft unconsolidated surface sediments. This approach would remove hot spot areas to the extent practicable. Prior to excavation, debris and vegetation will be removed. During remedial design the exact slopes, methods for regrading and removal, depth, and extent of excavation will be determined. | Volume removed: 1,020 CY | Removed material will be dewatered and stabilized as needed (e.g., drying agents) and transported to an offsite disposal facility. The proper facility for disposal (i.e., Subtitle C or Subtitle D landfill) will be determined by material profiling. Trucks, train, and/or barge may be used depending on the volume of removed material in conjunction with other PAAs. During remedial design the dewatering, stabilization, and transportation methods will be determined. | Rem stabil and tra The pr C or S materia may l remov | |
| RAA-1: No Action | | | | | | | |
| RAA-2: Armored Reactive Cap | x | | | | | | |
| RAA-3: Regrade Streambank, Limited Removal Action, Upland Consolidation, and Armored Reactive Capping of Residuals | x | | x | Х | | | |
| RAA-4: Regrade Streambank, Limited Removal Action, Offsite Disposal, and Armored Reactive Capping of Residuals | х | | X | | | | |
| Notes: 1 CY | See Staff Report Section 5 Cubic Yards | | | | | | |

H:Vhorizontal distance to vertical risePAAPriority Action AreaRAARemedial Action AlternativeSFSquare Feet

Selected Remedial Action

| noved material will be dewatered and |
|--|
| ilized as needed (e.g., drying agents) |
| ansported to an offsite disposal facility. |
| proper facility for disposal (i.e., Subtitle |
| Subtitle D landfill) will be determined by |
| ial profiling. Trucks, train, and/or barge |
| be used depending on the volume of |
| oved material in conjunction with other |
| PAAs. |
| |

| х | |
|---|--|

Table 12. Area 1 Upland PAA remedial action alternatives rankings and rationale for evaluation criteria.

| | | | | | Balancing Factors ¹ | | | | Ad | ditional Factors | | Overall | | |
|---------------------|--|--|---|---|--|--|---|--|--|--|-----------------------------|---|---|---|
| RAO | Remedial Action Alternative | Protectiveness | Effectiveness 5 = high effectiveness | Long-Term Reliability 5 = high reliability | Implementability 5 = high implementability | Implementation Risk 5 = low risk | Cost 5 = low cost | Initial Total Score ² | Hot Spot Treatment 5 = likely | Green Remediation 5 = highly sustainable | Total Score ³ | Unknowns | Advantages | Disadvantages |
| None | RAA-1: No Action | | Alternative does not provide p | rotection because no action is take | n to remove, treat, or contain res | sidual contamination at the Site. | | 00010 | As a result, | Alternative 1 was not carrie | d forward o | r evaluated for criteria a | s it does not achieve | protectiveness. |
| 5, 6, 7, | B RAA-2: Impervious Surface Cap, MNA | This alternative is considered protective. The upland PAA is already essentially capped with fill material to prevent direct exposure to COCs by human and ecological receptors. An impermeable cap and stormwater treatment would reduce groundwater recharge through contaminated soil and thereby reduce the migration of creosote NAPL and impacted groundwater towards the in-water PAAs. Managing stormwater would also reduce the impact of erosion caused by stormwater runoff. | Constructing the impervious surface cap and stormwater management system would limit infiltration through contaminated soil and reduce contaminated soil and reduce contaminate surface to the in-water areas. However, the overall effectiveness is unclear as no impacted soil would be treated or removed and a cap alone would not address the connection with further upland infiltration and groundwater flow that may contribute to the migration of creosote NAPL and impacted water. Additionally, it is not evident that MNA would achieve RAOs in a reasonable time period. As evidenced by current plume extent, MNA is not effectively occurring under current conditions. All risks remain on-site, including impacts to beneficial water use. The RAA would rely on the ICs to remain effective. | Conditions should get no worse than current, and may improve with time. However, the long term reliability of an impervious cap alone to address advective mass flux has yet to be determined. This alternative is sensitive to subsurface hydrogeologic and geochemical conditions. Monitoring and surface cap maintenance will be required in perpetuity with annual inspections and periodic maintenance. | Stratigraphy may challenge technology though local experience is available. Equipment and materials are standard and readily available. Installation, inspections, and maintenance would be easy to implement. | Contaminated soil should not be encountered during construction of the impervious surface cap and/or stormwater management system. However, there is still a low potential for worker exposure to creosote (e.g., dust exposure) and an inhalation risk for vapors. Few toxic degradation products; however, long- term risks to receptors will not change rapidly, as overall process is slow. Longer-term risks to workers will be managed by ICs. | \$5,565,000 Low capital cost, some long-term O&M. | 16 | No treatment or removal of hot spots though hot spot mobility would be reduced by limiting infiltration through contaminated soils and reducing groundwater mass flux. However, it is difficult to estimate the reduction in groundwater flux with the impervious cap alone as groundwate upgradient from the site may be of influence. | This alternative relies on passive technologies and greenhouse gas emissions would only be associated with construction and maintenance. The long-term monitoring of groundwater for natural attenuation would generate a minimal volume of IDW. | 22 | Groundwater flow and COC migration dynamics in groundwater have not been well characterized. Hydraulic connection to upgradient groundwater and associated influence on contaminant migration has not been determined. | Passive and cost effective, allows for some property reuses. | Less likely to effectively reduce risks compared to other alternatives and does not address hot spots in critical source areas and may not reduce risk. |
| 3, 4, 5, 6, 7, 8 | RAA-3: Impervious Surface Cap, Hydraulic Containment, Enhanced Bioremediation, MNA | This alternative is considered protective. The cap limits infiltration, the hydraulic containment system limits contaminant migration to potential receptors, and enhanced bioremediation reduces contaminant concentrations. However, the ability of the hydraulic containment and enhanced bioremediation system to be successful has not been determined. | The actual effectiveness of the hydraulic containment and groundwater recirculation system, particularly within the fine-grained soils at the Site, would need to be determined through extensive field pilot-scale testing. Most COCs are aerobically biodegradable. However, it is not evident that enhanced bioremediation will achieve RAOs in a reasonable time period and it is unclear how effective the system would be at reducing concentrations. | This alternative uses proven technologies in highly conductive soil, which exists in the fill. Equipment and maintenance needs can easily be met. Fouling issues may impede the reliability of the hydraulic containment system. Provides treatment only while active. Progress is difficult to measure and the rate of remediation may be very slow. Treatment and system maintenance must continue indefinitely pending source mitigation. | This alternative ranked lower for implementability than RAA-2 due to the additional infrastructure necessary for the extraction and treatment system. Though specialized, the equipment for the system would be relatively easy to obtain. Local experience to guide treatment system testing and design are available. The treatment system would require significantly more O&M than passive alternatives, including a monitoring well network. | Requires long-term monitoring and maintenance of the active extraction and treatment system. Few toxic 2 degradation products. 2 Pumping may cause vertical or lateral creosote NAPL mobility. Treatment plant must operate efficiently to be successful. | \$15,097,000 Installation cost is reasonable, long term O&M costly as the extraction and treatment system would have to operate indefinitely. Relies on functional water treatment plant. | 13 | This alternative would treat hot spots through enhanced bioremediation and reduce hot spot mobility through groundwater extraction. | This alternative ranks lower than RAA-2 due to the greenhouse gas emissions associated with long-term operation of the extraction and treatment system, which uses high amounts of energy over the long-term as well as treatment- associated wastes generated. | 19 | The ability of the extraction and treatment system to reduce contamination is unknown as the following have not been determined: desorption potential of COCs from soil/TOC, nutrient requirements to support aerobic degradation, and flow rate required to provide hydraulic containment. | Contains COC migration, removes some mass from subsurface, may reduce hot spot concentrations over a long-period of time. | High cost long term, and large energy consumption, may be ineffective at mass removal due to low hydrologic conductivity in deeper aquifer and to fact that dissolved mass can only be recovered in large volumes of water. |
| 3, 4, 5, 6, 7, 8 | RAA-4: Impervious Surface Cap, MNA, Permeable Reactive Barrier | This alternative is considered protective. The cap limits infiltration and the PRB would add a further barrier to prevent migration of upland sources towards in-water locations, though would not include source area treatment. | Similar to Alternative 3, a PRB would be designed to capture nearly all groundwater COC flux to the river, as supported by initial CapSim modeling. However, creosote NAPL may migrate into the basalt. Predominant risks remain on-site in upland, including impacts to beneficial water use | Captures, contains, and treats majority of groundwater leaving source area. Reliable method to deflect and guide groundwater flow, if flow is already mostly perpendicular to wall. Cap maintenance/monitoring will be required in perpetuity. Some downgradient monitoring will be needed to assess performance of barrier wall. Possible need to change-out media (not currently included/assumed to be necessary). | This alternative is relatively implementable given the length and depth of the PRB and the surface access. Some wastes would be produced in installing the reactive media. A small monitoring well network beyond the wall would be required. Meets ARARs. | Site geology may make installation difficult. PRB may have differing effects on groundwater and lead to additional fractured flow to the river, resulting in the need for additional understanding of Site hydraulics. Potential to 3 mobilize creosote NAPL during installation. Large construction project implemented over a short time period. Some risks associated with long-term PRB performance and the need to replace adsorptive media due to media fouling and reduced permeability. | \$7,890,000 Installation cost is high. Long term O&M cost low over time, but goes on indefinitely at least until sources removed. Disposal costs for carbon may also be high, if determined to be necessary in future (not included in costs). | 17 | No treatment or removal of hot spots but is expected to reduce hot spot mobility through reduced groundwate flux associated with the cap and groundwater treatment as the water migrates through the PRB. | Larger excavated volumes than RAA-3, but the PRB is a passive system with reduced greenhouse gas emissions associated with long term O&M. | 23 | Long term impact of COCs on media breakthrough, total extent to contain plume, propensity for groundwater to flow towards creek versus to the bay. | Protects cove area interaction with groundwater through absorption within the barrier wall. Short time- period for installation. | High capital cost. No hot spot treatment or removal. |

Table 12. Area 1 Upland PAA remedial action alternatives rankings and rationale for evaluation criteria.

| | | | | | Balancing Factors | | | | Addit | ional Factors | | | Overall | |
|--------------------|--|---|---|--|---|---|--|--|---|--|-----------------------------|---|---|---|
| RAO | Remedial Action Alternative | Protectiveness | Effectiveness 5 = high effectiveness | Long-Term Reliability 5 = high reliability | Implementability 5 = high implementability | Implementation Risk 5 = Iow risk | Cost 5 = low cost | Initial Total Score ² | Hot Spot Treatment 5 = likely 5 | Green Remedy 5 = highly sustainable | Total Score ³ | Unknowns | Advantages | Disadvantages |
| 3, 4, 5 6, 7, 8 | RAA-5: Impervious Surface Cap, MNA, Impermeable Isolation Wall (NAPL Area) | This alternative is considered protective. The impermeable bentonite slurry wall would surround the inferred extent of upland creosote NAPL area and extend vertically through the contaminated zone into the upper surface of basalt bedrock. The combination of the impermeable cap and isolation wall would prevent the migration of creosote NAPL and dissolved phase constituents by sealing off the inferred extent of upland creosote NAPL from Milton Creek and Scappoose Bay. | If keyed into basalt, reduces most flux from source area. Creosote NAPL may migrate into the basalt. Possible for diffusive transport of COCS through wall. Does not remove source so some risk pathways remain on-site. however, the full upland lateral and vertical extent of creosote NAPL is not known. | Reliable method to isolate source. May cause groundwater mounding and reduce effectiveness. Strong acids, bases, salt solutions, and organic chemicals may degrade grout. Cap maintenance/monitoring will be required in perpetuity. Some long-term monitoring required for groundwater to ensure effectiveness/integrity of wall is maintained. | Assume floorless injected from surface. Good surface access. Produces additional waste due to the greater excavation required. In locations with deeper basalt bedrock, would require less common excavation equipment to key into the basalt bedrock. | Site geology may make installation difficult. Additional risk due to the larger extent of excavation required. Wall may cause groundwater mounding, resulting in the need for additional understanding of Site hydraulics. | \$8,494,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. May not address all risks due to underlying fractures as pathway through basalt. | 15 | No hot spot treatment of removal, but reduces hot spot mobility by containment. | This alternative would generate additional excavated material and take a longer amount of time to implement. However, emissions associated with long term O&M may be reduced as a result of increased reliability. | 20 | Long term impact of COCs on wall integrity, full upland lateral and vertical extent of creosote NAPL, propensity for source to move vertically into underlying fractured basalt. | Diversion of groundwater outside of source area is a straightforward way to protect bay from source of creosote NAPL. Short time- period for installation. | High capital cost. Hydrogeologic modeling has not been completed, so unknowns are high. |
| 3, 4, 5 6, 7, 8 | RAA-6: Excavation and Offsite Disposal, Impervious Surface Cap (over residuals), MNA | This alternative is considered protective. The excavation and offsite disposal of all accessible upland soil containing creosote NAPL would directly remove contamination and eliminate long-term O&M. | This alternative is the most effective as it removes the majority of hot spots and associated risks, resulting in shorter time to meet upland RAOs. | Highly reliable. Established method with few uncertainties. Long-term O&M needs are reduced or eliminated. | Uses established technologies, though the large extent of excavation reduces the implementability. Generation of significant volumes of excavated material and increase in construction time period. Would require significant truck loads to remove material or barge and/or rail cars. Dewatering is expected to be needed, and associated treatment of water. | The large volume of excavated material corresponds to increased risk associated with soil handling and increased risk to the community and environment related to 1 truck traffic emissions and the increased potential for spills during transport. The excavation depth may require shoring (e.g., soil freezing). Some uncertainty with long-term liability at landfill. | \$60,386,000. Extremely high cost to implement (disposal). Mitigates risks so no long term O&M. Relies on permitted landfill. | 13 | 5 Includes extensive hot , spot removal. | The extent of excavation is considerably larger than the other alternatives, resulting in increased greenhouse gas emissions during construction and disposal and larger volumes of waste material. | 19 | Ability to remove all creosote NAPL and potential for displacement during excavation. | Quick to implement, satisfies RAOs quickly and effectively meets hot spot rules. | Disproportionately high capital cost due to transport and disposal costs. Increased community and environmental impacts due to truck traffic. |

Notes: 1

2

3

Staff Report Sections 6.2 and 6.3 provide evaluation criteria considered for each of the balancing factors.

Initial score includes the combined scores from the 5 balancing factors described in OAR 340-122-0090(3)

The final score includes additional scores from additional preferred criteria, including hot spot treatment and green remediation.

ARARs Applicable or Relevant and Appropriate Requirements Constituents of Concern

COCs

Institutional Control IC IDW

Investigation Derived Waste MNA Monitored Natural Recovery

NAPL O&M Non-Aqueous Phase Liquid Operations and Maintenance

PAA Priority Action Area

PRB Permeable Reactive Barrier

RAA RAO Remedial Action Alternative

Remedial Action Objective

TOC Total Organic Carbon

| | Demodial Action | | | Balancing Factors ¹ | | | | | | | | | |
|---------------|--|--|---|--|--|-----------|--|---|---|-------------------------------------|--|--|--|
| RAO | Alternative | Protectiveness | | Effectiveness 5 = high effectiveness | Long-Term Reliability 5 = high reliability | | Implementability 5 = high implementability | Implementation Risk 5 = Iow risk | Cost 5 = low cost | Initial Total Score ² | | | |
| None | RAA-1: No Action | | | Alt | ernative does not provide protection because no ac | tion is t | aken to remove, treat, or contain residual conta | mination at the Site. | | | | | |
| 1a, 1b, 1c, 2 | RAA-2: Armored Reactive Cap | This alternative is considered protective. Some removal of contaminated wood debris to aid in cap installation. The cap is expected to reduce contaminant mass flux and provide a barrier to direct contact. However, this portion of the Site's shoreline is subjected to stronger river current compared to the other in-water PAAs and additional armoring will likely be required. | 2 | The cap would effectively shield contaminants from entering the river, including from any remaining contaminated wood debris. Uncertainty posed by significant amount of remaining contaminated wood debris. | Long-term reliability impacted by ability to remove wood debris prior to cap installation. Long-term monitoring and maintenance required. Differential settlement of sediment may require and gas ebullition from decomposing wood debris may impact long- term reliability. Significant amounts of contaminated wood debris left in place below the mudline. | 4 | Proven technologies and placement methods. Any specialized materials or equipment expected to be readily available. Additional on-site grading may be needed pending results of the flood rise analysis. | Implementation risk is lower than RAA-3 and RAA-4 because the reduced scope of construction and removal. Some risk associated with the removal of timber pilings and wood debris. | \$5,382,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. May not address all risks due to no hot spot treatment or removal. | 16 | | | |
| 1a, 1b, 1c, 2 | RAA-3: Nearshore Removal Action, Upland Consolidation, and Armored Reactive Capping of Residuals | This alternative is considered protective. The cap will reduce contaminant mass flux and some hot spot removal (nearshore contaminated sediment and wood debris) will take place, though some of the PAA (approximately 20%) is not accessible by land-based equipment. Uncertainty related to ability to remove wood debris due to the age and highly weathered nature of the material. Uncertainty related to contamination with depth, and is possible that removing surface material may expose greater contamination concentrations and increase risk. Material consolidated on site presents different risk. | 3 | Approximately 80% of nearshore surface hot spot material removed from PAA. However, the ability to remove wood debris source material is unclear due to the significant volume and age. It is likely that much of the wood debris would remain. Additionally, impact of removal on effectiveness is unclear due to data gaps related to contaminant concentration with depth. Removing surface material removes contamination but may expose deeper contamination of greater concentrations. Material consolidated on site requires additional institutional and engineering controls. | Long-term reliability may be improved as hot spot material is removed. However, uncertainties related to ability to remove wood debris and contamination depth profile. Additional institutional and engineering controls needed compared to off site disposal. Robust long-term monitoring and maintenance, including sediment and porewater sampling, needed as creosote NAPL-containing sediment would be left in place. | 2 | Increased construction and complexity associated with removal due to volume of buried waste. Removal of wood debris may be challenging due to non- heterogeneous subsurface. Construction issues may occur due to strong river currents. Additional permitting and stakeholder negotiations needed for on site consolidation, particularly as material would be placed within the 100-year flood plain. | Greater scope of construction, greater complexity associated with significant amount of wood debris. Potential risk associated with potentially exposing greater contamination with surface material removal. Disturbance of contaminated sediment during the removal action has a high potential for accidental release to the environment. Robust BMPs would be required for implementation. Managing material on site would reduce implementation risk to the community from truck traffic but would present longer-term risk based on the material remaining on site. | \$8,138,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. May not address all risks due to incomplete hot spot treatment or removal. | 14 | | | |
| 1a, 1b, 1c, 2 | RAA-4: Nearshore Removal Action, Offsite Disposal, and Armored Reactive Capping of Residuals | This alternative is considered protective. Same considerations as RAA-3, but material removed from site reduces risk. | 4 | Same considerations as RAA-3 except material removed from site removes associated risk. | Same considerations as RAA-3 except removing contaminated material off site improves long-term reliability. | 3 | Same considerations as RAA-3 except reduced permitting and stakeholder negotiations as contaminated material would be disposed off site. | 2 Same considerations as RAA-3 except disposing of material off site would increase implementation risk to the community and environment related to truck traffic emissions and the increased potential for spills during transport. However, off site disposal reduces long term risk by removing contamination from the site. | \$9,528,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. May not address all risks due to incomplete hot spot treatment or removal. | 15 | | | |
| 1a, 1b, 1c, 2 | RAA-5: Complete Removal and Offsite Disposal | This alternative is considered protective. The excavation and offsite disposal of all creosote NAPL-impacted sediment would directly remove contamination and reduce or eliminate long-term O&M. However, depth of contamination has not been fully characterized. | 5 | This alternative is the most effective as it removes the majority of hot spots and associated risks, resulting in shorter time to meet upland RAOs. However, depth of contamination has not been fully characterized. | Highly reliable. Established method with few uncertainties. However, depth of contamination has not been fully characterized. Long-term O&M needs are reduced or eliminated. | 1 | Uses established technologies. Large construction scope and extent of excavation reduces the implementability. Ability to remove all wood debris is unlikely, significantly impacting ability to remove all sediment. Specialized equipment to shore the excavation area. Generation of significant volumes of excavated material and increase in construction time period. Would require significant truck loads to remove material or barge and/or rail cars. Dewatering is expected to be needed, and associated treatment of water. | Greater extent and complexity of construction. The large volume of excavated material corresponds to increased risk associated with material handling, increased potential for releases outside the work area, and increased risk to the community and environment related to truck traffic emissions and the increased potential for spills during transport. Some uncertainty with long-term liability at landfill. Risk associated with removal of significant amount of wood debris. Wood debris will impede removal. | \$29,181,000. Extremely high cost to implement (disposal). Mitigates risks so reduced or eliminated long term O&M. Relies on permitted landfill. | ; 13 | | | |

Table 13. Area 1 Dock PAA remedial action alternatives rankings and rationale for evaluation criteria.

| | | Ad | Iditional Factors | | Overall | | | | | |
|---------------|--|--|--|-----------------------------|--|--|---|--|--|--|
| RAO | Remedial Action Alternative | Hot Spot Treatment 5 = likely | Green Remediation 5 = highly sustainable | Total Score ³ | Unknowns | Advantages | Disadvant | | | |
| None | RAA-1: No Action | | As a result, Alternative 1 wa | as not carried | ied forward or evaluated for criteria as it does not achieve protectiveness. | | | | | |
| 1a, 1b, 1c, 2 | RAA-2: Armored Reactive Cap | Some contamination removal associated with removal of wood and pilings. However, generally no hot spot treatment or removal, but reduces hot spot mobility. | Low amount of construction and associated emissions and minimal generation of waste, resulting in low effect on environment. | 21 | Bench and/or pilot tests needed to determine the specific materials and thicknesses needed for the cap, particularly considering the stronger river currents in this PAA. Additional characterization and cap modeling needed for pre-design investigation. Additional geotechnical investigation needed for pre-design investigation to address soft sediment and contaminated wood waste. The extent of debris and timber piles. | Established technology and materials, less construction requirements. | No hot spot treatmen | | | |
| 1a, 1b, 1c, 2 | RAA-3: Nearshore Removal Action, Upland Consolidation, and Armored Reactive Capping of Residuals | Aim to remove timber piles and approximately 80% of creosote-impacted surface sediment hot spot material. Cap reduces hot spot mobility. However, some hot spot material would remain and potentially greater concentrations could be exposed from removal of surface material. Additionally, it is unclear how successful removal of wood debris is due to the significant volume and age. | Some construction, including personnel and equipment, is required for nearshore removal action and the on site consolidation system. Waste management would result in additional impact during dewatering. However, on site consolidation would require less truck traffic and associated emissions. Opportunities for green remediation efforts, such as low sulfur fuels, etc. | 20 | The extent of hot spot material remaining after removal. The ability to remove pilings and wood debris given significant volume and age. Data gaps related to contamination with depth. Additional characterization and cap modeling needed for pre-design investigation. Additional geotechnical investigation needed for pre- design investigation to address soft sediment and contaminated wood waste. Potential permitting or stakeholder negotiations complications related to on site consolidation. | Established technology and materials. Includes some hot spot removal. Less truck traffic associated with off site disposal. | Increased constructio compared to RAA-2. Pot deeper, buried, conta removal of surface se consolidation of contar presents risk. Robust lon and maintenance, includ porewater sampling, ner NAPL-containing sedime place. | | | |
| 1a, 1b, 1c, 2 | RAA-4: Nearshore Removal Action, Offsite Disposal, and Armored Reactive Capping of Residuals | 3 Same considerations as RAA-3. | Same considerations as RAA-3 except off site disposal would require additional truck traffic and associated emissions. | 20 | Same considerations as RAA-3 except those related to on site consolidation. | Established technology and materials. Includes some hot spot removal. Removal of contaminated material from site. | Increased constructio compared to RAA-2. Pot deeper contamination. traffic required for off | | | |
| 1a, 1b, 1c, 2 | RAA-5: Complete Removal and Offsite Disposal | 5 Includes extensive hot spot removal. | The extent of excavation is considerably larger than the other alternatives, resulting in increased greenhouse gas emissions during construction and disposal and larger volumes of waste material. Higher energy consumption related to longer time frame required for implementation. | 19 | The ability to remove pilings and wood debris given significant volume and age. Data gaps related to contamination with depth. Additional characterization needed for pre-design investigation. Additional geotechnical investigation needed for pre-design investigation to address soft sediment and contaminated wood waste. | Satisfies RAOs quickly and effectively meets hot spot rules. | Increased constructio compared to RAA-2, RA Potential exposur contamination. Disprop capital cost due to trans costs. Increased co environmental impacts d | | | |

Notes:

| 1 | Staff Report Sections 6.2 and 6.3 provide evaluation criteria considered for each of the balancing factors. | O&M Operations and Mainten |
|------|---|-----------------------------|
| 2 | Initial score includes the combined scores from the 5 balancing factors described in OAR 340-122- 0090(3) | PAA Priority Action Area |
| 3 | The final score includes additional scores from additional preferred criteria, including hot spot treatment and green remediation. | RAA Remedial Action Alterna |
| BMP | Best Management Practice | RAO Remedial Action Objecti |
| NAPL | Non-aqueous Phase Liquid | % Percent |
| | | |

| ages | |
|--|--|
| | |
| nt or removal. | |
| n requirements cential exposure of amination with diment. On site ninated material g-term monitoring ling sediment and eded as creosote ent would be left in | |
| n requirements ential exposure of Increased truck f site disposal. | |
| n requirements NA-3, and RAA-4. e of deeper portionately high port and disposal mmunity and lue to truck traffic. | |

nance

ative

tive

Table 14. Area 2 Dock PAA remedial action alternatives rankings and rationale for evaluation criteria.

| RAO | Remedial Action Alternative | | Balancing Factors ¹ | | | | | |
|---------------|---|--|---|---|--|---|---|----------------------|
| | | Remedial Action Alternative Protectiven | Protectiveness | Effectiveness 5 = high effectiveness | Long-Term Reliability 5 = high reliability | Implementability 5 = high implementability | Implementation Risk 5 = Iow risk | Cost 5 = low cost |
| None | RAA-1: No Action | Alternative does not provide protection because no action is taken to remove, treat, or contain residual contamination at the Site. | | | | | | |
| 1a, 1b, 1c, 2 | RAA-2: Armored Reactive Cap | This alternative is considered protective. Some removal of contaminated wood debris to aid in cap installation. The cap is expected to reduce contaminant mass flux and provide a barrier to direct contact. However, this portion of the Site's shoreline may be subjected to stronger river current compared to the other in-water PAAs and additional armoring may be required. | The cap would effectively shield contaminants from entering the river, including from any remaining contaminated wood debris. Uncertainty in the amount of remaining contaminated wood debris. | 2 Long-term reliability impacted by ability to remove wood debris prior to cap installation. Long-term monitoring and maintenance required. | Proven technologies and placement 4 methods. Any specialized materials or equipment expected to be readily available. | Implementation risk is lower than RAA-3 and RAA-4 because the reduced scope of construction and removal. | \$1,309,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. May not address all risks due to no hot spot treatment or removal. | |
| 1a, 1b, 1c, 2 | RAA-3: Nearshore Removal Action, Upland Consolidation, and ENR Capping of Residuals | This alternative is considered protective. The cap will reduce contaminant mass flux and hot spot removal (approximately 84%) will take place. However, some of the PAA (approximately 16%) is not accessible by land based equipment. Material consolidated on site presents different risk. | Approximately 84% of hot spot material removed from PAA, which would greatly increase the effectiveness of the remedial action. Material consolidated on site requires additional institutional and engineering controls. Uncertainty related to the effectiveness of a sand-only cap in an area with stronger river currents. | Improved long-term reliability as large percentage of hot spot material is removed. Additional institutional and engineering controls needed compared to off site disposal. Long- term monitoring and maintenance needed. Uncertainty related to the long-term reliability of a sand-only cap in an area with stronger river currents. | Increased construction and complexity associated with streambank regrading and removal action. River current control, robust BMPs needed to construct in area of stronger river currents. Additional permitting and stakeholder negotiations needed for on site consolidation, particularly as material would be placed within the 100-year flood plain. | Disturbance of contaminated sediment during the removal action has a higher potential for accidental release to the environment. Managing material on site would reduce implementation risk to the community from truck traffic but would present longer-term risk based on the material remaining on site. | \$1,596,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. May not address all risks due to incomplete hot spot treatment or removal. | . 14 |
| 1a, 1b, 1c, 2 | RAA-4: Nearshore Removal Action, Offsite Disposal, and Capping of Residuals | This alternative is considered protective. Same considerations as RAA-3, but material removed from site reduces risk. | Same considerations as RAA-3 except 4 material removed from site removes associated risk. | Same considerations as RAA-3 except 4 removing contaminated material off site improves long-term reliability. | 3 Same considerations as RAA-3 except reduced permitting and stakeholder negotiations as contaminated material would be disposed off site. | Same considerations as RAA-3 except disposing of material off site would increase implementation risk to the community and environment related to truck traffic emissions and the increased potential for spills during transport. However, off site disposal reduces long term risk by removing contamination from the site. | \$1,604,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. May not address all risks due to incomplete hot spot treatment or removal. | 16 |
Table 14. Area 2 Dock PAA remedial action alternatives ran

| | | Additional Factors | | | | | Overall | | | | |
|---------------|---|--------------------|--|---|--|-----------------------------|---|--|---|--|--|
| RAO | Remedial Action Alternative | | Hot Spot Treatment 5 = likely | | Green Remediation 5 = highly sustainable | Total Score ³ | Unknowns | Advantages | I | | |
| None | RAA-1: No Action | | | | As a result, Alternative 1 was not | carried forwa | d or evaluated for criteria as it does not achieve protectiveness. | | | | |
| 1a, 1b, 1c, 2 | RAA-2: Armored Reactive Cap | 1 | Some contamination removal associated with removal of wood and pilings. However, generally no hot spot treatment or removal, but reduces hot spot mobility. | 4 | Minimal construction and associated emissions and low volume of waste generated, resulting in lower overall impacts to the environment. | 21 | Bench and/or pilot tests needed to determine the specific materials and thicknesses needed for the cap, particularly considering the stronger river currents in this PAA. The extent of debris and timber piles. | Established technology and materials, less construction requirements. | No hot s | | |
| 1a, 1b, 1c, 2 | RAA-3: Nearshore Removal Action, Upland Consolidation, and ENR Capping of Residuals | 4 | Timber piles and approximately 84% of creosote-impacted surface sediment hot spot material will be removed. Cap reduces hot spot mobility. However, some hot spot material would remain. | 3 | Short distances for truck traffic for on-site consolidation, resulting in low emissions and overall low impacts to the environment | 21 | The extent of hot spot material remaining after removal. Potential permitting or stakeholder negotiations complications related to on site consolidation. Ability of sand-only cap to effectively remain in place in an area with stronger river currents. | Established technology and materials. Includes some hot spot removal. Less truck traffic associated with off site disposal. | Increased compared to of contamir | | |
| 1a, 1b, 1c, 2 | RAA-4: Nearshore Removal Action, Offsite Disposal, and Capping of Residuals | 4 | Same considerations as RAA-3. | 2 | Same considerations as RAA-3 except off site disposal would require additional truck traffic and associated emissions. | 22 | The extent of hot spot material remaining after removal. | Established technology and materials. Includes some hot spot removal. Removal of contaminated material from site. | Increased compared to requir | | |

Notes:

1 2

3

Staff Report Sections 6.2 and 6.3 provide evaluation criteria considered for each of the balancing factors.

Initial score includes the combined scores from the 5 balancing factors described in OAR 340-122-0090(3)

The final score includes additional scores from additional preferred criteria, including hot spot treatment and green remediation.

- BMPBest Management PracticeENREnhanced Natural RecoveryO&MOperations and Maintenance
- PAA Priority Action Area
- RAA Remedial Action Alternative
- RAO Remedial Action Objective
- % Percent

| Disadvantages |
|--|
| |
| spot treatment or removal. |
| d construction requirements RAA-2. On site consolidation nated material presents risk. |
| l construction requirements RAA-2. Increased truck traffic red for off site disposal. |

| | | | Balancing Factors ¹ | | | | | | | | | |
|---------------------|--|--|--|--|--|--|--|------|--|--|--|--|
| RAO | Remedial Action Alternative | Alternative Protectiveness Effectiveness Long-Term Reliability Implementability 5 = high effectiveness 5 = high reliability 5 = high implementability | | Implementation Risk 5 = low risk | Cost 5 = low cost | Initial Total Score ² | | | | | | |
| None | RAA-1: No Action | | Alternati | ve does not provide protection because no action is | s taken to remove, treat, or contain residual cont | amination at the Site. | | | | | | |
| 1-4 | RAA-2: Armored Reactive Cap | This alternative is considered protective. An amended cap is expected to reduce contaminant mass flux from sediment to surface water and provide a barrier to direct contact. Sediment porewater and surface water quality outside the cap boundary is expected to continue naturally recovering as new, cleaner sediment is deposited in the Cove (0.2 to 1.4 in/yr.). | 2 The cap would effectively shield contaminants from entering the river, supported by preliminary CapSim modeling. No impacted material would be removed, resulting in less certainty in the effectiveness. | Preliminary CapSim modeling indicates long- term cap effectiveness (500 years). The Cove's off-channel backwater setting provides protection against wake and wind driven waves. New sediment is expected to accumulate over the in-water portions of the cap. Long-term monitoring and maintenance required. | Proven technologies and placement methods. Potential for differential settlement below an armored reactive cap can be addressed with specialized capping materials and placement techniques as developed at other sites with similar conditions. Materials and equipment are expected to be available in the Pacific Northwest. On-site grading needed to maintain the flood-carrying capacity of the waterway if the results of the flood rise analysis exceed the City of St. Helen's threshold. Mitigation needed to offset the loss of intertidal shallow water habitat. | Implementation risk is lower than RAA-2, RAA-3, and RAA-5 because there would be no in-water removal of sediment. 4 Detential risks associated with differential settlement below the cap can be mitigated using capping materials that evenly distribute the weight of armoring (e.g., marine mattress). | \$3,551,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. May not address all risks due to no hot spot treatment or removal. | 16 | | | | |
| 1-4 | RAA-3: Riverbank Restoration, Nearshore Removal Action, Upland Consolidation, and Reactive Capping of Residuals | This alternative is considered protective. The cap will reduce contaminant mass flux and some hot spot removal will take place. Improved cap performance with streambank regrading. Uncertainty related to contamination with depth, and is possible that removing surface material may expose greater contamination concentrations and increase risk. Material left on site presents long- term risk. | Regrading facilitates improved access by land-based equipment, some hot spot removal (up to 90% contaminated surface sediment), more controlled placement of cap material, reduced erosional forces imposed on the cap. The temporary cofferdam allows an extended excavation area. Impact of removal on effectiveness is unclear due to data gaps related to contaminant concentration with depth. Removing surface material removes contamination but may expose deeper contamination of greater concentrations. Material consolidated on site requires additional institutional and engineering controls. | Constructing the cap on a flatter streambank is more reliable. Hot spot removal, though there are uncertainties related to the contamination depth profile and total volume that may remain in the stranded wedge. Additional institutional and engineering controls needed for on-site containment site disposal. Future water levels expected to rise, so long-term reliability may be affected. Long-term monitoring and maintenance needed. | 2 Local resources available for streambank regrading and removal action. Additional permitting and stakeholder negotiations needed for on site consolidation, particularly as material would be placed within the 100-year flood plain. | Flattening the streambank would improve cap constructability and construction worker safety. Greater scope of construction. Potential risk associated with potentially exposing greater contamination with surface material removal. Disturbance of contaminated sediment during the removal action has a potential for accidental release to the environment. Managing material on site would reduce implementation risk to the community from truck traffic but would present longer-term risk based on the material remaining on site. | \$4,836,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. Most hot spot material removed, but may not address all risks due to incomplete hot spot treatment or removal. | . 14 | | | | |
| 1a, 1b, 1c, 2, 3, 4 | RAA-4: Riverbank Restoration, Nearshore Removal Action, Offsite Disposal, and Reactive Capping of Residuals | This alternative is considered protective. Same considerations as RAA-3, but material removed from site reduces risk. | Same considerations as RAA-3 except 4 material removed from site removes associated risk. | Constructing the cap on a flatter streambank is more reliable. Hot spot removal, though there are uncertainties related to the contamination depth profile and total volume that may remain in the stranded wedge. Additional institutional and engineering controls needed for cap. Long-term monitoring and maintenance needed. | 3 Same considerations as RAA-3 except reduced permitting and stakeholder negotiations as contaminated material would be disposed off site. | Same considerations as RAA-3 except disposing of material off site would increase implementation risk to the community and environment related to truck traffic emissions and the increased potential for spills during transport. However, off site disposal reduces long term risk by removing contamination from the site. | \$5,984,000 Same considerations as RAA-3. | 15 | | | | |
| 1a, 1b, 1c, 2, 3, 4 | RAA-5: Complete Removal and Offsite Disposal | This alternative is considered protective. The excavation and offsite disposal of all creosote NAPL-impacted sediment would directly remove contamination and reduce or eliminate long-term O&M. However, depth of contamination has not been fully characterized. | This alternative is the most effective as it removes the majority of hot spots and associated risks, resulting in shorter time to meet upland RAOs. However, depth of contamination has not been fully characterized. | Highly reliable. Established method with few uncertainties. However, depth of contamination has not been fully characterized. Long-term O&M needs are reduced or eliminated. | Uses established technologies. Large construction scope and extent of excavation reduces the implementability. Specialized equipment to shore the excavation area. Generation of significant volumes of excavated material and increase in construction time period. Would require significant truck loads to remove material or barge and/or rail cars. Dewatering is expected to be needed, and associated treatment of water. | Greater extent and complexity of construction. The large volume of excavated material corresponds to increased risk associated with material handling, increased potential for releases outside the work area, and increased risk to the community and environment related to truck traffic emissions and the increased potential for spills during transport. Some uncertainty with long- term liability at landfill. | \$17,606,000. Extremely high cost to implement (disposal). Mitigates risks so reduced or eliminated long term O&M. Relies on permitted landfill. | 13 | | | | |

Table 15. Cove Area PAA remedial action alternatives rankir Table 15. Cove Area PAA remedial action alternatives rankings and rationale for evaluation criteria¹.

| | | | Addit | ional | Factors | Overall | | | | |
|---------------------|--|--|--|-------|---|-----------------------------|--|--|--|--|
| RAO | Remedial Action Alternative | | Hot Spot Treatment 5 = likely | | Green Remediation 5 = highly sustainable | Total Score ³ | Unknowns | Advantages | Di | |
| None | RAA-1: No Action | As a result, Alternative 1 was not carried forward or ev | | | or evaluated for criteria as it does not achieve | protectiveness. | | | | |
| 1-4 | RAA-2: Armored Reactive Cap | 1 | Riverbank regrading limited to slope stabilization and is not expected to include removal of a significant volume of soil within the riverbank seepage area. Thus no hot spot treatment or removal. | 4 | Construction and associated emissions are low due to small scope of work. No expected generation of waste decreases effects on the environment. | 21 | Bench and/or pilot tests needed to determine the specific materials and thicknesses needed for the cap. The extent of debris and timber piles. | Established technology and materials, less construction requirements. | No hot spo | |
| 1-4 | RAA-3: Riverbank Restoration, Nearshore Removal Action, Upland Consolidation, and Reactive Capping of Residuals | 4 | Up to 90% of surface sediment currently exhibiting a moderate to heavy petroleum sheen is expected to be removed with the remaining 10% beyond the reach of land- based equipment, even with the use of a temporary cofferdam system. Potentially greater concentrations could be exposed from removal of surface material. Cap reduces remaining hot spot mobility. | 3 | Construction for streambank regrading and the on site consolidation system will result in emissions. However, on site consolidation would require less truck traffic and associated emissions. | 21 | The extent of hot spot material remaining after removal. Data gaps related to contamination with depth. Additional characterization needed for remedial design Potential permitting or stakeholder negotiations complications related to on site consolidation | Established technology and materials. Includes some hot spot removal. Better cap integrity. Less truck traffic associated with off site disposal. | Increased co compared to R deeper co consolidation F | |
| 1a, 1b, 1c, 2, 3, 4 | RAA-4: Riverbank Restoration, Nearshore Removal Action, Offsite Disposal, and Reactive Capping of Residuals | 4 | Same considerations as RAA-3. | 2 | Additional construction is required compared to RAA-2 for streambank regrading. Off site disposal would require additional truck traffic and associated emissions. | 21 | The extent of hot spot material remaining after removal. | Established technology and materials. Includes some hot spot removal. Better cap integrity. Removal of contaminated material from site. | Increased increased emi | |
| 1a, 1b, 1c, 2, 3, 4 | RAA-5: Complete Removal and Offsite Disposal | 5 | Includes extensive hot spot removal. | 1 | The extent of excavation is considerably larger than the other alternatives, resulting in increased greenhouse gas emissions during construction and disposal and larger volumes of waste material. Higher energy consumption related to longer-time frame required for implementation. | 19 | Ability to remove all creosote NAPL and potential for displacement during excavation | Satisfies RAOs quickly and effectively meets hot spot rules. | Increased co compared to F Potential contamination capital cost du costs. Incr environmental | |

Notes:

| 1 | Staff Report Sections 6.2 and 6.3 provide evaluation criteria considered for each of the balancing factors. | O&M | Operations and |
|-------|--|-----|-------------------|
| 2 | Initial score includes the combined scores from the 5 balancing factors described in OAR 340-122-0090(3) | PAA | Priority Action A |
| 3 | The final score includes additional scores from additional preferred criteria, including hot spot treatment and green remediation. | RAA | Remedial Action |
| in/yr | inches per year | RAO | Remedial Action |
| NAPL | Non-aqueous Phase Liquid | % | percent |

| isadvantages |
|--|
| |
| ot treatment or removal. |
| construction requirements RAA-2. Potential exposure of contamination. On site n of contaminated material presents risk. |
| l truck traffic resulting in nissions required for off site disposal. |
| construction requirements RAA-2, RAA-3, and RAA-4. al exposure of deeper on. Disproportionately-high ue to transport and disposal reased community and I impacts due to truck traffic. |
| nd Maintenance |

- ion Alternative
- ion Objective

| | | | Balancing Factors ¹ | | | | | | | | |
|---------------------|---|--|--|--|--|---|--|--|--|--|--|
| RAO | Remedial Action Alternative | Protectiveness | Effectiveness 5 = high effectiveness | Long-Term Reliability 5 = high reliability | Implementability 5 = high implementability | Implementation Risk 5 = low risk | Cost 5 = low cost | Initial Total Score ² | | | |
| None | RAA-1: No Action | | A | Iternative does not provide protection because no action is ta | aken to remove, treat, or contain residual con | tamination at the Site. | | | | | |
| 1a, 1b, 1c, 2, 3, 4 | RAA-2: Armored Reactive Cap | This alternative is considered protective. The cap over creosote-impacted groundwater seeps is expected to reduce contaminant mass flux and provide a barrier to direct contact. | The cap would effectively shield contaminants from entering the creek. However, without regrading the cap may be less stable. No impacted material would be removed, resulting in less certainty in the effectiveness. | The long-term reliability of the armoring to protect the cap when Milton Creek is subject to peak discharge event has yet to be determined. Long-term cap inspection and maintenance, possibly greater maintenance required than alternatives including streambank regrading. | Established technology, common materials. Logistical challenges associated with constructing a reactive cap along the outside bend of a steep channelized streambank. | 4 Less intensive construction activities, but the steepness of the streambank and concerns about stability represents higher risk to construction workers. | \$898,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. May not address all risks due to no hot spot treatment or removal. | s : 15 t | | | |
| 1a, 1b, 1c, 2, 3, 4 | RAA-3: Regrade Streambank, Limited Removal Action, Upland Consolidation, and Armored Reactive Capping of Residuals | This alternative is considered protective. The cap will reduce contaminant mass flux to the creek and some hot spot removal will take place. Improved cap performance with streambank regrading. Material left on site presents different risk. | Flattening the steep east bank of the creek facilitates some hot spot removal, more controlled placement of cap material, and reduced erosional forces imposed on the cap. Material consolidated on site requires additional institutional and engineering controls. | Constructing the cap on a flatter streambank more reliable. Additional institutional and engineering controls needed compared to off site disposal. Long-term monitoring and maintenance needed. | Established technology, common materials. Increased construction associated with streambank regrading compared to RAA-2. Additional permitting and stakeholder negotiations needed for on site consolidation, particularly as material would be placed within the 100-year flood plain. | Flattening the streambank would improve cap constructability and construction worker safety. Managing material on site would reduce implementation risk to the community from truck traffic but would present longer-term risk based on the material remaining on site. | \$1,146,000 Installation cost is high, long term O&M costs low. Monitoring goes on indefinitely. Some hot spot material removed, but may not address all risks due to incomplete hot spot treatment or removal. | 16 | | | |
| 1a, 1b, 1c, 2, 3, 4 | RAA-4: Regrade Streambank, Limited Removal Action, Offsite Disposal, and Armored Reactive Capping of Residuals | Same considerations as RAA-3 except material removed from site removes associated risk. | Same considerations as RAA-3 except 4 material removed from site removes associated risk. | 4 Constructing the cap on a flatter streambank more reliable. Removing contaminated material off site improves long-term reliability. Long-term monitoring and maintenance needed. | Same considerations as RAA-3 except reduced permitting and stakeholder negotiations as contaminated material would be disposed off site. | Flattening the streambank would improve cap constructability and construction worker safety. Disposing of material off site would increase implementation risk to the community and environment related to truck traffic emissions and the increased potential for spills during transport. However, off site disposal reduces long term risk by removing contamination from the site. | 2 \$1,216,000 Same considerations as RAA-3. | 17 | | | |

Table 16. Upper Milton Creek PAA remedial action alternativ

| | | | Ac | Iditio | nal Factors | | Overall | | |
|---------------------|---|---|--|--------|--|-----------------------------|---|--|---|
| RAO | Remedial Action Alternative | | Hot Spot Treatment 5 = likely | | Green Remediation 5 = highly sustainable | Total Score ³ | Unknowns | Advantages | Disa |
| None | RAA-1: No Action | | As a result, Alternative 1 was not carried forw | | | carried forwa | rd or evaluated for criteria as it does not achie | ve protectiveness. | |
| 1a, 1b, 1c, 2, 3, 4 | RAA-2: Armored Reactive Cap | 1 | No hot spot treatment or removal, but reduces hot spot mobility. | 4 | Compared to RAA-3 and RAA-4 the extent of construction is less, corresponding to less greenhouse gas emissions. | 20 | The impacts of peak flow in Milton Creek and lack of streambank regrading on cap integrity. | Established technology and materials, less construction requirements. | No hot spot tre certainty |
| 1a, 1b, 1c, 2, 3, 4 | RAA-3: Regrade Streambank, Limited Removal Action, Upland Consolidation, and Armored Reactive Capping of Residuals | 2 | Streambank regrading allows the removal of some hot spot material. However, much of the hot spot contamination would remain. | 3 | Additional construction is required compared to RAA-2 for streambank regrading and the on site consolidation system. However, on site consolidation would require less truck traffic and associated emissions. | 21 | The extent of hot spot material remaining after removal. Potential permitting or stakeholder negotiations complications related to on site consolidation | Established technology and materials. Includes some hot spot removal. Better cap integrity. Less truck traffic associated with off site disposal. | Increased con compared to RA of contaminate |
| 1a, 1b, 1c, 2, 3, 4 | RAA-4: Regrade Streambank, Limited Removal Action, Offsite Disposal, and Armored Reactive Capping of Residuals | 2 | Streambank regrading allows the removal of some hot spot material. However, much of the hot spot contamination would remain. | 2 | Additional construction is required compared to RAA-2 for streambank regrading. Off site disposal would require additional truck traffic and associated emissions. | 21 | The extent of hot spot material remaining after removal. | Established technology and materials. Includes some hot spot removal. Better cap integrity. Removal of contaminated material from site. | Increased con compared to RA/ required fo |
| | | | Notes: | | , | 1 | • | | |
| | | | 1 | Staf | Report Sections 6.2 and 6.3 provide evalu | ation criteria | a considered for each of the balancing fac | ctors. | |

Initial score includes the combined scores from the 5 balancing factors described in OAR 340-122-0090(3)

The final score includes additional scores from additional preferred criteria, including hot spot treatment and green remediation.

Operations and Maintenance

PAA Priority Action Area

2

3

O&M

RAO

RAA Remedial Action Alternative

Remedial Action Objective

| advantages |
|--|
| |
| atment or removal, less y of cap integrity. |
| nstruction requirements A-2. On site consolidation d material presents risk. |
| nstruction requirements A-2. Increased truck traffic or off site disposal. |

Table 17. Summary of evaluation criteria scoring for each remedial action alternative for all PAAs.

| ΡΑΑ | RAA | Protective- ness | Effective- ness | Long-Term Reliability | Implement- ability | Implement- ation Risk | Cost | Initial Score | Hot Spot Treatment | Green Remediation | Final Score |
|---------------|---|---------------------|--|--------------------------|-----------------------|--------------------------|-----------------------|------------------|-----------------------|----------------------|----------------|
| | RAA-1: No Action | No | | | | Not furth | er evaluated as not p | rotective. | | | |
| | RAA-2: Impervious Surface Cap, MNA | Yes | 2 | 2 | 4 | 4 | 4 | 16 | 2 | 4 | 22 |
| | RAA-3: Impervious Surface Cap, Hydraulic Containment, Enhanced Bioremediation | Yes | 4 | 3 | 2 | 2 | 2 | 13 | 4 | 2 | 19 |
| Area 1 Upland | RAA-4: Impervious Surface Cap, MNA, Permeable Reactive Barrier | Yes | 4 | 4 | 3 | 3 | 3 | 17 | 3 | 3 | 23 |
| | RAA-5: Impervious Surface Cap, MNA, Impermeable Isolation Wall (NAPL Area) | Yes | 4 | 4 | 2 | 2 | 3 | 15 | 3 | 2 | 20 |
| | RAA-6: Excavation and Offsite Disposal, Impervious Surface Cap (over residuals), MNA | Yes | 5 | 5 | 1 | 1 | 1 | 13 | 5 | 1 | 19 |
| | RAA-1: No Action | No | | | | Not furth | er evaluated as not p | rotective. | | | |
| | RAA-2: Installation of Armored Reactive Cap | Yes | 2 | 2 | 4 | 4 | 4 | 16 | 1 | 4 | 21 |
| Area 1 Dock | RAA-3: Nearshore Removal Action, Upland Consolidation, and Armored Reactive Capping of Residuals | Yes | 3 | 3 | 2 | 3 | 3 | 14 | 3 | 3 | 20 |
| | RAA-4: Nearshore Removal Action, Offsite Disposal, and Armored Reactive Capping of Residuals | Yes | 4 | 4 | 3 | 2 | 2 | 15 | 3 | 2 | 20 |
| | RAA-5: Complete Removal and Offsite Disposal | Yes | 5 | 5 | 1 | 1 | 1 | 13 | 5 | 1 | 19 |
| | RAA-1: No Action | No | | | | Not furth | er evaluated as not p | rotective. | | | |
| | RAA-2: Installation of Armored Reactive Cap | Yes | 2 | 2 | 4 | 4 | 4 | 16 | 1 | 4 | 21 |
| Area 2 Dock | RAA-3: Nearshore Removal Action, Upland Consolidation, and Capping of Residuals | Yes | 3 | 3 | 2 | 3 | 3 | 14 | 4 | 3 | 21 |
| | RAA-4: Nearshore Removal Action, Offsite Disposal, and Capping of Residuals | Yes | 4 | 4 | 3 | 2 | 3 | 16 | 4 | 2 | 22 |
| | RAA-1: No Action | No | Not further evaluated as not protective. | | | | | | | | |
| | RAA-2: Installation of Armored Reactive Cap | Yes | 2 | 2 | 4 | 4 | 4 | 16 | 1 | 4 | 21 |
| Cove Area | RAA-3: Riverbank Restoration, Nearshore Removal Action, Upland Consolidation, and Reactive Capping of Residuals | Yes | 3 | 3 | 2 | 3 | 3 | 14 | 4 | 3 | 21 |
| | RAA-4: Riverbank Restoration, Nearshore Removal Action, Offsite Disposal, and Reactive Capping of Residuals | Yes | 4 | 4 | 3 | 2 | 2 | 15 | 4 | 2 | 21 |
| | RAA-5: Complete Removal and Offsite Disposal | Yes | 5 | 5 | 1 | 1 | 1 | 13 | 5 | 1 | 19 |
| | RAA-1: No Action | No | | | | Not furth | er evaluated as not p | rotective. | | | |
| Upper Milton | RAA-2: Installation of Armored Reactive Cap | Yes | 2 | 2 | 3 | 4 | 4 | 15 | 1 | 4 | 20 |
| Creek | RAA-3: Regrade Streambank, Limited Removal Action, Upland Consolidation, and Armored Reactive Capping of Residuals | Yes | 3 | 3 | 3 | 4 | 3 | 16 | 2 | 3 | 21 |
| | RAA-4: Regrade Streambank, Limited Removal Action, Offsite Disposal, and Armored Reactive Capping of Residuals | Yes | 4 | 4 | 4 | 3 | 2 | 17 | 2 | 2 | 21 |

Notes:

RAA Remedial Action Alternative PAA

Priority Action Area

Selected Remedial Action









| | | | | L |
|--|---|--------------------------|---|----------------|
| | | | | Vent Burner |
| Mil | ton Creek Logging Rollrood | Line and and and a state | Pond | |
| WILLIAM MILLIAM | Retort House Pump House Retorts Retorts | Transfer Elo | Doc | X |
| Shorelling | - - | | | SCAPPOOSE |
| NOTE: The locations of all features shown are approximate. Site features from GeoEngineers Figures circa 1998. | CLIENT: PORT OF ST. HEL | ENS | DWN BY: SD CHK'D BY: MR | PROJECT: FC |
| | AMEC 7376 S.W. Durham Road Portland, OR, U.S.A. 97224 | amec | DATUM: NAD83 PROJECTION: OR SP N. Ft. SCALE: 1 inch = 125 feet | TITLE: 1919 |

Page 191 of 240





| CLIENT: PORT OF ST. HEL | ENS | DWN BY: SD CHK'D BY: MR | PROJECT: | FORMER POPE & TALBOT WOOD-TREATING SITE | DATE: APRIL 2014 PROJECT NO.: 9-61M-112202.09 |
|---|------|---------------------------------------|----------|--|--|
| AMEC 7376 S.W. Durham Road Portland, OR, U.S.A. 97224 | amec | PROJECTION: SCALE: NOT TO SCALE | TITLE: | 1929 | rev. no.: Figure no.: 5 |



Page 193 of 240

| | 8-inch C Pipalinc | (Above Grou |
|-----------------------|----------------------|---------------------------------|
| veyors House | \checkmark | |
| POOSE BAY | MULTY | OMAH |
| , 0 | 62.5 125 | 250' |
| RMER POPE & TALBOT | | DATE: APRIL 2014 |
| VOOD-TREATING SITE | | PROJECT NO.: 9-61M-112202.09 |
| 3 SITE LAYOUT DIAGRAM | | FIGURE NO.: 6 |





| <:\11000\11200\1 | 1220\112202\DWG\ | 2014 SRI | Report\Figure A-5 | - 1948 Aerial.mxd |
|------------------|------------------|----------|-------------------|-------------------|





Page 197 of 240





| all features shown are approximate. Site features from GeoEngineers Figures circa 1998. | PORT OF ST. HEL | ENS | SD CHK'D BY: MR | FOR |
|---|---|------|---|--------|
| | AMEC 7376 S.W. Durham Road Portland, OR, U.S.A. 97224 | amec | NAD83 PROJECTION: OR SP N. Ft. SCALE: 1 inch = 125 feet | TITLE: |

Page 199 of 240



Page 200 of 240



| PORT OF ST. HELENS | | DWN BY: SD CHK'D BY: MR | PROJECT: |)= \ |
|---|------|--|----------|---------|
| AMEC 7376 S.W. Durham Road Portland, OR, U.S.A. 97224 | amec | NAD83 PROJECTION: OR SP N. Ft. SCALE: | TITLE: | |

Page 201 of 240

| | | and a state of the | 8 5 | 10 |
|--|---|--|--|----------|
| | | 100 | 17.76 | |
| | | and the | | |
| | William Grads | allroad | | |
| | Office/Shop | - Pole Peeling Plant | | |
| | | Dock Remnants | 3 | |
| | , B B O O | 5 8 8 | * * | |
| | \$ [©] | and the second se | 6 5 6 | |
| | | | © , | 1 Par |
| NOTE: The locations of all features shown are approximate. Site features from GeoEngineers Figures circa 1998. | CLIENT: | | DWN BY: PROJECT: | 2 |
| | PORT OF ST. HEI | LENS | | FOI W |
| | AMEC 7376 S.W. Durham Road Portland, OR, U.S.A. 97224 | amec | NAD83 TITLE: PROJECTION: OR SP N. Ft. SCALE: 1 inch = 500 feet | |





| Daga | 202 | ~f | 240 |
|------|-----|----|-----|
| Page | 203 | OI | 240 |

| ORMER POPE & TALBOT WOOD-TREATING SITE | DATE: APRIL 2014 PROJECT NO.: 9-61M-112202.09 |
|---|--|
| 2013 | REV. NO.: FIGURE NO.: 16 |

SCALE: NOT TO SCALE

| | DATE: |
|--|---|
| RMER POPE & TALBOT | DATE: APRIL 2014 |
| RMER POPE & TALBOT VOOD-TREATING SITE | DATE: APRIL 2014 PROJECT NO.: |
| RMER POPE & TALBOT VOOD-TREATING SITE | DATE: APRIL 2014 PROJECT NO.: 9-61M-112202.09 |
| RMER POPE & TALBOT VOOD-TREATING SITE | DATE: APRIL 2014 PROJECT NO.: 9-61M-112202.09 REV. NO.: |



K:\11000\11220\11220\112203\dwg_SRI_Report_2019\4-2 - Cross Section C-C'.dwg - - - May. 28, 2019 7:00am - brian.johnson



LEGEND:

- Existing Monitoring Well
- Direct-Push Boring
- 🛦 Seep

March 2011 Phase 1 TarGOST Boring:

- No Petroleum Sheen or NAPL Signal
- + Petroleum Sheen Slight
- Petroleum Sheen Moderate to Heavy

July 2011 Phase 2 Sediment Boring:

- No Petroleum Sheen or NAPL Observation
- Petroleum Sheen Slight
- Petroleum Sheen Moderate to Heavy (NAPL)
- Visible Creosote Sheen Observed in 0 Surface Sediments (2010, 2011, 2012)

July/October 2012 Sampling Event:

Porewater Sample Location

- Sheen Degree Unknown
- No Sheen
- ♦ Petroleum Sheen Slight
- Petroleum Sheen Moderate to Heavy

Limited Sediment Investigation (LSI) Sample Location

- Sheen Degree Unknown
- No Sheen
- Petroleum Sheen Slight
- Petroleum Sheen Moderate to Heavy

October 2013 Dock Area Investigation:

- ▼ Hand Auger without NAPL Observation
- Pothole without NAPL Observation
- Pothole with NAPL Observation
- DAI Boring without NAPL Observation
- DAI Boring with NAPL Observation
- Phase 3 Sampling Location with Surface Water Cage
- Phase 3 Bulk Sediment and Porewater Sample Location
- ----- Former Site Feature
- Inferred Extent of Upland NAPL
- Area of Intermittent Groundwater/NAPL Seeps
- Approximate Upland Site Boundary (Ordinary High Water 14 ft. NAVD 88)
- ---- Cross Section Location







Page 207 of 240



K:\11000\11200\11220\112203\dwg_SRI_Report_2019\4-2 - Cross Section C-C'.dwg - - - May. 28, 2019 7:00am - brian.johnson

| Ma Cascadia | PROJECT NUMBER | 0034-001-005 | FIGURE |
|-----------------|-------------------|--------------|--------|
| Associates, LLC | DECE | MBER 2019 | 21 |



Page 209 of 240





K:\11000\11220\112203\dwg_SRI_Report_2019\4-2 - Cross Section C-C'.dwg - - - May. 28, 2019 7:00am - brian.johnson

EXAGGERATION

Page 211 of 240

| Cascadia | PROJECT NUMBER | 0034-001-005 | FIGURE | |
|-----------------|-------------------|--------------|--------|--|
| Associates, LLC | DECE | MBER 2019 | 24 | |

CROSS SECTION E-E'



K:\11000\11200\11220\112203\dwg_SRI_Report_2019\4-2 - Cross Section C-C'.dwg - - - May. 28, 2019 7:00am - brian.johnson

- SP Poorly graded sand, gravelly sand, little or no fines
- SW-SM Fine to coarse sand and/or silty sand, sand silt mixtures
- SP-SM Fine to medium and/or fine to coarse sand, sand with some silt

| [| | | |
|---|-------------------|--------------|--------|
| CROSS S | ECTION | F-F' | |
| FORMER POPE & TALBOT WOOD-TREATING SITE | | | |
| M Cascadia | PROJECT NUMBER | 0034-001-005 | FIGURE |
| Associates, LLC | DECE | MBER 2019 | 25 |



K:\11000\11200\11220\112203\dwg_SRI_Report_2019\4-2 - Cross Section C-C'.dwg - - - May. 28, 2019 7:00am - brian.johnso

Page 213 of 240







FORMER POPE & TALBOT WOOD-TREATING SITE

| | PROJECT | 0024 001 005 | FIGURE |
|-----------------|---------|--------------|--------|
| Cascadia | NUMBER | 0034-001-005 | |
| Associates, LLC | DECE | MBER 2019 | 27 |

LEGEND:

Existing Monitoring Well

March 2011 Phase 1 TarGOST Boring:

- No Petroleum Sheen or NAPL Signal
- Minor Signal (Petroleum Sheen Slight)
- NAPL Signal
- July 2011 Phase 2 Sediment Boring:
- No Petroleum Sheen or NAPL Observation
- Petroleum Sheen Slight
- Petroleum Sheen Moderate to Heavy (NAPL)

July/October 2012 Sampling Event:

- ♦ Sheen Degree Unknown
- No Sheen
- ◇ Petroleum Sheen Slight
- Petroleum Sheen Moderate to Heavy

October 2013 Dock Area Investigation:

▼ Hand Auger without NAPL

- Pothole without NAPL Observation
- Pothole with NAPL Observation
- OAI Boring without NAPL Observation

DAI Boring with NAPL Observation

- Inferred Basalt Contour (4-foot interval) (darker shading indicates lower elevation)
- ----- Former Site Feature
- Areas ما مان Wood Debris Areas of Significant Creosote Contaminated
- Inferred Extent of Upland NAPL

Approximate Upland Site Boundary (Ordinary High Water 14 ft. NAVD 88)





Page 215 of 240



Page 216 of 240


| Page | 217 | of | 240 |
|-------|-----|----------|-----|
| ' ago | | <u> </u> | |

| | ° 7 | | | |
|---|-----------------------------------|--|--|--|
| Former 8-inch Creosote Pipeline (above ground) | | | | |
| peline | | | | |
| PR RE FR | | | | |
| | 8 - | | | |
| ner Pipeline circa 1919 | | | | |
| rmer Railroad ding Platform | | | | |
| | | | | |
| | 0 | | | |
| | | | | |
| Former Transfer | | | | |
| Table Dock | | | | |
| | - | | | |
| AMISTIC BOD | | | | |
| mppoose L | Ŀ | | | |
| s scar | Alle. | | | |
| er Hog Fuel | | | | |
| ng b er 0 | 8 | | | |
| 2 | 16 | | | |
| | and the second | | | |
| 40 B | AND AND | | | |
| <u>40</u> 0 75 150' | | | | |
| | | | | |
| DRMER POPE & TALBOT | DATE: MAY 2014 PROJECT NO.: | | | |
| VOOD-TREATING SITE | 9-61M-112202.09 REV. NO.: | | | |
| GRAPHY AND BATHYMETRY (2010) | 1 FIGURE NO.: | | | |
| | 30 | | | |



Page 218 of 240



LEGEND: Existing Monitoring Well Direct-Push Boring 🔺 Seep ▲ RI Sample Location Surface Water Level Measurement Station • October 2011 Shoreline Observation: March 2011 Phase 1 TarGOST Boring July 2011 Phase 2 Sediment Boring SDO July/October 2012 Sampling Event: Sediment Porewater Sample Location Limited Sediment Investigation (LSI) Sample Location October 2013 Dock Area Investigation: PWS-1 Hand Auger Pothole SEEP-PWS-1 • DAI Boring Ssed-M September 2017 Sampling Event: Phase 3 Data Gap Investigation Location Ssed-MC ---- Former Site Feature PWS-29

SD006

PWS-090617-2

MC-04

PWS-090517-3

0

SD010

SD007

S-2

SD008 (\bullet)

-

9010

S03

S018B

S018

SB-1

SD033

• SB-24

PWS-22

SD034

0

PWS-24

SD044

PWS-09

AW-3A

\$037

GP-26

SD049 SB-22

PWS-07

SB-28 /

WS-090517-11 SB-29

Ð

PWS-090417-14

SD047

PWS-06

SB-27

SB-26

SD036

SB-31

0

PWS-090617-15 + SD046

DAI SB 20 S024 • MW

DAI SB 12

PWS-05

SB-21

DAI SB 18

DALSB 1

S033

MW-6A

SD04

SB-25

SD03

PWS-090517

SB-12 🕥

SD029

/W-48

GD 21

PWS-

0

WS-090517-5

SD017

SD021

SB-07

PWS-23

SD02

SB-04

PWS-090517

SD025

50020

SD06

SD016

SB-02

PWS-16B SD030

PWS-090517-6

SD026

PWS-12

SD032

PWS-09051

PWS-090617-1

PWS-090517-4 0

SD028

SD027

PWS-13

D024 SB-0

SD018

SD013

PWS-27

SD019

SB-01

GP-24

Approximate Upland Site Boundary



Page 220 of 240



LEGEND:

- Existing Monitoring Well
- March 2011 Phase 1 TarGOST Boring:
- No Petroleum Sheen or NAPL Signal
- + Petroleum Sheen Slight

+ Petroleum Sheen - Moderate to Heavy

- July 2011 Phase 2 Sediment Boring:
- No Petroleum Sheen or NAPL Observation
- Petroleum Sheen Slight
- Petroleum Sheen Moderate to Heavy (NAPL) 🔶 No Sheen •
- Visible Creosote Sheen Observed in Surface Sediments (2010, 2011, 2012) 0

July/October 2012 Sampling Event:

Porewater_Sample_Location

- Sheen Degree Unknown
- No Sheen
- Petroleum Sheen Slight
- Petroleum Sheen Moderate to Heavy
- Limited Sediment Investigation (LSI) Sample Location September 2017 Sampling Event:
- Sheen Degree Unknown
- Petroleum Sheen Slight
- Petroleum Sheen Moderate to Heavy

October 2013 Dock Area Investigation:

- \triangledown Hand Auger without NAPL Observation
- Pothole without NAPL Observation
- Pothole with NAPL Observation
- DAI Boring without NAPL Observation
- DAI Boring with NAPL Observation

• Phase 3 Data Gap Investigation Location (No Sheen)

----- Former Site Feature

- Areas of Significant Creosote Contaminated Wood Debris
- Inferred Extent of Upland NAPL
- Area of Intermittent Groundwater/NAPL Seeps
- Approximate Offshore Areas
- Approximate Upland Site Boundary (Ordinary High Water 14 ft. NAVD 88)









Page 224 of 240



C:\DATA\POCC_0034_001\Projects\SRI_Dec_2019\Fig_05_04_PAHs_GW.mxd - ErikStrandhagen - 12/12/2019 - 7:49:38 AM

Page 225 of 240



| 1 - Contraction of the second | | | 1 | | |
|---|----------------------------|--------------------------------------|------------|--|--|
| | | | | | |
| | | | with a | | |
| MW-13B | | | | | |
| MW-13A C | | | • | | |
| | | - | - | | |
| | - | | 144.00 | | |
| | | | | | |
| 8/8/2011 0.234 U 0.467 U 2/20/2012 0.187 U 0.374 U | | | | | |
| 5/3/2012 0.187 U 0.374 U 8/14/2012 0.187 U 0.374 U | Star 1 |] | | | |
| MW-128 MW-12A | ~ | | | | |
| -K- | | | | | |
| | | | | | |
| | | | | | |
| *Isocontours | estimated | using average deter | cted | | |
| concentratio "B" wells exc | n at each w cluded from | ell. Shallower interv contouring. | val | | |
| - | | a mall | n) | | |
| U V | Date J.R. | atons (. ang tons (. | | | |
| e samo | e Dies dio | OH-HYDROCC | | | |
| 8/9/2011 2/20/2012 | 4.48 N 4.56 N | 0.472 U 0.519 N | | | |
| 5/4/2012 8/14/2012 | 5.37 N 8.49 N | 0.755 U 0.503 N | | | |
| NOTE | | | | | |
| BOLD indicates analy above the reporting lin | te was deteo nit | cted at or | | | |
| Shading indicates an exceedance of the Freshwater Ecotox Habitat Goal Level (HGL) for diesel-range hydrocarbons | | | | | |
| (0.64 mg/L). HGL from San Francisco Bay Regional Water Quality Board (Rev. 2, 2019). Nearshore monitoring wells used to assess potential | | | | | |
| risk from discharge of groundwater to surface water | | | | | |
| mg/L = milligrams per liter U = not detected at or above the stated level | | | | | |
| N = Presumptively identified result NJ = Presumptively identified and estimated result | | | | | |
| 2018 Imagery provided by Oregon Geospatial Enterprise Office. | | | | | |
| TOTAL PETROLEUM HYDROCARBONS | | | | | |
| DETECTED IN GROUNDWATER | | | | | |
| 2011-2012 FORMER POPE & TALBOT WOOD-TREATING SITE | | | | | |
| | PROJECT | 0034-001-005 | FIGURE | | |
| Associates, LLC | NUMBER DECE | EMBER 2019 | 39 | | |



K\1100011120011122011122031dwg_SRLReport_201914-2 - Cross Section C-C'.dwg --- May. 28, 2019 7:00am - brian.johnson

Page 227 of 240



K1110001122001122001122031dwgl_SRL Report_2019110-2 - Ecologica: Ris: Assessment Concept 🗈 Site Mode 20vg --- J Ch 12, 2019 10:55am - brian.johnson







Page 231 of 240





Vertical Datum: North American Vertical Datum of 1988 (NAVD88)



Legend

- Angular Riprap with Voids Filled by Smooth, Rounded Rock (e.g., Fish Mix)

50/50 Mixture of Sand and Organoclay

Ordinary High Water Mark (14 ft.)



1. The subsurface conditions shown are based on interpolation between 2. This figure is for informational purposes only. It is intended to assist in the

Vertical Datum: North American Vertical Datum of 1988 (NAVD88)

subsurface conditions may vary from those shown.

will serve as the official document of record.

widely spaced explorations and should be considered approximate; actual

identification of features discussed in a related document. Data were

compiled from sources as listed in this figure. The data sources do not

guarantee these data are accurate or complete. There may have been

updates to the data since the publication of this figure. This figure is a copy







Legend

- Angular Riprap with Voids Filled by Smooth, Rounded Rock (e.g., Fish Mix)
- 50/50 Mixture of Sand and Organoclay
 - NAPL Seep
 - ---- Ordinary High Water Mark (14 ft.)



Vertical Datum: North American Vertical Datum of 1988 (NAVD88)

subsurface conditions may vary from those shown.

will serve as the official document of record.

1. The subsurface conditions shown are based on interpolation between

widely spaced explorations and should be considered approximate; actual

2. This figure is for informational purposes only. It is intended to assist in the

identification of features discussed in a related document. Data were

compiled from sources as listed in this figure. The data sources do not

guarantee these data are accurate or complete. There may have been

updates to the data since the publication of this figure. This figure is a copy



Legend

- Angular Riprap with Voids Filled by Smooth, Rounded Rock (e.g., Fish Mix)
- 50/50 Mixture of Sand and Organoclay
 - NAPL Seep
 - ---- Ordinary High Water Mark (14 ft.)



Vertical Datum: North American Vertical Datum of 1988 (NAVD88)

subsurface conditions may vary from those shown.

will serve as the official document of record.

1. The subsurface conditions shown are based on interpolation between

widely spaced explorations and should be considered approximate; actual

2. This figure is for informational purposes only. It is intended to assist in the

identification of features discussed in a related document. Data were

compiled from sources as listed in this figure. The data sources do not

guarantee these data are accurate or complete. There may have been

updates to the data since the publication of this figure. This figure is a copy





Legend

- Angular Riprap with Voids Filled by Smooth, Rounded Rock (e.g., Fish Mix)
- 50/50 Mixture of Sand and Organoclay

NAPL Seep

Ordinary High Water Mark (14 ft.)



Vertical Datum: North American Vertical Datum of 1988 (NAVD88)

subsurface conditions may vary from those shown.

will serve as the official document of record.

1. The subsurface conditions shown are based on interpolation between

widely spaced explorations and should be considered approximate; actual

2. This figure is for informational purposes only. It is intended to assist in the

identification of features discussed in a related document. Data were

compiled from sources as listed in this figure. The data sources do not guarantee these data are accurate or complete. There may have been

updates to the data since the publication of this figure. This figure is a copy



STAFF REPORT

Scappoose Airport West Side Pavement Maintenance Bid Approval

| | Not To Exceed Total of \$347,893.05 |
|-------|---|
| RE: | Scappoose Airport West Side Pavement Maintenance Bid Approval |
| FROM: | Lacey Tolles, Airport Manager |
| TO: | Commission Board |
| DATE: | July 24, 2024 |

Discussion

The Scappoose Airport Master Plan identified Westside Pavement Maintenance as a key airport project for 2023-24. The pavement sealing and crack repairs are recommended based upon a 2021 study commissioned by the Oregon Department of Aviation ("ODAV").

In 2023, Port Airport Engineer Century West submitted a Federal Aviation Administration ("FAA") grant on behalf of the Port for Phase II of the West Side Pavement Maintenance Project. Following a public Invitation to Bid process, Century West reviewed the bids under criteria established by the FAA which directs that the lowest cost, most responsible bidder be selected. The lowest cost, most responsible bid was from C.R. Contracting of Bend, Oregon.

Cost Breakdown

| Total: | \$347,893.05 |
|----------|----------------|
| FAA 90% | (\$313,103.74) |
| ODAV 9%: | (\$31,310.38) |
| Port 1%: | (\$3,478.93) |

Recommendation

Adopt Resolution 2024-16, authorizing the Executive Director to execute a contract for the West Side Pavement Maintenance project with C.R. Contracting, LLC of \$347,893.05.

RESOLUTION NO. 2024-16

A RESOLUTION APPROVING THE SELECTION OF C.R. CONTRACTING FOR THE SCAPPOOSE AIRPORT WEST SIDE PAVEMENT MAINTENANCE PROJECT

WHEREAS, the Federal Aviation Administration ("FAA") provides financial support to airports in the form of Airport Improvement Program grants; and

WHEREAS, the Scappoose Airport Master Plan identified Westside Pavement Maintenance including pavement sealing and crack repairs as a key airport project for 2023-24. These repairs were also recommended in a 2021 study commissioned by the Oregon Department of Aviation; and

WHEREAS, Port staff and Century West ("Port Airport Engineer of Record") solicited an Invitation to Bid under FAA rules and received one bid; and

WHEREAS, Port staff and the Port Airport Engineer of Record thoroughly evaluated the bid and selected C.R. Contracting as the lowest cost, responsible bidder who provides lasting quality, excellent workmanship and professional service; and

WHEREAS, Port staff finds the Scope of Work reasonable at \$347,893.05.

BE IT RESOLVED by the Board of Commissioners of the Port of Columbia County as follows:

The Board authorizes the Executive Director to execute a contract for the West Side Pavement Maintenance Project with C.R. Contracting, LLC in the amount of \$347,893.05.

PASSED AND ADOPTED this ____ day of July 2024, by the following vote:

AYES: ____ NAYS: ____ Port of Columbia County
ABSTAIN:

By:

President

Attested By:

Secretary